

Recent Advances in Wave Function-Based Methods of Molecular-Property Calculations

Trygve Helgaker,^{*,†} Sonia Coriani,[‡] Poul Jørgensen,[§] Kasper Kristensen,[§] Jeppe Olsen,[§] and Kenneth Ruud[∥]

⁺Centre for Theoretical and Computational Chemistry, Department of Chemistry, University of Oslo, P.O. Box 1033 Blindern, N-0315 Oslo, Norway

^{*}Dipartimento di Scienze Chimiche e Farmaceutiche, Università degli Studi di Trieste, Via Giorgieri 1, I-34127 Trieste, Italy

[§]Lundbeck Center for Theoretical Chemistry, Department of Chemistry, Aarhus University, 8000 Aarhus C, Denmark

^{II}Centre for Theoretical and Computational Chemistry, Department of Chemistry, University of Tromsø, N-9037 Tromsø, Norway

CONTENTS

1.	Introduction	545	
2.	Electronic Hamiltonian		
	2.1. Many-Electron Hamiltonian	545	
	2.1.1. Classical Particles in an Electromagnetic		
	Field	545	
	2.1.2. Classical Hamiltonian	546	
	2.1.3. Schrödinger Hamiltonian	546	
	2.1.4. Pauli Hamiltonian	547	
	2.1.5. Breit—Pauli Hamiltonian	547	
	2.2. Molecular Electronic Breit—Pauli Hamiltonian	548	
	2.2.1. Nuclear Electromagnetic Potentials and		
	Fields	548	
	2.2.2. Molecular Electronic Breit—Pauli		
	Hamiltonian	549	
	2.2.3. Kinetic Energy	549	
	2.2.4. Coulomb Interactions	550	
	2.2.5. External Electric-Field Interactions	550	
	2.2.6. Zeeman Interactions	550	
	2.2.7. Spin—Orbit Interactions	550	
	2.2.8. Spin-Spin Interactions	551	
	2.2.9. Orbit-Orbit Interactions	551	
	2.2.10. Diamagnetic Interactions	551	
	2.3. Second Quantization		
	2.3.1. Second Quantization in the Spin-Orbital		
	Basis	552	
	2.3.2. Second Quantization in the Orbital Basis	553	
	2.4. Perturbation-Dependent Basis Sets	554	
	2.4.1. Perturbation-Dependent Atomic Orbitals	554	
	2.4.2. Orbital Connections	554	
	2.4.3. Perturbation Dependence of the Creation		
	and Annihilation Operators	555	
	2.4.4. One-Index Transformations	555	
	2.4.5. London Atomic Orbitals	556	
3.	Response Theory for Exact States	557	
	3.1. Quasi-Energy Formulation	557	

	2 1 1	Time a Demonstrate Over al Engeneration and Ita	
	3.1.1.	Time-Dependent Quasi-Energy and its	
		Variation Principle	557
	3.1.2.	Time-Averaged Quasi-Energy and Its	
		Variation Principle	558
	3.1.3.	Quasi-Energy Lagrangian and Its	
		Variation Principle	559
3.2.	Pertu	rbation Expansion of the Wave Function	560
	3.2.1.	Perturbation Expansion with a General	
		Reference State	560
	3.2.2.	Jacobian Matrix	561
	3.2.3.	Perturbation Expansion with an	
		Eigenvector Reference State	562
	3.2.4.	Wave Function Corrections As Functions	
		of the Reference State	562
3.3.	Perturbation Expansion of the Quasi-Energy		
3.4.	Response Functions		
	3.4.1.	Definition of Response Functions	564
	3.4.2.	Response Functions As Derivatives of the	
		Quasi-Energy	565
	3.4.3.	Explicit Expressions for the Linear, Quad-	
		ratic, and Cubic Response Functions	565
	3.4.4.	Symmetry Properties of Response	
		Functions	566
	3.4.5.	Residues of Response Functions	566
	3.4.6.	Equations of Motion for Response	
		Functions	566
3.5.	Respo	onse Functions As Derivatives of the Quasi-	
	Energy Lagrangian		
	3.5.1.	Perturbation Expansions of the	
		Lagrangian Quasi-Energies	567
	3.5.2.	Zero-Order Lagrange Multipliers	568
3.6.	3.6. Elimination Rules		

Special Issue: 2012 Quantum Chemistry

 Received:
 June 15, 2011

 Published:
 January 11, 2012

ACS Publications © 2012 American Chemical Society

	3.	.6.1.	2 <i>n</i> + 1 Rule	569
	3.	6.2.	2 <i>n</i> + 2 Rule	570
	3.	6.3.	Alternative Elimination Schemes	571
	3.7. D	amp	ed Response Theory	571
	3.	7.1.	Phenomenological Damping of Excited	
			States	572
	3.	.7.2.	Damped Linear Response theory	572
	3.	7.3.	Higher Order Damped Response theory	573
4.	Respo	nse	Theory for Approximate States	573
	4.1. H	lartre	ee–Fock Theory	573
	4.	1.1.	Quasi-Energy Response Theory in a Density-	
			Matrix Formulation	574
	4.	1.2.	Quasi-Energy Gradient	574
	4.	1.3.	Perturbed Density Matrix	575
	4.	1.4.	Response Functions	576
	4.2. N	lultio	configurational Self-Consistent Field	
	T	heor	У	576
	4.	2.1.	MCSCF Quasi-Energy	577
	4.	2.2.	Second-Order MCSCF Quasi-Energy	577
	4.	.2.3.	First-Order MCSCF Response Equation	579
	4.	2.4.	Linear MCSCF Response Function	579
	4.3. C	oup	led-Cluster Theory	580
	4.	3.1.	Coupled-Cluster Model	580
	4.	3.2.	Coupled-Cluster Quasi-Energy	
			Lagrangian	581
	4.	.3.3.	First-Order Molecular Properties	582
	4.	3.4.	Second-Order Molecular Properties	582
	4.	3.5.	Excitation Energies and Residues	583
	4.	.3.6.	Response Functions for the CCn Models	583
	4.	3.7.	Equation of Motion Coupled-Cluster	
			Method	584
	4.4. C)rbita	al-Relaxed Response Functions	586
	4.	.4.1.	Orbital-Relaxed Response Functions	586
	4.	.4.2.	Orbital-Relaxed MP2 Response Functions	586
	4.5. R	espo	onse Functions with Perturbation-	
	Dependent Basis Sets			
	4.6. C)verv	view of Developments and	
	In	nple	mentations	589
	4.	.6.1.	Expectation Values and First-Order	
			Properties	589
	4.	.6.2.	Hartree—Fock and MCSCF Response	
			Theory	589
	4.	.6.3.	Coupled-Cluster Response Theory	590
	4.	.6.4.	Explicit Correlation	591
	4.	.6.5.	Second-Order Polarization-Propagator	501
			Approximation	591
	4.	.6.6.	Algebraic-Diagrammatic-Construction	F 0 1
		< -	Method	591
~	4.	.6./.	Relativistic Corrections	591
5.	Survey	y of I	violecular Properties	592
	5.1. G	eom	netrical Derivatives	592
	5.	1.1.	Wolecular Gradients and Hessians	592
	5.	1.2.	Molecular Higher Order Derivatives	593

	nг	~ //	-	Λ.

		5.1.3. Property Geometrical Derivatives	594
	5.2.	Nuclear Magnetic Resonance	594
		5.2.1. Nuclear Shielding Constants	594
		5.2.2. Nuclear Spin—Spin Couplings	597
		5.2.3. Nuclear Spin—Rotation Constants	598
	5.3.	Electron Paramagnetic Resonance	599
		5.3.1. Electronic g Tensors	599
		5.3.2. Zero-Field Splittings	600
	5.3.3. Hyperfine Coupling Tensors		601
	5.4.	Electric Multipole Moments	601
		5.4.1. Electric Dipole Moments	601
		5.4.2. Electric Quadrupole Moments	601
	5.5.	Linear Response Properties	602
		5.5.1. Dipole Polarizabilities	602
		5.5.2. Magnetizabilities	602
		5.5.3. Rotational g Tensors	603
		5.5.4. Optical Rotation Tensors	604
	5.6.	Nonlinear Response Properties	605
		5.6.1. Hyperpolarizabilities and Nonlinear Optics	605
		5.6.2. Mixed Hyperpolarizabilities and	
		Birefringences	605
		5.6.3. Dispersion Coefficients	606
	5.7.	Interaction Properties	607
		5.7.1. Weak Interactions and Long-Range	
		Dispersion-Interaction Coefficients	607
		5.7.2. Interaction-Induced Properties	607
	5.8.	Excited-State Energies, Properties, and	
		Transition Properties	608
		5.8.1. Excitation Energies and One-Photon	
		Transition Strengths	608
		5.8.2. Electronic Circular Dichroism	608
		5.8.3. Multiphoton Absorption and Dichroism	608
		5.8.4. Magnetic Circular Dichroism	609
		5.8.5. Excited-State Molecular Gradients and	
		First-Order Properties	609
		5.8.6. Phosphorescence	610
		5.8.7. Spin—Orbit Coupling Constants	610
		588 Finite Lifetimes	610
6.	Pure	Vibrational Corrections	611
7.	Mol	ecular Properties in the Liquid and Solid	••••
	Pha	ses	611
	7.1.	Supermolecular Models	612
	7.2.	Multilevel Models	613
	7.3.	Continuum Models	615
	74	Molecular Properties with Periodic Boundary	0.0
		Conditions	616
8.	Con	cluding Remarks	617
Aı	uthor	Information	617
Bio	oara	ohies	617
Ac	:knov	wledgment	618
Ac	ronv	ims	618
References			619
Note Added in Proof			631

1. INTRODUCTION

Our understanding of the electronic structure of molecules often comes from spectroscopic investigations in which electromagnetic radiation is applied to a molecule and the scattering or absorption of the radiation is measured.^{1,2} These experimental techniques probe the structure of molecules by observing their response to applied electromagnetic perturbations. Such measurements provide a detailed picture of molecular systems, often rich in detail and sometimes difficult to interpret.

Over the last few decades, molecular electronic-structure theory has developed to a stage where it can provide invaluable help in the interpretation of experimental measurements of a broad range of molecular properties of importance in rotational and vibrational spectroscopies,³ ultraviolet—visible spectroscopies,^{4,5} magnetic-resonance spectroscopies,^{6–8} linear and nonlinear optics,^{1,9,10} and so on. In all these fields, spectroscopic constants can be calculated at various levels of electronic-structure theory, capable of approaching the exact value in a controlled, systematic manner, using established hierarchical levels of theory. At the same time, methodological developments toward larger systems are being made in different laboratories, promising to make calculations of molecular properties of systems containing hundreds and thousands of atoms routine in the near future. $^{11-14}$ In the present review, we discuss these developments in computational molecular electronicstructure theory. Our emphasis is on molecular response theory based on construction of a many-electron wave function, concentrating on developments during the past decade. As such, we do not discuss the various electronic-structure models themselves in any depth nor do we consider optimization of the energy and wave functions. Also, we do not discuss here calculations of molecular properties based on density-functional theory (DFT) except indirectly in our discussion of techniques for Hartree-Fock theory.

This review is divided into eight sections. Following the Introduction, we discuss in section 2 the molecular electronic Hamiltonian, including an overview of the Breit-Pauli Hamiltonian and the second-quantization formalism, extensively used in the remaining sections. Next, in section 3, we review response theory for exact states, providing a flexible quasi-energy framework subsequently used for approximate states in section 4, which contains detailed expositions of response theory for Hartree-Fock self-consistent field (SCF) theory, multiconfigurational SCF (MCSCF) theory, coupled-cluster theory, and Møller–Plesset theory. Following this exposition of response theory, we give a survey of molecular properties in section 5, covering geometrical properties and spectroscopic constants, magnetic resonance parameters, linear and nonlinear electromagnetic properties including nonlinear optics and birefringences, excitation energies, and excited-state properties. Following discussions of pure vibrational corrections in section 6 and molecular properties in liquids and solids in section 7, some concluding remarks are given in section 8.

2. ELECTRONIC HAMILTONIAN

In the present section, we discuss the electronic Hamiltonian with emphasis on those aspects that are important for evaluation of molecular properties. First, in section 2.1, we discuss the nonrelativistic electronic Hamiltonian in an external electromagnetic field and survey the Breit—Pauli relativistic corrections to second order in the fine-structure constant. In section 2.2, we introduce nuclear electromagnetic fields into the electronic Hamiltonian, thereby arriving at the molecular electronic Breit—Pauli Hamiltonian appropriate for molecular response calculations. Much of the exposition of molecular response theory given in the present review is presented using the formalism of second quantization. In section 2.3, we give a brief introduction to this formalism, in particular, we transform the molecular electronic Breit—Pauli Hamiltonian developed in section 2.2 to the second-quantization representation. Finally, in section 2.4, we discuss the effects of perturbation-dependent basis sets for the second-quantization representation of the molecular Hamiltonian and calculation of molecular properties.

2.1. Many-Electron Hamiltonian

In this subsection, we discuss the nonrelativistic Hamiltonian of an electronic system in the presence of an electromagnetic field and its relativistic corrections within the framework of the Breit—Pauli Hamiltonian. Particular attention is paid to the description of electromagnetic interactions and introduction of electron spin. The electronic Hamiltonian developed in this subsection forms the basis for the molecular electronic Hamiltonian discussed in section 2.2, where we introduce the effects of nuclear charges and nuclear magnetic moments. For a more thorough discussion of the electronic Hamiltonian, see ref 15.

2.1.1. Classical Particles in an Electromagnetic Field. In the presence of an electric field $E(\mathbf{r},t)$ and a magnetic field (or magnetic induction) $B(\mathbf{r},t)$, a classical particle of charge *z* moving with velocity **v** experiences the Lorentz force **F** (in Cartesian coordinates)

$$\mathbf{F} = z(\mathbf{E} + \mathbf{v} \times \mathbf{B}) \tag{1}$$

which is velocity dependent and hence nonconservative. The motion of the particle is determined by Newton's equation of motion (subject to boundary conditions)

$$\mathbf{F} = m\mathbf{a} \tag{2}$$

where m is the mass of the particle and \mathbf{a} its acceleration. The electric and magnetic fields \mathbf{E} and \mathbf{B} satisfy Maxwell's equations

$$\nabla \cdot \mathbf{E} = 4\pi\rho \tag{3}$$

$$\nabla \times \mathbf{B} - c^{-2} \partial_t \mathbf{E} = 4\pi c^{-2} \mathbf{J}$$
(4)

$$\nabla \cdot \mathbf{B} = \mathbf{0} \tag{5}$$

$$\nabla \times \mathbf{E} \,+\, \partial_t \mathbf{B} = \mathbf{0} \tag{6}$$

in the short-hand notation $\partial_t = \partial/\partial t$. Here and in the following, we employ SI-based atomic units¹⁶ where $m_e = 1$, e = 1, and $\hbar = 1$. In addition, the electric and magnetic constants satisfy $4\pi\varepsilon_0 = 1$ and $4\pi\mu_0 = c^2$, respectively. When the sources, that is, the charge density $\rho(\mathbf{r},t)$ and the current density $\mathbf{J}(\mathbf{r},t)$ generated by all particles in the system, are known, Maxwell's equations can be solved for $\mathbf{E}(\mathbf{r},t)$ and $\mathbf{B}(\mathbf{r},t)$. Conversely, since the particles are driven by the Lorentz force, $\rho(\mathbf{r},t)$ and $\mathbf{J}(\mathbf{r},t)$ depend on $\mathbf{E}(\mathbf{r},t)$ and $\mathbf{B}(\mathbf{r},t)$. In principle, therefore, we must simultaneously solve Maxwell's equations for the electromagnetic field and the classical equations of motion for the particles.

Maxwell's equations consist of two distinct pairs of equations: the inhomogeneous equations in eqs 3 and 4 and the homogeneous equations in eqs 5 and 6. The homogeneous equations are exact relations between the components of E and B, independent of the charges and currents in the system. Indeed, the homogeneous equations are automatically satisfied by expressing the fields in terms of a scalar potential $\phi(\mathbf{r},t)$ and a vector potential $\mathbf{A}(\mathbf{r},t)$ such that

$$\mathbf{E}(\mathbf{r}, t) = -\nabla \phi(\mathbf{r}, t) - \partial_t \mathbf{A}(\mathbf{r}, t)$$
(7)

$$\mathbf{B}(\mathbf{r}, t) = \nabla \times \mathbf{A}(\mathbf{r}, t) \tag{8}$$

The potentials (ϕ, \mathbf{A}) , which contain four rather than six components as in (\mathbf{E}, \mathbf{B}) , are subsequently obtained by solving the first, inhomogeneous pair of Maxwell's equations subject to boundary conditions (usually that the fields go to zero at an infinite distance from the sources). However, the scalar and vector potentials ϕ and \mathbf{A} are not unique in the sense that gauge transformations of the potentials

$$\phi' = \phi - \partial_t f \tag{9}$$

$$\mathbf{A}^{'} = \mathbf{A} + \nabla f \tag{10}$$

where the gauge function $f(\mathbf{r},t)$ depends on \mathbf{r} and t, do not affect the physical fields generated from the potentials by eqs 7 and 8. We are thus free to choose f so as to make ϕ and \mathbf{A} satisfy additional conditions. In the Coulomb gauge, the gauge function is chosen such that the vector potential becomes divergenceless

$$\nabla \cdot \mathbf{A} = \mathbf{0} \tag{11}$$

In the following, we shall always work in the Coulomb gauge, in which the scalar potential is given by the instantaneous Coulomb interaction, corresponding to the strict nonrelativistic limit of electrodynamics.¹⁷ Retardation and magnetic interactions appear as relativistic corrections through the purely solenoidal (divergenceless) vector potential. Indeed, as the speed of light *c* tends to infinity in Maxwell's equations in eqs 3-6, the divergence and curl of the magnetic field vanish, which combined with the boundary conditions suggests that, in the strict nonrelativistic limit, the magnetic field is zero everywhere. Non-Coulomb gauges have been explored for ab initio calculation of magnetic properties,^{18–21} but these have not gained widespread use.

2.1.2. Classical Hamiltonian. In classical Hamiltonian mechanics, a system of particles is described in terms of their positions q_i and conjugate momenta p_i . For each system, there exists a scalar Hamiltonian function $H(q_i, p_i)$ such that the classical equations of motion are given by

$$\dot{q}_i = \partial_{p_i} H, \, \dot{p}_i = -\partial_{q_i} H$$
 (12)

where $\partial p_i = \partial/\partial p_i$ and $\partial q_i = \partial/\partial q_i$. The corresponding equations for continuous systems are obtained by replacing the Hamiltonian function with the Hamiltonian density, the positions and momenta by the corresponding field variables, and the partial derivatives with functional derivatives. For a system of particles and fields, the Hamiltonian separates into three parts

$$H = H_{\rm p} + H_{\rm f} + H_{\rm int} \tag{13}$$

corresponding to the particles, the fields, and their interaction. The corresponding equations of motion are rarely solved selfconsistently. Rather, the degrees of freedom associated with either particles or fields are frozen, that is, treated as parameters, and the equations of motions are developed for the remaining variables. If the particles are treated as sources, then the particle term H_p drops out of the equations of motion, which then reduce to Maxwell's equations. When calculating molecular properties, on the other hand, we consider the response of molecules to REVIEW

drops out of the equations of motion. The classical Hamiltonian for a single particle of mass m and charge z moving in an external electromagnetic field E and B is

obtained from the free-particle Hamiltonian by the substitutions

$$\mathbf{p} \rightarrow \boldsymbol{\pi} = \mathbf{p} - z\mathbf{A}, \ H \rightarrow H + z\phi(\mathbf{r}, t)$$
 (14)

corresponding to the principle of minimal electromagnetic coupling^{22,23} based on a relativistic coupling of particles and fields.¹⁷ The coupling is minimal in the sense that it is the minimal coupling of particles and fields consistent with gauge invariance of the corresponding Lagrangian density.²⁴ In the nonrelativistic case, we then obtain the Hamiltonian

$$H(\mathbf{r}, \mathbf{p}) = \frac{\pi^2}{2m} + z\phi(\mathbf{r}, t)$$
(15)

The first of Hamilton's equations in eq 12 allows identification of the nonrelativistic kinetic momentum as

$$\boldsymbol{\pi} = m\mathbf{v} \tag{16}$$

Its total time derivative is the Lorentz force in eq 1, as can be shown from the second of Hamilton's equations.

2.1.3. Schrödinger Hamiltonian. To arrive at a quantummechanical description of the electron (i.e., a particle with charge z = -e = -1 au and mass $m = m_e = 1$ au) in an electromagnetic field, we perform the substitutions $\mathbf{p} \rightarrow -i\nabla$ and $H \rightarrow i\partial_t$ in eq 15 followed by multiplication with the wave function $\Psi(\mathbf{r})$ from the right, yielding the time-dependent Schrödinger equation

$$i\partial_t \Psi = H_{1c} \Psi \tag{17}$$

with the electronic Schrödinger Hamiltonian

$$H_{1c} = \frac{\pi^2}{2} - \phi, \ \boldsymbol{\pi} = -i\nabla + \mathbf{A}$$
(18)

We may expand the kinetic-energy operator $\pi^2/2$ in eq 18, yielding the following expression for the electronic Hamiltonian in the Coulomb gauge

$$H_{1c} = \frac{1}{2}p^2 + \mathbf{A} \cdot \mathbf{p} + \frac{1}{2}A^2 - \mathbf{\phi}$$
(19)

In an electromagnetic field, the vector potential thus makes both a linear paramagnetic contribution $\mathbf{A} \cdot \mathbf{p}$ and a quadratic diamagnetic contribution $A^2/2$ to the Hamiltonian. The diamagnetic contribution may be viewed as a small positive correction $A^2/2$ to the interaction $-\phi$ of the electron with the scalar potential.

Consider now a uniform magnetic field **B**. In the Coulomb gauge, such a field may be represented by the vector potential

$$\mathbf{A}_{\mathbf{O}}(\mathbf{r}) = \frac{1}{2}\mathbf{B} \times \mathbf{r}_{\mathbf{O}} = \frac{1}{2}\mathbf{B} \times (\mathbf{r} - \mathbf{O})$$
(20)

which vanishes at the gauge origin **O**. Using this potential, we may write the paramagnetic interaction in the form

$$\mathbf{A}_{\mathbf{O}} \cdot \mathbf{p} = \frac{1}{2} \mathbf{B} \cdot \mathbf{l}_{\mathbf{O}}, \ \mathbf{l}_{\mathbf{O}} = \mathbf{r}_{\mathbf{O}} \times \mathbf{p}$$
(21)

Thus, the paramagnetic interaction is proportional to the component of the orbital angular momentum I_O of the electron about the gauge origin **O** and along the direction of the external magnetic field **B**. It turns out, however, that this treatment of the interaction of the electron with the external magnetic field is inadequate as it ignores a fundamental property of the electron, namely, its spin.

2.1.4. Pauli Hamiltonian. The two-by-two Pauli spin matrices

$$\sigma_{\mathbf{x}} = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}, \ \sigma_{\mathbf{y}} = \begin{pmatrix} 0 & -\mathbf{i} \\ \mathbf{i} & 0 \end{pmatrix}, \ \sigma_{z} = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}$$
(22)

were introduced by Pauli in 1927^{25} and are matrix representations of twice the spin operator s for s = 1/2. They satisfy the algebra

$$\sigma_i \sigma_i = \delta_{ii} + i\varepsilon_{iik}\sigma_k \tag{23}$$

where the Kronecker delta δ_{ij} is 1 for identical indices and 0 otherwise whereas the Levi–Civita antisymmetric symbol ε_{ijk} is +1 for even permutations of the indices, -1 for odd permutations, and 0 if two or more indices are identical. Using the relation in eq 23, we easily deduce the Dirac identity

$$(\boldsymbol{\sigma} \cdot \mathbf{A})(\boldsymbol{\sigma} \cdot \mathbf{B}) = \mathbf{A} \cdot \mathbf{B} + \mathbf{i} \boldsymbol{\sigma} \times (\mathbf{A} \times \mathbf{B})$$
(24)

of which a particular instance is the simple relation

$$(\boldsymbol{\sigma} \cdot \mathbf{p})^2 = p^2 \tag{25}$$

It suggests that spin is hidden in the nonrelativistic free-particle Hamiltonian. However, a difference between the left- and the righthand sides of the above identity occurs when external electromagnetic fields are introduced through the minimal substitutions, eq 14. Starting from the left-hand side of the identity, we obtain a Hamiltonian of the form

$$H_{2c} = \frac{1}{2} (\boldsymbol{\sigma} \cdot \boldsymbol{\pi})^2 - \phi = \frac{1}{2} \pi^2 + \frac{1}{2} (\boldsymbol{\sigma} \cdot \mathbf{B}) - \phi \qquad (26)$$

whereas the right-hand side leads to a Hamiltonian in which the spin-Zeeman interaction (the second term above) is absent.

For a proper quantum-mechanical description of the electron, we therefore need a two-component wave function, which in the Pauli representation of eq 22 satisfies the Schrödinger equation

$$i\partial_{t} \begin{pmatrix} \Psi_{\alpha}(\mathbf{r}) \\ \Psi_{\beta}(\mathbf{r}) \end{pmatrix}$$

$$= \begin{pmatrix} \frac{1}{2}\pi^{2} - \phi + B_{z} & B_{x} - iB_{y} \\ B_{x} + iB_{y} & \frac{1}{2}\pi^{2} - \phi - B_{z} \end{pmatrix} \begin{pmatrix} \Psi_{\alpha}(\mathbf{r}) \\ \Psi_{\beta}(\mathbf{r}) \end{pmatrix} \quad (27)$$

The two components are coupled only in the presence of an external magnetic field. Alternatively, we may equip the electron with an additional discrete coordinate $\sigma = \pm 1/2$ in the manner $\Psi(\mathbf{x}) = \Psi(\mathbf{r}, \sigma)$, where $\Psi(\mathbf{r}, \pm 1/2)$ are two states whose lower and upper components vanish, respectively. The spin interaction is interpreted by associating a magnetic moment with the spin of the electron

$$\mathbf{m} = g_e \mu_{\rm B} \mathbf{s}, g_{\rm e} = -2, \ \mu_{\rm B} = \frac{e}{2m_{\rm e}} = \frac{1}{2} \, {\rm au}$$
 (28)

where we introduced the electron g factor g_e and the Bohr magneton μ_B . Here, we set the electron g factor equal to the value predicted by the Dirac equation, whereas its recommended value, including quantum-electrodynamics corrections, is $g_e =$ -2.0023193043622(15).²⁶ The spin paramagnetic term may now be written as a Zeeman interaction $-\mathbf{B} \cdot \mathbf{m}$ of the external induction with the magnetic moment of the electron. Thus, the two-component Hamiltonian eq 26 differs from the onecomponent Hamiltonian eq 18 only in the presence of the Zeeman term

$$H_{2c} = H_{1c} - \mathbf{B} \cdot \mathbf{m} \tag{29}$$

In the absence of an external magnetic field, the one- and twocomponent nonrelativistic Hamiltonians are equivalent and the two components of the wave function are not coupled. The total paramagnetic Zeeman interaction with a uniform magnetic field may therefore be written in the form

$$H_z = -\mathbf{B} \cdot \mathbf{m}_{\text{tot}} \ \mathbf{m}_{\text{tot}} = -\frac{1}{2} (\mathbf{l}_0 + 2\mathbf{s})$$
(30)

where $l_{\rm O}$ is the orbital-angular momentum of the electron about the gauge origin **O**, see eq 21. We note the anomalous double weight of the spin angular momentum in the magnetic dipolemoment operator \mathbf{m}_{tot} in eq 30.

From the one-electron Pauli Hamiltonian in eq 26, we obtain the *N*-electron Pauli Hamiltonian by adding pairwise Coulomb interactions to the sum of *N* one-electron Hamiltonians

$$H^{\text{NR}} = \sum_{i=1}^{N} \left(\frac{1}{2} (\boldsymbol{\sigma}_{i} \cdot \boldsymbol{\pi}_{i})^{2} - \boldsymbol{\phi}_{i} \right) + \sum_{i>j=1}^{N} \frac{1}{r_{ij}}$$
$$= \sum_{i=1}^{N} \left(\frac{1}{2} p_{i}^{2} + \mathbf{A}_{i} \cdot \mathbf{p}_{i} + \mathbf{B}_{i} \cdot \mathbf{s}_{i} + \frac{1}{2} A_{i}^{2} - \boldsymbol{\phi}_{i} \right)$$
$$+ \sum_{i>j=1}^{N} \frac{1}{r_{ij}}$$
(31)

adopting the short-hand notation $\phi_i = \phi(\mathbf{r}_{i}, t)$ and so on for the potentials and fields at the position of electron *i*. In nonrelativistic quantum mechanics, a many-electron system is described by an antisymmetric *N*-particle wave function $\Psi(\mathbf{x}_1, \mathbf{x}_2, ... \mathbf{x}_N)$ that satisfies the time-dependent Schrödinger equation

$$i\partial_t \Psi(\mathbf{x}_1, \dots \mathbf{x}_N, t) = H^{NR} \Psi(\mathbf{x}_1, \dots \mathbf{x}_N, t)$$
(32)

using the nonrelativistic electronic Pauli Hamiltonian of eq 31. Before we consider the potentials ϕ and **A** characteristic of molecular systems, we shall briefly consider the relativistic corrections to this Hamiltonian.

2.1.5. Breit–Pauli Hamiltonian. In a relativistic treatment of many-electron systems, the Dirac–Coulomb–Breit Hamiltonian is often used^{15,27}

$$H^{\text{DCB}} = \sum_{i} [c(\boldsymbol{\alpha}_{i} \cdot \boldsymbol{\pi}_{i}) - \phi_{i} + \beta_{i}c^{2}] \\ + \frac{1}{2} \sum_{i \neq j} \frac{1}{r_{ij}} - \frac{1}{2} \sum_{i \neq j} \left[\frac{c\boldsymbol{\alpha}_{i} \cdot c\boldsymbol{\alpha}_{j}}{c^{2}r_{ij}} + \frac{(c\boldsymbol{\alpha}_{i} \cdot \nabla_{i})(c\boldsymbol{\alpha}_{j} \cdot \nabla_{j})}{2c^{2}} \right]$$
(33)

where $\pmb{\alpha}$ and β are the four-by-four Dirac matrices, which in terms of two-by-two submatrices take the form

$$\boldsymbol{\alpha} = \begin{pmatrix} 0 & \boldsymbol{\sigma} \\ \boldsymbol{\sigma} & 0 \end{pmatrix}, \, \boldsymbol{\beta} = \begin{pmatrix} I & 0 \\ 0 & -I \end{pmatrix}$$
(34)

For a fully relativistic treatment of the electronic system, we may now solve the Dirac-Coulomb-Breit equation

$$i\partial_t \Psi(\mathbf{r}_i, t) = H^{\text{DCB}} \Psi(\mathbf{r}_i, t)$$
 (35)

which, due to the presence of negative-energy states, is done with the explicit or implicit use of projection operators onto the positive-energy states defined with respect to the current or some reference potential.²⁷ We note that this equation is only approximately Lorentz invariant, the two-electron part being correct only to second order in the fine-structure constant

$$\alpha = \frac{e^2}{c} \approx 7.297 \times 10^{-3} \tag{36}$$

The zero-order two-electron term in eq 33 is the Coulomb term, which is constructed from the interaction of one electron with the scalar potential of another electron defined with respect to the nuclear framework; it has the form of an instantaneous charge—charge interaction. Although it has the same form as in the nonrelativistic domain, the physical content is different in that it gives rise to the spin—same-orbit interaction. The second-order term in eq 33, the Breit term, arises from the interaction of one electron with the lowest order contribution of the corresponding vector potential and gives rise to the spin—other-orbit interaction. It has the form of a current—current interaction, as seen from the presence of the relativistic velocity operator $c\alpha$.

Reducing the Dirac-Coulomb-Breit Hamiltonian to twocomponent form, we obtain the Breit-Pauli operator

$$H^{\rm BP} = H^{\rm NR} + H^{\rm MV} + H^{\rm DW} + H^{\rm SsO} + H^{\rm SoO} + H^{\rm Ss} + H^{\rm OO} + O(\alpha^4)$$
(37)

where the leading term is the usual Pauli Hamiltonian in eq 31

$$H^{\rm NR} = \sum_{i} \frac{1}{2} \pi_i^2 + \sum_{i} \mathbf{B}_i \cdot \mathbf{s}_i - \sum_{i} \phi_i + \frac{1}{2} \sum_{i \neq j} \frac{1}{r_{ij}} \qquad (38)$$

and the lowest order relativistic corrections are given by

$$H^{\rm MV} = -\frac{\alpha^2}{8} \sum_{i} (\pi_i^4 + B_i^2) - \frac{\alpha^2}{4} \sum_{i} (\pi_i^2 \mathbf{B}_i \cdot \mathbf{s}_i + \mathbf{B}_i \cdot \mathbf{s}_i \pi_i^2)$$
(39)

$$H^{\rm DW} = \frac{\alpha^2}{8} \sum_{i} \left(\nabla_i \cdot \mathbf{E}_i \right) - \frac{\alpha^2 \pi}{2} \sum_{i \neq j} \delta(\mathbf{r}_{ij})$$
(40)

$$H^{\text{SsO}} = \frac{\alpha^2}{4} \sum_{i} \mathbf{s}_i \cdot (\mathbf{E}_i \times \boldsymbol{\pi}_i - \boldsymbol{\pi}_i \times \mathbf{E}_i) - \frac{\alpha^2}{2} \sum_{i \neq j} \frac{\mathbf{s}_i \cdot \mathbf{r}_{ij} \times \boldsymbol{\pi}_i}{r_{ij}^3}$$
(41)

$$H^{\text{SoO}} = -\alpha^2 \sum_{i \neq j} \frac{\mathbf{s}_j \cdot \mathbf{r}_{ij} \times \boldsymbol{\pi}_i}{r_{ij}^3}$$
(42)

$$H^{SS} = \frac{\alpha^2}{2} \sum_{i \neq j} \frac{r_{ij}^2 \mathbf{s}_i \cdot \mathbf{s}_j - 3\mathbf{s}_i \cdot \mathbf{r}_{ij} \mathbf{r}_{ij} \cdot \mathbf{s}_j}{r_{ij}^5} - \frac{4\alpha^2 \pi}{3} \sum_{i \neq j} \delta(\mathbf{r}_{ij}) \mathbf{s}_i \cdot \mathbf{s}_j$$
(43)

$$H^{\rm OO} = -\frac{\alpha^2}{4} \sum_{i \neq j} \frac{\boldsymbol{\pi}_i \cdot \boldsymbol{r}_{ij}^2 \boldsymbol{\pi}_j + \boldsymbol{\pi}_i \cdot \mathbf{r}_{ij} \mathbf{r}_{ij} \cdot \boldsymbol{\pi}_j}{r_{ij}^3}$$
(44)

Whereas the mass—velocity operator in eq 39 provides a correction to the kinetic energy, the Darwin operator in eq 40 corrects the potential energy of the electrons and their two-electron repulsion energy for the charge smearing caused by their Zitterbewegung.¹⁵ The spin—same-orbit operator or simply spin—orbit operator in eq 41 couples the spin of each electron to its own orbital motion in the presence of the external potential and the other electrons. The last three terms in eqs 42—44 couple different electrons to each other: the spin other-orbit operator in eq 42, the spin—spin operator in eq 43, and the orbit—orbit operator in eq 44.

2.2. Molecular Electronic Breit–Pauli Hamiltonian

Thus far, we have treated the electrons without specifying the scalar and vector potentials ϕ and **A** in which these particles move. In molecular systems, the most important contributions to these potentials are from the nuclei. In addition, there may be contributions from externally applied electromagnetic fields. To arrive at the molecular electronic Hamiltonian, we shall identify the nuclear contributions to the potentials ϕ_{nuc} and A_{nuc} and then perform the substitutions

$$\phi(\mathbf{r}) \rightarrow \phi_{\rm nuc}(\mathbf{r}) + \phi(\mathbf{r}) \tag{45}$$

$$\mathbf{A}(\mathbf{r}) \rightarrow \mathbf{A}_{\mathrm{nuc}}(\mathbf{r}) + \mathbf{A}(\mathbf{r}) \tag{46}$$

in the Breit–Pauli Hamiltonian eq 37, yielding the molecular electronic Breit–Pauli Hamiltonian.¹⁵ On a phenomenological basis, we also include in this Hamiltonian purely nuclear terms, which involve internuclear interactions or interactions of the nuclei with external electromagnetic fields. However, staying within the Born–Oppenheimer approximation, we treat the nuclei as stationary particles and do not introduce terms that involve the motion of the nuclei.

2.2.1. Nuclear Electromagnetic Potentials and Fields. Nuclei are complicated many-body systems. For our purposes, they are sufficiently well characterized by their charges Z_{K} , their radial extents \mathscr{R}_{K} their traceless quadrupole moments Θ_{K} and their magnetic moments \mathbf{M}_{K} . From these charges and moments, we arrive at the following multipole expansions of the nuclear contributions to the potentials in eqs 45 and 46

$$\phi_{\text{nuc}}(\mathbf{r}) = \sum_{K} \frac{Z_{K}}{r_{K}} - \frac{2\pi}{3} \sum_{K} Z_{K} \mathscr{R}_{K}^{2} \delta(\mathbf{r}_{K}) + \frac{1}{3} \sum_{K} \frac{\text{tr}\Theta_{K}(3\mathbf{r}_{K}\mathbf{r}_{K}^{T} - r_{K}^{2}\mathbf{I}_{3})}{r_{K}^{5}}$$
(47)

$$\mathbf{A}_{\mathrm{nuc}}(\mathbf{r}) = \alpha^2 \sum_{K} \frac{\mathbf{M}_{K} \times \mathbf{r}_{K}}{r_{K}^3}$$
(48)

where we ignored higher order moments as well as all nonelectromagnetic interactions such as those arising from parity violation, see ref 28 and references therein. To leading orders, the corresponding nuclear electric and magnetic fields are given by

$$\mathbf{E}_{\rm nuc}(\mathbf{r}) = \sum_{K} \frac{Z_{K} \mathbf{r}_{K}}{r_{K}^{3}}$$
(49)

$$\mathbf{B}_{\text{nuc}}(\mathbf{r}) = \alpha^2 \sum_{K} \frac{3(\mathbf{M}_{K} \cdot \mathbf{r}_{K})\mathbf{r}_{K} - r_{K}^{2}\mathbf{M}_{K}}{r_{K}^{5}} + \frac{8\pi}{3}\alpha^2 \sum_{K} \delta(\mathbf{r}_{K})\mathbf{M}_{K}$$
(50)

For symmetry (parity) reasons, there are no odd-order electric moments and no even-order magnetic moments in the expansions of the nuclear potentials.¹⁵ Among the different contributions to the nuclear potentials in eqs 47 and 48, by far the largest is the electrostatic contribution from the nuclear charge Z_{KJ} indeed, the resulting nuclear point-charge potential is responsible for generating stable molecular structures. The remaining contributions to the potentials are orders of magnitude smaller, representing small corrections to the point-charge nuclear model and being responsible for the hyperfine interactions in atoms and molecules.

The contact and quadrupolar interactions in eq 47 depend on the size and shape of the nucleus as represented by the radial extent and traceless quadrupolar moments

$$Z_K \ \mathscr{R}_K^2 = \int r^2 \rho_K(\mathbf{r}) \mathrm{d}\mathbf{r}$$
(51)

$$\Theta_{K} = \frac{1}{2} \int (3\mathbf{r}\mathbf{r}^{\mathrm{T}} - r^{2}\mathbf{I}_{3})\rho_{K}(\mathbf{r})\mathrm{d}\mathbf{r}$$
(52)

where $\rho_K(\mathbf{r})$ is the nuclear charge distribution. From eq 51, we note that the finite size reduces the potential at the nucleus, increasing the total energy. Since nuclear sizes are of the order of 1 fm, the correction is small, in particular, for light nuclei. The finite nuclear size may alternatively be corrected for by representing the nucleus by a Gaussian charge distribution^{27,29}

$$\rho_K(\mathbf{r}_K) = Z_K \left(\frac{\eta_K}{\pi}\right)^{3/2} \exp(-\eta_K r_K^2)$$
(53)

where $\eta_K = (3/2) \mathcal{R}_K^{-2}$. Gaussian distributions or linear combinations of such distributions are used since they simplify calculation of integrals over Gaussian atomic orbitals (AOs).³⁰ Integrated over all space, the potential from the Gaussian nucleus becomes

$$\int \frac{\rho_K(\mathbf{r}_K)}{r_K} d\mathbf{r} = \frac{Z_K \operatorname{erf}(\sqrt{\eta_K} r_K)}{r_K}$$
(54)

where the strictly increasing error function $0 \le erf(x) \le 1$ reduces the potential at the nucleus. The traceless quadrupolar moments in eq 52 interact with the field gradients at the position of the nuclei in eq 47, providing a correction for nonspherical charge distribution.³¹

The nuclear magnetic moments \mathbf{M}_K in eq 48 arise from the nuclear spins \mathbf{I}_K , to which they are related as

$$\mathbf{M}_{K} = g_{K} \mu_{N} \mathbf{I}_{K} = \gamma_{K} \hbar \mathbf{I}_{K}$$
(55)

where $\mu_N = e\hbar/2m_p$ is the nuclear magneton, g_K is the nuclear g value, and γ_K is the magnetogyric ratio of the nucleus. The nuclear magneton is numerically small, about 2.7×10^{-4} au. The nuclear g value g_K is a dimensionless empirical constant, different for each nucleus. It is on the order of unity (its absolute value never exceeds six) and may be positive or negative. The nuclear magnetic moments are therefore roughly 3 orders of magnitude smaller than the electronic magnetic moment **m** in eq 28. Only nuclei with spin 1/2 or greater have a nonvanishing magnetic dipole moment.

Finally, we take the purely nuclear contributions to the electronic Hamiltonian to be of the form

$$H^{\text{nuc}} = \frac{1}{2} \sum_{K \neq L} \frac{Z_K Z_L}{R_{KL}} + \sum_K Z_K \phi_K - \sum_K \mathbf{B} \cdot \mathbf{M}_K + \frac{\alpha^2}{2} \sum_{K \neq L} \frac{R_{KL}^2 (\mathbf{M}_K \cdot \mathbf{M}_L) - 3(\mathbf{M}_K \cdot \mathbf{R}_{KL}) (\mathbf{R}_{KL} \cdot \mathbf{M}_L)}{R_{KL}^5}$$

where we included the pairwise Coulomb repulsion between nuclei of charges Z_K and Z_L and separation R_{KL} , the interaction with an external scalar potential $\phi_K = \phi(\mathbf{R}_K)$ at the nuclear position \mathbf{R}_K , the Zeeman interaction of the nuclear magnetic moments \mathbf{M}_K with a uniform external magnetic induction **B**, and the pairwise dipolar magnetic interactions between nuclei of magnetic moments \mathbf{M}_K and \mathbf{M}_L and relative positions $\mathbf{R}_{KL} = \mathbf{R}_K - \mathbf{R}_L$. Unlike in the electronic spin—spin operator in eq 43, there is no nuclear—nuclear contact term since the stationary nuclei never occupy the same point in space.

2.2.2. Molecular Electronic Breit–Pauli Hamiltonian. Substituting the nuclear potentials in eqs 47 and 48 and the fields in eqs 49 and 50 into the electronic Breit–Pauli Hamiltonian in eq 37 and adding the purely nuclear contributions in eq 56 we arrive at the molecular electronic Breit–Pauli Hamiltonian. There are many ways to collect the terms in the resulting Hamiltonian. Our classification is based on the nature of the physical mechanisms and interactions rather than on the number and kinds of particles involved in each interaction, giving

$$H_{\text{mol}}^{\text{BP}} = \begin{cases} H_{\text{kin}} &\leftarrow \text{ kinetic energy} \\ +H_{\text{cou}} &\leftarrow \text{ Coulomb interactions} \\ +H_{\text{ef}} &\leftarrow \text{ external electric interactions} \\ +H_{z} &\leftarrow \text{ Zeeman interactions} \\ +H_{so} &\leftarrow \text{ spin-orbit interactions} \\ +H_{\text{ss}} &\leftarrow \text{ spin- orbit interactions} \\ +H_{oo} &\leftarrow \text{ orbit-orbit interactions} \\ +H_{\text{dia}} &\leftarrow \alpha^{4} \text{ diamagnetic interactions} \end{cases}$$
(57)

Before considering the individual terms, we recall that the nuclei are treated as stationary sources of electromagnetic fields, thereby excluding all terms involving nuclear motion. Moreover, although the electrons are treated relativistically only to order α^2 , we retain diamagnetic terms (which are on the order of α^4 , assuming that the external fields are on the order of α^2) since they are spectroscopically important and needed for calculation of molecular magnetic properties as well as to maintain formal gauge invariance. With these remarks, we are ready to survey the various contributions to the molecular electronic Breit–Pauli Hamiltonian.

2.2.3. Kinetic Energy. The Breit–Pauli kinetic-energy operator in eq 57 is given by

$$H_{\rm kin} = -\frac{1}{2} \sum_{i} \nabla_i^2 - \frac{\alpha^2}{8} \sum_{i} \nabla_i^4$$
(58)

The first term is the usual Newtonian kinetic-energy operator. The second term is the mass-velocity term, which arises because of the relativistic dependence of mass on velocity. The mass-velocity correction is always negative and represents, together with the one-electron Darwin correction described shortly, the major contribution to the relativistic energy correction for slow electrons. The mass-velocity term is unbounded from below. It (and the other small terms in the Breit-Pauli Hamiltonian) should therefore not be used in variational calculations.^{32,33}

(56)

2.2.4. Coulomb Interactions. In the Breit—Pauli Hamiltonian in eq 57, the Coulomb interactions are represented by

$$H_{cou} = -\sum_{iK} \frac{Z_K}{r_{iK}} + \frac{1}{2} \sum_{i \neq j} \frac{1}{r_{ij}} + \frac{1}{2} \sum_{K \neq L} \frac{Z_K Z_L}{R_{KL}} + \frac{\alpha^2 \pi}{2} \sum_{iK} Z_K \delta(\mathbf{r}_{iK}) - \frac{\alpha^2 \pi}{2} \sum_{i \neq j} \delta(\mathbf{r}_{ij}) + \frac{2\pi}{3} \sum_{iK} Z_K R_K^2 \delta(\mathbf{r}_{iK}) - \frac{1}{3} \sum_{iK} \frac{\text{tr}\Theta_K (3\mathbf{r}_{iK} \mathbf{r}_{iK}^{\text{T}} - r_{iK}^2 \mathbf{I}_3)}{r_{iK}^5}$$
(59)

The first three operators are the usual nonrelativistic pointcharge interactions among electrons of charge -1 and nuclei of charge Z_{K_3} followed by the Darwin corrections (fourth and fifth operators) caused by the Zitterbewegung of the electrons. The one-electron Darwin correction provides, together with the mass—velocity correction in eq 58, the dominant relativistic correction to the total energies of light systems—it is almost as large as the mass—velocity correction but positive, increasing the total energy. The negative two-electron Darwin term is much less important (by about 2 orders of magnitude)³⁴ since the repulsion between the electrons reduces their probability of occupying the same point in space. There is no nuclear—nuclear Darwin correction in eq 59 since the nuclei are treated as stationary in the Born—Oppenheimer approximation.

As noted in section 2.2.1, the nuclear point-charge model is not always adequate. The finite size and nonspherical charge distributions of the nuclei may then be corrected for by using the second last and last operators in eq 59, respectively. In these operators, R_K represents the nuclear extent as defined in eq 51 and Θ_K is the nuclear quadrupole moment, interacting with the field gradient at the nuclear quadrupole resonance (NQR) studies³⁵ as well as hyperfine interactions in high-resolution rotational spectra.³ They are also important in magnetic resonance studies since the quadrupole moment is aligned with the nuclear magnetic moment, thereby providing a coupling between nuclear spins and electronic field gradients, thus providing an important relaxation pathway in nuclear magnetic resonance (NMR) spectroscopy.³⁶

2.2.5. External Electric-Field Interactions. The interactions of a molecule with an externally applied scalar potential $\phi(\mathbf{r})$ are in the Breit–Pauli Hamiltonian in eq 57 represented by the operator

$$H_{\rm ef} = -\sum_{i} \phi_{i} + \sum_{K} Z_{K} \phi_{K} + \frac{\alpha^{2}}{8} \sum_{i} (\nabla_{i} \cdot \mathbf{E}_{i})$$
(60)

According to Coulomb's law in eq 3, the Darwin correction vanishes in vacuum. Since macroscopically applied fields are fairly uniform on a molecular scale, it is expedient to expand the Hamiltonian in eq 60 in multipoles

$$H_{\rm ef} = Q_{\rm tot} \phi_0 - \boldsymbol{\mu}_{\rm tot} \cdot \mathbf{E}_0 - \frac{1}{2} \operatorname{tr} \mathbf{Q}_{\rm tot} \mathbf{V}_0 + \dots$$
(61)

where q_{tot} is the total charge of the molecule while μ_{tot} and Q_{tot} are the dipole and second moments, respectively

$$\mu_{\text{tot, }\alpha} = -\sum_{i} r_{i\alpha} + \sum_{K} Z_{K} R_{K\alpha}$$
(62)

$$Q_{\text{tot, }\alpha\beta} = -\sum_{i} r_{i\alpha}r_{i\beta} + \sum_{K} Z_{K}R_{K\alpha}R_{K\beta}$$
(63)

Since the field gradient generated by an external source is traceless, the traceless quadrupole moment

$$\Theta = \frac{3}{2}\mathbf{Q} - \frac{1}{2}(\operatorname{tr} \mathbf{Q})\mathbf{I}_3 \tag{64}$$

is often used instead of **Q**. In the multipole expansion of eq 61, the electric potential ϕ_0 , the electric field \mathbf{E}_0 , and the electric field gradient \mathbf{V}_0 are evaluated at the origin of the expansion. Higher order multipoles are rarely needed to describe macroscopic fields but are needed for the highly nonuniform fields generated by neighboring molecular systems, in particular, for calculation of weak intermolecular interactions and longrange Coulomb interactions.³⁷ The external scalar potential ϕ also contributes to the spin—orbit interaction, as discussed in section 2.2.7.

2.2.6. Zeeman Interactions. The Zeeman term in the Breit–Pauli Hamiltonian in eq 57 describes the paramagnetic interactions of the molecule with an externally applied magnetic field B

$$H_{z} = -\mathbf{B} \cdot \sum_{i} \left(-\frac{1}{2} \mathbf{l}_{i0} - \mathbf{s}_{i} + \frac{1}{2} \alpha^{2} \mathbf{s}_{i} \nabla_{i}^{2} \right) - \mathbf{B} \cdot \sum_{K} \mathbf{M}_{K}$$
(65)

Since the nuclear moments are on the order of 10^{-3} in atomic units, the nuclear part is much smaller than the electronic part but it is important in NMR spectroscopy, where it determines the position of the (unshielded) resonance lines in the spectra. We also note that, in the Breit–Pauli Hamiltonian in eq 57, there is, apart from the paramagnetic interactions in H_z discussed here, a corresponding set of diamagnetic (quadratic) interactions in H_{dia} discussed later.

There are three distinct contributions to the electronic part of H_z in eq 65, the first of which represents the Zeeman interaction with the magnetic moment generated by the orbital angular momentum of the electrons relative to the gauge origin $\mathbf{l}_{iO} = \mathbf{r}_{iO} \times \mathbf{p}_i$. The second and third electronic contributions to H_z constitute the Zeeman interaction with the spin of the electrons. In addition to a dominant nonrelativistic contribution from eq 38, there is a small relativistic correction from eq 39, which contributes to the *g* shift of electron paramagnetic resonance (EPR) spectroscopy. Note that because of its dependence on the gauge origin **O** the Zeeman operator is not uniquely defined.

The expectation value of an imaginary Hermitian operator Ω vanishes for orbitally nondegenerate states, since their wave function may be chosen real

$$\langle \operatorname{real} | \Omega | \operatorname{real} \rangle = \langle \operatorname{real} | \Omega | \operatorname{real} \rangle^{^{*}} = \langle \operatorname{real} | \Omega^{^{+}} | \operatorname{real} \rangle$$

= - \langle \text{real} \Omega | \Omega | \text{real} \rangle \text{ (66)}

For such states, the expectation value of $\mathbf{L}_{O} = \sum_{i} \mathbf{l}_{iO}$ therefore vanishes and the orbital angular momentum is said to be quenched. Likewise, the expectation value of any triplet operator such as the spin angular momentum operator $\mathbf{S} = \sum_{i} \mathbf{s}_{i}$ vanishes for singlet states. Closed-shell molecules therefore do not interact with external magnetic fields to first order, but the second-order diamagnetic interactions never vanish.

2.2.7. Spin–Orbit Interactions. Collecting those terms in the Breit–Pauli Hamiltonian of eq 57 that, to second order in the fine-structure constant, couple the motion of the electrons to

particle spins (magnetic moments), we obtain

$$H_{so} = \frac{\alpha^{2}}{2} \sum_{iK} \frac{Z_{K} \mathbf{s}_{i} \cdot \mathbf{l}_{iK}}{r_{iK}^{3}} - \frac{\alpha^{2}}{2} \sum_{i \neq j} \frac{\mathbf{s}_{i} \cdot \mathbf{l}_{ij}}{r_{ij}^{3}} - \alpha^{2} \sum_{i \neq j} \frac{\mathbf{s}_{j} \cdot \mathbf{l}_{ij}}{r_{ij}^{3}} + \alpha^{2} \sum_{iK} \frac{\mathbf{M}_{K} \cdot \mathbf{l}_{iK}}{r_{iK}^{3}} + \frac{\alpha^{2}}{4} \sum_{i} \mathbf{s}_{i} \cdot (\mathbf{E}_{i} \times \mathbf{p}_{i} - \mathbf{p}_{i} \times \mathbf{E}_{i})$$

$$(67)$$

The spin—orbit interaction arises from the coupling of electron spin to the magnetic fields induced by other charges in relative motion. In the Born—Oppenheimer approximation, all particle motion is with respect to the nuclear framework. The first term in eq 67 is the nuclear spin—orbit operator arising from the motion of the electrons relative to the nuclei. It should be emphasized that the angular momentum operator l_{iK} accordingly represents the relative motion of the electron and nucleus. The second term above is the spin—same-orbit operator arising from the orbital motion of the reference electron with respect to the nuclear frame, whereas the third term is the spin—other-orbit operator, which arises from the relative motion of the other electron with the respect to the nuclear frame.

Because of their triplet nature, the spin—orbit operators do not contribute to the total energy of closed-shell systems to order α^2 . On the other hand, they are responsible for the splitting of otherwise degenerate states in open-shell systems, producing, for example, the fine structure of atomic spectra. The spin—orbit operators are also important in providing a mechanism for intersystem crossings and phosphorescence, that is, spin-forbidden radiationless and radiative transitions between molecular electronic energy levels. Finally, in magnetic-resonance spectroscopies, the spin—orbit operators contribute to second order to g values and hyperfine coupling constants and to zero-field splittings in EPR and often provide the leading order relativistic corrections to heavy-atom effects on the shielding constants of light nuclei.

The fourth operator in eq 67, which arises from substitution of \mathbf{A}_{nuc} of eq 48 into the $\sum_i \mathbf{A}_i \cdot \mathbf{p}_i$ part of eq 38, provides a mechanism for coupling the nuclear magnetic moments \mathbf{M}_K to the orbital motion of the electrons. It is known as the orbital hyperfine operator or the paramagnetic spin—orbit operator; it is similar to the spin—other-orbit operator, which likewise couples the spin and orbital motion of different particles. Like all hyperfine operators, the orbital hyperfine interaction is weak (10^{-8} au). It contributes to nuclear shielding constants and indirect nuclear spin—spin coupling constants in NMR and in second order to hyperfine coupling constants in EPR. The last term in eq 67 depends on the external electric field and is unimportant for free molecules.

2.2.8. Spin–Spin Interactions. Collecting all terms in the Breit–Pauli operator of eq 57 that involve interactions between the spins (magnetic moments) of two particles, we obtain

$$H_{ss} = \frac{\alpha^2}{2} \sum_{i \neq j} \left[\frac{r_{ij}^2 \mathbf{s}_i \cdot \mathbf{s}_j - 3 \mathbf{s}_i \cdot \mathbf{r}_{ij} \mathbf{r}_{ij} \cdot \mathbf{s}_j}{r_{ij}^5} - \frac{8\pi}{3} \delta(\mathbf{r}_{ij}) \mathbf{s}_i \cdot \mathbf{s}_j \right] -\alpha^2 \sum_{iK} \left[\frac{r_{iK}^2 \mathbf{s}_i \cdot \mathbf{M}_K - 3 \mathbf{s}_i \cdot \mathbf{r}_{iK} \mathbf{r}_{iK} \cdot \mathbf{M}_K}{r_{iK}^5} - \frac{8\pi}{3} \delta(\mathbf{r}_{iK}) \mathbf{s}_i \cdot \mathbf{M}_K \right] + \frac{\alpha^2}{2} \sum_{K \neq L} \frac{R_{KL}^2 (\mathbf{M}_K \cdot \mathbf{M}_L) - 3(\mathbf{M}_K \cdot \mathbf{R}_{KL}) (\mathbf{R}_{KL} \cdot \mathbf{M}_L)}{R_{KL}^5}$$

consisting of classical dipolar and Fermi-contact interaction terms. Together with the Darwin operator of eq 59, the twoelectron Fermi-contact operator in the first line of eq 68 provides the largest two-electron relativistic correction to the energy of light systems. The two-electron spin—spin operator also contributes to the zero-field splitting observable in EPR spectroscopy of triplet species (biradicals)^{38,39} discussed in section 5.3.2. Whereas the two-electron spin—spin interaction dominates the zero-field splittings in organic biradicals, the spin—orbit interaction dominates the splittings in systems containing heavier elements, eq 612.

The hyperfine interaction operator in the second line of eq 68 couples the electronic and nuclear spins and arises from substitution of \mathbf{B}_{nuc} of eq 50 into $\sum_i \mathbf{B}_i \cdot \mathbf{s}_i$ of eq 38. Together with the orbital hyperfine operator in eq 67 it contributes to the indirect coupling of nuclear spins observed in NMR for liquids, as discussed in section 5.2. The isotropic Fermi-contact term is particularly important, coupling the rapidly tumbling nuclear moments in high-resolution NMR. Finally, the dipolar nuclear spin—spin operator in the last line of eq 68 is responsible for important direct coupling of nuclear spins observed in NMR in solids but makes no contribution to the isotropic coupling in liquids. We omitted the nuclear—nuclear contact term, which does not contribute for stationary nuclei.

2.2.9. Orbit—**Orbit Interactions.** The two-electron orbit orbit operator in eq 57 arises from eq 44 and represents a relativistic correction to the two-electron Coulomb interaction due to the relative motion of the electrons

$$H_{\rm oo} = -\frac{\alpha^2}{4} \sum_{i \neq j} \frac{\mathbf{p}_i \cdot r_{ij}^2 \mathbf{p}_j + \mathbf{p}_i \cdot \mathbf{r}_{ij} \mathbf{r}_{ij} \cdot \mathbf{p}_j}{r_{ij}^3}.$$
 (69)

The orbit—orbit correction is smaller than the two-electron Darwin and spin—spin corrections but becomes more important in heavier atoms.³⁴ Moreover, the orbit—orbit interaction does not split levels and does not cause transitions. It is therefore usually of less interest than the two-electron spin—orbit and spin—spin interactions.

2.2.10. Diamagnetic Interactions. The diamagnetic interactions in the Breit–Pauli Hamiltonian of eq 57 are terms of order α^4 or smaller (keeping in mind that the external magnetic induction is typically of order α^2), arising from expansion of the vector potential in the kinetic-momentum operators of eq 37. Although the Breit–Pauli Hamiltonian as such is correct only to order α^2 , the small diamagnetic interactions are included in our discussion since they are important for many spectroscopic parameters

$$H_{\text{dia}} = \begin{cases} H_{\text{BB}} &\leftarrow \text{magnetizability} \\ +H_{\text{BM}} &\leftarrow \text{nuclear shielding (NMR)} \\ +H_{\text{MM}} &\leftarrow \text{indirect nuclear spin-spin coupling (NMR)} \\ +H_{\text{MVBs}} \leftarrow \text{electronic } g \text{ factor shift (EPR)} \\ +H_{\text{Bs}} &\leftarrow \text{electronic } g \text{ factor shift (EPR)} \\ +H_{\text{Ms}} &\leftarrow \text{hyperfine coupling (EPR)} \end{cases}$$

$$(70)$$

The first three operators in H_{dia} arise from expansion of $A^2/2$ with the vector potential given by eq 46

$$H_{\rm BB} = \frac{1}{8} \sum_{i} \left(\mathbf{B} \times \mathbf{r}_{i\rm O} \right)^2 \tag{71}$$

(68)

$$H_{\rm BM} = \frac{\alpha^2}{2} \sum_{iK} \frac{(\mathbf{B} \times \mathbf{r}_{iO}) \cdot (\mathbf{M}_K \times \mathbf{r}_{iK})}{r_{iK}^3}$$
(72)

$$H_{\rm MM} = \frac{\alpha^4}{2} \sum_{iKL} \frac{\left(\mathbf{M}_K \times \mathbf{r}_{iK}\right) \cdot \left(\mathbf{M}_L \times \mathbf{r}_{iL}\right)}{r_{ik}^3 r_{iL}^3}$$
(73)

They contribute diamagnetically to the magnetizability tensor, to the nuclear shielding tensor, and to the indirect nuclear spin—spin tensor, respectively. The fourth operator in H_{dia} is the spin-dependent part of the mass—velocity operator in eq 39

$$H_{\text{MVBs}} = -\frac{\alpha^2}{4} \sum_i \left(\pi_i^2 \mathbf{B}_i \cdot \mathbf{s}_i + \mathbf{B}_i \cdot \mathbf{s}_i \pi_i^2 \right)$$
(74)

Finally, the last two operators in H_{dia} arise from expansion of eq 46 in the spin—orbit and spin—other-orbit operators in eqs 41 and 42 with a simultaneous substitution of the nuclear electric field in eq 49

$$H_{\text{Bs}} = \frac{\alpha^2}{4} \sum_{iK} \frac{Z_K (\mathbf{B} \times \mathbf{r}_{i0}) \cdot (\mathbf{s}_i \times \mathbf{r}_{iK})}{r_{iK}^3} - \frac{\alpha^2}{4} \sum_{i \neq j} \frac{(\mathbf{B} \times \mathbf{r}_{i0}) \cdot (\mathbf{s}_i + 2\mathbf{s}_j) \times \mathbf{r}_{ij}}{r_{ij}^3}$$
(75)

$$H_{\rm Ms} = \frac{\alpha^4}{2} \sum_{i, \ KL} \frac{Z_K(\mathbf{M}_L \times \mathbf{r}_{iL}) \cdot (\mathbf{s}_i \times \mathbf{r}_{iK})}{r_{iK}^3 r_{iL}^3} - \frac{\alpha^4}{2} \sum_{K, \ i \neq j} \frac{(\mathbf{M}_K \times \mathbf{r}_{iK}) \cdot (\mathbf{s}_i + 2\mathbf{s}_j) \times \mathbf{r}_{ij}}{r_{iK}^3 r_{ij}^3}$$
(76)

These operators couple the spin of the electron to the external magnetic field and the nuclear magnetic moments, respectively, contributing in second order to the *g* shifts and nuclear hyperfine coupling constants in EPR spectroscopy⁴⁰ as well as often being the leading order relativistic correction in heavy-element shieldings.^{41–43}

2.3. Second Quantization

We introduce in this subsection second quantization with emphasis on the representation of the molecular electronic Breit–Pauli operator in this formalism. For a more extensive treatment of second quantization in quantum chemistry, see the monograph by Helgaker, Olsen, and Jørgensen.³⁰

2.3.1. Second Quantization in the Spin-Orbital Basis. In second quantization, the elementary operators are the creation and annihilation operators a_P^{\dagger} and a_P , respectively. The annihilation operator a_P annihilates an electron in spin orbital ψ_P , whereas the creation operator a_P^{\dagger} creates an electron in the same spin orbital. In a basis of orthonormal spin orbitals, these operators satisfy the anticommutation relations

$$[a_{p}^{\dagger}, a_{Q}^{\dagger}]_{+} = 0 \tag{77}$$

$$[a_{P}, a_{Q}]_{+} = 0 \tag{78}$$

$$[a_P^\dagger, a_Q]_+ = \delta_{PQ} \tag{79}$$

From these simple relations all other algebraic properties of the second-quantization formalism follow. In a nonorthonormal basis of spin orbitals, the anticommutation relations would be the same except for the last relation, which becomes

$$[a_{P}^{\dagger}, a_{Q}]_{+} = \langle \psi_{Q} | \psi_{P} \rangle \tag{80}$$

where $\langle \psi_O | \psi_P \rangle$ is the overlap between the two spin orbitals.

In the Fock space generated by an orthonormal basis of M spin orbitals ψ_P , the basis vectors are the occupation-number vectors $|\mathbf{k}\rangle$, where k_P is the occupation number of spin orbital P in $|\mathbf{k}\rangle$. The simplest such state is the vacuum state $|vac\rangle$, which contains no electrons, that is, all occupation numbers k_P are zero. Any occupation-number vector can be generated from the vacuum state by application of a string of creation operators to the vacuum state

$$|\mathbf{k}\rangle = [\prod_{k=1}^{M} (a_{p}^{\dagger})^{k_{p}}] |\text{vac}\rangle$$
(81)

whereas application of an annihilation operator to the vacuum state yields

$$a_p |\mathrm{vac}\rangle = 0$$
 (82)

A general *N*-electron state is represented as a linear combination of such occupation-number vectors, each containing *N* electrons. By applying strings of creation and annihilation operators to a given *N*-electron state, we generate new states, possibly containing more or fewer particles, the anticommutation relations in eqs 77-79 ensuring that all such manipulations are consistent with the Pauli principle.

An important class of operators is the number-conserving operators, containing the same number of creation and annihilation operators, in particular, the single- and double-excitation operators $a_P^+ a_Q$ and $a_P^+ a_Q^+ a_R a_S$, in terms of which we may, for example, construct the nonrelativistic field-free Hamiltonian operator

$$\hat{H} = \sum_{PQ} h_{PQ} a_P^{\dagger} a_Q + \frac{1}{2} \sum_{PQRS} g_{PQRS} a_P^{\dagger} a_R^{\dagger} a_S a_Q + h_{\text{nuc}}$$
(83)

where

$$h_{PQ} = \int \psi_P^*(\mathbf{x}) \left(-\frac{1}{2} \nabla^2 - \sum_I \frac{Z_I}{r_I} \right) \psi_Q(\mathbf{x}) d\mathbf{x}$$
(84)

$$g_{PQRS} = \iint \frac{\psi_P^*(\mathbf{x}_1)\psi_R^*(\mathbf{x}_2)\psi_Q(\mathbf{x}_1)\psi_S(\mathbf{x}_2)}{r_{12}} \, \mathrm{d}\mathbf{x}_1 \, \mathrm{d}\mathbf{x}_2 \qquad (85)$$

$$h_{\rm nuc} = \frac{1}{2} \sum_{I \neq J} \frac{Z_I Z_J}{R_{IJ}} \tag{86}$$

The form of this second-quantization operator may be interpreted in the following way: When applied to an electronic state, the Hamiltonian produces a linear combination of the original state with states generated by single- and double-electron excitations from this state; with each such excitation there is an associated probability amplitude h_{PQ} and g_{PQRS} calculated from spin orbitals according to eqs 84 and 85, respectively. The expectation value with respect to the state $|0\rangle$ is given by

$$\langle 0|H|0\rangle = \sum_{PQ} D_{PQ} h_{PQ} + \frac{1}{2} \sum_{PQRS} d_{PQRS} g_{PQRS} + h_{nuc} \quad (87)$$

where the one- and two-electron density-matrix elements

$$D_{PQ} = \langle 0 | a_P^{\dagger} a_Q | 0 \rangle \tag{88}$$

$$d_{PQRS} = \langle 0 | a_P^{\dagger} a_R^{\dagger} a_S a_Q | 0 \rangle \tag{89}$$

are expectation values of the excitation operators.

2.3.2. Second Quantization in the Orbital Basis. In eq 87, the energy is calculated in a basis of spin orbitals. Often it is more convenient to work in a basis of orthonormal orbitals, typically molecular orbitals (MOs) expanded in AOs. A general spin orbital may be written as a linear combination of two orbitals multiplied by spin functions

$$\psi_P(\mathbf{r}, m_s) = \phi_P^{\alpha}(\mathbf{r})\alpha(m_s) + \phi_P^{\beta}(\mathbf{r})\beta(m_s)$$
(90)

where the spin functions α and β are eigenfunctions of the totaland projected-spin angular-momentum operators

$$S^{2}\alpha(m_{\rm s}) = \frac{3}{4}\alpha(m_{\rm s}), \quad S_{\rm z}\alpha(m_{\rm s}) = \frac{1}{2}\alpha(m_{\rm s}) \tag{91}$$

$$S^{2}\beta(m_{\rm s}) = \frac{3}{4}\beta(m_{\rm s}), \ S_{\rm z}\beta(m_{\rm s}) = -\frac{1}{2}\beta(m_{\rm s})$$
 (92)

and where $m_{\rm s} = \pm 1/2$ is a discrete spin coordinate such that $\alpha(1/2) = \beta(-1/2) = 1$ and $\alpha(-1/2) = \beta(1/2) = 0.^{30}$ In nonrelativistic theory, it is common to use spin orbitals in the more restricted form

$$\psi_{p\sigma}(\mathbf{r}, \, m_{\rm s}) = \phi_p(\mathbf{r})\sigma(m_{\rm s}) \tag{93}$$

so that a given spin orbital consists of an orbital multiplied by a spin function. Spin orbitals with different spins are orthogonal.

In the orbital basis the elementary creation and annihilation operators are $a_{p\sigma}^{\dagger}$ and $a_{p\sigma}$, from which all other operators can be constructed. It is then convenient to classify all second quantization operators according to their spin properties. A spin-tensor operator of integral or half-integral rank *S* is a set of 2S + 1 operators $T^{S,M}$, where *M* runs from -S to *S* in unit increments and which fulfill the relations

$$[S_{\pm}, T^{S, M}] = \sqrt{S(S+1) - M(M\pm 1)}T^{S, M\pm 1}$$
(94)

$$[S_{z}, T^{S, M}] = MT^{S, M}$$
(95)

where we assume that $T^{S,S+1} = T^{S,-S-1} = 0$. A tensor operator working on the vacuum state generates a set of spin eigenfunctions with total and projected spins *S* and *M* (provided it does not annihilate the vacuum state). Two examples of doublet spin-tensor operators $\{T^{1/2,1/2}, T^{1/2,-1/2}\}$ are the creation operators $\{a_{p\alpha}^{+}, a_{p\beta}^{+}\}$ and the annihilation operators $\{-a_{p\beta}, a_{p\alpha}\}$. Important singlet and triplet spin-tensor operators $\{T^{0,0}\}$ and $\{T^{1,1}, T^{1,0}, T^{1,-1}\}$ are the single-excitation operators

$$S_{pq}^{0, 0} = \frac{1}{\sqrt{2}} (a_{p\alpha}^{\dagger} a_{q\alpha} + a_{p\beta}^{\dagger} a_{q\beta})$$
(96)

and the triplet single-excitation operators

$$T_{pq}^{1,\ 1} = -a_{p\alpha}^{\dagger}a_{q\beta} \tag{97}$$

$$T_{pq}^{1, 0} = \frac{1}{\sqrt{2}} \left(a_{p\alpha}^{\dagger} a_{q\alpha} - a_{p\beta}^{\dagger} a_{q\beta} \right)$$
(98)

$$T_{pq}^{1,-1} = a_{p\beta}^{\dagger} a_{q\alpha} \tag{99}$$

From the triplet spin-tensor operators we may, for example, construct the spin-shift and spin-projection operators $S_{+} = -\sum_{p} T_{pp}^{1,1}$, $S_{-} = \sum_{p} T_{pp}^{1,-1}$, and $S_{z} = (1/2)^{1/2} \sum_{p} T_{pp}^{1,0}$. For the

representation of the electronic Hamiltonian, we shall use the

$$E_{pq} = \sqrt{2}S_{pq}^{0, 0} = a_{p\alpha}^{\dagger}a_{q\alpha} + a_{p\beta}^{\dagger}a_{q\beta}$$
(100)

and also the triplet operators in the Cartesian form

singlet excitation operators in the form

$$T_{pq} = \begin{pmatrix} T_{pq}^{x} \\ T_{pq}^{y} \\ T_{pq}^{z} \end{pmatrix} = \begin{pmatrix} \frac{1}{2}(a_{p\alpha}^{\dagger}a_{q\beta} + a_{p\beta}^{\dagger}a_{q\alpha}) \\ \frac{1}{2i}(a_{p\alpha}^{\dagger}a_{q\beta} - a_{p\beta}^{\dagger}a_{q\alpha}) \\ \frac{1}{2}(a_{p\alpha}^{\dagger}a_{q\alpha} - a_{p\beta}^{\dagger}a_{q\beta}) \end{pmatrix}$$
$$= \begin{pmatrix} -\frac{1}{2} & \frac{1}{2} & 0 \\ -\frac{1}{2i} & -\frac{1}{2i} & 0 \\ 0 & 0 & \frac{1}{\sqrt{2}} \end{pmatrix} \begin{pmatrix} T_{pq}^{1,1} \\ T_{pq}^{1,-1} \\ T_{pq}^{1,0} \end{pmatrix}$$
(101)

In terms of these excitation operators the second-quantization representation of the Breit–Pauli Hamiltonian of eqs 38–44 becomes

$$H^{\rm NR} = \sum_{pq} \left\langle p \left| \frac{1}{2} \pi^2 - \phi \right| q \right\rangle E_{pq} + \sum_{pq} \mathbf{B} \cdot \mathbf{T}_{pq} + \frac{1}{2} \sum_{pqrs} \left\langle pr | r_{12}^{-1} | qs \right\rangle (E_{pq} E_{rs} - \delta_{qr} E_{ps})$$
(102)

$$H^{\rm MV} = -\frac{\alpha^2}{8} \sum_{pq} \langle p | \pi^4 + B^2 | q \rangle E_{pq} - \frac{\alpha^2}{2} \sum_{pq} \langle p | \pi^2 | q \rangle \mathbf{B} \cdot \mathbf{T}_{pq}$$
(103)

$$H^{\rm DW} = \frac{\alpha^2}{8} \sum_{pq} \langle p | (\nabla \cdot \mathbf{E}) | q \rangle E_{pq}$$
$$- \frac{\alpha^2 \pi}{2} \sum_{pqrs} \langle pr | \delta(r_{12}) | qs \rangle (E_{pq} E_{rs} - \delta_{qr} E_{ps}) \qquad (104)$$

$$H^{\text{SsO}} = \frac{\alpha^2}{4} \sum_{pq} \langle p | \mathbf{E} \times \boldsymbol{\pi} - \boldsymbol{\pi} \times \mathbf{E} | q \rangle \cdot \mathbf{T}_{pq} + \frac{\alpha^2}{4} \sum_{pqrs} \langle pr | (\nabla_1 r_{12}^{-1}) \times \boldsymbol{\pi}_1 | qs \rangle \cdot (\mathbf{T}_{pq} E_{rs} - \delta_{qr} \mathbf{T}_{ps})$$
(105)

$$H^{\text{SoO}} = -\frac{\alpha^2}{2} \sum_{pqrs} \langle pr | (\nabla_1 r_{12}^{-1}) \times \boldsymbol{\pi}_2 | qs \rangle \cdot (\mathbf{T}_{pq} E_{rs} - \delta_{qr} \mathbf{T}_{ps})$$
(106)

$$H^{\rm SS} = \frac{\alpha^2}{8} \sum_{pqrs} \sum_{\mu\nu} \langle pr | [(\mathbf{I}_3 \nabla_1^2 - \nabla_1 \nabla_1^{\rm T}) r_{12}^{-1}]_{\mu\nu} | qs \rangle (4T_{pq}^{\mu} T_{rs}^{\nu} - \delta_{\mu\nu} \delta_{qr} E_{ps})$$
(107)

$$4 \sum_{pqrs} \langle pr | r_1 | (r_3 \vee_1 \vee_1 \vee_1 \vee_1 \vee_1) r_{12} r_2 | q_2 / (L_{pq} L_{rs} \circ_{qr} L_{ps}) \rangle$$
(108)

For a discussion of these operators and their integral evaluation, see Coriani et al. 34

2.4. Perturbation-Dependent Basis Sets

In second quantization, the electronic Hamiltonian is expressed in terms of one- and two-electron integrals over MOs expanded in AOs, see eqs 102–108. In calculations of molecular properties, the use of this Hamiltonian presents a complication not encountered in energy calculations, namely, that the AOs employed often depend explicitly on the external parameters representing the perturbations. In this section we discuss the second-quantization Hamiltonian with emphasis on the consequences of such perturbation-dependent basis sets.

2.4.1. Perturbation-Dependent Atomic Orbitals. In calculations of molecular properties the AOs often depend explicitly on the external parameters that describe the system. Here, the AOs are taken to be solid-harmonic Gaussian functions with angular-momentum quantum numbers l and m and exponent a fixed to an atomic position K and equipped with a complex phase factor that depends on the external magnetic field **B** and the gauge origin **O**⁴⁴ in the manner

$$\chi_{lm}(\mathbf{r}; a, \mathbf{K}, \mathbf{B}, \mathbf{O}) = e^{i(1/2)\mathbf{B} \times \mathbf{O}_{K} \cdot \mathbf{r}} S_{lm}(\mathbf{r}_{K}) e^{-ar_{K}^{2}}$$
(109)

where $O_K = O - K$, $\mathbf{r}_K = \mathbf{r} - \mathbf{K}$, and $S_{lm}(\mathbf{r}_K)$ is a solid-harmonic function. By fixing the AOs to the nuclei we obtain a balanced treatment of the electronic system at different molecular geometries, introducing into the AOs the observed strong coupling of the atomic charge distributions to the nuclei. By including a field-dependent phase factor in the AOs we ensure that our results become independent of the gauge origin **O** and furthermore improve basis-set convergence by introducing into the AOs the correct first-order response of the electrons to an applied external magnetic field, as discussed in section 2.4.5. We note here that AOs with an explicit dependence on the external electric field have also been proposed^{45,46} but are much less widespread, as the added computational complexity does not compensate for the rather moderate improvements in basis-set convergence observed.

Use of perturbation-dependent AOs is essential for reducing the AO basis sets to a manageable size while maintaining high accuracy in the calculations. However, their use also introduces certain complications, in particular, in the evaluation of molecular integrals. Moreover, since the AOs change with the external parameters ε , we are in fact at each value of ε faced with a different second-quantization representation $H(\varepsilon)$ of the Hamiltonian. To calculate molecular properties with such basis sets we must establish a connection between Hamiltonians at different values of ε .⁴⁷

2.4.2. Orbital Connections. Consider the second-quantization representation of an electronic Hamiltonian in a basis of perturbation-dependent AOs $\chi_{\mu}(\mathbf{r};\varepsilon)$, where ε represents the external parameters (e.g., the molecular geometry or the magnetic field strength) upon which the AOs in eq 109 depend. For the unperturbed system, for which we conventionally set $\varepsilon = 0$, we calculate a set of orthonormal MOs

$$\phi_p(\mathbf{r}; \mathbf{0}) = \sum_{\mu} C_{p\mu}^{(0)} \chi_{\mu}(\mathbf{r}; \mathbf{0})$$
(110)

from which the Hamiltonian operator is constructed in the usual manner

$$H(0) = \sum_{pq} h_{pq}(0) E_{pq}(0) + \frac{1}{2} \sum_{pqrs} g_{pqrs}(0) [E_{pq}(0) E_{rs}(0) - \delta_{rs} E_{ps}(0)]$$
(111)

The one- and two-electron integrals are obtained by integration over the orthonormal MOs in eq 110

$$h_{pq}(0) = \langle \phi_p(0) | h_1(0) | \phi_q(0) \rangle \tag{112}$$

$$g_{pqrs}(0) = \langle \phi_p(0)\phi_r(0)|r_{12}^{-1}|\phi_q(0)\phi_s(0)\rangle$$
(113)

whereas the excitation operators

$$E_{pq}(0) = a_{p\alpha}^{\dagger}(0)a_{q\alpha}(0) + a_{p\beta}^{\dagger}(0)a_{q\beta}(0)$$
(114)

are constructed in the usual manner from the creation and annihilation operators that obey the anticommutation relations in eqs 77-79.

Let us now consider a perturbed system with $\varepsilon \neq 0$. For this system we would like to set up a second-quantization representation that connects smoothly with the representation of the unperturbed system.^{48,49} We define a set of nonorthogonal unmodified MOs (UMOs) by retaining the MO coefficients of the unperturbed system

$$\phi_p(\mathbf{r};\varepsilon) = \sum_{\mu} C_{p\mu}^{(0)} \chi_{\mu}(\mathbf{r};\varepsilon)$$
(115)

This UMO basis connects smoothly with the unperturbed basis of eq 110 at $\varepsilon = 0$ but is nonorthonormal at $\varepsilon \neq 0$ since the overlap integrals of the AOs depend on the perturbation

$$S_{pq}(\varepsilon) = \langle \phi_p(\varepsilon) | \phi_q(\varepsilon) \rangle$$

= $\sum_{\mu\nu} C_{p\mu}^{(0)} C_{q\nu}^{(0)} \langle \chi_\mu(\varepsilon) | \chi_\nu(\varepsilon) \rangle \neq \delta_{pq}$ (116)

We therefore cannot represent our Hamiltonian in this basis at $\varepsilon \neq 0$ without affecting the algebra of the creation and annihilation operators, having to replace the simple anticommutation relation in eq 79 with the more complicated relation in eq 80. To avoid this complication we construct a set of orthonormalized MOs (OMOs) by an orthonormalization of the UMOs

$$\tilde{\phi}_{p}(\varepsilon) = \sum_{q} \phi_{q}(\varepsilon) T_{qp}(\varepsilon)$$
(117)

where $\mathbf{T}(\varepsilon)$ is the connection matrix, $^{47-49}$ taken to satisfy the relation

$$\Gamma^{\dagger}(\varepsilon)\mathbf{S}(\varepsilon)\mathbf{T}(\varepsilon) = \mathbf{I}$$
(118)

The connection matrix can be chosen in infinitely many ways, each of which establishes a particular orbital connection between orthonormalized MOs at different ε . For example, by decomposing **T** into its real and imaginary parts $\mathbf{T} = \mathbf{T}^{\mathrm{R}} + i\mathbf{T}^{\mathrm{I}}$ with $T_{pq}^{\mathrm{R}} = 0$ for p > q and $T_{pq}^{\mathrm{I}} = 0$ for $p \ge q$ we arrive at the Gram–Schmidt connection.^{47,50} A physically better motivated connection is obtained by requiring the OMOs $\tilde{\phi}_p(\mathbf{r};\varepsilon)$ to resemble as closely as possible some target functions $f_p(\mathbf{r};\varepsilon)$. Introducing the overlap matrix between the UMOs and the target functions

$$W_{pq}(\varepsilon) = \langle f_p(\varepsilon) | \phi_q(\varepsilon) \rangle \tag{119}$$

it can be shown that the connection matrix is given by

$$\mathbf{T}(\varepsilon) = \mathbf{W}^{-1}(\varepsilon) [\mathbf{W}(\varepsilon) \mathbf{S}^{-1}(\varepsilon) \mathbf{W}^{\dagger}(\varepsilon)]^{1/2}$$
(120)

Requiring the OMOs to resemble the perturbed UMOs by setting $f_p(\mathbf{r};\varepsilon) = \phi_p(\mathbf{r};\varepsilon)$,⁴⁸ we obtain the symmetric orthonormalization⁵¹

$$\mathbf{T}(\varepsilon) = \mathbf{S}^{-1/2}(\varepsilon) \tag{121}$$

Alternatively, we may require the OMOs to resemble the unperturbed UMOs by setting $f_p(\mathbf{r};\varepsilon) = \phi_p(\mathbf{r};0)$, yielding instead the natural connection.⁴⁷ The natural connection ensures that the OMOs in terms of which the Hamiltonian is constructed change as little as possible with ε , improving numerical stability.⁵²

In the OMO basis of eq 117 we may now construct a Hamiltonian operator that is defined for all ε in the usual manner

$$H(\varepsilon) = \sum_{\widetilde{pq}} \tilde{h}_{\widetilde{p}\widetilde{q}}(\varepsilon) E_{\widetilde{p}\widetilde{q}}(\varepsilon) + \frac{1}{2} \sum_{\widetilde{pqrs}} \tilde{g}_{\widetilde{p}\widetilde{q}\widetilde{r}\widetilde{s}}(\varepsilon) [E_{\widetilde{p}\widetilde{q}}(\varepsilon) E_{\widetilde{rs}}(\varepsilon) - \delta_{\widetilde{rs}} E_{\widetilde{p}\widetilde{s}}(\varepsilon)]$$
(122)

where the OMO integrals are denoted by the tilde and may be expanded in UMO integrals

$$\begin{aligned}
\dot{h}_{\tilde{p}\tilde{q}}(\varepsilon) &= \sum_{pq} h_{pq}(\varepsilon) T_{\tilde{p}p}(\varepsilon) T_{\tilde{q}q}^{*}(\varepsilon) \\
&= \sum_{pq} \langle \phi_{p}(\varepsilon) | h_{1}(\varepsilon) | \phi_{q}(\varepsilon) \rangle T_{\tilde{p}p}(\varepsilon) T_{\tilde{q}q}^{*}(\varepsilon)
\end{aligned} \tag{123}$$

$$\begin{split} \tilde{g}_{\tilde{p}\tilde{q}\tilde{r}\tilde{s}}(\varepsilon) &= \sum_{pqrs} g_{pqrs}(\varepsilon) T_{\tilde{p}p}(\varepsilon) T_{\tilde{q}q}^*(\varepsilon) T_{\tilde{r}r}(\varepsilon) T_{\tilde{s}\tilde{s}}^*(\varepsilon) \\ &= \sum_{pqrs} \langle \phi_p(\varepsilon) \phi_r(\varepsilon) | r_{12}^{-1} | \phi_q(\varepsilon) \phi_s(\varepsilon) \rangle T_{\tilde{p}p}(\varepsilon) T_{\tilde{q}q}^*(\varepsilon) T_{\tilde{r}r}(\varepsilon) T_{\tilde{s}\tilde{s}}^*(\varepsilon) \end{split}$$

$$(124)$$

The excitation operators are constructed in the usual manner

$$E_{\tilde{p}\tilde{q}}(\varepsilon) = a^{\dagger}_{\tilde{p}\alpha}(\varepsilon)a_{\tilde{q}\alpha}(\varepsilon) + a^{\dagger}_{\tilde{p}\beta}(\varepsilon)a_{\tilde{q}\beta}(\varepsilon)$$
(125)

from the creation and annihilation operators that obey the anticommutation relations in eqs 77-79. Above, we restricted ourselves to singlet perturbations; the corresponding expressions for triplet perturbations are obtained by a simple generalization of the scheme for singlet perturbations.

2.4.3. Perturbation Dependence of the Creation and Annihilation Operators. The dependence of the creation and annihilation operators $a_P^+(\varepsilon)$ and $a_P(\varepsilon)$ on ε in eq 125 arises from the dependence of the OMOs in eq 117 on ε . However, for the purpose of calculating derivatives of electronic energies we may treat these operators as being independent of ε because the creation and annihilation operators always appear in transition expectation values such as

$$\langle \mathbf{K} | a_{P}^{\dagger} a_{Q} | \mathbf{L} \rangle = \langle \operatorname{vac} | a_{K_{1}} a_{K_{2}} \dots a_{K_{N}} a_{P}^{\dagger} a_{Q} a_{L_{N}}^{\dagger} \dots a_{L_{2}}^{\dagger} a_{L_{1}}^{\dagger} | \operatorname{vac} \rangle$$
(126)

which, by means of the anticommutation relations in eqs 77–79, may be reduced to sums of products of Kronecker deltas. Since these are independent of ε we may ignore the perturbation dependence of the creation and annihilation operators altogether

$$H(\varepsilon) = \sum_{\widetilde{pq}} \tilde{h}_{\widetilde{pq}}(\varepsilon) E_{\widetilde{pq}} + \frac{1}{2} \sum_{\widetilde{pqrs}} \tilde{g}_{\widetilde{pqrs}}(\varepsilon) (E_{\widetilde{pq}} E_{\widetilde{rs}} - \delta_{\widetilde{rs}} E_{\widetilde{ps}})$$
(127)

If instead the Hamiltonian had been constructed from nonorthonormal spin orbitals, the vacuum expectation values would reduce to sums of products of perturbation-dependent overlap integrals $S_{pq}(\varepsilon)$, making it necessary to account for the perturbation dependence of the creation and annihilation operators in calculation of energies and other properties of the system.

The above arguments are no longer valid when we consider, for example, nonadiabatic coupling matrix elements,⁵³ the intensities of electronic circular dichroism (ECD),⁵⁴ and vibrational circular dichroism (VCD).^{55,56} We are then interested in calculating elements of the form $\langle \mathbf{K} | \mathbf{d} / d\varepsilon | \mathbf{L} \rangle$, where $\langle \mathbf{K} |$ and $| \mathbf{L} \rangle$ are different electronic states. The creation and annihilation operators in the bra and ket states are no longer the same, and transition densities cannot be reduced to simple Kronecker deltas. In such calculations the perturbation dependence of the excitation operators cannot be neglected. In these cases, the natural connection provides a particularly attractive approach for calculating the relevant matrix elements.^{47,54,56}

2.4.4. One-Index Transformations. For calculation of molecular properties, we need to expand the OMO integrals in eqs 123 and 124 in orders of the perturbation. To express these expansions in a compact form we introduce the integrals (restricting ourselves to the one-electron case and omitting the argument ε for ease of presentation)

$$h_{pq}(t) = \sum_{rs} h_{rs} [\exp(t \ln \mathbf{T})]_{pr} [\exp(t \ln \mathbf{T}^*)]_{qs}$$
(128)

which include the UMO integrals (t = 0) and the OMO integrals (t = 1) as special cases. Expanding around t = 0 and setting t = 1 we obtain an expansion of the OMO integrals in the UMO integrals

$$\tilde{\mathbf{h}} = \mathbf{h} + \{\ln \mathbf{T}, \, \mathbf{h}\} + \frac{1}{2}\{\ln \mathbf{T}, \, \ln \mathbf{T}, \, \mathbf{h}\} + \dots$$
 (129)

where we introduced a brace notation for one-index transformations

$$\{\mathbf{M}, \mathbf{h}\}_{pq} = \sum_{o} (M_{po}h_{oq} + M^*_{qo}h_{po})$$
(130)

with the following notation for symmetrized multiple one-index transformations

$$\{\mathbf{M}_{1}, \mathbf{M}_{2}, \mathbf{h}\} = \frac{1}{2} \{\mathbf{M}_{1}, \{\mathbf{M}_{2}, \mathbf{h}\}\} + \frac{1}{2} \{\mathbf{M}_{2}, \{\mathbf{M}_{1}, \mathbf{h}\}\}$$
(131)

An order-by-order expansion in the perturbation now yields to second ${\rm order}^{48,57-59}$

$$\tilde{\mathbf{h}}(\varepsilon) = \mathbf{h}^{(0)} + \mathbf{h}^{(1)}\varepsilon + \frac{1}{2}\mathbf{h}^{(2)}\varepsilon^2 + \dots$$
(132)

with the expansion coefficients

$$\tilde{\mathbf{h}}^{(0)} = \mathbf{h}^{(0)}$$
 (133)

$$\tilde{\mathbf{h}}^{(1)} = \mathbf{h}^{(1)} + \left\{ \mathbf{T}^{(1)}, \, \mathbf{h}^{(0)} \right\}$$
(134)

$$\tilde{\mathbf{h}}^{(2)} = \mathbf{h}^{(2)} + \frac{1}{2} \Big\{ \mathbf{T}^{(2)} - \mathbf{T}^{(1)} \mathbf{T}^{(1)}, \, \mathbf{h}^{(0)} \Big\} \\ + \Big\{ \mathbf{T}^{(1)}, \, \mathbf{h}^{(1)} \Big\} + \Big\{ \mathbf{T}^{(1)}, \, \mathbf{T}^{(1)}, \, \mathbf{h}^{(0)} \Big\}$$
(135)

$$\ln T = (T - I) - \frac{1}{2}(T - I)^{2} + \frac{1}{3}(T - I)^{3} - \dots$$
 (136)

to re-express ln T in terms of the derivatives of T, noting that T(0) = I. The advantage of the expressions in eqs 133–135 over those obtained by a straightforward expansion of the OMO integrals in eq 123 is that they maintain the symmetry of the integrals upon differentiation. The two-electron integrals are obtained in the same manner using the one-index transformations

$$\{\mathbf{M}, \mathbf{g}\}_{pqrs} = \sum_{o} \left(M_{po}g_{oqrs} + M_{qo}^*g_{pors} + M_{ro}g_{pqos} + M_{so}^*g_{pqro} \right)$$
(137)

as a generalization of the one-index transformations for one-electron integrals in eq 130. Finally, we note that the lowest order connection matrices are given by⁴⁷

$$\mathbf{T}^{(1)} = \begin{pmatrix} -\frac{1}{2}\mathbf{S}^{(1)}, & \text{symmetric connection} \\ -\mathbf{W}^{(1)}, & \text{natural connection} \end{pmatrix}$$
(138)
$$\mathbf{T}^{(2)} = \begin{pmatrix} -\frac{1}{2}\mathbf{S}^{(2)} + \frac{3}{4}\mathbf{S}^{(1)}\mathbf{S}^{(1)}, & \text{symmetric connection} \\ -\frac{1}{2}\mathbf{S}^{(2)} - \frac{1}{4}(\mathbf{w}^{(2)\dagger} - \mathbf{w}^{(2)}) \end{pmatrix}$$

$$\Gamma^{(2)} = \begin{pmatrix} -\frac{1}{2}\mathbf{S}^{(2)} - \frac{1}{2}(\mathbf{W}^{(2)\dagger} - \mathbf{W}^{(2)}) \\ + 2\mathbf{W}^{(1)}\mathbf{W}^{(1)} + \mathbf{W}^{(1)\dagger}\mathbf{W}^{(1)}, & \text{natural connection} \end{cases}$$
(139)

where

$$W_{pq}(\varepsilon) = \langle \phi_p(0) | \phi_q(\varepsilon) \rangle \tag{140}$$

in the symmetric and natural connections.

2.4.5. London Atomic Orbitals. We recall from section 2.1.4 that, in calculations involving an externally applied uniform magnetic field **B**, the field is represented by a vector potential $A_O(r)$ that vanishes at the gauge origin **O**

$$\mathbf{B} = \nabla \times \mathbf{A}_{\mathbf{O}}(\mathbf{r}), \, \mathbf{A}_{\mathbf{O}}(\mathbf{r}) = \frac{1}{2}\mathbf{B} \times (\mathbf{r} - \mathbf{O})$$
(141)

The representation of the magnetic field is not unique, being dependent on our choice of gauge origin **O**. The translation of the gauge origin from **O** to **K** represents a gauge transformation of eqs 9 and 10

$$\mathbf{A}_{\mathbf{K}}(\mathbf{r}) = \mathbf{A}_{\mathbf{O}}(\mathbf{r}) - \mathbf{A}_{\mathbf{O}}(\mathbf{K}) = \mathbf{A}_{\mathbf{O}}(\mathbf{r}) + \nabla f(\mathbf{r})$$
(142)

with the gauge function

$$f(\mathbf{r}) = -\mathbf{A}_{\mathbf{O}}(\mathbf{K}) \cdot \mathbf{r} \tag{143}$$

In exact calculations, such gauge transformations do not affect the value of the calculated observables. To see this invariance we note that a general gauge transformation, eqs 9 and 10, with gauge function f represents a unitary transformation of the operator $H - i\partial_t$

$$H' - i\partial_t = e^{-if(r)}(H - i\partial_t)e^{if(r)}$$
(144)

In order for the Schrödinger equation to still be satisfied

$$(H' - \mathrm{i}\partial_t)\Psi'(\mathbf{r}) \Leftrightarrow (H - \mathrm{i}\partial_t)\Psi(\mathbf{r})$$
(145)

the new wave function must be related to the old one by a compensating unitary transformation

$$\Psi'(\mathbf{r}) = e^{-if(\mathbf{r})}\Psi(\mathbf{r}) \tag{146}$$

No observable properties such as the electron density are then affected by the transformation

$$\rho'(\mathbf{r}) = \Psi'(\mathbf{r})^* \Psi'(\mathbf{r}) = (\Psi(\mathbf{r})e^{-if(\mathbf{r})})^* (e^{-if(\mathbf{r})}\Psi(\mathbf{r}))$$
$$= \Psi^*(\mathbf{r})\Psi(\mathbf{r}) = \rho(\mathbf{r})$$
(147)

In exact calculations the wave function has the ability to perform the transformation in eq 146, leaving all observables unaffected. In particular, for the gauge transformation in eq 143 the exact wave function transforms as

$$\Psi_{K}^{\text{exact}}(\mathbf{r}) = e^{i\mathbf{A}_{O}(K) \cdot \mathbf{r}} \Psi_{O}^{\text{exact}}(\mathbf{r})$$
(148)

giving gauge-invariant results. By contrast, approximate wave functions are in general unable to perform this transformation

$$\Psi_{K}^{approx}(\mathbf{r}) \neq \ e^{i\mathbf{A}_{O}(K) \cdot \mathbf{r}} \Psi_{O}^{approx}(\mathbf{r}) \tag{149}$$

As a result, different gauge origins may give different results. We note that attaching an explicit phase factor to the approximate wave function

$$\Psi_{K}^{\text{approx}}(\mathbf{r}) \stackrel{\text{def}}{=} e^{i\mathbf{A}_{O}(K) \cdot \mathbf{r}} \Psi_{O}^{\text{approx}}(\mathbf{r})$$
(150)

does not solve this problem; it merely produces the same result as with the gauge origin at O.⁶⁰ However, no natural, best gauge origin can usually be identified (except for atoms). In any case, we might as well have carried out the calculation with the origin at O.

By contrast, when applied to individual AOs, this approach becomes useful, as can be seen by first assuming that we have AOs positioned at K with the properties⁵⁵

$$H_{0}\psi_{lm}^{\mathbf{K}} = E_{0}\psi_{lm}^{\mathbf{K}}, \quad L_{z}^{\mathbf{K}}\psi_{lm}^{\mathbf{K}} = m_{l}\psi_{lm}^{\mathbf{K}},$$
$$\mathbf{L}^{\mathbf{K}} = -\mathbf{i}(\mathbf{r} - \mathbf{K}) \times \nabla$$
(151)

Typical examples of such AOs are traditional field-independent spherical-harmonic Gaussians. With the gauge origin at K and vector potential $A_{K}(\mathbf{r})$ these AOs are correct to first order in B

$$H_{\mathbf{K}}(\mathbf{B})\psi_{lm}^{\mathbf{K}} = \left[H_{0} + \frac{1}{2}BL_{z}^{\mathbf{K}} + O(B^{2})\right]\psi_{lm}^{\mathbf{K}}$$
$$= \left[E_{0} + \frac{1}{2}m_{l}B + O(B^{2})\right]\psi_{lm}^{\mathbf{K}}$$
(152)

where we omitted the Zeeman spin contribution BS_z . On the other hand, with the gauge origin at $\mathbf{O} \neq \mathbf{K}$ and vector potential $\mathbf{A}_{\mathbf{O}}(\mathbf{r})$ the AOs at **K** are correct only to zero order in **B**

$$H_{\mathbf{O}}(\mathbf{B})\psi_{lm}^{\mathbf{K}} = \left[H_{0} + \frac{1}{2}BL_{z}^{\mathbf{O}} + O(B^{2})\right]\psi_{lm}^{\mathbf{K}}$$
$$\neq \left[E_{0} + \frac{1}{2}m_{l}B + O(B^{2})\right]\psi_{lm}^{\mathbf{K}}$$
(153)

Clearly, standard AOs positioned at K are biased toward the gauge origin at K in the sense that they give a better description of the electronic system with this origin than with all other gauge origins. Therefore, by attaching to each AO a phase factor that represents the transformation of the gauge origin from the global origin O to the AO center K

$$\chi_{lm}^{\mathbf{K}} = e^{i\mathbf{A}_{\mathbf{K}}(\mathbf{O})\cdot\mathbf{r}}\psi_{lm}^{\mathbf{K}} = e^{i(1/2)\mathbf{B}\times(\mathbf{O}-\mathbf{K})\cdot\mathbf{r}}\psi_{lm}^{\mathbf{K}}$$
(154)

we obtain a basis set where the AOs are correct to first order in the perturbation⁵⁵ for any gauge origin

$$H_{O}(\mathbf{B})\chi_{lm}^{\mathbf{K}} = \left[H_{0} + \frac{1}{2}BL_{z}^{\mathbf{O}} + O(B^{2})\right]\chi_{lm}^{\mathbf{K}}$$
$$= \left[E_{0} + \frac{1}{2}m_{l}B + O(B^{2})\right]\chi_{lm}^{\mathbf{K}}$$
(155)

The calculations become gauge-origin independent, and uniform (good) quality follows. These are the London orbitals (1937),^{44,61,62} also known as GIAOs (gauge-origin independent AOs or gauge-origin including AOs), widely used in molecular calculations involving an external magnetic field.

3. RESPONSE THEORY FOR EXACT STATES

In the present section we develop molecular response theory for exact wave functions, in particular, we derive response functions and study their properties. In section 4, the results obtained here for exact wave functions will be used to develop response theory for approximate electronic-structure models.

We begin our development in section 3.1 by studying various forms of the time-dependent electronic Schrödinger equation and introducing the time-dependent quasi-energy, which is a generalization of the energy of standard time-independent perturbation theory. Two formulations for the quasi-energy are developed: a formulation analogous to the standard Hermitian expectation value of the energy and a formulation based on a non-Hermitian Lagrangian. For periodic Hamiltonians it is shown that the time-dependent Schrödinger equation may be written as a variation principle for the time-averaged quasienergy. In section 3.2, the time-dependent wave function is expanded in orders of the perturbation and expressions are developed for the wave function corrections of different orders. This development is followed in section 3.3 by perturbation expansions of the time-dependent and time-averaged quasienergy in the Hermitian form. In section 3.4, the response functions associated with an observable are identified from expansion of the expectation value of the corresponding operator. It is shown how the response functions may be identified with terms in the perturbation expansion of the quasi-energy. Using the Hermitian formulation of the quasienergy, explicit expressions for the lowest order response functions are given; in section 3.5, alternative expressions for the response functions are given, based on the Lagrangian formulation of the quasi-energy.

In developing these Hermitian and Lagrangian expressions for the response functions, a generalization of Wigner's 2n + 1 rule to time-dependent theory is repeatedly invoked; this generalization is discussed in section 3.6. Finally, in section 3.7, a phenomenological treatment is given to describe the broadening of absorption spectra arising from the finite lifetimes of excited states and from interactions with other molecules.

3.1. Quasi-Energy Formulation

A convenient framework for molecular response theory is the quasi-energy formulation,^{63–66} which provides a uniform treatment of periodic time-dependent perturbations and timeindependent perturbations. In the present subsection we introduce the time-dependent and time-averaged quasi-energies and develop the variation principles for these quantities, laying the foundation for our development of response functions in later subsections. Importantly, our treatment is not restricted to the quasi-energy calculated as an expectation value from a normalized wave function; we also consider the quasi-energy calculated by projection from an intermediately normalized wave function. The general framework developed here is therefore equally applicable to variational model theories such as MCSCF theory and to nonvariational model theories such as coupled-cluster theory, both of which are treated in section 4.

3.1.1. Time-Dependent Quasi-Energy and Its Variation Principle. The time development of the electronic wave function $|0\rangle$ is determined by the time-dependent Schrödinger equation

$$H|0\rangle = i\partial_t|0\rangle \tag{156}$$

where the Hermitian Hamiltonian consists of a dominant timeindependent part H_0 and a time-dependent perturbation V(t)

$$H = H_0 + V(t)$$
(157)

From the Hermiticity of the Hamiltonian it follows from eq 156 that the norm $\langle 0|0\rangle$ is independent of time; here, we assume normalization to unity

$$0|0\rangle = 1 \tag{158}$$

In the following it will be convenient to separate the wave function into a product of a phase factor and a regular wave function $|\tilde{0}\rangle^{63}$

$$|0\rangle = e^{-iF(t)}|\tilde{0}\rangle \tag{159}$$

where the phase F(t) is a real-valued function that depends only on time. The regular wave function $|\tilde{0}\rangle$ is taken to be a normalized wave function whose overlap with a chosen time-independent normalized reference state $|R\rangle$ is real and positive: $\langle R|\tilde{0}\rangle > 0$. Neither the reference state $|R\rangle$ nor the states $|i\rangle$ are required to be eigenfunctions of H_0 . Introducing an orthonormal basis $\{|R\rangle_i |i\rangle\}$, we may write the regular wave function as

$$|\tilde{0}\rangle = \frac{|\mathbf{R}\rangle + \sum_{i} c_{i}|i\rangle}{\sqrt{1 + \sum_{i} |c_{i}|^{2}}}$$
(160)

The restriction that the overlap $\langle R|\tilde{0}\rangle$ is nonzero means that the above ansatz is not fully general. However, this restriction does not pose any problems in the current context, where we are only interested in perturbation expansions rather than in the explicit time development for an arbitrary time-dependent perturbation. We note that the separation in eq 159 is not unique, depending on our choice of regular wave function and phase factor. Consider an alternative separation, $|0\rangle = e^{-iF'(t)}|\tilde{0}'\rangle$, different from that in eq 159. Comparing with eq 159, we find that $|\tilde{0}\rangle$ and $|\tilde{0}'\rangle$ are related by a time-dependent phase factor

$$|\tilde{0}'\rangle = e^{i(F'(t) - F(t))}|\tilde{0}\rangle$$
(161)

We shall later see that this nonuniqueness does not affect the central quantity of response theory: the time-averaged quasi-energy.

<

Let us now consider the Schrödinger equation for the phaseseparated wave function. By substituting eq 159 into eq 156 we obtain the time-dependent Schrödinger equation for $|\tilde{0}\rangle$ and F(t)

$$(H - i\partial_t - \dot{F}(t))|\tilde{0}\rangle = 0$$
 (162)

Introducing in eq 162 the projection operator P_x for the space orthogonal to x

$$P_{\rm x} = 1 - \frac{|{\rm x}\rangle\langle{\rm x}|}{\langle{\rm x}|{\rm x}\rangle} \tag{163}$$

we arrive at the time-dependent Schrödinger equation for the regular wave function

$$P_{\tilde{0}}(H - i\partial_t)|\tilde{0}\rangle = 0 \tag{164}$$

If we instead project eq 162 onto $|\tilde{0}\rangle$, we obtain an equation for the phase factor

$$Q(t) = \dot{F}(t) = \langle \tilde{0} | H - i\partial_t | \tilde{0} \rangle$$
(165)

By first solving eq 164 for the regular wave function and subsequently determining the phase factor by integrating eq 165 we obtain the full solution to the Schrödinger equation in eq 156. In the timeindependent case, with V(t) = 0 and $H_0|0\rangle = E_0|0\rangle$, we obtain Q(t) = E_0 and hence $F(t) = E_0 t + k$. For this reason Q(t) is known as the time-dependent quasi-energy. To distinguish the expression for the time-dependent quasi-energy in eq 165 from other expressions derived later, this form is termed the Hermitian form of the timedependent quasi-energy. The time-dependent quasi-energy is a real quantity, being the derivative of the real function F(t); alternatively, the real valuedness of Q(t) follows explicitly from the Hermiticity of H and the constant norm of $|0\rangle$ in eq 165. The quasi-energy is not a unique quantity; it depends on the choice of the regular wave function and phase factor: Q(t) = F(t). However, in section 3.1.2 it is demonstrated that, for periodic perturbations, the time-averaged quasi-energy is uniquely defined.

For the development of response theory we need to characterize the quasi-energy and regular wave function in terms of a variation principle. From eq 162 we obtain the Frenkel variation principle⁶⁷ for an arbitrary variation $|\delta \tilde{0}\rangle$ in the regular wave function

$$\langle \delta \tilde{0} | H - i \partial_t - Q(t) | \tilde{0} \rangle = 0$$
 (166)

Taking the complex conjugate of eq 166 and adding it to the same equation we obtain

$$\delta \langle \tilde{0} | H - i \partial_t | \tilde{0} \rangle + i \frac{d}{dt} \langle \tilde{0} | \delta \tilde{0} \rangle = 0$$
(167)

where we used $\delta \langle \tilde{0} | Q(t) | \tilde{0} \rangle = Q(t) \delta \langle \tilde{0} | \tilde{0} \rangle = 0$, which follows from the fixed normalization of the regular wave function. To simplify eq 167 it is useful to separate a general variation $|\delta \tilde{0} \rangle$ into a component $|\delta \tilde{0}^{\perp} \rangle$ orthogonal to $|\tilde{0} \rangle$ and a component $i\alpha | \tilde{0} \rangle$ parallel to $|\tilde{0} \rangle$

$$|\delta\tilde{0}\rangle = |\delta\tilde{0}^{\perp}\rangle + i\alpha|\tilde{0}\rangle \tag{168}$$

From the normalization of the regular wave function it follows that the amplitude α in eq 168 is real. Inserting eq 168 into eq 167, we obtain the variation principle

$$\begin{split} \langle \tilde{0} | H - i\partial_t | \delta \tilde{0}^{\perp} \rangle + \langle \delta \tilde{0}^{\perp} | H - i\partial_t | \tilde{0} \rangle + \dot{\alpha} \langle \tilde{0} | \tilde{0} \rangle - \dot{\alpha} \langle \tilde{0} | \tilde{0} \rangle \\ &= \langle \tilde{0} | H - i\partial_t | \delta \tilde{0}^{\perp} \rangle + \langle \delta \tilde{0}^{\perp} | H - i\partial_t | \tilde{0} \rangle = 0 \end{split}$$
(169

similar to the standard variation principle of time-independent quantum mechanics. The variation principles in eqs 167 and 169 are equivalent, and either may be used to determine the time-dependent wave function. In particular, eq 167 constitutes the time-dependent variation principle of Langhoff, Epstein, and Karplus.⁶³ Note that it is not a variation principle in the usual sense in which the variation in the function to be determined vanishes at the solution. To establish such a variation principle we shall henceforth restrict our attention to periodic perturbations.

3.1.2. Time-Averaged Quasi-Energy and Its Variation Principle. Let us now consider a periodic time-dependent perturbation of period T and frequency ω

$$V(t + T) = V(t), \ \omega = \frac{2\pi}{T}$$
(170)

noting that this periodicity does not require all components of V(t) to oscillate with the same frequency ω , only to oscillate with a frequency $n\omega$, where n is an integer (negative, zero, or positive). It may be shown that the regular wave function oscillates with the same period T

$$|\tilde{0}(t+T)\rangle = |\tilde{0}(t)\rangle \tag{171}$$

For such periodic perturbations, we introduce the time-averaged quasi-energy or simply the quasi-energy as

$$\mathcal{Q} = \{\mathbf{Q}\}_T = \{\langle \tilde{\mathbf{0}} | H - \mathbf{i} \partial_t | \tilde{\mathbf{0}} \rangle \}_T$$
(172)

where the time average of a periodic function g(t) is defined by

$$\{g\}_{T} = \frac{1}{T} \int_{0}^{T} g(t) dt$$
 (173)

Importantly, the time-averaged quasi-energy \mathcal{Q} is independent of our choice of regular wave function. To see this, let \mathcal{Q}' and \mathcal{Q} be associated with the two representations $|\tilde{0}'\rangle$ and $|\tilde{0}\rangle$, which are connected via eq 161. From eqs 161 and 171 it follows that F'(t) - F(t) and hence $\dot{F}'(t) - \dot{F}(t)$ are periodic in t, yielding

$$\mathcal{Q}' - \mathcal{Q} = \{ Q'(t) - Q(t) \}_T = \{ \dot{F}'(t) - \dot{F}(t) \}_T = 0$$
(174)

demonstrating that the choice of regular wave function does not affect the time-averaged quasi-energy.

Let us now consider the variation principle for the timeaveraged quasi-energy. Taking the time average of the variation principle in eq 167 and invoking the periodicity of $|\tilde{0}\rangle$ and $|\delta \tilde{0}\rangle$, we obtain the variation principle for the quasi-energy⁶³

$$\delta \mathcal{Q} = 0 \tag{175}$$

Conversely, to see that the time-dependent variation principle in eq 169 follows from that in eq 175 we introduce eq 168 into eq 175, obtaining

$$\{\langle \tilde{0}|H - i\partial_t |\delta \tilde{0}^{\perp}\rangle\}_T + \{\langle \delta \tilde{0}^{\perp}|H - i\partial_t |\tilde{0}\rangle\}_T + \{\dot{\alpha}\}_T = 0$$
(176)

As α is periodic, the term $\{\dot{\alpha}\}_T$ vanishes. Furthermore, as $|\delta \tilde{0}^{\perp}\rangle$ may have an arbitrary time dependence in the interval [0,T], eq 176 shows that $\langle \tilde{0}|H-i\partial_t|\delta \tilde{0}^{\perp}\rangle + \langle \delta \tilde{0}^{\perp}|H-i\partial_t|\tilde{0}\rangle$ vanishes at all times, giving eq 169. For periodic perturbations, the timedependent variation principles in eqs 167 and 169 are thus equivalent to the time-averaged variation principle of eq 175.

Assume next that the Hamiltonian contains a term that depends on the perturbation strength ε . By differentiating the

quasi-energy of eq 172 and using the stationary condition in eq 175 we obtain

$$\frac{\mathrm{d}\mathscr{Q}}{\mathrm{d}\varepsilon} = \{\langle \tilde{0} | \partial_{\varepsilon} H | \tilde{0} \rangle \}_{T} + \{\langle \partial_{\varepsilon} \tilde{0} | H - \mathrm{i} \partial_{t} | \tilde{0} \rangle \}_{T}
+ \{\langle \tilde{0} | H - \mathrm{i} \partial_{t} | \partial_{\varepsilon} \tilde{0} \rangle \}_{T}
= \{\langle \tilde{0} | \partial_{\varepsilon} H | \tilde{0} \rangle \}_{T} + \delta_{\mathscr{Q}} |_{\partial \tilde{0} = \partial_{\varepsilon} \tilde{0}} = \{\langle \tilde{0} | \partial_{\varepsilon} H | \tilde{0} \rangle \}_{T}$$
(177)

which constitutes a generalization of the Hellmann–Feynman theorem^{68,69} to time-dependent perturbations.⁶³ The quasi-energy variation principle eq 175 and the associated Hellmann–Feynman theorem in eq 177 will play an important role in our development of perturbation theory; formally, they are equivalent to the variation principle and Hellmann–Feynman theorem for time-independent perturbations, respectively, greatly simplifying development of response theory.

3.1.3. Quasi-Energy Lagrangian and Its Variation Principle. In the theory developed in sections 3.1.1 and 3.1.2 the time-dependent quasi-energy was calculated as an expectation value according to eq 165 from the explicitly normalized wave function $|\tilde{0}\rangle$ in eq 160. Often a different formulation is more convenient, where the energy is calculated by projection rather than as an expectation value, from an intermediately normalized wave function $|\hat{0}\rangle$ with unit overlap with the reference state

$$\langle \mathbf{R}|\hat{\mathbf{0}}\rangle = 1 \tag{178}$$

Such an approach is taken in many-body perturbation theory and in coupled-cluster theory, for example. By analogy with the normalized wave function $|\tilde{0}\rangle$ in eq 160 the intermediately normalized wave function is expanded in the orthonormal basis $\{|\mathbf{R}\rangle, |i\rangle\}$

$$|\hat{0}\rangle = |\mathbf{R}\rangle + \sum_{i} c_{i}|i\rangle$$
 (179)

and is related to the regular wave function in eq 160 by a real normalization constant $N_{\rm R}$

$$\begin{split} |\tilde{0}\rangle &= N_{\rm R} |\hat{0}\rangle, \\ N_{\rm R} &= \langle {\rm R} |\tilde{0}\rangle = \left(1 + \sum_{i} |c_i|^2\right)^{-1/2} = \langle \hat{0} |\hat{0}\rangle^{-1/2} \end{split} \tag{180}$$

As the normalization constant $N_{\rm R}$ varies with time, the intermediately normalized wave function does not fulfill the usual time-dependent Schrödinger equation. Instead, inserting eq 180 into eq 162 we obtain the following Schrödinger equation for the intermediately normalized wave function

$$\left(H - i\partial_t - Q(t) - i\frac{\dot{N}_R(t)}{N_R(t)}\right)|\hat{0}\rangle = 0$$
(181)

where the quasi-energy Q(t) is defined in terms of $|0\rangle$, see eq 165. To express Q(t) in terms of $|0\rangle$ we multiply eq 181 with $\langle R|$ from the left. From the intermediate normalization in eq 178 we obtain

$$\langle \mathbf{R}|H|\hat{\mathbf{0}}\rangle - Q(t) - \mathrm{i}\frac{N_R(t)}{N_R(t)} = 0$$
(182)

Given that $iN_R(t)/N_R(t)$ is imaginary and Q(t) is real we may identify the time-dependent quasi-energy in the intermediate normalization with the real part of the complex quantity $\langle R|H|\hat{0}\rangle$

$$Q(t) = \operatorname{Re}\langle \mathbf{R}|H|\hat{\mathbf{0}}\rangle \tag{183}$$

where $|\hat{0}\rangle$ is a solution to the time-dependent Schrödinger equation in eq 181. The form of the time-dependent quasienergy given in eq 183 is termed the non-Hermitian form in contrast to the Hermitian form of eq 165.

Let us now consider the solution of the Schrödinger equation in the intermediate normalization. Using the definition of the time-dependent quasi-energy in eq 165 and the relation between the regular and the intermediately normalized wave functions in eq 180 it follows from projection of $|\hat{0}\rangle$ onto the time-dependent Schrödinger equation, eq 181, that

$$\langle \hat{0}|H - \mathrm{i}\partial_t - Q(t) - \mathrm{i}rac{\dot{N}_{\mathrm{R}}(t)}{N_{\mathrm{R}}(t)}|\hat{0}
angle = N_{\mathrm{R}}^{-1}\langle \hat{0}|H - \mathrm{i}\partial_t - Q(t)|\tilde{0}
angle$$

$$= N_{\rm R}^{-2} \langle \tilde{0} | H - i\partial_t - Q(t) | \tilde{0} \rangle = 0$$
(184)

We therefore need to solve eq 181 only in the space orthogonal to the intermediately normalized wave function. Projecting the Schrödinger equation in eq 181 from the left with the projector of eq 163 with $x = \hat{0}$ we obtain the following Schrödinger equation for the intermediately normalized wave function

$$P_{0}(H - i\partial_{t})|\hat{0}\rangle = 0$$
(185)

similar to that for the regular wave function in eq 164. Alternatively, we can use the coefficients c of eq 179 to introduce a nonorthogonal basis $|\hat{i}\rangle$ for the space orthogonal to $|\hat{0}\rangle$

$$|\hat{i}\rangle = |i\rangle - c_i^* |\mathbf{R}\rangle, \, \langle \hat{i} |\hat{\mathbf{0}}\rangle = 0$$
 (186)

and write the projected Schrödinger equation as

$$\langle \hat{i}|H - i\partial_t|\hat{0}\rangle = 0$$
 (187)

To establish a variation principle for the quasi-energy in the intermediate normalization we note that the quasi-energy is calculated from eq 183 subject to the constraint that the intermediately normalized state satisfies the projected Schrödinger equation in eq 187. Invoking Lagrange's method we construct a Lagrangian by adding to $\langle R|H|\hat{0}\rangle$ the constraints in eq 187 multiplied by the undetermined multipliers \bar{c}_i . Introducing the notation

$$\langle \overline{0} | = \sum_{i} \overline{c}_{i}^{*} \langle \hat{i} | \tag{188}$$

we obtain the time-dependent complex-valued quasi-energy Lagrangian

$$L^{c}(t) = \langle \mathbf{R} | H | \hat{\mathbf{0}} \rangle + \langle \overline{\mathbf{0}} | H - \mathbf{i} \partial_{t} | \hat{\mathbf{0}} \rangle$$
(189)

By taking the real part of $L^{c}(t)$ we obtain the time-dependent quasi-energy Lagrangian

$$L(t) = \operatorname{Re}(\langle \mathbf{R}|H|\hat{\mathbf{0}}\rangle + \langle \overline{\mathbf{0}}|H - \mathrm{i}\partial_t|\hat{\mathbf{0}}\rangle)$$
(190)

which is equal to the quasi-energy Q(t) in eq 183 when the projected equations in eq 187 are satisfied. The Lagrangian is required to be stationary with respect to variations of the Lagrange multipliers \bar{c}_{ij} which implies eq 187. To show this implication, we assume stationarity of eq 190 with respect to variations in the real and imaginary parts of the multipliers, $\bar{c}_i = \bar{c}_i^{\rm R} + i\bar{c}_i^{\rm I}$ yielding the stationary conditions

$$\frac{\partial L(t)}{\partial \overline{c}^{\mathrm{R}}} = \mathrm{Re}\langle \hat{i} | H - \mathrm{i}\partial_t | \hat{\mathbf{0}} \rangle = 0$$
(191)

which are equivalent to eq 187.

The stationary conditions above are to be solved at each time *t*. By analogy with eq 175, we now introduce the time-averaged quasi-energy Lagrangian

$$\mathcal{L} = \{L\}_T$$
$$= \operatorname{Re}(\{\langle R|H|\hat{0}\rangle\}_T + \{\langle \overline{0}|H - i\partial_t|\hat{0}\rangle\}_T)$$
(193)

where $\langle \overline{0} | \overline{0} \rangle = 0$, as follows from eqs 186 and 188. Henceforth, \mathcal{L} will be called the quasi-energy Lagrangian or the non-Hermitian form of the quasi-energy. The Lagrange multipliers and the intermediately normalized state are now determined from the time-averaged stationary condition

$$\delta \mathscr{J} = 0 \tag{194}$$

corresponding to the stationary conditions

$$\frac{\partial \mathcal{L}}{\partial c_i} = \frac{\partial \mathcal{L}}{\partial \overline{c_i}} = 0 \tag{195}$$

However, for eq 194 to be a valid stationary condition we must also show that it implies the time-dependent equation, eq 187. Consider in eq 194 a variation $\langle \delta \overline{0} |$, yielding the condition

$$\delta \mathscr{J} = \operatorname{Re}\{\langle \delta \overline{0} | H - \mathrm{i} \partial_t | \hat{0} \rangle\}_T = 0$$
(196)

As the variation $\langle \delta \overline{0} |$ may have an arbitrary time dependence in the interval [0,*T*], eq 196 implies that the equation

$$\operatorname{Re}\langle \delta \overline{0} | H - \mathrm{i} \partial_t | \hat{0} \rangle = 0 \tag{197}$$

holds at all times. Finally, using the fact that $\langle \overline{\partial 0} |$ and $i \langle \overline{\partial 0} |$ are both allowed variations in $\langle \overline{\partial 0} |$, eq 187 is obtained from eq 197. We have thus shown that eqs 194 and 187 are equivalent. The time-averaged Lagrangian variation principle of eq 194 will later be used to derive compact expressions for the response functions in coupled-cluster theory.

From the stationarity of the Lagrangian with respect to c and \overline{c} , we obtain for the derivative of the Lagrangian with respect to a perturbation parameter ε in the Hamiltonian the relation

$$\frac{\mathrm{d}\mathcal{F}}{\mathrm{d}\varepsilon} = \operatorname{Re}(\{\langle \mathbf{R}|\partial_{\varepsilon}H|\hat{\mathbf{0}}\rangle\}_{T} + \{\langle \overline{\mathbf{0}}|\partial_{\varepsilon}H|\hat{\mathbf{0}}\rangle\}_{T})$$
(198)

which constitutes an alternative form of the Hellmann-Feynman theorem, eq 177.

3.2. Perturbation Expansion of the Wave Function

The development in section 3.1 allows us to determine the solution to the time-dependent Schrödinger equation in the time domain. Although such a scheme is needed to treat the interaction between a molecule and an arbitrarily strong external field in a general manner, we now restrict ourselves to cases where the interaction with the external field may be treated by perturbation theory, expanding the wave function in orders of a periodic time-dependent perturbation V(t).

We begin by specifying the form of the perturbation operator. Allowing for several periodic components we write V(t) in the general form

$$V(t) = \sum_{B} \varepsilon_{B} V_{B} e^{-i\omega_{B}t}$$
(199)

where the summation over *B* runs symmetrically over both positive and negative indices. From the Hermiticity requirement, $V(t) = V^{\dagger}(t)$, it follows that all components of nonzero frequency occur in pairs of opposite frequencies $\omega_{\pm B}$, where we adopted the notation

$$\omega_{-B} = -\omega_B \tag{200}$$

Allowing for complex perturbation strengths, the Hermiticity of V(t) gives the following symmetry relations

$$V_B^{\dagger} = V_{-B} \tag{201}$$

$$\varepsilon_B^* = \varepsilon_{-B} \tag{202}$$

Instead of using the real and imaginary parts of ε_B as the independent parameters, it will be convenient to consider ε_B and ε_B^* as independent parameters. For symmetry reasons, static perturbations ($\omega = 0$) are also included with both a positive and a negative perturbation index. In this way, all perturbations entering eq 199 are treated on the same footing and satisfy the relations in eqs 200–202. When evaluating response functions we shall later examine expectation values of the specific perturbation component with index B_0 (frequency $\omega_{B_{\alpha'}}$ perturbation strength $\varepsilon_{B_{\alpha'}}$ and operator V_{B_0}).

To ensure that V(t) is periodic, all ω_B must be equal to some chosen frequency ω multiplied by an integer. In general, the frequencies ω_B correspond to some experimental setup and are not multiples of some common frequency ω . However, for an arbitrarily small $\tau \ge 0$ we can always find integers n_B and a frequency ω such that $|n_B\omega_B - \omega| \le \tau$.

3.2.1. Perturbation Expansion with a General Reference State. Having specified V(t), we now consider the expansion of the wave function defined by eq 187, which by substitution of the Hamiltonian of eq 157 becomes

$$\langle \hat{i}|H_0 + V(t) - i\partial_t|\hat{0}\rangle = 0$$
 (203)

Using intermediate normalization, the coefficients $\mathbf{c}(t)$ of the wave function $|\hat{0}\rangle$ in eq 179 may be written as the sum of a perturbation-independent zero-order part $\mathbf{c}^{(0)} = \{1, c_i^0\}$ and a perturbation-dependent correction

$$\Delta \mathbf{c}(t) = \mathbf{c}(t) - \mathbf{c}^{(0)} \tag{204}$$

with coefficients $\{0,c_i(t) - c_i^{(0)}\}$. The zero-order wave function and a basis for its orthogonal complement are then given by

$$|\hat{0}^{(0)}\rangle = |\mathbf{R}\rangle + \sum_{i} c_{i}^{(0)}|i\rangle$$
(205)

$$|\hat{i}^{(0)}\rangle = |i\rangle - c_i^{(0)*}|R\rangle$$
(206)

where $\langle \hat{i}^{(0)} | \hat{0}^{(0)} \rangle = 0$ follows from $\langle i | R \rangle = 0$. Retaining only the zero-order terms in eq 203 we obtain the zero-order equation

$$\langle \hat{i}^{(0)} | H_0 | \hat{0}^{(0)} \rangle = 0$$
 (207)

demonstrating that $H_0|\hat{0}^{(0)}\rangle$ has no components orthogonal to $|\hat{0}^{(0)}\rangle$ and hence that $|\hat{0}^{(0)}\rangle$ is an eigenfunction of H_0 with eigenvalue E_0

$$H_0|\hat{\mathbf{0}}^{(0)}\rangle = E_0|\hat{\mathbf{0}}^{(0)}\rangle \tag{208}$$

Usually the zero-order state is taken to be the ground-state wave function. However, in the following we do not make this assumption, requiring only that the zero-order state is an eigenstate of H_0 with a nonzero overlap with $|\mathbf{R}\rangle$, not necessarily the ground state.

We now consider the perturbation-dependent correction to the zero-order wave function. Inserting $|\hat{i}\rangle$ written as the sum of a zero-order part and a correction term

$$|\hat{i}\rangle = |\hat{i}^{(0)}\rangle + \left(|\hat{i}\rangle - |\hat{i}^{(0)}\rangle\right) = |\hat{i}^{(0)}\rangle - \Delta c_i^*(t)|\mathbf{R}\rangle$$
(209)

into eq 203 and using eq 207 we obtain the following equations for the corrections $\Delta c_i(t)$

$$\begin{split} \langle \hat{i}^{(0)} | V(t) | \hat{0}^{(0)} \rangle &- \mathrm{i} \partial_t \Delta c_i(t) + \sum_j \left(\langle \hat{i}^{(0)} | H_0 | j \rangle - \delta_{ij} E_0 \right) \Delta c_j(t) \\ &+ \sum_j \left(\langle \hat{i}^{(0)} | V(t) | j \rangle - \delta_{ij} \langle \mathbf{R} | V(t) | \hat{0}^{(0)} \rangle \right) \Delta c_j(t) \\ &- \Delta c_i(t) \sum_j \langle \mathbf{R} | V(t) | j \rangle \Delta c_j(t) - \Delta c_i(t) \sum_j \langle \mathbf{R} | H_0 | j \rangle \Delta c_j(t) = 0 \end{split}$$
(210)

Next, by inserting the expansion of the perturbation V(t) given in eq 199 and introducing the vectors

$$\left(\hat{H}^{\mathbb{R}[1]}\right)_{i} = \langle \hat{i}^{(0)} | H_{0} | \mathbb{R} \rangle \tag{211}$$

$$\left(\hat{V}_{B}^{[1]}\right)_{i} = \langle \hat{i}^{(0)} | V_{B} | \hat{\boldsymbol{0}}^{(0)} \rangle \tag{212}$$

$$\left(\hat{V}_{B}^{R[1]}\right)_{i} = \langle \hat{i}^{(0)} | V_{B} | \mathbf{R} \rangle \tag{213}$$

and the matrices

$$A_{ij} = \langle \hat{i}^{(0)} | H_0 | j \rangle - \delta_{ij} E_0 \tag{214}$$

$$\left(\hat{V}_{B}^{[2]}\right)_{ij} = \langle \hat{i}^{(0)} | V_{B} | j \rangle - \delta_{ij} \langle R | V_{B} | \hat{\mathbf{0}}^{(0)} \rangle$$
(215)

where **A** is the Jacobian matrix, we arrive at the Schrödinger equation for $\Delta \mathbf{c}(t)$

$$(\mathbf{A} - \mathrm{i}\partial_t)\Delta\mathbf{c}(t) = -\sum_B \varepsilon_B \mathrm{e}^{-\mathrm{i}\omega_B t} \Big[\hat{\mathbf{V}}_B^{[1]} + \hat{\mathbf{V}}_B^{[2]}\Delta\mathbf{c}(t) -\Delta\mathbf{c}(t) \Big(\hat{\mathbf{V}}_{-B}^{\mathrm{R}[1]\dagger} \Delta\mathbf{c}(t) \Big) \Big] + \Delta\mathbf{c}(t) \Big(\hat{\mathbf{H}}^{\mathrm{R}[1]\dagger} \Delta\mathbf{c}(t) \Big)$$
(216)

All terms in this equation follow straightforwardly from eq 210 except the term containing $\hat{V}_{-B}^{R[1]\dagger}$, which is obtained in the following manner

$$\sum_{j} \langle R|V|j \rangle \Delta c_{j}(t) = \sum_{j} \langle j|V|R \rangle^{*} \Delta c_{j}(t)$$

$$= \sum_{j, B} \varepsilon_{B}^{*} \langle j|V_{B}|R \rangle^{*} \Delta c_{j}(t) \mathbf{e}^{\mathrm{i}\omega_{B}t}$$

$$= \sum_{j, -B} \varepsilon_{B} \langle j|V_{-B}|R \rangle^{*} \Delta c_{j}(t) \mathbf{e}^{-\mathrm{i}\omega_{B}t}$$

$$= \sum_{B} \varepsilon_{B} \mathbf{e}^{-\mathrm{i}\omega_{B}t} \left(\hat{\mathbf{V}}_{-B}^{\mathrm{R}[1]\dagger} \Delta \mathbf{c}(t) \right) \quad (217)$$

where the Hermiticity of V(t) and eqs 200 and 202 have been used.

From the Schrödinger equation in eq 216 we may now determine equations for the wave function parameters to different orders in the perturbation

$$\Delta \mathbf{c}(t) = \sum_{B_1} \varepsilon_{B_1} \mathbf{c}_{B_1}^{(1)} \mathbf{e}^{-i\omega_{B_1}t} + \sum_{B_1B_2} \varepsilon_{B_1} \varepsilon_{B_2} \mathbf{c}_{B_{1, B_2}}^{(2)} \mathbf{e}^{-i(\omega_{B_1} + \omega_{B_2})t} + \dots$$
$$+ \sum_{B^n} (\prod_{k=1}^n \varepsilon_{B_k}) \mathbf{c}_{B^n}^{(n)} \mathbf{e}^{-i\sum_{k=1}^n \omega_{B_k}t} + \dots$$
(218)

where B^n is an *n*-dimensional multi-index of perturbation indices B_i

$$B^n = (B_1, B_2, ..., B_n)$$
(219)

The multi-index B^n is not ordered; consequently, there are, for instance, six third-order corrections B^3 corresponding to the six permutations of B_1 , B_2 , and B_3 . However, corrections referring to the same set of indices are required to be identical, for example, $\mathbf{c}_{B_1,B_2}^{(2)} = \mathbf{c}_{B_2,B_1}^{(2)}$. By inserting the expansion of the wave function parameters in eq 218 in the Schrödinger equation eq 216 and collecting terms proportional to ε_{B1} and to $\varepsilon_{B_1}\varepsilon_{B_2}$ we obtain the first- and second-order corrections, respectively

$$\mathbf{c}_{B_1}^{(1)} = -(\mathbf{A} - \omega_{B_1} \mathbf{I})^{-1} \hat{\mathbf{V}}_{B_1}^{[1]}$$
(220)

$$\mathbf{c}_{B_{1}, B_{2}}^{(2)} = -P_{[1, 2]}[\mathbf{A} - (\omega_{B_{1}} + \omega_{B_{2}})\mathbf{I}]^{-1} \\ \times \left[\hat{\mathbf{V}}_{B_{1}}^{[2]} \mathbf{c}_{B_{2}}^{(1)} - \mathbf{c}_{B_{1}}^{(1)} \left(\hat{\mathbf{H}}^{\mathbb{R}[1]\dagger} \mathbf{c}_{B_{2}}^{(1)} \right) \right]$$
(221)

where we used the identity $\mathbf{c}_{B_1,B_2}^{(2)} = \mathbf{c}_{B_2,B_1}^{(2)}$ and introduced $P_{[1,2]}$, which averages over the two permutations of 1 and 2

$$P_{[1, 2]}f_{B_1B_2} = \frac{1}{2!} (f_{B_1B_2} + f_{B_2B_1})$$
(222)

In a similar way, the general correction of order $n \ge 2$ becomes

$$\mathbf{c}_{B^{n}}^{(n)} = -P_{[1, n]} (\mathbf{A} - \sum_{k=1}^{n} \omega_{B_{k}} \mathbf{I})^{-1} \\ \times \left(\hat{\mathbf{V}}_{B_{1}}^{[2]} \mathbf{c}_{B_{2}, \dots, B_{n}}^{(n-1)} - \sum_{m=1}^{n-2} \mathbf{c}_{B_{2}, \dots, B_{m+1}}^{(m)} \hat{\mathbf{V}}_{-B_{1}}^{\mathbf{R}[1]\dagger} \mathbf{c}_{B_{m+2}, \dots, B_{n}}^{(n-m-1)} \\ - \sum_{m=1}^{n-1} \mathbf{c}_{B_{1}, \dots, B_{m}}^{(m)} \hat{\mathbf{H}}^{\mathbf{R}[1]\dagger} \mathbf{c}_{B_{m+1}, \dots, B_{n}}^{(n-m)} \right)$$
(223)

where $P_{[1,n]}$ averages over the *n*! permutations of the integers 1, 2, ..., *n*, imposing the required permutational symmetry on the wave function corrections.

3.2.2. Jacobian Matrix. In the preceding subsection the Jacobian was introduced in eq 214 and the wave function corrections were obtained as solutions to linear equations involving this matrix, see, for example, eq 223. We shall now study the Jacobian in more detail and, in particular, show that its eigenvalues are the excitation energies from the zero-order state.

Consider an eigenstate $|n^{(0)}\rangle$ of the Hamiltonian H_0 with energy E_n , assuming that this eigenstate has a nonzero overlap with the reference state $|R\rangle$. Using the nonorthogonal basis $\{|\hat{0}^{(0)}\rangle, |i\rangle\}$ rather than the orthonormal basis $\{|R\rangle, |i\rangle\}$, this eigenvector can be expanded as

$$|n^{(0)}\rangle = |\hat{\mathbf{0}}^{(0)}\rangle + \sum_{i} C_{i}^{n}|i\rangle$$
(224)

The condition that $|n^{(0)}\rangle$ is an eigenstate of H_0 with eigenvalue E_n may then be written as

$$(H_0 - E_0) \left(|\hat{\mathbf{0}}^{(0)}\rangle + \sum_i C_i^n |i\rangle \right)$$
$$= (E_n - E_0) \left(|\hat{\mathbf{0}}^{(0)}\rangle + \sum_i C_i^n |i\rangle \right)$$
(225)

Projecting eq 225 from the left against the complete basis $\{|R\rangle, |\hat{i}^{(0)}\rangle\}$ we obtain the matrix eigenvalue problem

$$\mathbf{H}^{\mathbf{b}}\begin{pmatrix}1\\\mathbf{C}^{n}\end{pmatrix} = (E_{n} - E_{0})\mathbf{S}^{\mathbf{b}}\begin{pmatrix}1\\\mathbf{C}^{n}\end{pmatrix}$$
(226)

where the shifted Hamiltonian and overlap matrices are given by (in a shorthand sub-block notation)

$$\mathbf{H}^{\mathrm{b}} = \begin{pmatrix} \langle \mathbf{R} | H_0 - E_0 | \hat{\mathbf{0}}^{(0)} \rangle & \langle \mathbf{R} | H_0 - E_0 | j \rangle \\ \langle \hat{i}^{(0)} | H_0 - E_0 | \hat{\mathbf{0}}^{(0)} \rangle & \langle \hat{i}^{(0)} | H_0 - E_0 | j \rangle \end{pmatrix}$$
$$= \begin{pmatrix} \mathbf{0} & \langle \mathbf{R} | H_0 - E_0 | j \rangle \\ \mathbf{0} & A_{ij} \end{pmatrix}$$
(227)

$$\mathbf{S}^{\mathrm{b}} = \begin{pmatrix} \langle \mathbf{R} | \hat{\mathbf{0}}^{(0)} \rangle & \langle \mathbf{R} | j \rangle \\ \langle \hat{i}^{(0)} | \hat{\mathbf{0}}^{(0)} \rangle & \langle \hat{i}^{(0)} | j \rangle \end{pmatrix} = \begin{pmatrix} 1 & 0 \\ 0 & \delta_{ij} \end{pmatrix}$$
(228)

The first column of the Hamiltonian \mathbf{H}^{b} vanishes because of eq 208; in the overlap matrix \mathbf{S}^{b} the off-diagonal blocks vanish because of the orthonormality of $\{|\mathbf{R}\rangle, |i\rangle\}$ and $\{|0^{(0)}\rangle, |\hat{i}^{(0)}\rangle\}$, whereas the unity of the diagonal blocks follows from eqs 205 and 206. Expansion of the Schrödinger equation in eq 226 now gives the Jacobian eigenvalue problem

$$\mathbf{AC}^n = (E_n - E_0)\mathbf{C}^n \tag{229}$$

whose eigenvalues are the excitation energies $E_n - E_0$ and whose eigenvectors define the excited eigenstates $|n^{(0)}\rangle$. In approximate theories the eigenvalues of the approximate Jacobian will be used to define excitation energies.

3.2.3. Perturbation Expansion with an Eigenvector Reference State. In the standard formulation of time-dependent perturbation theory, the reference state $|R\rangle$ is chosen as a normalized eigenfunction $|0^{(0)}\rangle$ of the unperturbed Hamiltonian H_0 with energy E_0 . The zero-order amplitudes then vanish, $\mathbf{c}^{(0)} = \mathbf{0}$, yielding $|\hat{0}^{(0)}\rangle = |0^{(0)}\rangle$ and $|\hat{i}^{(0)}\rangle = |i\rangle$ so that the vectors and matrices in eqs 211–215 reduce to

$$\left(\hat{H}^{\mathbb{R}[1]}\right)_i = 0 \tag{230}$$

$$\left(\hat{V}_{B}^{[1]}\right)_{i} = \left(\hat{V}_{B}^{R[1]}\right)_{i} = \left(V_{B}^{[1]}\right)_{i} = \langle i|V_{B}|\mathbf{0}^{(0)}\rangle \tag{231}$$

$$A_{ij} = E_{ij}^{[2]} = \langle i | H_0 | j \rangle - \delta_{ij} E_0$$
(232)

$$\left(\hat{V}_{B}^{[2]}\right)_{ij} = \left(V_{B}^{[2]}\right)_{ij} = \langle i|V_{B}|j\rangle - \delta_{ij}\langle 0^{(0)}|V_{B}|0^{(0)}\rangle \qquad (233)$$

Note, in particular, how the asymmetric Jacobian matrix **A** of the general theory now becomes the Hermitian Hessian matrix $\mathbf{E}^{[2]}$. Inserting the identifications of eqs 230 and 233 into the wave function corrections in eqs 220, 221, and 223, which are valid for any reference state, we obtain the following wave function corrections valid for an eigenvector reference state

$$\mathbf{c}_{B_1}^{(1)} = -\left(\mathbf{E}^{[2]} - \omega_{B_1}\mathbf{I}\right)^{-1}\mathbf{V}_{B_1}^{[1]}$$
(234)

$$\mathbf{c}_{B_1, B_2}^{(2)} = -P_{[1, 2]} (\mathbf{E}^{[2]} - (\omega_{B_1} + \omega_{B_2}) \mathbf{I})^{-1} \mathbf{V}_{B_1}^{[2]} \mathbf{c}_{B_2}^{(1)}$$
(235)

$$\mathbf{z}_{B^{n}}^{(n)} = -P_{[1, n]} \left(\mathbf{E}^{[2]} - \sum_{k=1}^{n} \omega_{B_{k}} \mathbf{I} \right)^{-1} \\ \times \left(\mathbf{V}_{B_{1}}^{[2]} \mathbf{c}_{B_{2}, \dots, B_{n}}^{(n-1)} - \sum_{m=1}^{n-2} \mathbf{c}_{B_{2}, \dots, B_{m+1}}^{(m)} \mathbf{V}_{-B_{1}}^{[1]\dagger} \mathbf{c}_{B_{m+2}, \dots, B_{n}}^{(n-m-1)} \right)$$
(236)

For static perturbations with $\omega_{B_k} = 0$ these responses reduce to the standard time-independent perturbation corrections.

3.2.4. Wave Function Corrections As Functions of the Reference State. The corrections to the wave function depend on the choice of reference state $|R\rangle$. However, all observables must be independent of this choice. In section 3.5.2, we will show explicitly that the lowest order observables are independent of the reference state. To prepare for this analysis we here study the relations for the zero-order state and also examine the first-order wave function corrections obtained using different reference states.

We first note that the intermediately normalized zero-order wave function $|\hat{0}^{(0)}(R)\rangle$ of eq 205 (whose dependence on the reference state is now made explicit) and the normalized zero-order state $|0^{(0)}\rangle$ are related as

$$|\hat{0}^{(0)}(R)\rangle = \sqrt{\langle \hat{0}^{(0)}(R) | \hat{0}^{(0)}(R) \rangle} | 0^{(0)} \rangle$$
(237)

Consider next the first-order corrections, where we examine the relation between the first-order correction using $|R\rangle$ and the normalized eigenstate $|0^{(0)}\rangle$ as reference states. For this purpose the expressions in eqs 220, 221, and 223 are inconvenient as they refer to expansions in the orthonormal basis $\{|R\rangle, |i\rangle\}$, which depends on the choice of the reference state $|R\rangle$. To avoid this dependence we work instead directly in terms of the first-order state expressed as

$$|\hat{0}_{B_1}^{(1)}(\mathbf{R})\rangle = \sum_i c_{B_1, i}^{(1)}(\mathbf{R})|i\rangle$$
 (238)

To obtain the state representation of the first-order correction we note that the wave function corrections by construction are orthogonal to the reference state. The first-order correction $|\hat{D}_{B_L}^{(1)}(R)\rangle$ of eq 238 is therefore orthogonal to the reference state, $\langle R|\hat{O}_{B_L}^{(1)}(R)\rangle = 0$, giving the condition

$$P_{\rm R}|\hat{0}_{B_1}^{(1)}({\rm R})\rangle = |\hat{0}_{B_1}^{(1)}({\rm R})\rangle$$
(239)

The first-order equation in eq 220 may be expanded using the definitions of **A** and $\mathbf{V}_{B_i}^{[1]}$ in eqs 212 and 214, respectively, giving

$$\langle \hat{i}^{(0)} | \left[\left(H_0 - E_0 - \omega_{B_1} \right) \sum_{j} c^{(1)}_{B_{1,j}} | j \rangle + V_{B_1} | \hat{\mathbf{0}}^{(0)} \rangle \right] = 0$$
(240)

The first-order equations are thus solved in the space spanned by $\{\langle i^{(0)} | \}$, which is the space orthogonal to $\langle \hat{0}^{(0)} |$. The first-order equations can therefore be recast in the basis-independent form

$$P_{\hat{0}^{(0)}}(H_{0} - E_{0} - \omega_{B_{1}})|\hat{0}^{(1)}_{B_{1}}(\mathbf{R})\rangle = -\sqrt{\langle \hat{0}^{(0)}(\mathbf{R})|\hat{0}^{(0)}(\mathbf{R})\rangle} P_{\hat{0}^{(0)}}V_{B_{1}}|0^{(0)}\rangle$$
(241)

where we used eq 237. The action of the matrix $P_{\hat{0}^{(0)}}(H_0 - E_0 - \omega_{B_1})$ on the eigenvector $|0^{(0)}\rangle$ vanishes; thus, eq 241 determines $|\hat{0}_{B_1}^{(1)}(R)\rangle$ to within a component of $|0^{(0)}\rangle$, which is subsequently determined by eq 239. To find the form of $|\hat{0}_{B_1}^{(1)}(R)\rangle$ that fulfills eqs 239 and 241 we first note that the ansatz

$$|\hat{0}_{B_{1}}^{(1)}(\mathbf{R})\rangle = \sqrt{\langle \hat{0}^{(0)}(\mathbf{R}) | \hat{0}^{(0)}(\mathbf{R}) \rangle} (|\hat{0}_{B_{1}}^{(1)}(0^{(0)})\rangle + \alpha | 0^{(0)} \rangle)$$
(242)

solves the first-order equations in eq 241 for any α . Insertion of the ansatz in eq 242 into eq 239 identifies α , yielding the first-order correction

$$\hat{\boldsymbol{\theta}}_{B_{1}}^{(1)}(\mathbf{R}) \rangle = \sqrt{\langle \hat{\boldsymbol{0}}^{(0)}(\mathbf{R}) | \hat{\boldsymbol{0}}^{(0)}(\mathbf{R}) \rangle} \\ \times (| \hat{\boldsymbol{0}}_{B_{1}}^{(1)}(\mathbf{0}^{(0)}) \rangle - \langle \mathbf{R} | \mathbf{0}^{(0)} \rangle^{-1} \langle \mathbf{R} | \mathbf{0}_{B_{1}}^{(1)}(\mathbf{0}^{(0)}) \rangle | \mathbf{0}^{(0)} \rangle)$$

$$(243)$$

3.3. Perturbation Expansion of the Quasi-Energy

Having determined the perturbation expansion of the wave function parameters in section 3.2 we are now ready to consider the expansion of the quasi-energy (in the Hermitian form) in orders of the external perturbations. We thereby obtain the timedependent analogue of the standard perturbation expansion of the energy in time-independent perturbation theory; this expansion is subsequently used in section 3.4 to identify expressions for the response functions.

We assume that the reference state is the eigenstate $|0^{(0)}\rangle$ of the zero-order Hamiltonian, giving the wave function corrections of eqs 234–236 rather than the somewhat more complicated wave function corrections for a general reference state in eqs 220, 221, and 223. In terms of the intermediately normalized wave function the quasi-energy of eq 172 becomes

$$\mathcal{Q} = \left\{ \frac{\langle \hat{\mathbf{0}} | H - \mathbf{i} \partial_t | \hat{\mathbf{0}} \rangle}{\langle \hat{\mathbf{0}} | \hat{\mathbf{0}} \rangle} \right\}_T$$
(244)

Remembering that $\mathbf{c}^{(0)}$ is the expansion for the state $|\mathbf{0}^{(0)}\rangle$, here $\{1,0\}$ and introducing the matrices \mathbf{H}_0 and \mathbf{V}_B for the operators H_0 and V_B , respectively, in the orthonormal basis $\{|\mathbf{0}^{(0)}\rangle, |i\rangle\}$, the numerator and denominator of the time-dependent quasi-energy may be written in the form

$$E(t) = \langle \hat{\mathbf{0}} | H - i\partial_t | \hat{\mathbf{0}} \rangle$$

= $\left(\mathbf{c}^{(0)} + \Delta \mathbf{c} \right)^{\dagger} \left(\mathbf{H}_0 + \sum_B \varepsilon_B \mathbf{V}_B \mathbf{e}^{-i\omega_B t} - i\partial_t \right) \left(\mathbf{c}^{(0)} + \Delta \mathbf{c} \right)$
(245)

$$S(t) = \langle \hat{\mathbf{0}} | \hat{\mathbf{0}} \rangle = 1 + \Delta \mathbf{c}^{\dagger} \Delta \mathbf{c}$$
(246)

The time-dependent quasi-energy Q(t) and the quantities E(t) and S(t) may be expanded in the external perturbations as

$$Q(t) = \sum_{n=0}^{\infty} \sum_{B^n} \left(\prod_{k=1}^n \varepsilon_{B_k} \right) Q_{B^n}^{(n)} e^{-i\sum_{k=1}^n \omega_{B_k} t}$$
(247)

$$E(t) = \sum_{n=0}^{\infty} \sum_{B^n} \left(\prod_{k=1}^n \varepsilon_{B_k} \right) E_{B^n}^{(n)} e^{-i\sum_{k=1}^n \omega_{B_k} t}$$
(248)

$$S(t) = \sum_{n=0}^{\infty} \sum_{B^n} \left(\prod_{k=1}^n \varepsilon_{B_k} \right) S_{B^n}^{(n)} e^{-i \sum_{k=1}^n \omega_{B_k} t}$$
(249)

To accommodate the zero-order terms in these expressions we conventionally set $\prod_{k=1}^{0} \varepsilon_{B_k} = 1$. We furthermore require that the corrections $Q_{B^n}^{(n)}$, $E_{B^n}^{(n)}$, and $S_{B^n}^{(n)}$ are permutationally symmetric, for instance, $Q_{B_1,B_2}^{(2)} = Q_{B_2,B_1}^{(2)}$. The *n*th-order corrections are then proportional to the *n*th-order derivatives of the quasienergy

$$\frac{\mathrm{d}^{n}Q}{\mathrm{d}\varepsilon_{B_{1}}...\mathrm{d}\varepsilon_{B_{n}}} = \frac{1}{n!} Q_{B^{n}}^{(n)}$$
(250)

and likewise for $E_{B^n}^{(n)}$ and $S_{B^n}^{(n)}$.

Inserting the expansion of the intermediate wave function from eq 218 in eqs 248 and 249, the expansions of E(t) and S(t) become

$$E_{B^{n}}^{(n)} = P_{[1, n]} \sum_{m=0}^{n} \mathbf{c}_{-B_{1}, \dots, -B_{m}}^{(m)+} \left(\mathbf{H}_{0} - \sum_{k=m+1}^{n} \omega_{B_{k}} \mathbf{I} \right) \mathbf{c}_{B_{m+1}, \dots, B_{n}}^{(n-m)} + P_{[1, n]} \sum_{m=0}^{n-1} \mathbf{c}_{-B_{2}, \dots, -B_{m+1}}^{(m)+} \mathbf{V}_{B_{1}} \mathbf{c}_{B_{m+2}, \dots, B_{n}}^{(n-m-1)}$$
(251)

$$S_{B^n}^{(n)} = P_{[1, n]} \sum_{m=0}^{n} \mathbf{c}_{-B_1, \dots, -B_m}^{(m)+} \mathbf{c}_{B_{m+1}, \dots, B_n}^{(n-m)}$$
(252)

Using eq 178 we obtain for the lowest order terms in the expansion of S(t) the contributions

$$S^{(0)} = \mathbf{c}^{(0)\dagger} \mathbf{c}^{(0)} = 1$$
(253)

$$S_{B_1}^{(1)} = \mathbf{c}_{-B_1}^{(1)\dagger} \mathbf{c}^{(0)} + \mathbf{c}^{(0)\dagger} \mathbf{c}_{B_1}^{(1)} = \mathbf{0}$$
(254)

Expansion of the time-dependent quasi-energy Q(t) is most conveniently determined from Q(t)S(t) = E(t), which by eqs 253 and 254 leads to the identification

$$Q_{B^{n}}^{(n)} = P_{[1, n]} \left(E_{B^{n}}^{(n)} - \sum_{m=0}^{n-2} Q_{B_{1}, \dots, B_{m}}^{(m)} S_{B_{m+1}, \dots, B_{n}}^{(n-m)} \right)$$
(255)

For the zero-order correction we trivially obtain that

$$Q^{(0)} = E_0 (256)$$

Next, given that $\mathbf{c}^{(0)}$ is an eigenvector of \mathbf{H}_0 and that the wave function correction $\mathbf{c}_{B_1}^{(1)}$ is orthogonal to the zero-order state $\mathbf{c}^{(0)}$, we obtain the following first-order correction to the quasi-energy

$$Q_{B_{1}}^{(1)} = E_{B_{1}}^{(1)} = \mathbf{c}^{(0)\dagger} (\mathbf{H}_{0} - \omega_{B_{1}} \mathbf{I}) \mathbf{c}_{B_{1}}^{(1)} + \mathbf{c}_{-B_{1}}^{(1)\dagger} \mathbf{H}_{0} \mathbf{c}^{(0)} + \mathbf{c}^{(0)\dagger} \mathbf{V}_{B_{1}} \mathbf{c}^{(0)} = \mathbf{c}^{(0)\dagger} \mathbf{V}_{B_{1}} \mathbf{c}^{(0)}$$
(257)

To obtain the second-order correction to the time-dependent quasienergy we first write

$$Q_{B_{1},B_{2}}^{(2)} = P_{[1,2]}(\mathbf{c}^{(0)\dagger}(\mathbf{H}_{0} - (\omega_{B_{1}} + \omega_{B_{2}})\mathbf{I})\mathbf{c}_{B_{1},B_{2}}^{(2)} + \mathbf{c}_{-B_{1},-B_{2}}^{(2)\dagger}\mathbf{H}_{0}\mathbf{c}^{(0)} + \mathbf{c}_{B_{1}}^{(1)\dagger}(\mathbf{H}_{0} - \omega_{B_{2}}\mathbf{I})\mathbf{c}_{B_{2}}^{(1)} + \mathbf{c}_{-B_{1}}^{(0)\dagger}\mathbf{V}_{B_{2}}\mathbf{c}^{(0)} - E_{0}\left(\mathbf{c}^{(0)\dagger}\mathbf{c}_{B_{1},B_{2}}^{(2)} + \mathbf{c}_{-B_{1},-B_{2}}^{(2)\dagger}\mathbf{c}^{(0)} + \mathbf{c}_{B_{1}}^{(1)\dagger}\mathbf{c}_{B_{2}}^{(1)}\right)\right) \\ = P_{[1,2]}\left(\mathbf{c}_{-B_{1}}^{(1)\dagger}\left(\mathbf{E}^{[2]} - \omega_{B_{2}}\mathbf{I}\right)\mathbf{c}_{B_{2}}^{(1)} + \mathbf{c}_{-B_{1}}^{(1)\dagger}\mathbf{V}_{B_{2}}^{[1]} + \mathbf{V}_{-B_{2}}^{[1]\dagger}\mathbf{c}_{B_{1}}^{(1)}\right)\right)$$
(258)

where we used the definitions of $\mathbf{E}^{[2]}$ and $\mathbf{V}_{B}^{[1]}$ of eq 233 and the orthogonality of the wave function corrections to the zero-order state. From the first-order equations in eq 234 we find that the first two terms in the final expression in eq 258 cancel, yielding the following simple expression for the second-order time-dependent quasi-energy

$$Q_{B_{1}, B_{2}}^{(2)} = P_{[1, 2]} \mathbf{V}_{-B_{1}}^{[1]\dagger} \mathbf{c}_{B_{2}}^{(1)}$$
(259)

We note that the second-order energy depends only on the firstorder wave function, while the zero- and first-order energies in eqs 256 and 257 depend only on the zero-order wave function.

We next consider the corresponding perturbation expansion of the time-averaged quasi-energy \mathcal{Q} . From the relation

$$\frac{1}{T} \int_{0}^{T} e^{in\omega t} dt = \frac{1}{T} \int_{0}^{T} e^{in2\pi t/T} dt = \delta_{n, 0}$$
(260)

we observe that the time average of $Q_{B^n}^{(n)} e^{i \Sigma_k \omega_{B_k} t}$ vanishes unless $\Sigma_k \omega_{Bk} = 0$. We therefore obtain the following expansion of the time-averaged quasi-energy

$$\mathcal{Q} = \sum_{n} \sum_{B^{n}: \sum_{k} \omega_{B_{k}} = 0} (\prod_{k} \varepsilon_{B_{k}}) \mathcal{Q}_{B^{n}}^{(n)}$$
(261)

with trivial relations between the time-averaged and time-dependent quasi-energy corrections

$$\mathcal{Q}_{B^{n}}^{(n)} = \begin{cases} Q_{B^{n}}^{(n)} & \text{if } \sum_{k} \omega_{B_{k}} = 0\\ 0 & \text{otherwise} \end{cases}$$
(262)

The time-averaged quasi-energy \mathcal{Q} is variational in each of the nonvanishing components of the perturbation expansion. In particular, because $\mathcal{Q}_{B^{2n}}^{(2n)}$ and $\mathcal{Q}_{B^{2n+1}}^{(2n+1)}$ are linear in the wave function corrections $\mathbf{c}^{(k)}$ of orders k > n, these corrections do not contribute. Thus, to determine the time-averaged quasi-energy to order 2n + 1 it is only necessary to include the wave function corrections up to order n. This is an example of the 2n + 1 rule, which is proved in section 3.6.

With this observation we are ready to develop compact forms for the lower order corrections to the quasi-energy. In the following we consider the perturbation component with index B_0 since the expectation value of the V_{B_0} operator will be the central quantity when identifying the response functions in section 3.4. The corresponding frequency ω_{B_0} is chosen such that the sum of all the frequencies for a given correction vanishes. The first- and second-order quasi-energies become

$$\mathcal{Q}_{B_0}^{(1)} = \mathbf{c}^{(0)\dagger} \mathbf{V}_{B_0} \mathbf{c}^{(0)}, \quad \omega_{B_0} = 0$$
(263)

$$\mathcal{Q}_{B_{0}, B_{1}}^{(2)} = P_{[0, 1]} \mathbf{V}_{-B_{0}}^{[1]\dagger} \mathbf{c}_{B_{1}}^{(1)}, \, \omega_{B_{0}} = -\omega_{B_{1}}$$
(264)

Using the first-order equations in eq 220 and $\omega_{-B_0} = \omega_{B_1}$ the second-order quasi-energy correction may alternatively be written in the form

$$\mathcal{Q}_{B_{0}, B_{1}}^{(2)} = P_{[0, 1]} \mathbf{V}_{-B_{0}}^{[1]\dagger} \mathbf{c}_{B_{1}}^{(1)} = P_{[0, 1]} \mathbf{V}_{-B_{0}}^{[1]\dagger} (\mathbf{E}^{[2]} - \omega_{-B_{0}} \mathbf{I})^{-1} \mathbf{V}_{B_{1}}^{[1]}$$
$$= P_{[0, 1]} \mathbf{c}_{-B_{0}}^{(1)\dagger} \mathbf{V}_{B_{1}}^{[1]}$$
(265)

To obtain the third-order quasi-energy we expand eq 255, including only wave function corrections to first order, and then use the expression for $\mathbf{V}^{[2]}$ in eq 233

$$\mathcal{Q}_{B_{0}, B_{1}, B_{2}}^{(3)} = P_{[0, 2]} \left(\mathbf{c}_{-B_{0}}^{(1)\dagger} \mathbf{V}_{B_{1}} \mathbf{c}_{B_{2}}^{(1)} - E_{B_{1}}^{(1)} \mathbf{c}_{-B_{0}}^{(1)\dagger} \mathbf{c}_{B_{2}}^{(1)} \right)$$

$$= P_{[0, 2]} \mathbf{c}_{-B_{0}}^{(1)\dagger} \mathbf{V}_{B_{1}}^{[2]} \mathbf{c}_{B_{2}}^{(1)}, \ \omega_{B_{0}} = -\left(\omega_{B_{1}} + \omega_{B_{2}}\right)$$

(266)

The fourth-order energy is obtained by first expanding eq 255 using only wave function corrections to second order and then the definitions of $\mathbf{E}^{[2]}$ and $\mathbf{V}^{[2]}$ in eqs 232 and 233 yielding

$$\begin{aligned} \mathscr{Q}_{B_{0}, B_{1}, B_{2}, B_{3}}^{(4)} &= P_{[0, 3]} \Big(\mathbf{c}_{-B_{0}, -B_{1}}^{(2)\dagger} \big(\mathbf{H}_{0} - \big(\omega_{B_{2}} + \omega_{B_{3}} \big) \mathbf{I} \big) \mathbf{c}_{B_{2}, B_{3}}^{(2)} \\ &+ \mathbf{c}_{-B_{0}, -B_{1}}^{(2)\dagger} \mathbf{V}_{B_{2}} \mathbf{c}_{B_{3}}^{(1)} + \mathbf{c}_{-B_{0}}^{(1)\dagger} \mathbf{V}_{B_{1}} \mathbf{c}_{B_{2}, B_{3}}^{(2)} \\ &- E^{(0)} \mathbf{c}_{-B_{0}, -B_{1}}^{(2)\dagger} \mathbf{c}_{B_{2}, B_{3}}^{(2)} - E_{B_{1}}^{(1)} \mathbf{c}_{-B_{0}}^{(2)\dagger} \mathbf{c}_{B_{2}, B_{3}}^{(2)} \\ &- E_{B_{2}}^{(1)} \mathbf{c}_{-B_{0}, -B_{1}}^{(2)\dagger} \mathbf{c}_{B_{3}}^{(1)} - \mathbf{Q}_{B_{0}, B_{1}}^{(2)} \mathbf{S}_{B_{2}, B_{3}}^{(2)} \Big) \\ &= P_{[0, 3]} \Big(\mathbf{c}_{-B_{0}, -B_{1}}^{(2)\dagger} \mathbf{c}_{B_{3}}^{(1)} - \big(\omega_{B_{2}} + \omega_{B_{3}} \big) \mathbf{I} \Big) \mathbf{c}_{B_{2}, B_{3}}^{(2)} \\ &+ P_{[2, 3]} \mathbf{c}_{-B_{0}, -B_{1}}^{(2)\dagger} \mathbf{V}_{B_{2}}^{[2]} \mathbf{c}_{B_{3}}^{(1)} \\ &+ \mathbf{c}_{-B_{0}}^{(1)\dagger} \mathbf{V}_{B_{1}}^{[2]} \mathbf{c}_{B_{2}, B_{3}}^{(2)} - \mathbf{Q}_{B_{0}, B_{1}}^{(2)} \mathbf{S}_{B_{2}, B_{3}}^{(2)} \Big) \tag{267}$$

which by eq 235 simplifies to the following expression

$$\mathcal{Q}_{B_{0}, B_{1}, B_{2}, B_{3}}^{(4)} = P_{[0, 3]} \Big(\mathbf{c}_{-B_{0}}^{(1)\dagger} \mathbf{V}_{B_{1}}^{[2]} \mathbf{c}_{B_{2}, B_{3}}^{(2)} - Q_{B_{0}, B_{1}}^{(2)} S_{B_{2}, B_{3}}^{(2)} \Big),$$

$$\omega_{B_{0}} = - \big(\omega_{B_{1}} + \omega_{B_{2}} + \omega_{B_{3}} \big)$$
(268)

for the fourth-order quasi-energy.

3.4. Response Functions

To introduce response functions we now consider the expectation value of the operator V_{B_0} and the expansion of this expectation value in orders of the external perturbation. Subsequently, we identify the response functions with derivatives of the quasi-energy and use these identifications to obtain explicit expressions for response functions to third order. Having examined the symmetries of the response functions we finally examine their residues.

3.4.1. Definition of Response Functions. To introduce response functions we consider the Hamiltonian eq 157, where terms with indices B_0 and $-B_0$ are omitted from the perturbation

$$V(t) = \sum_{B \neq \pm B_0} \varepsilon_B V_B e^{-i\omega_B t} = \sum_B' \varepsilon_B V_B e^{-i\omega_B t}$$
(269)

Here and in the following the prime indicates that terms with

indices B_0 and $-B_0$ are omitted, unlike the perturbation in eq 199. The response functions are now defined as terms in the perturbation expansion of the expectation value of an operator V_{B_0}

$$\langle V_{B_0} \rangle'(t) = \langle \langle V_{B_0} \rangle \rangle + \sum_{B_1} \langle \varepsilon_{B_1} \langle \langle V_{B_0}; V_{B_1} \rangle \rangle_{\omega_{B_1}} e^{-i\omega_{B_1}t} + \frac{1}{2!} \sum_{B_1, B_2} \langle \varepsilon_{B_1} \varepsilon_{B_2} \langle \langle V_{B_0}; V_{B_1}, V_{B_2} \rangle \rangle_{\omega_{B_1}, \omega_{B_2}} e^{-i(\omega_{B_1} + \omega_{B_2})t} + \frac{1}{3!} \sum_{B_1, B_2, B_3} \langle \varepsilon_{B_1} \varepsilon_{B_2} \varepsilon_{B_3} \langle \langle V_{B_0}; V_{B_1}, V_{B_2}, V_{B_3} \rangle \rangle_{\omega_{B_1}, \omega_{B_2}, \omega_{B_3}} \times e^{-i(\omega_{B_1} + \omega_{B_2} + \omega_{B_3})t} + \dots$$
(270)

The terms $\langle \langle V_{B_0} \rangle \rangle$, $\langle \langle V_{B_0}; V_{B_1} \rangle \rangle_{\omega_{B_1}}$, $\langle \langle V_{B_0}; V_{B_1}, V_{B_2} \rangle \rangle_{\omega_{B_1}, \omega_{B_2}}$, and $\langle \langle V_{B_0}; V_{B_1}, V_{B_2}, V_{B_3} \rangle \rangle_{\omega_{B_1}, \omega_{B_2}, \omega_{B_3}}$ are the zero-order, linear, quadratic, and cubic response functions, respectively,⁷⁰ which describe how the expectation value of an operator V_{B_0} responds to a set of external perturbations to zero, first, second, and third orders in the perturbation strengths. Note that the response function, in particular, they do not depend on whether a Hermitian (eq 177) or a non-Hermitian (eq 198) form of the expectation value is used. For example, using the Hermitian form it follows that $\langle V_{B_0} \rangle'(t) = \langle 0 | V_{B_0} | 0' \rangle$, where $| 0 \rangle$ is the time-dependent wave function in eq 156 for the perturbation in eq 269, while the zero-order response function is a simple expectation value for the normalized eigenfunction of the unperturbed state: $\langle \langle V_{B_0} \rangle = \langle 0^{(0)} | V_{B_0} | 0' \rangle$.

3.4.2. Response Functions As Derivatives of the Quasi-Energy. To establish a connection between the expansion of the expectation value in response functions and the expansion of the quasi-energy we first take the time average of the expectation value of $V_{B_0}e^{-i\omega_{B_0}t}$. Using eq 260 we obtain

$$\left\{ \langle V_{B_0} \rangle' e^{-i\omega_{B_0}t} \right\}_T = \langle \langle V_{B_0} \rangle \rangle + \sum_{\substack{B_1: \\ \omega_{B_0} + \omega_{B_1} = 0}} \varepsilon_{B_1} \langle \langle V_{B_0}; V_{B_1} \rangle \rangle_{\omega_{B_1}} \\ + \frac{1}{2!} \sum_{\substack{B_1, B_2: \\ \omega_{B_0} + \omega_{B_1} + \omega_{B_2} = 0}} \varepsilon_{B_1} \varepsilon_{B_2} \langle \langle V_{B_0}; V_{B_1}, V_{B_2} \rangle \rangle_{\omega_{B_1}, \omega_{B_2}} \\ + \frac{1}{3!} \sum_{\substack{B_1, B_2: \\ \omega_{B_0} + \omega_{B_1} + \omega_{B_2} + \omega_{B_3} = 0}} \varepsilon_{B_1} \varepsilon_{B_2} \varepsilon_{B_3} \langle \langle V_{B_0}; V_{B_1}, V_{B_2}, V_{B_3} \rangle \rangle_{\omega_{B_1}, \omega_{B_2}, \omega_{B_3}} + \dots$$

$$(271)$$

An alternative identification of the time-averaged expectation value $\{\langle V_{B_0} \rangle e^{-i\omega_{B_0} t}\}_T$ is obtained using the time-dependent Hellmann–Feynman theorem eq 177 for the perturbation of eq 199 followed by eq 261

$$\left\{ \langle V_{B_0} \rangle' e^{-i\omega_{B_0}t} \right\}_T = \frac{d\mathcal{Q}}{d\varepsilon_{B_0}} \bigg|_{\varepsilon_{B_0} = \varepsilon_{-B_0} = 0} = \mathcal{Q}_{B_0}^{(1)} \delta_{\omega_{B_0}, 0}$$

$$+ 2 \sum_{\substack{B_1:\\ \omega_{B_0} + \omega_{B_1} = 0}} \varepsilon_{B_1} \mathcal{Q}_{B_0, B_1}^{(2)} + 3 \sum_{\substack{B_1, B_2:\\ \omega_{B_0} + \omega_{B_1} + \omega_{B_2} = 0}} \varepsilon_{B_1} \varepsilon_{B_2} \mathcal{Z}_{B_0, B_1, B_2}^{(3)}$$

$$+ 4 \sum_{\substack{B_1, B_2:\\ \omega_{B_0} + \omega_{B_1} + \omega_{B_2} + \omega_{B_3} = 0}} \varepsilon_{B_1} \varepsilon_{B_2} \varepsilon_{B_3} \mathcal{Q}_{B_0, B_1, B_2, B_3}^{(4)}$$

$$(272)$$

Comparing eqs 271 and 272 we obtain the identifications

$$\langle\langle V_{B_0} \rangle\rangle = \frac{\mathrm{d}\mathcal{Q}}{\mathrm{d}\varepsilon_{B_0}} = \mathcal{Q}_{B_0}^{(1)} \tag{273}$$

$$\langle\langle V_{B_0}; V_{B_1} \rangle\rangle_{\omega_{B_1}} = \frac{\mathrm{d}^2 \mathscr{Q}}{\mathrm{d}\varepsilon_{B_0} \mathrm{d}\varepsilon_{B_1}} = 2 \mathscr{Q}_{B_0, B_1}^{(2)}$$
(274)

$$\langle \langle V_{B_0}; V_{B_1}, V_{B_2} \rangle \rangle_{\omega_{B_1}, \omega_{B_2}} = \frac{d^3 \mathcal{Q}}{d\varepsilon_{B_0} d\varepsilon_{B_1} d\varepsilon_{B_2}}$$
$$= 3! \mathcal{Q}_{B_0, B_1, B_2}^{(3)}$$
(275)

$$\langle \langle V_{B_0}; V_{B_1}, V_{B_2}, V_{B_2} \rangle \rangle_{\omega_{B_1}, \omega_{B_2}, \omega_{B_3}} = \frac{d^4 \mathcal{Q}}{d\varepsilon_{B_0} d\varepsilon_{B_1} d\varepsilon_{B_3} d\varepsilon_{B_3}} = 4! \mathcal{Q}_{B_0, B_1, B_2, B_3}^{(4)}$$
(276)

and conclude that the response functions are the derivatives of the quasi-energy (in the Hermitian form) with respect to the perturbation parameters. We shall later extend this identification to the quasi-energy in the Lagrangian form.

3.4.3. Explicit Expressions for the Linear, Quadratic, and Cubic Response Functions. In section 3.4.2, we expressed the response functions in terms of the quasi-energy corrections $Q_{B^n}^{(n)}$. We now combine these identifications with the results of section 3.3 to express the response functions in terms of wave function corrections.

If we are only interested in lower order response functions, up to the cubic response function, for instance, then it is convenient to introduce a more compact notation for the operators and frequencies. We use *A*, *B*, *C*, and *D* for the operators V_{B_0} , V_{B_1} , V_{B_2} , and V_{B_3} , respectively, and ω_A , ω_B , ω_C , and ω_D for the frequencies ω_{B_0} , ω_{B_1} , ω_{B_2} , and ω_{B_3} , respectively. The paired operators occur again with opposite frequencies, for example, $\omega_{A^+} = -\omega_A$. We furthermore assume that the basis vectors $|i\rangle$ are eigenfunctions of H_0 with energies E_{ij} in which case the matrix $\mathbf{E}^{[2]}$ in eq 232 becomes diagonal

$$E_{ij}^{[2]} = \delta_{ij}\omega_i, \ \omega_i = E_i - E_0 \tag{277}$$

Using eq 276 and the form of the response functions in eqs 264, 266, and 268 we arrive at the spectral representation of the response functions⁷¹

$$\langle \langle A; B \rangle \rangle_{\omega_B} = 2! P^{\omega_A}_{[A, B]} \mathbf{V}^{[1]\dagger}_{A^{\dagger}} \mathbf{c}^{(1)}_B$$

$$= 2! P^{\omega_A}_{[A, B]} \sum_i \frac{\langle \mathbf{0}^{(0)} | A | i \rangle \langle i | B | \mathbf{0}^{(0)} \rangle}{\omega_B - \omega_i}$$

$$(278)$$

$$\langle \langle A; B, C \rangle \rangle_{\omega_{B}, \omega_{C}} = 3! P^{\omega_{A}}_{[A, B, C]} \mathbf{c}^{(1)\dagger}_{A^{\dagger}} \mathbf{V}^{[2]}_{B} \mathbf{c}^{(1)}_{C}$$

$$= -3! P^{\omega_{A}}_{[A, B, C]} \sum_{ij} \frac{\langle 0^{(0)} | A | i \rangle \langle i | \overline{B} | j \rangle \langle j | C | 0^{(0)} \rangle}{(\omega_{A} + \omega_{i})(\omega_{C} - \omega_{j})}$$

$$(279)$$

$$= -4! P_{[A, B, C, D]}^{\omega_{A}} \left(\sum_{ijk} \frac{\langle 0^{(0)} | A | i \rangle \langle i | \overline{B} | j \rangle \langle j | \overline{C} | k \rangle \langle k | D | 0^{(0)} \rangle}{(\omega_{A} + \omega_{i})(\omega_{C} + \omega_{D} - \omega_{j})(\omega_{D} - \omega_{k})} - \sum_{ij} \frac{\langle 0^{(0)} | A | i \rangle \langle i | B | 0^{(0)} \rangle}{\omega_{B} - \omega_{i}} \frac{\langle 0^{(0)} | C | j \rangle \langle j | D | 0^{(0)} \rangle}{(\omega_{C} + \omega_{j})(\omega_{D} - \omega_{j})} \right)$$
(280)

where we introduced the shorthand notation

$$\overline{X} = X - \langle 0^{(0)} | X | 0^{(0)} \rangle \tag{281}$$

and where $P_{[A,B,C,D]}^{\omega_A}$ averages over all permutations of *A*, *B*, *C*, *D* and adjusts the frequency ω_A such that all frequencies add up to zero: $\omega_A + \omega_B + \omega_C + \omega_D = 0$. As a result, we obtain from eq 278 the following well-known expression for the linear response function in the spectral representation

$$\langle\langle A;B\rangle\rangle_{\omega_{B}} = \sum_{i} \frac{\langle 0^{(0)}|A|i\rangle\langle i|B|0^{(0)}\rangle}{\omega_{B} - \omega_{i}} - \sum_{i} \frac{\langle 0^{(0)}|B|i\rangle\langle i|A|0^{(0)}\rangle}{\omega_{B} + \omega_{i}}$$
(282)

3.4.4. Symmetry Properties of Response Functions. A number of symmetry relations and identities exist between response functions of the same order. First, as the response functions are symmetric in the permutation of the indices we have

$$\langle\langle A; B \rangle\rangle_{\omega_B} = \langle\langle B; A \rangle\rangle_{\omega_A} = \langle\langle B; A \rangle\rangle_{-\omega_B}$$
(283)

$$\langle \langle A; B, C \rangle \rangle_{\omega_{B}, \omega_{C}} = \langle \langle A; C, B \rangle \rangle_{\omega_{C}, \omega_{B}} = \langle \langle C; A, B \rangle \rangle_{\omega_{A}, \omega_{B}}$$
(284)

$$\langle \langle A; B, C, D \rangle \rangle_{\omega_{B}, \omega_{C}, \omega_{D}} = \langle \langle A; C, B, D \rangle \rangle_{\omega_{C}, \omega_{B}, \omega_{D}} = \langle \langle D; A, B, C \rangle \rangle_{\omega_{A}, \omega_{B}, \omega_{C}}$$
(285)

Next, assuming that the operators *A*, *B*, *C*, *D* are Hermitian, additional relations are obtained by setting $\langle A \rangle^* = \langle A \rangle$ in eq 270. Using eqs 200 and 202 we then obtain the following relations

$$\langle\langle A; B \rangle\rangle_{\omega_{\rm B}} = \langle\langle A; B \rangle\rangle_{-\omega_{\rm B}}^{*}$$
 (286)

$$\langle\langle A; B, C \rangle\rangle_{\omega_{B}, \omega_{C}} = \langle\langle A; B, C \rangle\rangle_{-\omega_{B}, -\omega_{C}}^{*}$$
(287)

$$\langle\langle A; B, C, D \rangle\rangle_{\omega_{B}, \omega_{C}, \omega_{D}} = \langle\langle A; B, C, D \rangle\rangle^{*}_{-\omega_{B}, -\omega_{C}, -\omega_{D}}$$
(288)

between response functions at positive and negative frequencies.

3.4.5. Residues of Response Functions. From the spectral representations of the linear, quadratic, and cubic response functions given in eqs 278–280 it is seen that these tend to $\pm \infty$ when the frequency parameters ω_{A} , ω_{B} , ... tend to plus or minus an excitation energy, $\pm \omega_{f}$, of the unperturbed system. For example, the linear response function of eq 278 becomes infinite for $\omega_B \rightarrow \omega_f$. However, if the linear response function is multiplied by $\omega_B - \omega_f$ then the resulting function $(\omega_B - \omega_f)\langle\langle A; B \rangle\rangle_{\omega_B}$ is finite for $\omega_B \rightarrow \omega_f$ and the linear response function is said to have a first-order pole with an associated residue $\lim_{\omega_B \rightarrow \omega_f} \omega_{\mu}$. Higher order response functions have a more complicated pole structure, for example, the cubic response

function $\langle\langle A; B, C, D \rangle\rangle_{\omega_{B}\omega_{O}\omega_{D}}$ has poles at $\omega_{C} + \omega_{D} = \pm \omega_{f}$ see eq 280.

The residues of the response functions provide important information about the unperturbed system. From eq 278 we obtain the single residues of the linear response function as

$$\lim_{\omega_B \longrightarrow \omega_i} (\omega_B - \omega_i) \langle \langle A; B \rangle \rangle_{\omega_B} = \langle \mathbf{0}^{(0)} | A | i \rangle \langle i | B | \mathbf{0}^{(0)} \rangle \qquad (289)$$

$$\lim_{\omega_B \longrightarrow -\omega_i} (\omega_B + \omega_i) \langle \langle A; B \rangle \rangle_{\omega_B} = - \langle 0^{(0)} | B | i \rangle \langle i | A | 0^{(0)} \rangle$$
(290)

From these residues we obtain transition-matrix elements $\langle 0^{(0)} | A | i \rangle$ of an operator *A* between the reference state $| 0^{(0)} \rangle$ and states $| i \rangle$ different from the reference state. Assuming that these transition moments are known, the transition moments of an operator *A* between two states $| i \rangle$ and $| j \rangle$ both different from the reference state $| 0^{(0)} \rangle$ are obtained from the double residues of the quadratic response function in eq 279

$$\lim_{\omega_{B} \longrightarrow -\omega_{i}} (\omega_{B} + \omega_{i}) \lim_{\omega_{C} \longrightarrow \omega_{j}} (\omega_{C} - \omega_{j}) \langle \langle A; B, C \rangle \rangle_{\omega_{B}, \omega_{C}}$$
$$= - \langle 0^{(0)} |B| i \rangle \langle i |\overline{A}| j \rangle \langle j | C | 0^{(0)} \rangle$$
(291)

Finally, the following single residue of the quadratic response function

$$\lim_{\omega_{C} \longrightarrow \omega_{i}} (\omega_{C} - \omega_{i}) \langle \langle A; B, C \rangle \rangle_{\omega_{B}, \omega_{C}}$$

$$= -P_{[A, B]}^{\omega_{A}} \sum_{j} \frac{\langle 0^{(0)} |A| j \rangle \langle j |\overline{B}| i \rangle \langle i | C | 0^{(0)} \rangle}{\omega_{A} + \omega_{j}}$$
(292)

is important for calculation of induced transitions.

3.4.6. Equations of Motion for Response Functions. In this subsection we consider a set of relations between response functions known as the equations of motion.⁷² To obtain these relations we use the time-dependent Schrödinger equation in eq 156 to write the total time derivative of an expectation value of a time-independent operator V_{B_0} in the form

$$\frac{\mathrm{d}}{\mathrm{d}t} \langle 0(t) | V_{B_0} | 0(t) \rangle = \langle \partial_t 0(t) | V_{B_0} | 0(t) \rangle + \langle 0(t) | V_{B_0} | \partial_t 0(t) \rangle$$
$$= -\mathrm{i} \langle 0(t) | [V_{B_0}, H] | 0(t) \rangle$$
(293)

which constitutes the Ehrenfest equation for the operator V_{B_0} .⁷³ Assuming that neither V_{B_0} nor any terms in the Hamiltonian contains terms involving time differentiation, eq 159 may be used to write the expectation values of eq 293 in terms of the regular wave function $|\tilde{0}\rangle$

$$\begin{aligned} \frac{\mathrm{d}}{\mathrm{d}t} \langle \tilde{0} | V_{B_0} | \tilde{0} \rangle &= \langle \partial_t \tilde{0} | V_{B_0} | \tilde{0} \rangle + \langle \tilde{0} | V_{B_0} | \partial_t \tilde{0} \rangle \\ &= -\mathrm{i} \langle \tilde{0} | [V_{B_0}, H] | \tilde{0} \rangle \end{aligned}$$
(294)

Expanding the expectation values in eq 294 in response functions using eq 270 we obtain

$$-i \sum_{B_{1}} \omega_{B_{1}} \varepsilon_{B_{1}} \langle V_{B_{0}}; V_{B_{1}} \rangle \rangle_{\omega_{B_{1}}} e^{-i\omega_{B_{1}}t} - \frac{i}{2} \sum_{B_{1}, B_{2}} \varepsilon_{B_{1}} \varepsilon_{B_{2}} (\omega_{B_{1}} + \omega_{B_{2}}) \langle \langle V_{B_{0}}; V_{B_{1}}, V_{B_{2}} \rangle \rangle_{\omega_{B_{1}}, \omega_{B_{2}}} e^{-i(\omega_{B_{1}} + \omega_{B_{2}})t} + ... = -i \sum_{B_{1}} \varepsilon_{B_{1}} \left(\langle 0^{(0)} | [V_{B_{0}}, V_{B_{1}}] | 0^{(0)} \rangle + \langle \langle [V_{B_{0}}, H_{0}]; V_{B_{1}} \rangle \rangle_{\omega_{B_{1}}} \right) e^{-i\omega_{B_{1}}t} - \frac{i}{2} \sum_{B_{1}, B_{2}} \varepsilon_{B_{1}} \varepsilon_{B_{2}} \left(2 \langle \langle [V_{B_{0}}, V_{B_{1}}]; V_{B_{2}} \rangle \rangle_{\omega_{B_{2}}} + \langle \langle [V_{B_{0}}, H_{0}]; V_{B_{1}}, V_{B_{2}} \rangle \rangle_{\omega_{B_{1}}, \omega_{B_{2}}} \right) e^{-i(\omega_{B_{1}} + \omega_{B_{2}})t} + ...$$
(295)

By equating terms of the same order in the perturbation strength we obtain the following equations of motion

$$\omega_{B_1}\langle\langle V_{B_0}; V_{B_1} \rangle\rangle_{\omega_{B_1}} = \langle 0^{(0)} | [V_{B_0}, V_{B_1}] | 0^{(0)} \rangle + \langle\langle [V_{B_0}, H_0]; V_{B_1} \rangle\rangle_{\omega_{B_1}}$$
(296)

$$(\omega_{B_1} + \omega_{B_2}) \langle \langle V_{B_0}; V_{B_1}, V_{B_2} \rangle \rangle_{\omega_{B_1}, \omega_{B_2}}$$

$$= P_{[1, 2]} \langle \langle [V_{B_0}, V_{B_1}]; V_{B_2} \rangle \rangle_{\omega_{B_2}}$$

$$+ \langle \langle [V_{B_0}, H_0]; V_{B_1}, V_{B_2} \rangle \rangle_{\omega_{B_1}, \omega_{B_2}}$$

$$(297)$$

for the linear and quadratic response functions.

The equations of motion for the response functions may be used to obtain relations between transition-matrix elements. Consider, for example, the linear response function. The equation of motion in eq 296 and the finiteness of $\langle 0^{(0)} | [A,B] | 0^{(0)} \rangle$ give the relation

$$\lim_{\omega_{B} \longrightarrow \omega_{i}} (\omega_{B} - \omega_{i}) \omega_{B} \langle \langle A; B \rangle \rangle_{\omega_{B}}$$

$$= \omega_{i} \lim_{\omega_{B} \longrightarrow \omega_{i}} (\omega_{B} - \omega_{i}) \langle \langle [A, H_{0}]; B \rangle \rangle_{\omega_{B}}$$
(298)

which in combination with eq 290 yields

$$\omega_i \langle \mathbf{0}^{(0)} | A | i \rangle = \langle \mathbf{0}^{(0)} | [A, H_0] | i \rangle \tag{299}$$

This relation also follows trivially from the fact that $|0^{(0)}\rangle$ and $|i\rangle$ are both eigenfunctions of H_0 . However, the important point is that eq 299 was here obtained only by assuming that the time development fulfills the Ehrenfest equation in eq 294. Therefore, in any approximate theory that fulfills the Ehrenfest equation, transition moments and energies identified from the poles and residues of the linear response function satisfy the equation of motion, eq 299.

3.5. Response Functions As Derivatives of the Quasi-Energy Lagrangian

In section 3.4, response functions were developed from expansion of the quasi-energy in the Hermitian form, eq 244. However, in the development of response theory for coupledcluster theory it is important to use the quasi-energy in the non-Hermitian form, eq 193. In section 3.5.1, we therefore study the expansion of the time-dependent quasi-energy Lagrangian of eq 190 and the time-averaged quasi-energy Lagrangian of eq 193 in orders of the external perturbation.

Using an approach similar to that of section 3.4, the theory may be developed to arbitrary order in the external perturbation, see refs 74 and 75 for examples from coupled-cluster theory. We here restrict

our treatment to terms up to second order in the perturbation, which allows us to identify the linear response function. It will not be assumed that the reference state is an eigenstate of the zero-order Hamiltonian, so the wave function expressions will be those of eqs 220, 221, and 223. Although the expressions for the quasi-energy corrections and the response functions obtained in section 3.5.1 differ from those developed in section 3.3, they are equivalent, as may be demonstrated explicitly by a (tedious) comparison of terms.

3.5.1. Perturbation Expansions of the Lagrangian Quasi-Energies. By analogy with the expansion of the time-dependent quasi-energy and the averaged quasi-energy in the Hermitian form, we have the following expansions for the corresponding quasi-energy Lagrangians

$$L^{c}(t) = \sum_{n=0}^{\infty} \sum_{B^{n}} \left(\prod_{k=1}^{n} \varepsilon_{B_{k}} \right) L_{B^{n}}^{c(n)} e^{-i \sum_{k=1}^{n} \omega_{B_{k}} t}$$
(300)

$$L(t) = \sum_{n=0}^{\infty} \sum_{B^n} \left(\prod_{k=1}^n \varepsilon_{B_k} \right) L_{B^n}^{(n)} \mathrm{e}^{-\mathrm{i} \sum_{k=1}^n \omega_{B_k} t}$$
(301)

$$\mathscr{L} = \sum_{n} \sum_{B^{n}: \sum_{k} \omega_{B_{k}} = 0} \left(\prod_{k} \varepsilon_{B_{k}} \right) \mathscr{L}_{B^{n}}^{(n)}$$
(302)

Given that L(t) is the real part of $L^{c}(t)$ we obtain from eqs 300 and 301 the simple relation

$$L_{B^{n}}^{(n)} = \frac{1}{2} \left(L_{B^{n}}^{c(n)} + L_{-B^{n}}^{c(n)*} \right) = \frac{1}{2} C^{\pm} L_{B^{n}}^{c(n)}$$
(303)

where $-B^n$ contains the elements $-B_i$ and where the operator C^{\pm} is defined by

$$C^{\pm}X_{B_n} = X_{B^n} + X^*_{-B^n} \tag{304}$$

By analogy with eq 262, the terms in the time-dependent and time-averaged quasi-energy Lagrangian are related as

$$\mathscr{L}_{B^{n}}^{(n)} = \begin{cases} \frac{1}{2} C^{\pm} L_{B^{n}}^{c(n)} & \text{if } \sum_{k} \omega_{B_{k}} = 0\\ 0 & \text{otherwise} \end{cases}$$
(305)

The expansions of the quasi-energy in the Hermitian and non-Hermitian forms in eqs 261 and 302, respectively, are identical for all field strengths. The two expansions must therefore be identical term by term and have the same derivatives

$$\mathscr{L}_{B^n}^{(n)} = \mathscr{L}_{B^n}^{(n)} \tag{306}$$

Furthermore, as the expansion coefficients $\mathscr{D}_{B^n}^{(n)}$ are proportional to the response functions, see eqs 274–276, the response functions may also be expressed in terms of $\mathscr{L}_{B^n}^{(n)}$, yielding the following linear, quadratic, and cubic response functions in the Lagrangian form

$$\langle\langle V_{B_0}; V_{B_1} \rangle\rangle_{\omega_{B_1}} = \frac{d^2 \mathscr{L}}{d\varepsilon_{B_0} d\varepsilon_{B_1}} = 2 \mathscr{L}_{B_0, B_1}^{(2)} = C^{\pm} L_{B_0, B_1}^{c(2)}$$
(307)

$$\langle\langle V_{B_0}; V_{B_1}, V_{B_2} \rangle\rangle_{\omega_{B_1}, \omega_{B_2}} = \frac{d^3 \mathscr{L}}{d\varepsilon_{B_0} d\varepsilon_{B_1} d\varepsilon_{B_2}} = \frac{3!}{2} C^{\pm} L^{c(3)}_{B_0, B_1, B_2}$$
(308)

to be compared with the expressions for response functions in the Hermitian form in eqs 274-276.

To obtain the expressions for the terms in the expansion of the quasienergy Lagrangian, the coefficients of the multiplier state $|\overline{0}\rangle$ of eq 188 must be expanded in the perturbation. By analogy with the expansion of the wave function parameters the multipliers are expanded as

$$\overline{\mathbf{c}}(t) = \overline{\mathbf{c}}^{(0)} + \sum_{B_1} \varepsilon_{B_1} \overline{\mathbf{c}}_{B_1}^{(1)} e^{-i\omega_{B_1}t} + \sum_{B_1B_2} \varepsilon_{B_1} \varepsilon_{B_2} \overline{\mathbf{c}}_{B_1, B_2}^{(2)} e^{-i(\omega_{B_1} + \omega_{B_2})t} + ... + \sum_{n} \sum_{B^n} (\prod_{k=1}^n \varepsilon_{B_k}) \overline{\mathbf{c}}_{B^n}^{(n)} e^{-i\sum_{k=1}^n \omega_{B_k}t} + ...$$
(310)

We are now ready to derive the expressions for the zero-, first-, and second-order terms in the expansion of \mathcal{L} . As the Lagrangian quasienergy \mathcal{L} is variational in the wave function and multiplier parameters, these parameters to order *n* determine \mathcal{L} to order 2n + 1. Furthermore, the multiplier parameters occur linearly in \mathcal{L} , allowing for an additional reduction discussed in section 3.6: the multiplier parameters to order *n* determine the time-averaged Lagrangian quasi-energy to order 2n + 2. In short, to obtain the Lagrangian to second order, we need the wave function parameters to first order and the multipliers to zero order.

Introducing the expansion of eq 310 in the multiplier state $\langle \overline{0} |$ of eq 188 and the expansion of eq 218 in $\langle \hat{0} |$ we may expand the timeaveraged quasi-energy in orders of the perturbation. Using this result we obtain from the expansion of eq 189 and from eq 305 the terms contributing to the lowest order quasi-energy terms as

$$\mathscr{Z}_{0}^{(0)} = \operatorname{Re}\left(\langle \mathbf{R}|H_{0}|\hat{\mathbf{0}}^{(0)}\rangle + \langle \overline{\mathbf{0}}^{(0)}|H_{0}|\hat{\mathbf{0}}^{(0)}\rangle\right)$$
(311)

$$\mathscr{Z}_{B_{0}}^{(1)} = \frac{1}{2} C^{\pm} \left(\langle \mathbf{R} | V_{B_{0}} | \hat{\mathbf{0}}^{(0)} \rangle + \langle \overline{\mathbf{0}}^{(0)} | V_{B_{0}} | \hat{\mathbf{0}}^{(0)} \rangle \right)$$
(312)

$$\mathscr{L}_{B_{0},B_{1}}^{(2)} = \frac{1}{2} C^{\pm} P_{[0,1]} \bigg[\langle \mathbf{R} | V_{B_{0}} | \hat{\mathbf{0}}_{B_{1}}^{(1)} \rangle + \langle \overline{\mathbf{0}}^{(0)} | V_{B_{0}} | \hat{\mathbf{0}}_{B_{1}}^{(1)} \rangle - \big(\sum_{i} \overline{c}_{i}^{(0)*} c_{B_{1},i}^{(1)} \big) \bigg(\langle \mathbf{R} | H_{0} - \omega_{B_{0}} | \hat{\mathbf{0}}_{B_{0}}^{(1)} \rangle + \langle \mathbf{R} | V_{B_{0}} | \hat{\mathbf{0}}^{(0)} \rangle \bigg) \bigg]$$
(313)

These expressions provide alternative forms for the lowest order corrections to the time-averaged quasi-energy, which by definition are identical to those obtained previously. As the zero-order state is an eigenstate of H_0 (eq 208), the zero-order term becomes

$$\mathcal{Z}_{0}^{(0)} = E_{0} \tag{314}$$

The first-order correction of eq 312 must similarly be identical to that of eq 263, and insertion of eq 313 into eq 307 gives an alternative form of the linear response equation of eq 278. The equivalence between the forms of the first-order corrections is explicitly demonstrated in the following subsection, where we derive the expression for the zero-order Lagrange multipliers. **3.5.2. Zero-Order Lagrange Multipliers.** The zero-order Lagrange multipliers, which determine the quasi-energy Lagrangian to second order, are most conveniently identified by invoking the stationarity of the time-averaged zero-order Lagrangian of eq 311 with respect to the variations in the zero-order wave function parameters

$$\frac{\partial}{\partial c_k^{(0)}} \left(\sum_j \langle \mathbf{R} | H_0 | j \rangle c_j^{(0)} + \sum_{ij} \overline{c_i^{(0)*}} \langle i | H_0 | j \rangle c_j^{(0)} - \left(\sum_i \overline{c_i^{(0)*}} c_i^{(0)} \right) \sum_j \langle \mathbf{R} | H_0 | j \rangle c_j^{(0)} \right) = \mathbf{0}$$
(315)

which in terms of the Jacobian ${\bf A}$ of eq 214 may be written in the form

$$\langle \mathbf{R}|H_0|k\rangle + \sum_i \overline{c}_i^{(0)*} A_{ik} = 0$$
(316)

Introducing the shifted Hamiltonian H^b of eq 227, the zero-order multiplier equations may be written in matrix form

$$\begin{pmatrix} 1 & \overline{\mathbf{c}}^{(0)\dagger} \end{pmatrix} \mathbf{H}^{\mathsf{b}} = \mathbf{0} \tag{317}$$

showing that $\left(\overline{\mathbf{c}}^{(0)}\right)$ is a left eigenvector of the shifted

Hamiltonian $\mathbf{H}^{\mathbf{b}}$ with zero eigenvalue, corresponding to an eigenvalue E_0 of the unshifted Hamiltonian H_0 . Assuming that this eigenvalue is nondegenerate, we conclude that the state

defined by
$$\begin{pmatrix} 1 \\ \overline{c}^{(0)} \end{pmatrix}$$
 is proportional to the zero-order state

$$|\mathbf{R}\rangle + \sum_{i} \overline{c}_{i}^{(0)} |\hat{i}^{(0)}\rangle = \alpha \left(|\mathbf{R}\rangle + \sum_{i} c_{i}^{(0)} |i\rangle\right)$$
(318)

where α is some scalar. To determine α we insert the expansion in eq 206 into eq 318, enabling us to make the identifications

$$\overline{c}_{i}^{(0)} = \alpha c_{i}^{(0)}, \ \alpha = \frac{1}{1 + \sum_{i} |c_{i}^{(0)}|^{2}} = \frac{1}{\langle \hat{0}^{(0)} | \hat{0}^{(0)} \rangle}$$
(319)

The zero-order multiplier state $|\overline{0}{}^{(0)}\rangle$ occurs in the following together with the reference state, making it convenient to introduce the (zero-order) lambda state

$$|\Lambda^{(0)}\rangle = |\mathbf{R}\rangle + \sum_{i} \overline{c_i}^{(0)} |\hat{i}^{(0)}\rangle$$
(320)

which from eq 319 may be written in the form

$$|\Lambda^{(0)}\rangle = \frac{1}{\langle \hat{\boldsymbol{0}}^{(0)} | \hat{\boldsymbol{0}}^{(0)} \rangle} | \hat{\boldsymbol{0}}^{(0)} \rangle \tag{321}$$

which shows that $\langle \Lambda^{(0)} | \hat{0}^{(0)} \rangle = 1$.

As a first use of the explicit forms of the Lagrange multipliers we show that $\mathscr{L}^{(1)}$ of eq 312 is identical to $\mathscr{L}^{(1)}$ of eq 263, thereby also demonstrating that $\mathscr{L}^{(1)}$ is independent of the choice of the reference state even though the left- and right-hand states of eq 312 both depend on this state. The equivalence is easily established using eqs 237, 320, and 321 and the identification in eq 263

$$\begin{aligned} \mathscr{Z}_{B_{0}}^{(1)} &= \frac{1}{2} C^{\pm} \langle \Lambda^{(0)}(\mathbf{R}) | V_{B_{0}} | \hat{\mathbf{0}}^{(0)}(\mathbf{R}) \rangle = \frac{\langle \hat{\mathbf{0}}^{(0)}(\mathbf{R}) | V_{B_{0}} | \hat{\mathbf{0}}^{(0)}(\mathbf{R}) \rangle}{\langle \hat{\mathbf{0}}^{(0)}(\mathbf{R}) | \hat{\mathbf{0}}^{(0)}(\mathbf{R}) \rangle} &= \langle \mathbf{0}^{(0)} | V_{B_{0}} | \mathbf{0}^{(0)} \rangle = \mathbf{c}^{(0)} \mathbf{V}_{B_{0}} \mathbf{c}^{(0)} = \mathscr{Q}_{B_{0}}^{(1)} \end{aligned}$$
(322)

The higher order terms, such as the second-order corrections $\mathcal{L}^{(2)}$ of eq 313 and $\mathcal{Q}^{(2)}$ in eq 264, may similarly be shown to be identical.

3.6. Elimination Rules

In the calculation of the terms in the expansion of the quasienergy in sections 3.3 and 3.5 we made repeated use of the 2n + 1and 2n + 2 rules, which state that wave function parameters and Lagrange multipliers to order *n* determine the quasi-energy to order 2n + 1 and 2n + 2, respectively. These rules are developed in this section, where we also briefly discuss alternative elimination rules. We adopt a formalism that includes both exact and approximate wave functions. For an early exposition of these rules, see ref 76.

3.6.1. 2n + 1 Rule. Consider a time-averaged quasi-energy $\mathscr{Q}(\varepsilon,\lambda)$ that depends on the external parameters ε and the wave function parameters λ . The optimized quasi-energy is given by

$$\mathscr{Q}(\varepsilon) = \mathscr{Q}(\varepsilon, \lambda(\varepsilon)) \tag{323}$$

where the optimized wave function parameters $\lambda(\varepsilon)$ are determined from the variational condition

$$\frac{\partial \mathcal{Q}(\varepsilon, \lambda(\varepsilon))}{\partial \lambda(\varepsilon)} = 0 \tag{324}$$

In response theory, we are interested in the expansion of $\mathcal{Q}(\varepsilon,\lambda(\varepsilon))$ in orders of ε

$$\mathcal{Q}(\varepsilon) = \mathcal{Q}^{(0)} + \mathcal{Q}^{(1)}\varepsilon + \frac{1}{2}\mathcal{Q}^{(2)}\varepsilon^2 + \dots$$
(325)

In particular, we wish to determine expressions for the perturbed quasi-energies $\mathcal{L}^{(n)}$ in terms of the perturbed wave function parameters $\lambda^{(k)}$ of the expansion

$$\lambda(\varepsilon) = \lambda^{(0)} + \lambda^{(1)}\varepsilon + \frac{1}{2}\lambda^{(2)}\varepsilon^2 + \dots$$
(326)

To determine the perturbed quasi-energies, we expand the quasienergy in orders of ε and λ

$$\mathcal{Q}(\varepsilon, \lambda) = \mathcal{Q}^{(00)} + \mathcal{Q}^{(10)}\varepsilon + \mathcal{Q}^{(01)}\lambda + \frac{1}{2}\mathcal{Q}^{(20)}\varepsilon^{2} + \mathcal{Q}^{(11)}\varepsilon\lambda + \frac{1}{2}\mathcal{Q}^{(02)}\lambda^{2} + \dots$$
(327)

where we introduced a short-hand notation for quasi-energy derivatives

$$\mathcal{Q}^{(mn)} = \frac{\partial^{m+n} \mathcal{Q}(\varepsilon, \lambda)}{\partial \varepsilon^m \partial \lambda^n} |_{\varepsilon=0, \ \lambda=0}$$
(328)

By inserting the expansion in eq 326 into eq 327 and collecting terms of the same order of ε we obtain to lowest orders

$$\mathcal{Q}^{(0)} = \mathcal{Q}^{(00)} + \mathcal{Q}^{(01)}\lambda^{(0)} + \dots$$
(329)

$$\mathcal{Q}^{(1)} = \mathcal{Q}^{(10)} + \mathcal{Q}^{(11)} \lambda^{(0)} + \mathcal{Q}^{(01)} \lambda^{(1)} + \mathcal{Q}^{(02)} \lambda^{(0)} \lambda^{(1)} + \dots$$
(330)

$$\mathcal{Q}^{(2)} = \mathcal{Q}^{(20)} + 2\mathcal{Q}^{(21)}\lambda^{(0)} + 2\mathcal{Q}^{(11)}\lambda^{(1)} + 2\mathcal{Q}^{(12)}\lambda^{(0)}\lambda^{(1)} + \mathcal{Q}^{(01)}\lambda^{(2)} + \mathcal{Q}^{(02)}\lambda^{(0)}\lambda^{(2)} + \mathcal{Q}^{(02)}\lambda^{(1)}\lambda^{(1)} + \mathcal{Q}^{(03)}\lambda^{(0)}\lambda^{(1)}\lambda^{(1)} + \dots$$
(331)

 $a(\mathbf{1})$

a(20)

where the omitted terms are of second and higher orders in $\lambda^{(0)}$. From order considerations we note that $\lambda^{(n)}$ does not occur in quasi-energies of orders less than *n*, that $\lambda^{(n)}$ occurs only linearly in quasi-energies up to order 2n - 1, and that $\lambda^{(n)}$ appears nonlinearly in quasi-energies of order 2n and greater. The zeroorder parameters $\lambda^{(0)}$ have a special status and may appear to infinite order, depending on the form of $\mathcal{Q}(\varepsilon, \lambda)$. The structure of the perturbed quasi-energies is therefore

$$\mathcal{Q}^{(n)} = \mathcal{Q}^{(n)}(\underbrace{\lambda^{(0)}}_{\text{infinite}}, \underbrace{\lambda^{(1)}, \dots, \lambda^{(n/2)}}_{\text{parameters}}, \underbrace{\lambda^{([n/2]+1)}, \dots, \lambda^{(n)}}_{\text{parameters}})$$
(332)

where $\lfloor n/2 \rfloor$ is the greatest integer less than or equal to n/2. The quasi-energy $\mathcal{Q}^{(n)}$ thus depends linearly on $\lambda^{(k)}$ for $[n/2] < k \le n$.

We now wish to identify the stationary conditions for $\mathcal{Q}^{(n)}$ with respect to variations in the perturbed wave functions $\lambda^{(k)}$. Using the chain rule and the fact that $\lambda^{(k)}$ is independent of ε we obtain

$$\frac{\partial \mathcal{Q}^{(n)}}{\partial \lambda^{(k)}} = \frac{\partial^{n+1} \mathcal{Q}(\varepsilon, \lambda(\varepsilon))}{\partial \lambda^{(k)} \partial \varepsilon^{n}} \bigg|_{\varepsilon=0} = \frac{\partial^{n}}{\partial \varepsilon^{n}} \frac{\partial \mathcal{Q}(\varepsilon, \lambda(\varepsilon))}{\partial \lambda^{(k)}} \bigg|_{\varepsilon=0}$$
$$= \frac{\partial^{n}}{\partial \varepsilon^{n}} \frac{\partial \lambda(\varepsilon)}{\partial \lambda^{(k)}} \frac{\partial \mathcal{Q}(\varepsilon, \lambda(\varepsilon))}{\partial \lambda(\varepsilon)} \bigg|_{\varepsilon=0} = \frac{\partial^{n}}{\partial \varepsilon^{n}} \frac{\varepsilon^{k}}{k!} \frac{\partial \mathcal{Q}(\varepsilon, \lambda(\varepsilon))}{\partial \lambda(\varepsilon)} \bigg|_{\varepsilon=0}$$
(333)

where we inserted eq 326. From the variational condition in eq 324 it now follows that the *n*th-order quasi-energy $\mathcal{Q}^{(n)}$ is stable toward variations in all perturbed parameters $\lambda^{(k)}$

$$\frac{\partial \mathcal{Q}^{(n)}}{\partial \lambda^{(k)}} = 0 \tag{334}$$

According to eq 332 $\mathcal{Q}^{(n)}$ depends linearly on $\lambda^{(k)}$ with $k > \lfloor n/2 \rfloor$, while according to eq 334 $\mathcal{Q}^{(n)}$ is stationary with respect to $\lambda^{(k)}$. The coefficient multiplying $\lambda^{(k)}$ in the expression for $\mathcal{Q}^{(n)}$ must therefore be zero. We conclude that $\mathcal{Q}^{(n)}$ must be independent of $\lambda^{(k)}$ for $k > \lfloor n/2 \rfloor$

$$\mathcal{Q}^{(n)} = \mathcal{Q}^{(n)}(\boldsymbol{\lambda}^{(0)}, \, \boldsymbol{\lambda}^{(1)}, \, \dots \boldsymbol{\lambda}^{([n/2])})$$
(335)

We have now established the 2n + 1 rule: the wave function to order *n* determines the quasi-energy to order 2n + 1. Applied to the lower order quasi-energies in eqs 329-331 we obtain (in a parametrization with $\lambda^{(0)} = 0$) the following simplified expressions for the energies

$$\mathcal{Q}^{(0)} = \mathcal{Q}^{(00)} \tag{336}$$

$$\mathcal{Q}^{(1)} = \mathcal{Q}^{(10)}$$
 (337)

$$\mathscr{Q}^{(2)} = \mathscr{Q}^{(20)} + 2\mathscr{Q}^{(11)}\lambda^{(1)} + \mathscr{Q}^{(02)}\lambda^{(1)}\lambda^{(1)}$$
(338)

$$\mathcal{Q}^{(3)} = \mathcal{Q}^{(30)} + 3\mathcal{Q}^{(21)}\lambda^{(1)} + 3\mathcal{Q}^{(12)}\lambda^{(1)}\lambda^{(1)} + \mathcal{Q}^{(03)}\lambda^{(1)}\lambda^{(1)}\lambda^{(1)}$$
(339)

We note how the first-order wave function determines the quasienergy to third order.

Having established the structure of the quasi-energies $\mathcal{Q}^{(n)}$ in eq 335, we shall now use the variational conditions in eq 334 to determine $\lambda^{(k)}$. First, assuming that $k \leq n$ we obtain from eqs 333 and 334 the stationary conditions

$$\frac{\partial \mathcal{Q}^{(n)}}{\partial \lambda^{(k)}} = \frac{\partial^n}{\partial \varepsilon^n} \frac{\varepsilon^k}{k!} \frac{\partial \mathcal{Q}(\varepsilon, \lambda(\varepsilon))}{\partial \lambda(\varepsilon)} \bigg|_{\varepsilon=0} = \binom{n}{k} \frac{\partial^{n-k}}{\partial \varepsilon^{n-k}} \frac{\partial \mathcal{Q}(\varepsilon, \lambda(\varepsilon))}{\partial \lambda(\varepsilon)} \bigg|_{\varepsilon=0} = 0$$
(340)

where we used the Leibniz rule for the differentiation of a product of two functions

$$\frac{\partial^{n}[f(\varepsilon)g(\varepsilon)]}{\partial\varepsilon^{n}} = \sum_{i=0}^{n} \binom{n}{i} \frac{\partial^{i}f(\varepsilon)}{\partial\varepsilon^{i}} \frac{\partial^{n-i}g(\varepsilon)}{\partial\varepsilon^{n-i}}$$
(341)

Next, we note that the stationary conditions in eq 340 are not all independent, depending only on the value of n - k rather than separately on n and k. We therefore only need to consider the conditions

$$\frac{\partial \mathcal{Q}^{(n)}}{\partial \lambda^{(0)}} = \left. \frac{\partial^n}{\partial \varepsilon^n} \frac{\partial \mathcal{Q}(\varepsilon, \lambda(\varepsilon))}{\partial \lambda(\varepsilon)} \right|_{\varepsilon=0} = 0$$
(342)

By satisfying the stationary conditions with respect to $\lambda^{(0)}$ in eq 342 for all $n \ge 0$ we ensure that the expansion of $\partial \mathcal{Q}(\varepsilon, \lambda(\varepsilon)) / \partial \lambda(\varepsilon)$ about $\varepsilon = 0$ vanishes to all orders. The conditions in eq 342 for all $n \ge 0$ are therefore equivalent to the conditions in eq 324 for all values of ε .

Let us now consider the structure of the *n*th-order response equations $\partial \mathcal{Q}^{(n)} / \partial \lambda^{(0)} = 0$ of eq 342 in more detail. Differentiating eq 342 with respect to $\lambda^{(k)}$, using the chain rule, and inserting eq 326 we obtain

$$\frac{\partial}{\partial \lambda^{(k)}} \frac{\partial \mathcal{Q}^{(n)}}{\partial \lambda^{(0)}} \bigg|_{\varepsilon=0} = \frac{\partial^{n}}{\partial \varepsilon^{n}} \left(\frac{\partial \lambda(\varepsilon)}{\partial \lambda^{(k)}} \frac{\partial}{\partial \lambda(\varepsilon)} \frac{\partial \mathcal{Q}(\varepsilon, \lambda(\varepsilon))}{\partial \lambda(\varepsilon)} \right) \bigg|_{\varepsilon=0}$$
$$= \frac{\partial^{n}}{\partial \varepsilon^{n}} \left(\frac{\varepsilon^{k}}{k!} \frac{\partial^{2} \mathcal{Q}(\varepsilon, \lambda(\varepsilon))}{\partial \lambda(\varepsilon)^{2}} \right) \bigg|_{\varepsilon=0}$$
(343)

which may be simplified by application of the Leibniz rule in eq 341 to give

$$\frac{\partial}{\partial \lambda^{(k)}} \frac{\partial \mathcal{Q}^{(n)}}{\partial \lambda^{(0)}} \bigg|_{\varepsilon=0} = \begin{cases} \binom{n}{k} \frac{\partial^{n-k}}{\partial \varepsilon^{n-k}} \frac{\partial^2 \mathcal{Q}(\varepsilon, \lambda(\varepsilon))}{\partial \lambda(\varepsilon)^2} \bigg|_{\varepsilon=0} & (k < n) \\ \mathcal{Q}^{(02)} & (k = n) \\ 0 & (k > n) \end{cases}$$
(344)

We conclude that the stationary conditions in eq 342 may be written in the form $^{76}\,$

$$\frac{\partial \mathcal{Q}^{(n)}}{\partial \lambda^{(0)}} = \mathcal{Q}^{(02)} \lambda^{(n)} + R^{(n)} (\lambda^{(0)}, ..., \lambda^{(n-1)}) = 0 \quad (345)$$

where $R^{(n)}$ is a polynomial in $\lambda^{(k)}$ with k < n. To a given order n in the perturbation the response equations

$$\mathscr{Q}^{(02)}\lambda^{(k)} = -R^{(k)}(\lambda^{(0)}, ..., \lambda^{(k-1)})$$
(346)

are solved recursively for $k \le n$ beginning with $\lambda^{(1)}$, the solution $\lambda^{(k)}$ depending on the solutions $\lambda^{(l)}$ with l < k.

3.6.2. 2n + 2 Rule. In the intermediate normalization, the time-dependent quasi-energy Q(t) in eq 183 is not variationally determined and is therefore not suited for calculation of response functions. In contrast, the time-dependent Lagrangian L(t) in eq 190 provides us with an equivalent, variational formulation of the energy in the intermediate normalization, suitable for response theory, noting that the corresponding time-averaged quasi-energy \mathcal{L} in eq 193 satisfies the stationary conditions in eq 194. Likewise, in approximate theories the approximate quasi-energy is often not variational.⁶⁶ Rather, the wave function parameters satisfy some subsidiary condition $e(\varepsilon, \lambda(\varepsilon)) = 0$ such as the amplitude equations in coupled-cluster theory. In such cases, a variational quasi-energy Lagrangian may be constructed of the general form^{66,77}

$$\mathcal{L}(\varepsilon) = \mathcal{L}(\varepsilon, \lambda(\varepsilon), \overline{\lambda}(\varepsilon))$$
$$= \mathcal{L}(\varepsilon, \lambda(\varepsilon)) + \overline{\lambda}(\varepsilon)e(\varepsilon, \lambda(\varepsilon))$$
(347)

where $\overline{\lambda}$ (ε) is a collection of Lagrange multipliers. For the Lagrangian in the intermediate normalization in eq 193, $e(\varepsilon,\lambda(\varepsilon))$ and $\overline{\lambda}$ (ε) correspond to eqs 185 and 188, respectively. The Lagrangian is by construction stationary in the multipliers as well as in the original wave function parameters (omitting arguments for brevity)

$$\frac{\partial \mathcal{E}}{\partial \bar{\lambda}} = e = 0 \tag{348}$$

$$\frac{\partial \mathcal{L}}{\partial \lambda} = \frac{\partial \mathcal{Q}}{\partial \lambda} + \bar{\lambda} \frac{\partial e}{\partial \lambda} = 0$$
(349)

The variation with respect to $\overline{\lambda}$ in eq 348 gives the equation for λ and is trivially satisfied, whereas the variation in λ in eq 349 yields a set of linear equations that determine the multipliers. By analogy with the quasi-energy expansion in eq 325 we expand the Lagrangian in the perturbation strength

$$\mathscr{L}(\varepsilon) = \mathscr{L}^{(0)} + \mathscr{L}^{(1)}\varepsilon + \frac{1}{2}\mathscr{L}^{(2)}\varepsilon^2 + \dots$$
(350)

By construction, perturbation-strength derivatives of a quasienergy and its corresponding Lagrangian function are identical. However, by constructing a Lagrangian function we have the freedom to eliminate computationally expensive parameters by means of the variational conditions in eqs 348 and 349. We shall now examine the elimination rules for the quasi-energy expressed as a Lagrangian in eq 347.

Since the Lagrangian satisfies the variational conditions in eq 349 with respect to the wave function parameters λ , the 2n + 1 rule for these parameters follow in the same way as for the variational quasi-energy in section 3.6.1. Thus, the responses $\lambda^{(k)}$

with $k \leq \lfloor n/2 \rfloor$ are sufficient to evaluate the *n*th-order Lagrangian $\mathcal{L}^{(n)}$. Likewise, it is possible to eliminate some of the perturbed multipliers $\overline{\lambda}^{(i)}$ from $\mathcal{J}^{(n)}$, noting that the Lagrangian is both linear and stationary in the multipliers, see eqs 347 and 348. However, we cannot eliminate all multipliers $\overline{\lambda}^{(i)}$ independently because the preceding elimination of $\dot{\lambda}^{(k)}$ (in accordance with the 2n + 1 rule) has modified some of the terms containing $\overline{\lambda}^{(i)}$ $(i \le n - k)$ in eq 347. For example, elimination of $\lambda^{(k)}$ with $[n/2] < k \le n$ removes the factor multiplying $\lambda^{(k)}$ from $\mathcal{L}^{(n)}$. Since this factor contains $\overline{\lambda}^{(i)}$ and $\lambda^{(i)}$ with $i \leq n - k$, $\mathcal{L}^{(n)}$ is no longer variational in these parameters. In short, removal of $\lambda^{(k)}$ with $[n/2] < k \le n$ from $\mathcal{L}^{(n)}$ precludes subsequent removal of $\overline{\lambda}^{(i)}$ with $i \leq n - k$, binding them to $\mathcal{L}^{(n)}$.

Having made this observation we now examine which multipliers $\overline{\lambda}^{(i)}$ are affected when the perturbed wave function parameters $\lambda^{(k)}$ are eliminated from the quasi-energy Lagrangian in accordance with the 2n + 1 rule, following Kristensen et al.⁷⁸ For clarity of presentation,

- the 2n + 1 rule, following Kristensen et al. ⁷⁸ For clarity of presentation, we consider even- and odd-order quasi-energies separately. $\mathcal{A}^{(2n+1)}$: Application of the 2n + 1 rule to $\mathcal{A}^{(2n+1)}$ removes $\lambda^{(n+1)}$ and higher order parameters, binding $\overline{\lambda}^{(k)}$ with $k \leq 2n + 1 (n + 1) = n$ to $\mathcal{A}^{(2n+1)}$. Since $\overline{\lambda}^{(k)}$ with k > n may still be removed from $\mathcal{A}^{(2n+1)}$ by the 2n + 1 rule, $\overline{\lambda}^{(k)}$ with $k \leq n$ are sufficient to determine $\mathcal{A}^{(2n+1)}$. $\mathcal{A}^{(2n+2)}$: Application of the 2n + 1 rule to $\mathcal{A}^{(2n+2)}$ removes $\lambda^{(n+2)}$ and higher order parameters, binding $\overline{\lambda}^{(k)}$ with $k \leq 2n + 2 (n + 2) = n$ to $\mathcal{A}^{(2n+2)}$. Since $\overline{\lambda}^{(k)}$ with k > n may still be removed from $\mathcal{A}^{(2n+2)}$ by the 2n + 1 rule, $\overline{\lambda}^{(k)}$ with $k \leq n$ are sufficient to determine $\mathcal{A}^{(2n+2)}$. We have new established the 2n + 2 rule for Lagrange

We have now established the 2n + 2 rule for Lagrange multipliers: the multipliers to order n determine the Lagrangian to order 2n + 2.^{76,77} Together with the 2n + 1 rule, the 2n + 2 rule is important in reducing the cost of calculating perturbed quasienergies of variational and nonvariational methods.

3.6.3. Alternative Elimination Schemes. In the discussion of the 2n + 1 and 2n + 2 rules we eliminated response parameters for the full set of N perturbations responsible for a given molecular property. Alternatively, these elimination rules may be applied to a subset K of the full set of N perturbations that determine the property of interest.⁷⁸ In this way, we eliminate as many parameters as possible for the first subset of K perturbations at the price of calculating higher order parameters for the remaining subset of N - K perturbations. Such a strategy is useful when the molecular property involves a mixture of extensive and intensive perturbations.

For extensive perturbations the number of components is proportional to the system size (e.g., the $3N_{\rm atoms}$ geometric or nuclear-magnetic perturbations of a molecule), whereas the number of intensive perturbations is system independent (e.g., the three electric or magnetic dipole operators of a molecule). It is then advantageous to eliminate as many response parameters as possible involving the computationally expensive extensive perturbations. Consider, for example, a third-order molecular property expressed as a quasi-energy Lagrangian derivative involving one extensive differentiation (with $3N_{\text{atoms}}$ components) and two intensive differentiations (with 3 components), schematically denoted as \mathcal{J}^{eii} . Applying the 2n + 1 rule to all perturbations, all $2(3N_{\text{atoms}} + 3)$ first-order parameters λ^{e} , λ^{i} , $\overline{\lambda}^{\text{e}}$, and $\overline{\lambda}^{\text{i}}$ must be determined (disregarding zero-order parameters), making the calculations expensive for large molecules. Instead, by applying the 2n + 1 rule to the extensive perturbations only, we may eliminate all occurrences of λ^{e} and $\overline{\lambda}^{e}$ from L^{eii} at the small cost of not being able to remove any intensive perturbations and therefore having to determine 6 first-order parameters λ^1 and $\overline{\lambda}^1$ and 12 second-order parameters λ^{ii} and $\overline{\lambda}^{ii}$. In this manner the number of response equations to be solved is 18 (rather than $6N_{\text{atoms}} + 6$), independently of the number of atoms in the system, reducing the computational cost for large molecules.79,80

The above development described the formal analysis required to minimize the number of response equations to be solved. For even-order response functions we may alternatively develop expressions that ensure that the errors of these response functions are quadratic in the error of the highest wave function corrections. Following the development of Sellers for timeindependent perturbations⁸¹ this has been discussed for the linear response function by Koch and Harrison.⁸²

3.7. Damped Response Theory

Molecular response functions become singular when one or more of the optical frequencies are equal to an excitation energy, as discussed in section 3.5.2. These singularities give an unphysical behavior of the calculated molecular properties in the resonance region such as divergent dispersion curves and infinitely narrow absorption peaks, reflecting the infinite lifetime of excited states in standard molecular response theory. For a rigorous treatment of molecular properties near resonance, the finite lifetime of excited states must be accounted for. However, this is not readily achieved in traditional response theory. Instead, finite lifetimes are in damped response theory imposed by multiplying the excited states by a damping factor.⁷¹ The resulting phenomenological lifetimes are related to the widths of the absorption peaks by the energy-time uncertainty principle.

From a practical point of view, introduction of finite excited-state lifetimes in damped response theory allows broadened absorption (or dispersion) spectra in any frequency region to be calculated directly, without explicit reference to the transition strengths of individual excited states. The damped response functions describe standard absorption transition strengths, such as the residue of the linear response function in eq 290, with superimposed absorption (or dispersion) line-shape functions. Direct calculation of absorption spectra in damped response theory is particularly advantageous when many transitions need to be addressed in the traditional residue approach. Important cases include electronic transitions to high-lying excited states, for example, in X-ray spectra,^{83,84} and absorption spectra of large molecules in general,⁸⁵ where the excited-state density is particularly high. Another important use of damped response theory is calculation of the polarizability at different imaginary frequencies, giving direct access to the C_6 dispersion coefficients that govern weak intermolecular interactions.^{82,86–88}

In their complex propagator approach, Norman et al.⁸⁹ introduced a damping term into the Liouville equation to account for finite lifetimes and identified damped response functions from this equation. Using a quasi-energy formulation Kristensen et al.90 introduced finite lifetimes directly into the response functions in terms of complex excitation energies. The two formulations are equivalent, giving the same damped response functions. We note in passing that Jensen, Autschbach, and coworkers also presented linear response functions using finite lifetimes at the time-dependent density-functional level of theory,⁹¹ applying it, for instance, to the study of resonance Raman scattering.⁹² In all formulations finite excited-state lifetimes are introduced by means of an empirical damping parameter. This parameter is input to the response calculation; the

theory provides no recipe for determining the damping parameter. Below, we discuss damped response theory in terms of complex excitation energies.

3.7.1. Phenomenological Damping of Excited States. Let us assume that the excited-state wave function $|n\rangle$ is a solution to the time-independent Schrödinger equation

$$H_0|n\rangle = E_n|n\rangle \tag{351}$$

The corresponding time-dependent excited state given by

$$|n(t)\rangle = e^{-iE_n t}|n\rangle \tag{352}$$

then trivially satisfies the time-dependent Schrödinger equation

$$i\partial_t |n(t)\rangle = H_0 |n(t)\rangle = E_n |n(t)\rangle$$
 (353)

In particular, the norm of $|n(t)\rangle$ is constant in time

$$\langle n(t)|n(t)\rangle = \langle n|n\rangle = 1$$
 (354)

Therefore, no decay occurs from the excited state to the ground state (or to other excited states). In other words, the lifetime of the excited state $|n(t)\rangle$ is infinite.

In reality, an excited state has a finite lifetime but this lifetime cannot be accounted for by the Hamiltonian H_0 . Instead, a phenomenological description of the lifetime may be obtained by introducing a damped excited state $|\overline{n}(t)\rangle$ according to

$$|\overline{n}(t)\rangle = e^{-\gamma t}|n(t)\rangle = e^{-i(E_n - i\gamma)t}|n\rangle$$
 (355)

The norm of the damped excited state $|\overline{n}(t)\rangle$ decays exponentially in time

$$\langle \overline{n}(t) | \overline{n}(t) \rangle = e^{-2\gamma t}$$
 (356)

and $(2\gamma)^{-1}$ may therefore be interpreted as the effective lifetime of the excited state. Unlike the undamped state $|n(t)\rangle$ in eq 353, the damped state $|\overline{n}(t)\rangle$ does not possess a well-defined real energy

$$i\partial_t |\bar{n}(t)\rangle = (E_n - i\gamma) |\bar{n}(t)\rangle \tag{357}$$

Comparing eqs 353 and 357 we see that damped excited states may be introduced through complex excited-state energies

$$E_n \rightarrow E_n - i\gamma$$
 (358)

In the absence of external perturbations the ground-state lifetime is infinite and the damping parameter associated with the ground-state energy E_0 is zero. Substitution in eq 358 is therefore equivalent to introduction of complex excitation energies

$$\omega_n \rightarrow \omega_n - i\gamma, \quad \omega_n = E_n - E_0$$
 (359)

Conceptually, damped response theory only requires the substitutions in eq 359 to be performed in the standard response-function expressions to introduce finite exited-state lifetimes. The empirical damping parameter γ is an effective inverse lifetime common to all excited states, leading to broadening that in reality has a number of physical origins: spontaneous-emission broadening, Doppler broadening, vibrational broadening, and collisional broadening. It is difficult, if at all possible, to devise an accurate ab initio model that takes into account the different broadening effects and provides a γ value tailored to each excited state; instead, finite lifetimes are treated empirically by means of the single damping parameter γ common to all excited states.

Damped response functions are complex and describe broadened dispersion and absorption spectra, as illustrated for the complex polarizability in section 3.7.2. In section 3.7.3 we briefly discuss higher order damped response theory.

3.7.2. Damped Linear Response theory. By introducing complex excitation energies into the standard linear response function expression in eq 282 according to eq 359 the resulting damped linear response function $\langle\langle A;B \rangle\rangle_{\omega_n}$ may be written as

$$\overline{\langle\langle A;B\rangle\rangle}_{\omega_{B}} = -\sum_{j} \left(\frac{\langle 0^{(0)}|A|j\rangle\langle j|B|0^{(0)}\rangle}{(\omega_{j}-\mathrm{i}\gamma)-\omega_{B}} + \frac{\langle j|A|0^{(0)}\rangle\langle 0^{(0)}|B|j\rangle}{(\omega_{j}-\mathrm{i}\gamma)^{*}+\omega_{B}} \right)$$
(360)

Note that the excitation energies in the second term are complex conjugated. Formally this is done to retain with the underlying structure of the Hessian matrix $E^{[2]}$ in eq 232 also in damped response theory, see ref 90 for details. By performing the complex conjugation of the second term in eq 360 we obtain

$$\overline{\langle\langle A;B\rangle\rangle}_{\omega_{B}+i\gamma} = -\sum_{j} \left(\frac{\langle 0^{(0)}|A|j\rangle\langle j|B|0^{(0)}\rangle}{\omega_{j}-(\omega_{B}+i\gamma)} + \frac{\langle j|A|0^{(0)}\rangle\langle 0^{(0)}|B|j\rangle}{\omega_{j}+(\omega_{B}+i\gamma)} \right)$$
(361)

where the imaginary $i\gamma$ term in eq 360 is now effectively associated with the frequency ω_B rather than with the excitation energy ω_j . For this reason we added the $i\gamma$ term to the frequency argument of the linear response function. Thus, damped linear response theory effectively corresponds to introducing a complex optical frequency

$$\omega_{\rm B} \rightarrow \omega_{\rm B} + \mathrm{i}\gamma \tag{362}$$

Damped response theory therefore requires solving response equations with complex frequencies.^{89,93,90} By construction the damped linear response function in eq 361 satisfies the symmetry relation for the standard linear response function in eq 283

$$\overline{\langle\langle A;B\rangle\rangle}_{\omega_B + i\gamma} = \overline{\langle\langle B;A\rangle\rangle}_{-(\omega_B + i\gamma)}$$
(363)

Thus, phenomenological introduction of complex excitation energies as described above preserves this fundamental symmetry property of the linear response function.

To illustrate the underlying structure of the damped response function in eq 361 we consider the case where $A = B = \mu_{\alpha}$ is a component of the electric dipole operator, writing the real and imaginary components of the damped response function out in detail

$$\overline{\langle \langle \mu_{\alpha}; \mu_{\alpha} \rangle \rangle}_{\omega_{B}} = -\sum_{j} \left(\langle 0^{(0)} | \mu_{\alpha} | j \rangle \langle j | \mu_{\alpha} | 0^{(0)} \rangle D_{j}(\omega_{B}) \right. \\ \left. + \langle 0^{(0)} | \mu_{\alpha} | j \rangle \langle j | \mu_{\alpha} | 0^{(0)} \rangle D_{j}(-\omega_{B}) \right. \\ \left. + i \left[\langle 0^{(0)} | \mu_{\alpha} | j \rangle \langle j | \mu_{\alpha} | 0^{(0)} \rangle A_{j}(\omega_{B}) \right. \\ \left. - \langle 0^{(0)} | \mu_{\alpha} | j \rangle \langle j | \mu_{\alpha} | 0^{(0)} \rangle A_{j}(-\omega_{B}) \right] \right)$$
(364)

where the dispersion and absorption line-shape functions $D_j(\pm \omega)$ and $A_j(\pm \omega)$ are given by

$$D_{j}(\omega) = \frac{\omega_{j} - \omega}{(\omega_{j} - \omega)^{2} + \gamma^{2}},$$

$$D_{j}(-\omega) = \frac{\omega_{j} + \omega}{(\omega_{j} + \omega)^{2} + \gamma^{2}}$$
(365)

$$A_j(\omega) = rac{\gamma}{(\omega_j - \omega)^2 + \gamma^2}$$

$$A_j(-\omega) = \frac{\gamma}{\left(\omega_j + \omega\right)^2 + \gamma^2}.$$
(366)

The dispersion and absorption line-shape functions are plotted in Figure 1. We note that γ determines the width of the line-shape functions, in particular, 2γ is the full width at half-maximum for the Lorentzian absorption line-shape function.

A plot of the real component of $\langle \langle \mu_{\alpha j} \mu_{\alpha} \rangle \rangle_{\omega_B}$ against ω_B describes a spectrum of (minus) the polarizability, also close to resonance frequencies, where it has a physically correct dispersion shape D_j . In contrast, the standard response function in eq 278 diverges at resonance frequencies. The real part of the damped response function thus provides a physically motivated description of the polarizability at all frequencies, whereas the standard response function is well behaved only in the non-resonant region.

The imaginary part of $\langle \langle \mu_{\alpha i} \mu_{\alpha} \rangle \rangle_{\omega_B}$ plotted against ω_B describes an absorption spectrum with Lorentzian line-shape functions A_j superimposed on electric-dipole transition strengths $||\langle 0^{(0)}||\mu_{\alpha}|j\rangle||^2$. We note that damped linear response functions may be evaluated in any frequency interval to obtain the corresponding absorption spectrum. Thus, damped response theory enables direct determination of absorption spectra also when application of standard response theory is difficult because it would require determination of a large number of individual transition dipole strengths, for example, in X-ray spectra and absorption spectra of large molecules.

3.7.3. Higher Order Damped Response theory. For the damped linear response function discussed above, introduction of complex excitation energies is equivalent to introduction of complex frequencies, see eq 362. Likewise, for higher order response functions complex frequencies are effectively introduced in accordance with eq 362 for each frequency that enters the response function.^{89,90} However, the fundamental symmetry relations for higher order response functions, such as those in eqs 284 and 285, are in general not conserved in higher order damped response theory, as is the case in damped linear response theory, see eq 363.

Introduction of damping in quadratic and higher order response functions can be interpreted in a similar way as for the polarizability in section 3.7.2, where one component describes dispersion and the other absorption. For example, magnetic optical rotation (dispersion) and magnetic circular dichroism (MCD) (absorption) are described by the imaginary and real components, respectively, of the damped quadratic response function $\langle\langle \mu_{\alpha}, \mu_{\beta}, m_{\gamma} \rangle\rangle_{\alpha_{D}, 0}^{\infty}$ where m_{γ} is a component of the magnetic dipole operator, $^{94-96}_{96}$ and two-photon absorption (TPA) spectra can be obtained from a damped cubic response function as described in ref 97. We note that, in general, the residue spectrum of a conventional response function can be obtained from the imaginary (or real in the case of an imaginary operator) component of the corresponding damped response function.

4. RESPONSE THEORY FOR APPROXIMATE STATES

In this section, we use the variation principles developed in the previous section to determine molecular response functions for approximate wave functions. From these response functions (and their poles and residues) we determine molecular properties for ground and excited states and transition matrix elements between these states, where the ground state is represented by an



Figure 1. Dispersion $D_n(\omega)$ (left) and absorption $A_n(\omega)$ (right) lineshape functions of damped response theory.

approximate wave function. For the variational Hartree—Fock and MCSCF models in sections 4.1 and 4.2 we use the variation principle for the time-averaged quasi-energy to determine the response functions. For the nonvariational coupled-cluster model in section 4.3 we use the time-averaged Lagrangian to determine the response functions. In section 4.3, we also describe how response functions may be obtained for a coupled-cluster wave function using the equation of motion formalism. Next, in section 4.4 we consider Møller—Plesset perturbation theory, in which only static ground-state properties can be calculated.

In sections 4.1-4.4 we focus on the theoretical formulation of the wave function models mentioned above, where response functions are obtained as perturbation strength derivatives of the quasi-energy (or quasi-energy Lagrangian). Following the discussion of response functions for perturbation-dependent basis sets in section 4.5 we give in section 4.6 an overview of computational developments and implementations of response methods in a historical perspective.

4.1. Hartree–Fock Theory

The general theoretical basis for carrying out response theory at the SCF and MCSCF levels of theory was formulated by Olsen and Jørgensen in 1985,⁷⁰ followed by implementations of the linear,⁹⁸ quadratic,⁹⁹ and cubic^{100,101} response functions in the MO basis in the 1980s and 1990s. Since 2000, developments in SCF response theory have been aimed at large molecular systems and a reformulation of response equations and response functions in the AO basis^{13,14,102–106,141} using the elements of the AO density matrix as variational parameters instead of the MO coefficients. In the AO-based response formalism, all manipulations are reduced to elementary matrix operations such as additions and multiplications, making this formalism particularly convenient for developing linear-scaling algorithms when sparse matrix algebra can be applied.

Noting that the equations defining SCF response theory in the MO basis may be obtained as a special case of the MCSCF equations derived in section 4.2, we summarize in this section the AO-based quasi-energy formulation by Thorvaldsen et al.¹⁰⁶ The main advantage of this formulation is that perturbationdependent basis sets (which may also depend on time) are an integral part of the formulation, treating perturbations with perturbationdependent AOs (e.g., geometrical and magnetic perturbations with atom-fixed London orbitals, see section 2.4) on the same footing as perturbations where the field dependence is not included in the AOs (e.g., electric-field perturbations). This situation appears, for instance, when considering time-periodic London atomic orbitals.¹⁰⁷ Bearing in mind that DFT is not the focus of this review, we here restrict ourselves to Hartree-Fock theory but note that the same formulation applies to Kohn-Sham theory with the additional inclusion of exchange-correlation terms, which can be efficiently implemented and calculated

using automatic differentiation techniques.¹⁰⁸ We also note that the quasi-energy formulation by Thorvaldsen et al.¹⁰⁶ has been generalized to two- and four-component wave functions in the basis of four-component spinors.¹⁰⁹

4.1.1. Quasi-Energy Response Theory in a Density-Matrix Formulation. In the quasi-energy of eq 165 only the ket state is subject to time differentiation. The Hartree–Fock quasi-energy is therefore not symmetric with respect to operations on the bra and ket states

$$Q(t) = \langle HF|H - i\partial_t | HF \rangle$$
(367)

In particular, the second term in eq 367 may be evaluated as

$$\langle \widetilde{\mathrm{HF}}| - \mathrm{i}\partial_t | \widetilde{\mathrm{HF}} \rangle = -\mathrm{i} \sum_J \int \phi_J^*(\mathbf{r}, t) \partial_t \phi_J(\mathbf{r}, t) \mathrm{d}\mathbf{r}$$
 (368)

where the summation is over all occupied spin MOs (denoted with index *J*), differentiating only the ket MO $\phi_J(\mathbf{r},t)$. In contrast, the electron density

$$\rho(\mathbf{r}, t) = \sum_{J} \phi_{J}^{*}(\mathbf{r}, t)\phi_{J}(\mathbf{r}, t)$$
(369)

and its time derivative

$$\dot{\rho}(\mathbf{r}, t) = \sum_{J} \left(\phi_{J}^{*}(\mathbf{r}, t) \phi_{J}(\mathbf{r}, t) + \phi_{J}^{*}(\mathbf{r}, t) \phi_{J}(\mathbf{r}, t) \right)$$
(370)

are symmetric in $\phi_J^*(\mathbf{r},t)$ and $\phi_J(\mathbf{r},t)$. Given that the timederivative contribution to the quasi-energy in eq 368 contains only the second term in the time-differentiated electron density, it is not straightforward to express the quasi-energy in terms of $\rho(\mathbf{r},t)$ and $\dot{\rho}(\mathbf{r},t)$.

Let us now consider the matrix representation of $\rho(\mathbf{r},t)$ in the AO basis. By expanding the MOs in the AOs $\chi_{\mu\nu}$, which may depend on the external (possibly time-dependent) perturbation (see section 2.4)

$$\phi_J(\mathbf{r}, t) = \sum_{\mu} C_{\mu J}(t) \chi_{\mu}(\mathbf{r}, t)$$
(371)

we obtain

$$\rho(\mathbf{r}, t) = \sum_{\mu\nu} D_{\mu\nu} \dot{\chi}^*_{\mu}(\mathbf{r}, t) \chi_{\nu}(\mathbf{r}, t)$$
(372)

where we introduced the one-electron density matrix in the AO basis, **D**, whose elements are given by

$$D_{\mu\nu}(t) = \sum_{J} C^{*}_{\mu J}(t) C_{\nu J}(t)$$
(373)

The time derivative of $\rho(\mathbf{r},t)$ may thus be expressed in terms of the time derivative of the density matrix

$$\dot{\rho}(\mathbf{r}, t) = \sum_{\mu\nu} \left(\dot{D}_{\mu\nu} \dot{\chi}^*_{\mu}(\mathbf{r}, t) \chi_{\nu}(\mathbf{r}, t) + D_{\mu\nu} \dot{\chi}^*_{\mu}(\mathbf{r}, t) \chi_{\nu}(\mathbf{r}, t) + D_{\mu\nu} \dot{\chi}^*_{\mu}(\mathbf{r}, t) \dot{\chi}^*_{\nu}(\mathbf{r}, t) \right)$$
(374)

with contributions from the density matrix as well as the AOs.

Even though the Hartree–Fock quasi-energy cannot be expressed in terms of $\rho(\mathbf{r},t)$ and its time derivative (or, equivalently, in terms of **D** and its time derivative) it is possible to express the

perturbation-strength derivative of the time-averaged Hartree–Fock quasi-energy as a function of the density matrix **D** and its time derivative $\dot{\mathbf{D}}^{106}$

$$\frac{\mathrm{d}\mathcal{Q}}{\mathrm{d}\varepsilon_A} = \frac{\mathrm{d}\mathcal{Q}(\mathbf{D},\,\dot{\mathbf{D}})}{\mathrm{d}\varepsilon_A} \equiv \mathcal{Q}^A(\mathbf{D},\,\dot{\mathbf{D}}) \tag{375}$$

where the quasi-energy derivative $\mathcal{Q}^{A}(\mathbf{D},\dot{\mathbf{D}})$ (whose explicit form is given in section 4.1.2) contains all information needed to determine the response functions by further differentiation with respect to the perturbation strengths, similarly to eqs 274–276

$$\langle\langle A; B \rangle\rangle_{\omega_B} = \frac{\mathrm{d}\mathcal{Q}^A(\mathbf{D}, \dot{\mathbf{D}})}{\mathrm{d}\varepsilon_B}|_{\{\varepsilon\}=0}, \ \omega_A = -\omega_B \tag{376}$$

$$\langle \langle A; B, C \rangle \rangle_{\omega_{B}, \omega_{C}} = \frac{\mathrm{d}^{2} \mathcal{Q}^{A}(\mathbf{D}, \dot{\mathbf{D}})}{\mathrm{d}\varepsilon_{B} \mathrm{d}\varepsilon_{C}} |_{\{\varepsilon\}=0},$$

$$\omega_{A} = -\omega_{B} - \omega_{C}$$

$$(377)$$

and so on. The quasi-energy derivative $\mathscr{Q}^{A}(\mathbf{D},\dot{\mathbf{D}})$ therefore provides an alternative starting point to the quasi-energy for identifying response functions in terms of perturbed density matrices. Extending the notation in eq 375, we write the derivatives of a general function or matrix f in the manner

$$f^{B_1\dots B_m} = \frac{\mathrm{d}^m f}{\mathrm{d}\varepsilon_{B_1}\dots\mathrm{d}\varepsilon_{B_m}} \tag{378}$$

$$f^{n, B_1...B_m} = \frac{\partial^{n+m} f}{\partial (\mathbf{D}^{\mathrm{T}})^n \partial \varepsilon_{B_1}...\partial \varepsilon_{B_m}}$$
(379)

with special cases $f^{0,B_1...B_m} = \partial^m f / \partial \varepsilon_{B_1}...\partial \varepsilon_{B_m}$ and $f^{n,0} = \partial^n f / \partial (\mathbf{D}^T)^n$. In some cases, such derivatives are evaluated at zero perturbation strength, but this will always be clear from the context.

4.1.2. Quasi-Energy Gradient. When working with general time- and perturbation-dependent basis functions $\chi_{\mu\nu}$ it is convenient to introduce the generalized Hartree–Fock energy¹⁰⁶

$$E = \operatorname{tr}\left(\mathbf{h} + \mathbf{V} + \frac{1}{2}\mathbf{G}(\mathbf{D}) - \frac{\mathrm{i}}{2}\mathbf{T}\right)\mathbf{D} + h_{\mathrm{nuc}}$$
(380)

where **h** is the one-electron matrix without the external perturbation, **V** is the perturbation matrix (e.g., an electric dipole matrix), G(D) is the two-electron matrix containing Coulomb and exchange contributions, **T** is an anti-Hermitian time-differentiated overlap matrix

$$h_{\mu\nu} = \langle \chi_{\mu} | -\frac{1}{2} \nabla^2 - \sum_{K} Z_{K} r_{K}^{-1} | \chi_{\nu} \rangle$$
(381)

$$V_{\mu\nu} = \langle \chi_{\mu} | V(t) | \chi_{\nu} \rangle \tag{382}$$

$$G_{\mu\nu}(\mathbf{D}) = \sum_{\alpha\beta} D_{\beta\alpha} (g_{\mu\nu\alpha\beta} - g_{\mu\beta\alpha\nu})$$
(383)

$$g_{\mu\nu\alpha\beta} = \iint \frac{\chi_{\mu}^{*}(\mathbf{r}_{1})\chi_{\alpha}^{*}(\mathbf{r}_{2})\chi_{\nu}(\mathbf{r}_{1})\chi_{\beta}(\mathbf{r}_{2})}{r_{12}} \,\mathrm{d}\mathbf{r}_{1} \,\mathrm{d}\mathbf{r}_{2}$$
(384)

$$T_{\mu\nu} = \langle \chi_{\mu} | \dot{\chi}_{\nu} \rangle - \langle \dot{\chi}_{\mu} | \chi_{\nu} \rangle \tag{385}$$

while h_{nuc} is the nuclear repulsion term of eq 86. In the notation of eqs 378 and 379, the quasi-energy gradient $\mathcal{Q}^A = \mathcal{Q}^A(\mathbf{D},\dot{\mathbf{D}})$ may be written in terms of the partial derivatives of the generalized Hartree–Fock energy $E^{0,A}$ and a reorthonormalization term tr $\mathbf{S}^A \mathbf{W}^{106}$

$$\mathscr{Q}^{A} = E^{0, A} - \mathrm{tr}\mathbf{S}^{A}\mathbf{W}$$
(386)

where $E^{0,A}$ is given by

$$E^{0, A} = \operatorname{tr}\left(\mathbf{h}^{A} + \mathbf{V}^{A} + \frac{1}{2}\mathbf{G}^{A}(\mathbf{D}) - \frac{\mathrm{i}}{2}\mathbf{T}^{A}\right)\mathbf{D} + h_{\operatorname{nuc}}^{A} \qquad (387)$$

because $\mathbf{h}^{0,A} = \mathbf{h}^A$ and so on. Note that in $\mathbf{G}^A(\mathbf{D})$ only the twoelectron integrals $g_{\mu\nu\alpha\beta}$ are differentiated with respect to ε_A . The \mathbf{S}^A matrix in eq 386 is the derivative of the AO overlap matrix

$$S_{\mu\nu} = \langle \chi_{\mu} | \chi_{\nu} \rangle \tag{388}$$

and W is a generalized energy- and frequency-weighted density matrix

$$\mathbf{W} = \mathbf{DFD} + \frac{\mathbf{i}}{2} (\dot{\mathbf{D}SD} - \mathbf{DSD})$$
(389)

where **F** is the generalized time-dependent Fock matrix

$$\mathbf{F} = \mathbf{E}^{1, 0} = \mathbf{h} + \mathbf{G}(\mathbf{D}) + \mathbf{V} - \frac{\mathbf{i}}{2}\mathbf{T}$$
 (390)

We note the close relationship between eq 386 and the molecular gradient in the AO basis.¹¹⁰ The main difference lies in the inclusion of time-dependent terms in eq 386, where **W** is a generalization of the energy-weighted density matrix.¹¹¹ Indeed, the molecular gradient is recovered in the time-independent limit, when the time-dependent terms (i.e., the terms involving **T**, \mathbf{T}^A , and the two last terms in **W**) vanish. If the perturbation *A* is not described using perturbation-dependent basis sets, most of the terms in eq 386 vanish and the following simple expression is obtained

$$\mathcal{Q}^{A} = \operatorname{tr} \mathbf{V}^{A} \mathbf{D}$$
(391)

which represents the expectation value of the one-electron perturbation operator V^A .

The quasi-energy gradient in eq 386 provides the starting point for generating response functions by further differentiation, according to eqs 376 and 377. However, before considering response functions in section 4.1.4, we discuss the parametrization of the density matrix in section 4.1.3.

4.1.3. Perturbed Density Matrix. The density matrix for a single-determinant Hartree–Fock state must satisfy the trace, Hermiticity, and idempotency conditions

$$tr DS = N \tag{392}$$

$$\mathbf{D}^{\dagger} = \mathbf{D} \tag{393}$$

$$\mathbf{DSD} = \mathbf{D} \tag{394}$$

where N is the number of electrons in the molecular system. Whereas the trace and Hermiticity conditions are automatically satisfied in the formulation given below, the idempotency relation requires special attention. Furthermore, the time-dependent density matrix must be a solution to the time-dependent SCF (TDSCF) matrix equation¹⁰⁶

$$\left(\mathbf{F} - \frac{\mathbf{i}}{2}\mathbf{S}\partial_t\right)\mathbf{D}\mathbf{S} - \left(\mathbf{S}\mathbf{D}\mathbf{F} + \frac{\mathbf{i}}{2}\partial_t(\mathbf{S}\mathbf{D})\mathbf{S}\right) = \mathbf{0}$$
 (395)

which, in the absence of a time-dependent perturbation, reduces to the usual Fock stationary conditions FDS = SDF of Hartree–Fock theory.³⁰ We note that eq 395 was also derived in ref 102 but without taking into account the possible perturbation and time dependence of the AO basis functions.

The AO density matrix is now expanded in orders of the perturbation

$$\mathbf{D} = \mathbf{D}_0 + \mathbf{D}^{(1)} + \mathbf{D}^{(2)} + \dots$$
(396)

where D_0 is the optimized density matrix for the unperturbed system (the subscript 0 here and elsewhere denoting the unperturbed system) and where higher order density matrices are written in terms of their frequency and operator components following eq 218

$$\mathbf{D}^{(1)} = \sum_{B} \varepsilon_{B} \mathbf{D}^{B} \mathbf{e}^{-\mathbf{i}\omega_{B}t}$$
(397)

$$\mathbf{D}^{(2)} = \sum_{BC} \varepsilon_B \varepsilon_C \mathbf{D}^{BC} \mathbf{e}^{-\mathbf{i}(\omega_B + \omega_C)t}$$
(398)

The perturbed density matrices are partitioned into particular and homogeneous components.¹⁰⁶ The particular component is chosen such that the time-dependent density matrix satisfies the idempotency relation, whereas the homogeneous component ensures that the density matrix is a solution to the TDSCF equation.

To illustrate this partitioning scheme, we consider the firstorder perturbed density matrix D^{B} , which we decompose into particular and homogeneous components

$$\mathbf{D}^B = \mathbf{D}^B_{\mathrm{P}} + \mathbf{D}^B_{\mathrm{H}} \tag{399}$$

From eq 394 we find that the first-order idempotency relation is the inhomogeneous equation

$$\mathbf{D}^{B}\mathbf{S}_{0}\mathbf{D}_{0} + \mathbf{D}_{0}\mathbf{S}_{0}\mathbf{D}^{B} - \mathbf{D}^{B} = \mathbf{N}^{B}, \ \mathbf{N}^{B} = -\mathbf{D}_{0}\mathbf{S}^{B}\mathbf{D}_{0} \quad (400)$$

A particular solution $\mathbf{D}_{\mathrm{P}}^{B}$ to this equation is given by

$$\mathbf{D}_{\mathrm{P}}^{\mathrm{B}} = \mathbf{N}^{\mathrm{B}} \mathbf{S}_{0} \mathbf{D}_{0} + \mathbf{D}_{0} \mathbf{S}_{0} \mathbf{N}^{\mathrm{B}} - \mathbf{N}^{\mathrm{B}} = -\mathbf{D}_{0} \mathbf{S}^{\mathrm{B}} \mathbf{D}_{0} \qquad (401)$$

The homogeneous component \mathbf{D}_{H}^{B} is parametrized in terms of an unknown matrix \mathbf{X}^{B} in the form

$$\mathbf{D}_{\mathrm{H}}^{B} = \mathbf{D}_{0}\mathbf{S}_{0}\mathbf{X}^{B} - \mathbf{X}^{B}\mathbf{S}_{0}\mathbf{D}_{0}$$
(402)

ensuring that $\mathbf{D}^{\text{B}}_{\text{H}}$ is a solution to the homogeneous equation associated with eq 400

$$\mathbf{D}_{\mathrm{H}}^{\mathrm{B}}\mathbf{S}_{0}\mathbf{D}_{0} + \mathbf{D}_{0}\mathbf{S}_{0}\mathbf{D}_{\mathrm{H}}^{\mathrm{B}} - \mathbf{D}_{\mathrm{H}}^{\mathrm{B}} = \mathbf{0}$$

$$(403)$$

In this way, it is ensured that \mathbf{D}^{B} in eq 399 satisfies the idempotency relation to first order. The matrix \mathbf{X}^{B} in the homogeneous component \mathbf{D}_{H}^{B} is now determined so that \mathbf{D}^{B} is also a solution to the first-order TDSCF equation. Differentiating eq 395 with respect to ε_{B} at zero perturbation strengths the following equation for \mathbf{X}^{B} is obtained¹⁰⁶

$$\left(\mathbf{E}^{[2]} - \omega_B \mathbf{S}^{[2]}\right) \mathbf{X}^B = \mathbf{R}^B \tag{404}$$

where $\mathbf{E}^{[2]}$ and $\mathbf{S}^{[2]}$ are generalized Hessian and metric matrices, respectively. As eq 404 is solved iteratively, it is not necessary to construct these matrices explicitly; it is sufficient to know their products with a general matrix **X**

$$\begin{split} E^{[2]}X &= G(XS_0D_0 - D_0S_0X)D_0S_0 - S_0D_0G(XS_0D_0 - D_0S_0X) \\ &+ F_0(XS_0D_0 - D_0S_0X)S_0 - S_0(XS_0D_0 - D_0S_0X)F_0 \end{split} \label{eq:eq:expansion}$$

$$\mathbf{S}^{[2]}\mathbf{X} = \mathbf{S}_0(\mathbf{X}\mathbf{S}_0\mathbf{D}_0 - \mathbf{D}_0\mathbf{S}_0\mathbf{X})\mathbf{S}_0$$
(406)

where \mathbf{F}_0 is the Fock matrix of the unperturbed system. The righthand-side matrix of the response equation in eq 404 is given by

$$\mathbf{R}^{B} = \left[\mathbf{E}^{1, B} + \mathbf{E}^{2, 0}(\mathbf{D}_{P}^{B})\right]\mathbf{D}_{0}\mathbf{S}_{0} + \left(\mathbf{F}_{0} - \frac{\omega_{B}}{2}\mathbf{S}_{0}\right)\left(\mathbf{D}_{P}^{B}\mathbf{S}_{0} + \mathbf{D}_{0}\mathbf{S}^{B}\right) \\ - \mathbf{S}_{0}\mathbf{D}_{0}\left[\mathbf{E}^{1, B} + \mathbf{E}^{2, 0}(\mathbf{D}_{P}^{B})\right] - \left(\mathbf{S}_{0}\mathbf{D}_{P}^{B} + \mathbf{S}^{B}\mathbf{D}_{0}\right)\left(\mathbf{F}_{0} - \frac{\omega_{B}}{2}\mathbf{S}_{0}\right)$$
(407)

We note that the particular component $\mathbf{D}_{\mathrm{P}}^{B}$ only depends on the zero-order density matrix \mathbf{D}_{0} , whereas the homogeneous component $\mathbf{D}_{\mathrm{H}}^{B}$ requires solution of the response equation in eq 404.

The first-order analysis given above may be generalized to arbitrary orders, decomposing the *m*th-order density matrix into particular and homogeneous components

$$\mathbf{D}^{B_1\dots B_m} = \mathbf{D}_{\mathbf{p}}^{B_1\dots B_m} + \mathbf{D}_{\mathbf{H}}^{B_1\dots B_m}$$
(408)

The *m*th-order idempotency condition can be written in the form

$$\mathbf{D}^{B_1...B_m} \mathbf{S} \mathbf{D}_0 + \mathbf{D}_0 \mathbf{S} \mathbf{D}^{B_1...B_m} - \mathbf{D}^{B_1...B_m} = \mathbf{N}^{B_1...B_m}$$
(409)

where the inhomogeneity $\mathbf{N}^{B_1...B_m}$ contains perturbed density matrices of order less than *m*. A particular solution $\mathbf{D}_{\mathbf{P}}^{B_1...B_m}$ to eq 409 is given by

$$\mathbf{D}_{\mathrm{P}}^{B_{1}\ldots B_{m}} = \mathbf{N}^{B_{1}\ldots B_{m}}\mathbf{S}_{0}\mathbf{D}_{0} + \mathbf{D}_{0}\mathbf{S}_{0}\mathbf{N}^{B_{1}\ldots B_{m}} - \mathbf{N}^{B_{1}\ldots B_{m}}$$
(410)

Note that $\mathbf{D}_{\mathbf{P}}^{B_1...B_m}$ has the same structure as the first-order particular density matrix $\mathbf{D}_{\mathbf{P}}^B$ in eq 401. The *m*th-order homogeneous component $\mathbf{D}_{\mathbf{H}_1...B_m}^{B_1...B_m}$ is parametrized as in eq 402

$$\mathbf{D}_{\mathrm{H}}^{B_{1}\ldots B_{m}} = \mathbf{D}_{0}\mathbf{S}\mathbf{X}^{B_{1}\ldots B_{m}} - \mathbf{X}^{B_{1}\ldots B_{m}}\mathbf{S}\mathbf{D}_{0}$$
(411)

where $\mathbf{X}^{B_1...B_m}$ is determined so that the total density matrix in eq 408 satisfies the TDSCF equation to order *m* by solving the *m*th-order response equation

$$[\mathbf{E}^{[2]} - (\omega_{B_1} + \dots + \omega_{B_m})\mathbf{S}^{[2]}]\mathbf{X}^{B_1\dots B_m} = \mathbf{R}^{B_1\dots B_m}$$
(412)

This equation has the same form as the first-order equation in eq 404, with $\mathbf{R}^{B_1...B_m}$ containing only lower order density matrices. Since the response equations have the same structure to all orders they can be determined using the same solver, for example, the linear-scaling solver of Coriani et al.,¹⁴ making the formulation suitable for large molecular systems. We note that the partitioning of the density matrix presented here, where one component is a solution to the idempotency relation and another component solves the TDSCF equation, was first developed and implemented for molecular Hessians and magnetizabilities by Larsen et al.¹¹²

In summary, to determine the *m*th-order density matrices $\mathbf{D}^{B_1...B_m}$ we first construct the *m*th-order inhomogeneity $\mathbf{N}^{B_1...B_m}$ from lower order density matrices and determine the particular solutions $\mathbf{D}_{\mathrm{P}}^{B_1...B_m}$. Next, the right-hand sides $\mathbf{R}^{B_1...B_m}$ are constructed from $\mathbf{D}_{\mathrm{P}}^{B_1...B_m}$ and from lower order density matrices

followed by solution of the *m*th-order response equations to yield $\mathbf{X}^{B_1...B_m}$ and the homogeneous components $\mathbf{D}_{\mathrm{H}}^{B_1...B_m}$. Finally, $\mathbf{D}^{B_1...B_m}$ is determined according to eq 408. We have thus established a recursive procedure for determining perturbed density matrices to arbitrary order, where perturbation-dependent basis sets are an integrated part of the formulation. With the perturbed density matrices at hand it is straightforward to evaluate response functions, as discussed in the next subsection.

4.1.4. Response Functions. Response functions may be identified by differentiation of the quasi-energy gradient in eq 386 according to eqs 376 and 377. In the notation of eq 379 the derivative of the first term in eq 386 at zero interaction strength may be written as

$$\frac{\mathrm{d}E^{0, A}}{\mathrm{d}\varepsilon_{B}} = \frac{\partial E^{0, A}}{\partial \varepsilon_{B}} + \operatorname{tr} \frac{\partial E^{0, A}}{\partial \mathbf{D}^{\mathrm{T}}} \frac{\partial \mathbf{D}}{\partial \varepsilon_{B}} = E^{0, AB} + \operatorname{tr} \mathbf{E}^{1, A} \mathbf{D}^{B}$$
(413)

Including also the derivative of the second term in eq 386 we obtain the following expression for the linear response function

$$\langle\langle A; B \rangle\rangle_{\omega_{B}} = \mathscr{Q}^{AB} = E^{0, AB} + \operatorname{tr} \mathbf{E}^{1, A} \mathbf{D}^{B} - \operatorname{tr} \mathbf{S}^{AB} \mathbf{W}_{0} - \operatorname{tr} \mathbf{S}^{A} \mathbf{W}^{B}$$

$$(414)$$

According to eq 390 the differentiated Fock matrix becomes

$$\mathbf{E}^{1, A} = \mathbf{F}^{0, A} = \mathbf{h}^{A} + \mathbf{G}^{A}(\mathbf{D}_{0}) + \mathbf{V}^{A} - \frac{i}{2}\mathbf{T}^{A}$$
 (415)

and using eq 389 the W matrices are given by

$$\mathbf{W}_0 = \mathbf{D}_0 \mathbf{F}_0 \mathbf{D}_0 \tag{416}$$

$$\mathbf{W}^{B} = \mathbf{D}_{0}\mathbf{F}^{B}\mathbf{D}_{0} + \mathbf{D}^{B}\left(\mathbf{F}_{0} + \frac{\omega_{B}}{2}\mathbf{S}_{0}\right)\mathbf{D}_{0} + \mathbf{D}_{0}\left(\mathbf{F}_{0} - \frac{\omega_{B}}{2}\mathbf{S}_{0}\right)\mathbf{D}^{B}$$
(417)

Although not immediately apparent from the expression in eq 414, the linear response function satisfies the general symmetry relation in eq 283. When the basis set does not depend on the external perturbations *A* and *B* only the contribution involving V^A in the second term of eq 414 remains, reducing the linear response function to the simple expression

$$\langle\langle A; B \rangle\rangle_{\omega_B} = \operatorname{tr} \mathbf{V}^A \mathbf{D}^B \tag{418}$$

also obtained by Larsen et al.¹⁰²

The quadratic response function in eq 377 is given by¹⁰⁶

$$\begin{aligned} \langle\langle A; B, C \rangle\rangle_{\omega_{B, \omega_{C}}} &= \mathcal{Q}^{ABC} = E^{0, ABC} \\ &+ \operatorname{tr}(\mathbf{E}^{1, AC}\mathbf{D}^{B} + \mathbf{E}^{1, AB}\mathbf{D}^{C} + \mathbf{E}^{2, A}(\mathbf{D}^{B})\mathbf{D}^{C} + 2\mathbf{E}^{1, A}\mathbf{D}^{BC}) \\ &- \operatorname{tr}(\mathbf{S}^{ABC}\mathbf{W}_{0} + \mathbf{S}^{AC}\mathbf{W}^{B} + \mathbf{S}^{AB}\mathbf{W}^{C} + \mathbf{S}^{A}\mathbf{W}^{BC}) \end{aligned}$$
(419)

where the first-order density matrices \mathbf{D}^{B} and \mathbf{D}^{C} and the secondorder density matrices \mathbf{D}^{BC} are required to determine the quadratic response function. We also note that $\mathbf{E}^{2,A}(\mathbf{D}^{B})$ corresponds to the two-electron matrix in 383 with differentiated two-electron integrals. It is possible to obtain an expression for $\langle\langle A; B, C \rangle\rangle_{\omega_{B},\omega_{C}}$ that complies with the 2n + 1 rule so that only first-order density matrices \mathbf{D}^{A} , \mathbf{D}^{B} , and \mathbf{D}^{C} are required; see ref 106 for details.

4.2. Multiconfigurational Self-Consistent Field Theory

The multiconfigurational self-consistent field (MCSCF) model^{30,113,114} is a highly flexible electronic-structure model,

well suited to systems not dominated by a single electronic configuration, for example, the ground-state ozone molecule, molecular dissociation processes, and symmetry-forbidden reactions. With its flexible multiconfigurational ansatz the MCSCF model is able to recover the static correlation necessary for a qualitatively correct description of such systems; for a quantitatively correct description some correction for dynamic correlation is also necessary, as can be achieved by perturbation theory (e.g., second-order complete-active-space perturbation theory, CASPT2,¹¹⁵ or second-order *n*-electron valence-state perturbation interaction (MRCI)¹¹⁴ treatment. In the following, we discuss the calculation of molecular properties for the MCSCF model, restricting ourselves to linear response theory.

4.2.1. MCSCF Quasi-Energy. In terms of the unperturbed MCSCF reference state $|MC\rangle$, the regular time-dependent MCSCF wave function $|MC\rangle$ may be parametrized as⁷⁰

$$|\widetilde{\mathrm{MC}}\rangle = \mathrm{e}^{\mathrm{i}\kappa(t)}\mathrm{e}^{\mathrm{i}S(t)}|\mathrm{MC}\rangle$$
(420)

where $|\text{MC}\rangle$ is a linear combination of Slater determinants and $\kappa(t)$ and S(t) are Hermitian orbital-rotation and state-rotation operators, respectively

$$\kappa(t) = \sum_{\mu} \left(\kappa_{\mu}(t) q_{\mu}^{\dagger} + \kappa_{\mu}^{*}(t) q_{\mu} \right)$$
(421)

$$S(t) = \sum_{n} (S_{n}(t)R_{n}^{\dagger} + S_{n}^{*}(t)R_{n})$$
(422)

Here $\kappa(t)$ is expressed in terms of the orbital-excitation operators q_{μ}^{\dagger} corresponding to the singlet and triplet excitation operators in eqs 97–100, where μ is a compound MO index for orbital excitations. The state-transfer operator S(t) is expressed in terms of the operators

$$R_n^{\dagger} = |n\rangle \langle \mathrm{MC}| \tag{423}$$

where the $|n\rangle$ states span the orthogonal complement of the MCSCF reference state $|MC\rangle$. The reference state $|MC\rangle$ is determined by variationally optimizing the expectation value of the unperturbed Hamiltonian H_0 with respect to the orbital- and state-rotation parameters,⁷⁰ yielding the Brillouin conditions

$$\langle \mathrm{MC}|[H_0, q_{\mu}^{\dagger}]|\mathrm{MC}\rangle = 0 \tag{424}$$

$$\langle \mathrm{MC}|[H_0, R_n^{\dagger}]|\mathrm{MC}\rangle = 0 \tag{425}$$

Because of the exponential parametrization in eq 420 the timedependent MCSCF wave function is normalized by construction and it is not necessary to impose a normalization constraint. In this parametrization we introduce the time-averaged MCSCF quasi-energy

$$\mathcal{Q} = \{ \langle \widetilde{\mathrm{MC}} | H - \mathrm{i}\partial_t | \widetilde{\mathrm{MC}} \rangle \}_T$$

= $\{ \langle \mathrm{MC} | \mathrm{e}^{-\mathrm{i}S(t)} \mathrm{e}^{-\mathrm{i}\kappa(t)} (H - \mathrm{i}\partial_t) \mathrm{e}^{\mathrm{i}\kappa(t)} \mathrm{e}^{\mathrm{i}S(t)} | \mathrm{MC} \rangle \}_T$ (426)

where the time-dependent orbital- and state-rotation parameters are determined by optimizing \mathcal{Q} variationally

$$\delta \mathcal{Q} = 0 \tag{427}$$

In the following, we derive the MCSCF linear response function; for higher order response functions, see ref 70.

4.2.2. Second-Order MCSCF Quasi-Energy. The orbitaland state-rotation operators may be expanded in orders of the perturbation

$$\kappa(t) = \kappa^{(1)}(t) + \kappa^{(2)}(t) + \dots$$
(428)

$$S(t) = S^{(1)}(t) + S^{(2)}(t) + \dots$$
(429)

where the zero-order parameters vanish because the reference wave function $|\text{MC}\rangle$ is variationally optimized for the unperturbed system. Since the MCSCF quasi-energy is determined variationally, only the first-order response parameters are needed for the linear response function studied here, following the 2n + 1rule. The first-order operators may be expanded in terms of their frequency and operator components (similarly to the expansion in eq 218)

$$\kappa^{(1)}(t) = \sum_{B} \varepsilon_{B} \kappa^{B} e^{-i\omega_{B}t}$$
(430)

$$S^{(1)}(t) = \sum_{B} \varepsilon_{B} S^{B} e^{-i\omega_{B}t}$$
(431)

where *B* is a combined operator—frequency index. The operators κ^B and S^B are written in terms of the parameters κ^B_{μ} and S^B_m (equivalently to the expression in the time domain in eqs 421 and 422) as

$$\kappa^{B} = \sum_{\mu} \left(\kappa^{B}_{\mu} q^{\dagger}_{\mu} + \kappa^{-B*}_{\mu} q_{\mu} \right)$$
(432)

$$S^{B} = \sum_{n} \left(S_{n}^{B} R_{n}^{\dagger} + S_{n}^{-B*} R_{n} \right)$$
(433)

The paired structure of κ^B with respect to the *B* and -B indices in eq 432 ensures that $\kappa^{-B+} = \kappa^B$ such that $\kappa^{(1)}(t)$ is Hermitian (and likewise for the *S* operators).

We now expand the MCSCF quasi-energy in orders of the perturbation

$$\mathcal{Q} = \mathcal{Q}^{(0)} + \mathcal{Q}^{(1)} + \mathcal{Q}^{(2)} + \dots$$
(434)

Limiting ourselves to linear response theory, we need only consider the MCSCF quasi-energy up to second order. The zero-order quasi-energy is simply the MCSCF energy

$$\mathcal{Q}^{(0)} = \langle \mathrm{MC} | H_0 | \mathrm{MC} \rangle \tag{435}$$

Because of the 2n + 1 rule, first-order parameters do not contribute to the first-order quasi-energy, which reduces to a sum of simple expectation values. Furthermore, upon time averaging, only static terms in V(t) contribute to the first-order MCSCF quasi-energy

$$\mathcal{Q}^{(1)} = \left\{ \langle \mathrm{MC} | \sum_{B} \varepsilon_{B} V_{B} \mathrm{e}^{-\mathrm{i}\omega_{B}t} | \mathrm{MC} \rangle \right\}_{T}$$
$$= \sum_{B:\omega_{B}=0} \varepsilon_{B} \langle \mathrm{MC} | V_{B} | \mathrm{MC} \rangle$$
(436)

To determine the second-order MCSCF quasi-energy

$$\mathcal{Q}^{(2)} = \left\{ \langle \mathrm{MC} | \mathrm{e}^{-\mathrm{i}S(t)} \mathrm{e}^{-\mathrm{i}\kappa(t)} (H - \mathrm{i}\partial_t) \mathrm{e}^{\mathrm{i}\kappa(t)} \mathrm{e}^{\mathrm{i}S(t)} | \mathrm{MC} \rangle \right\}_T^{(2)}$$
(437)

we invoke the Baker-Campbell-Hausdorff (BCH) expansion

$$e^{-A}Be^{A} = B + [B, A] + \frac{1}{2!}[[B, A], A] + \frac{1}{3!}[[[B, A], A], A], A] + \dots$$

(438)

to write the part of $\mathcal{Q}^{(2)}$ containing the Hamiltonian $H = H_0 + V(t)$ in the form

$$\{\langle \widetilde{\mathrm{MC}} | H | \widetilde{\mathrm{MC}} \rangle \}_{\mathrm{T}}^{(2)} = \{ \langle \mathrm{MC} | \left(\mathrm{i} \left[V(t), S^{(1)}(t) + \kappa^{(1)}(t) \right] - \left[\left[H_{0}, \kappa^{(1)}(t) \right], S^{(1)}(t) \right] - \frac{1}{2} \left[\left[H_{0}, S^{(1)}(t) \right], S^{(1)}(t) \right] - \frac{1}{2} \left[\left[H_{0}, \kappa^{(1)}(t) \right], \kappa^{(1)}(t) \right] \right] |\mathrm{MC} \rangle \}_{\mathrm{T}}$$
(439)

where the terms involving second-order rotation parameters vanish by the 2n + 1 rule. Likewise, using the commutator relation $[\partial_t, X] = \dot{X}$ for a general operator *X* we find that the time-derivative term in eq 437 becomes

$$\begin{split} \left\{ \langle \widetilde{\mathrm{MC}} | -\mathrm{i}\partial_t | \widetilde{\mathrm{MC}} \rangle \right\}_T^{(2)} \\ &= \left\{ -\mathrm{i} \langle \mathrm{MC} | \left(-\frac{1}{2} \left[\dot{S}^{(1)}, S^{(1)} \right] - \frac{1}{2} \left[\dot{\kappa}^{(1)}, \kappa^{(1)} \right] \right. \\ &\left. - \left[\dot{\kappa}^{(1)}, S^{(1)} \right] \right) | \mathrm{MC} \rangle \right\}_T \end{split}$$
(440)

The second-order quasi-energy may be written in terms of the different frequencies and perturbations as

$$\mathcal{Q}^{(2)} = \sum_{B_1, B_2} \varepsilon_{B_1} \varepsilon_{B_2} \mathcal{Q}^{B_1 B_2}, \ \omega_{B_1} + \omega_{B_2} = 0$$
(441)

where $\mathcal{Q}^{B_1B_2} = \mathcal{Q}^{B_2B_1}$ and the frequency condition excludes contributions that in any case vanish upon time averaging.

Let us now consider one particular nonvanishing component of the second-order quasi-energy $\mathcal{Q}^{\;(2)}$

$$\mathcal{Q}^{B_1B_2} = \frac{1}{2} \frac{\mathrm{d}^2 \mathcal{Q}^{(2)}}{\mathrm{d}\varepsilon_{B_1}\mathrm{d}\varepsilon_{B_2}} \tag{442}$$

obtained by differentiation of eqs 439 and 440 with respect to ε_{B_1} and ε_{B_2} . From the first term in eq 439 we obtain

$$-\frac{i\frac{d^{2}\{\langle MC|[S^{(1)}(t) + \kappa^{(1)}(t), V(t)]|MC\rangle\}_{T}}{d\varepsilon_{B_{1}}d\varepsilon_{B_{2}}}$$

$$= -i\{\{\langle MC|[S^{B_{1}} + \kappa^{B_{1}}, V_{B_{2}}]|MC\rangle$$

$$+\langle MC|[S^{B_{2}} + \kappa^{B_{2}}, V_{B_{1}}]|MC\rangle\}e^{-i(\omega_{B_{1}} + \omega_{B_{2}})t}\}_{T}$$

$$= -2iP_{[1, 2]}\langle MC|[S^{B_{1}} + \kappa^{B_{1}}, V_{B_{2}}]|MC\rangle$$
(443)

where $P_{[1,2]}$ averages over permutations, see eq 222. To express eq 443 in a compact manner, we introduce the column vectors $\boldsymbol{\beta}^{B}$ and \mathbf{T}^{\dagger} containing the first-order orbital- and

state-rotation parameters and the corresponding operators, respectively

$$\boldsymbol{\beta}^{B} = \begin{pmatrix} \boldsymbol{\eta}^{B} \\ \boldsymbol{\eta}^{-B^{*}} \end{pmatrix} = \begin{pmatrix} \boldsymbol{\kappa}^{B} \\ \boldsymbol{S}^{B} \\ \boldsymbol{\kappa}^{-B^{*}} \\ \boldsymbol{S}^{-B^{*}} \end{pmatrix}, \quad \mathbf{T}^{\dagger} = \begin{pmatrix} \mathbf{t} \\ \mathbf{t}^{\dagger} \end{pmatrix} = \begin{pmatrix} \mathbf{q} \\ \mathbf{R} \\ \mathbf{q}^{\dagger} \\ \mathbf{R}^{\dagger} \end{pmatrix}$$
(444)

Noting that the conjugate row vectors

$$\boldsymbol{\beta}^{-B^{\dagger}} = \begin{pmatrix} \boldsymbol{\eta}^{-B^{*}} & \boldsymbol{\eta}^{B} \end{pmatrix} = \begin{pmatrix} \kappa^{-B^{*}} & \boldsymbol{S}^{-B^{*}} & \kappa^{B} & \boldsymbol{S}^{B} \end{pmatrix}$$
(445)

$$\mathbf{T} = (\mathbf{t}^{\dagger} \quad \mathbf{t}) = (\mathbf{q}^{\dagger} \quad \mathbf{R}^{\dagger} \quad \mathbf{q} \quad \mathbf{R})$$
(446)

contain the same elements as the column vectors $\boldsymbol{\beta}^{B}$ and \mathbf{T}^{\dagger} (but reordered) we may write $S^{B_{1}} + \kappa^{B_{1}}$ of eqs 432 and 433 in the following two equivalent forms

$$S^{B_1} + \kappa^{B_1} = \mathbf{T}\boldsymbol{\beta}^{B_1} = \boldsymbol{\beta}^{-B_1 +} \mathbf{T}^{\dagger}$$
(447)

Returning to eq 443, we now obtain

$$-2iP_{[1, 2]}\langle MC| [S^{B_{1}} + \kappa^{B_{1}}, V_{B_{2}}] |MC\rangle$$

$$= -2iP_{[1, 2]}\beta^{-B_{1}+}\langle MC| [T^{+}, V_{B_{2}}] |MC\rangle$$

$$= -2iP_{[1, 2]}\beta^{-B_{1}+}V_{B_{2}}^{[1]}$$
(448)

where we introduced the column vector

$$\mathbf{V}_{B_2}^{[1]} = \langle \mathrm{MC} | \left[\mathbf{T}^{\dagger}, \, V_{B_2} \right] | \mathrm{MC} \rangle \tag{449}$$

Likewise, we obtain after some algebra the following expression for the remaining terms in eq 439

$$-\frac{d^{2}}{d\varepsilon_{B_{1}}d\varepsilon_{B_{2}}} \Big\{ \langle \mathrm{MC} | \left[\left[H_{0}, \, \kappa^{(1)}(t) \right], \, S^{(1)}(t) \right] + \frac{1}{2} \Big[\left[H_{0}, \, S^{(1)}(t) \right], \, S^{(1)}(t) \Big] \\ + \frac{1}{2} \Big[\left[H_{0}, \, \kappa^{(1)}(t) \right], \, \kappa^{(1)}(t) \Big] | \mathrm{MC} \rangle \Big\}_{T} \\ = -2P_{[1, \ 2]} \langle \mathrm{MC} | \left[\left[H_{0}, \, \kappa^{B_{1}} \right], \, S^{B_{2}} \right] \\ + \frac{1}{2} \Big[\left[H_{0}, \, S^{B_{1}} \right], \, S^{B_{2}} \Big] + \frac{1}{2} \Big[\left[H_{0}, \, \kappa^{B_{1}} \right], \, \kappa^{B_{2}} \Big] | \mathrm{MC} \rangle \\ = P_{[1, \ 2]} \langle \mathrm{MC} | \left[\left[\kappa^{B_{1}}, \, H_{0} \right], \, S^{B_{2}} \right] + \left[S^{B_{1}}, \, \left[H_{0}, \, \kappa^{B_{2}} \right] \right] \\ + \left[S^{B_{1}}, \, \left[H_{0}, \, S^{B_{2}} \right] \right] + \left[\kappa^{B_{1}}, \, \left[H_{0}, \, \kappa^{B_{2}} \right] \right] | \mathrm{MC} \rangle \\ = P_{[1, \ 2]} \beta^{-B_{1}+} \mathbf{E}^{[2]} \beta^{B_{2}}$$
(450)

Here we introduced the electronic Hessian $E^{\left[2\right]}$, which in terms of the matrices

$$\mathbf{A} = \begin{pmatrix} \langle \mathrm{MC} | [\mathbf{q}, [H_0, \mathbf{q}^{\dagger}]] | \mathrm{MC} \rangle & \langle \mathrm{MC} | [[\mathbf{q}, H_0], \mathbf{R}^{\dagger}]] | \mathrm{MC} \rangle \\ \langle \mathrm{MC} | [\mathbf{R}, [H_0, \mathbf{q}^{\dagger}]] | \mathrm{MC} \rangle & \langle \mathrm{MC} | [\mathbf{R}, [H_0, \mathbf{R}^{\dagger}]] | \mathrm{MC} \rangle \end{pmatrix}$$

$$(451)$$

$$\mathbf{B} = \begin{pmatrix} \langle \mathrm{MC} | [\mathbf{q}, [H_0, \mathbf{q}]] | \mathrm{MC} \rangle & \langle \mathrm{MC} | [[\mathbf{q}, H_0], \mathbf{R}]] | \mathrm{MC} \rangle \\ \langle \mathrm{MC} | [\mathbf{R}, [H_0, \mathbf{q}]] | \mathrm{MC} \rangle & \langle \mathrm{MC} | [\mathbf{R}, [H_0, \mathbf{R}]] | \mathrm{MC} \rangle \end{pmatrix}$$

$$(452)$$
$$\mathbf{E}^{[2]} = \begin{pmatrix} \mathbf{A} & \mathbf{B} \\ \mathbf{B}^* & \mathbf{A}^* \end{pmatrix} \tag{453}$$

The Hessian $\mathbf{E}^{[2]}$ is Hermitian, as follows from the Hermiticity of **A** and symmetry of **B**

$$\mathbf{A} = \mathbf{A}^{\dagger}, \quad \mathbf{B} = \mathbf{B}^{\mathrm{T}}$$
(454)

assuming that the Brillouin conditions in eq 425 hold. Combining eqs 448 and 450 we find that the Hamiltonian part of the second-order quasi-energy $\mathcal{L}^{(2)}$ in eq 439 becomes

$$\frac{\mathrm{d}^{2}\left\{\langle \widetilde{\mathbf{MC}}|H|\widetilde{\mathbf{MC}}\rangle\right\}_{\mathrm{T}}^{(2)}}{\mathrm{d}\varepsilon_{B_{1}}\mathrm{d}\varepsilon_{B_{2}}}$$
$$=2P_{[1, 2]}\left[-\mathrm{i}\boldsymbol{\beta}^{-B_{1}\dagger}\mathbf{V}_{B_{2}}^{[1]}+\frac{1}{2}\boldsymbol{\beta}^{-B_{1}\dagger}\mathbf{E}^{[2]}\boldsymbol{\beta}^{B_{2}}\right]$$
(455)

for a specific set of operator-frequency indices (B_1, B_2) .

It remains to consider the part of the second-order quasienergy that contains the time derivative in eq 440. Noting that the time derivatives of the first-order parameters in eqs 430 and 431 are given by

$$\dot{\kappa}^{(1)}(t) = -i \sum_{B} \omega_{B} \varepsilon_{B} \kappa^{B} e^{-i\omega_{B} t}$$
(456)

$$\dot{S}^{(1)}(t) = -i \sum_{B} \omega_{B} \varepsilon_{B} S^{B} e^{-i\omega_{B}t}$$
(457)

we obtain upon time differentiation of eq 440 with respect to ε_{B_1} and ε_{B_2} the expression

$$\frac{d^{2}\{\langle \widetilde{\mathrm{MC}} | -\mathrm{i}\partial_{t} | \widetilde{\mathrm{MC}} \rangle\}_{T}}{d\varepsilon_{B_{1}} d\varepsilon_{B_{2}}}$$

$$= 2P_{[1, 2]} \omega_{B_{1}} \langle \mathrm{MC} | [\kappa^{B_{1}} + S^{B_{1}}, \kappa^{B_{2}} + S^{B_{2}}] | \mathrm{MC} \rangle$$

$$= -P_{[1, 2]} \omega_{B_{2}} \boldsymbol{\beta}^{-B_{1} \dagger} \mathbf{S}^{[2]} \boldsymbol{\beta}^{B_{2}} \qquad (458)$$

To obtain the first equality of eq 458 we used the relation

$$2P_{[1, 2]}\omega_{B_1}\langle \mathrm{MC}|[\kappa^{B_1}, S^{B_2}]|\mathrm{MC}\rangle$$

= $2P_{[1, 2]}\omega_{B_1}\langle \mathrm{MC}|[S^{B_1}, \kappa^{B_2}]|\mathrm{MC}\rangle$ (459)

while for the second equality we used eq 447, the frequency relation $\omega_{B_1} = -\omega_{B_2}$ and introduced the Hermitian metric matrix

$$\mathbf{S}^{[2]} = \langle \mathrm{MC} | [\mathbf{T}^{\dagger}, \mathbf{T}] | \mathrm{MC} \rangle = \begin{pmatrix} \mathbf{\Sigma} & \mathbf{\Delta} \\ -\mathbf{\Delta}^{*} & -\mathbf{\Sigma}^{*} \end{pmatrix}$$
(460)

whose submatrices

$$\boldsymbol{\Sigma} = \begin{pmatrix} \langle \mathrm{MC} | [\mathbf{q}, \, \mathbf{q}^{\dagger}] | \mathrm{MC} \rangle & \langle \mathrm{MC} | [\mathbf{q}, \, \mathbf{R}^{\dagger}] | \mathrm{MC} \rangle \\ \langle \mathrm{MC} | [\mathbf{R}, \, \mathbf{q}^{\dagger}] | \mathrm{MC} \rangle & \langle \mathrm{MC} | [\mathbf{R}, \, \mathbf{R}^{\dagger}] | \mathrm{MC} \rangle \end{pmatrix}$$
(461)

$$\mathbf{\Delta} = \begin{pmatrix} \langle \mathrm{MC} | [\mathbf{q}, \mathbf{q}] | \mathrm{MC} \rangle & \langle \mathrm{MC} | [\mathbf{q}, \mathbf{R}] | \mathrm{MC} \rangle \\ \langle \mathrm{MC} | [\mathbf{R}, \mathbf{q}] | \mathrm{MC} \rangle & \langle \mathrm{MC} | [\mathbf{R}, \mathbf{R}] | \mathrm{MC} \rangle \end{pmatrix}$$
(462)

are Hermitian and antisymmetric, respectively

$$\Sigma = \Sigma^{\dagger}, \ \Delta = -\Delta^{\mathrm{T}}$$
 (463)

Combining eqs 455 and 458 we find that the (B_1,B_2) component of the second-order MCSCF quasi-energy of eq 442 becomes

$$\mathcal{P}^{B_{1}B_{2}} = P_{[1, 2]} \left(-i\boldsymbol{\beta}^{-B_{1}\dagger} \mathbf{V}^{[1]}_{B_{2}} + \frac{1}{2} \boldsymbol{\beta}^{-B_{1}\dagger} \left(\mathbf{E}^{[2]} - \omega_{B_{2}} \mathbf{S}^{[2]} \right) \boldsymbol{\beta}^{B_{2}} \right) \\ = \frac{1}{2} \left(-i\boldsymbol{\beta}^{-B_{1}\dagger} \mathbf{V}^{[1]}_{B_{2}} - i\boldsymbol{\beta}^{-B_{2}\dagger} \mathbf{V}^{[1]}_{B_{1}} + \boldsymbol{\beta}^{-B_{1}\dagger} \left(\mathbf{E}^{[2]} - \omega_{B_{2}} \mathbf{S}^{[2]} \right) \boldsymbol{\beta}^{B_{2}} \right)$$

$$(464)$$

where we used the paired structure of the Hermitian matrices $\mathbf{E}^{[2]}$ and $\mathbf{S}^{[2]}$ in eqs 453 and 460 and the relation $\omega_{B_1} = -\omega_{B_2}$ to arrive at the last expression. In the next subsection we shall use this expression to derive the first-order MCSCF response equation.

4.2.3. First-Order MCSCF Response Equation. According to the quasi-energy variation principle in eq 427, the quasi-energy must be stationary to all orders with respect to variations in all frequency and operator components. In particular, the second-order quasi-energy component $\mathcal{Q}^{B_1B_2}$ must be stationary with respect to variations in $\boldsymbol{\beta}^{-B_1+}$

$$\frac{\partial \mathcal{L}^{B_1 B_2}}{\partial \boldsymbol{\beta}^{-B_1 \dagger}} = \mathbf{0} \tag{465}$$

From eq 464 it then follows that the first-order variations satisfy the set of linear equations

$$\left(\mathbf{E}^{[2]} - \boldsymbol{\omega}_{B_2} \mathbf{S}^{[2]}\right) \boldsymbol{\beta}^{B_2} = \mathbf{i} \mathbf{V}_{B_2}^{[1]}$$

$$\tag{466}$$

which is our final result for the MCSCF linear response equations. We note that if all matrix blocks involving statetransfer operators are removed then eq 466 reduces to the Hartree–Fock linear response equations in the MO basis, the MO counterpart to the response equations in the AO basis in eq 404.

4.2.4. Linear MCSCF Response Function. The MCSCF linear response function $\langle\langle A;B \rangle\rangle_{\omega_B}$ is equal to the second derivative of the time-averaged quasi-energy with respect to the perturbation strengths ε_A and ε_B . Using eqs 441 and 464 we thus obtain the following expression for the linear response function

$$\langle\langle A; B \rangle\rangle_{\omega_{B}} = \frac{\mathrm{d}^{2}\mathcal{Q}^{(2)}}{\mathrm{d}\varepsilon_{A}\mathrm{d}\varepsilon_{B}} = 2\mathcal{Q}^{AB} = -\mathrm{i}\boldsymbol{\beta}^{-A\dagger}\mathbf{V}_{B}^{[1]} - \mathrm{i}\boldsymbol{\beta}^{-B\dagger}\mathbf{V}_{A}^{[1]} + \boldsymbol{\beta}^{-A\dagger}\left(\mathbf{E}^{[2]} - \omega_{B}\mathbf{S}^{[2]}\right)\boldsymbol{\beta}^{B}, \quad \omega_{A} = -\omega_{B}$$

$$(467)$$

By inserting the response equations in eq 466 or in the equivalent equations for perturbation A, the MCSCF linear response function may be written in two alternative forms, in accordance with the general symmetry relations in eq 283

$$\langle \langle A; B \rangle \rangle_{\omega_{B}} = -\mathbf{i} \boldsymbol{\beta}^{-B\dagger} \mathbf{V}_{B}^{[1]} = -\mathbf{i} \boldsymbol{\beta}^{-A\dagger} \mathbf{V}_{A}^{[1]}$$

= $\langle \langle B; A \rangle \rangle_{\omega_{A}}, \ \omega_{A} = -\omega_{B}$ (468)

When the matrix blocks involving state-transfer operators are omitted, eq 468 reduces to the Hartree–Fock linear response function in the MO representation, which is the MO counterpart of the AO-based expression in eq 418. The MCSCF quadratic and cubic response functions are obtained by differentiation of the third- and fourth-order quasi-energies, respectively. For a detailed treatment of higher order response functions, see ref 70 as well as the papers describing the implementation of the MCSCF quadratic⁹⁹ and cubic¹⁰¹ response functions.

4.3. Coupled-Cluster Theory

Coupled-cluster theory^{30,117} provides the most successful framework so far developed for high-accuracy calculations of the electronic structure and properties of molecules. The theory describes the system in terms of virtual excitations from a reference state, typically taken to be the Hartree–Fock wave function. Using an exponential ansatz for the wave function, a size-extensive description is achieved. The full configuration-interaction (FCI) wave function is approached as increasingly higher order excitations are included in the treatment of the system.

In this section, we present a framework for evaluation of molecular properties in coupled-cluster theory, based on construction of a variational Lagrangian. In the development presented here, we restrict ourselves to the standard formulation of coupled-cluster theory, where the Hartree—Fock state is a good approximation to the reference state. These coupledcluster models are thus not sufficiently flexible to describe electronic structures characterized by large static correlation effects. Typically, therefore, these models cannot describe bond breaking and molecular dissociation. We do not treat multiconfigurational coupled-cluster theory nor explicitly correlated coupledcluster theory. Such formulations of coupled-cluster theory are becoming increasingly important but have so far not been extensively developed with respect to calculation of molecular properties.

4.3.1. Coupled-Cluster Model. In coupled-cluster theory the wave function is written in the form

$$|CC\rangle = \exp(T)|HF\rangle \tag{469}$$

where $|\text{HF}\rangle$ is the Hartree–Fock reference state and the cluster operator is given by

$$T = \sum_{\mu} t_{\mu} \tau_{\mu} \tag{470}$$

The operator manifold τ_{μ} constitutes a set of commuting excitation operators

$$[\tau_{\mu}, \tau_{\nu}] = 0 \tag{471}$$

which, when working on the Hartree-Fock reference state, generate excited electronic configurations

$$\tau_{\mu}|\mathrm{HF}\rangle = |\mu\rangle \tag{472}$$

In practice, the cluster operator is partitioned as

$$T = T_1 + T_2 + \dots (473)$$

where T_1 produces single excitations, T_2 double excitations, and so on

$$T_1 = \sum_{AI} t_I^A a_A^\dagger a_I \tag{474}$$

$$T_2 = \sum_{A > B} \sum_{I > J} t_{IJ}^{AB} a_A^{\dagger} a_I a_B^{\dagger} a_J$$

$$\tag{475}$$

With each excitation operator τ_{μ} in eq 470 there is an associated coupled-cluster amplitude t_{μ} whose squared value is proportional to the probability of the associated virtual excitation occurring.

The amplitudes of the coupled-cluster wave function in eq 469 are not determined variationally. Rather, they are determined by projection, rewriting the Schrödinger equation in the form

$$\exp(-T)H_0 \exp(T)|\text{HF}\rangle = E_{\text{CC}}|\text{HF}\rangle$$
(476)

This similarity-transformed Schrödinger equation is subsequently projected from the left by the manifold

$$\langle \mu | = \langle \mathrm{HF} | \tau_{\mu}^{\dagger} \tag{477}$$

to yield the nonlinear coupled-cluster amplitude equations

$$\langle \mu | \exp(-T) H_0 \exp(T) | \mathrm{HF} \rangle = 0 \tag{478}$$

whose solution yields the coupled-cluster amplitudes and wave function. Furthermore, projecting the Schrödinger equation in eq 476 against the Hartree–Fock state we obtain the coupledcluster energy

$$E_{\rm CC} = \langle \rm HF | \exp(-T) H_0 \exp(T) | \rm HF \rangle$$
(479)

In coupled-cluster theory, the energy is thus not calculated as an expectation value.

The different models of the coupled-cluster hierarchy differ in the truncation of the coupled-cluster operator T of eq 470. In particular, the coupled-cluster singles-doubles (CCSD) model¹¹⁸ is obtained by truncating the expansion after the double excitations

$$|\text{CCSD}\rangle = \exp(T_1 + T_2)|\text{HF}\rangle \tag{480}$$

The CCSD amplitude equations are obtained from eq 478 by projecting against the single $\langle \mu_1 |$ and double $\langle \mu_2 |$ excitation space. Using the BCH expansion in eq 438, we obtain the following equations for the singles and doubles amplitudes

$$\langle \mu_1 | H_0^{T_1} + \left[H_0^{T_1}, T_2 \right] | \text{HF} \rangle = 0$$
 (481)

$$\left\langle \mu_{2}|H_{0}^{T_{1}} + \left[H_{0}^{T_{1}}, T_{2}\right] + \frac{1}{2}\left[\left[H_{0}^{T_{1}}, T_{2}\right], T_{2}\right]|HF\right\rangle = 0$$
(482)

where we introduced the T_1 -transformed Hamiltonian

$$H_0^{T_1} = \exp(-T_1)H_0 \exp(T_1)$$
(483)

Alternatively, we may truncate the cluster operator after the triple excitations, giving the coupled-cluster single-double-triple (CCSDT) model.^{119,120} Optimization of the CCSD energy scales as n^6 and of the CCSDT energy as n^8 , where n is the number of orbitals. The high cost of the optimization makes the coupled-cluster hierarchy applicable only at low levels of theory. On the other hand, the coupled-cluster hierarchy converges rapidly to the FCI electronic energy, at least in the absence of static correlation.

In coupled-cluster theory, the scaling of each model is determined by the scaling of the amplitude equations for the highest excitations, for example, the n^6 scaling of the CCSD model arises from the last two terms of the doubles equations in

eq 482. As the contribution to the correlation energy decreases with increasing excitation levels, it makes sense to introduce an intermediate coupled-cluster hierarchy of a lower scaling by making approximations in the highest excitation amplitude equations. For this purpose, we introduce the Møller–Plesset partitioning of the Hamiltonian

$$H_0 = F + U \tag{484}$$

where *F* is the (zero-order) Fock operator and *U* the (first-order) fluctuation potential, representing the difference between the true electron–electron interactions and the averaged Fock potential. A hierarchy of intermediate coupled-cluster models is now introduced by approximating the highest excitation amplitude equations by retaining only the leading terms of Møller–Plesset theory, keeping the lower excitation amplitude equations unchanged. In the CC2 model,¹²¹ for example, we keep the CCSD singles equations eq 481, whereas the CCSD doubles equations in eq 482 are approximated by the lowest order Møller–Plesset terms

$$\langle \mu_2 | H_0^{T_1} + [F, T_2] | HF \rangle = 0 \tag{485}$$

thereby reducing the scaling to n^5 . Likewise, by approximating the CCSDT triples equations in the same manner we obtain the CC3 model, ^{122,123} with an n^7 rather than n^8 scaling. Note that in the resulting coupled-cluster response hierarchy consisting of the models CC2, CCSD, CC3, CCSDT, the cluster amplitudes are determined iteratively by solving nonlinear amplitude equations and the scaling increases by a factor of n at each level in the hierarchy.

A simpler coupled-cluster hierarchy is generated if, for each intermediate model, the contributions from the highest order excitations are calculated from perturbation theory rather than iteratively. The CC2 model then reduces to the second-order Møller–Plesset (MP2) model by setting the t_1 amplitudes to zero, that is, by omitting the singles amplitude equation and determining the doubles amplitudes from eq 485 where the T1-transformed Hamiltonian $H_0^{T_1}$ is replaced by H_0 .

In coupled-cluster theory, the triple excitations give contributions to the energy of orders four and higher in the fluctuation potential. In the perturbation-based intermediate model obtained from CCSDT theory, this fourth-order energy contribution is added (in a slightly modified form) to the CCSD energy, yielding the coupled-cluster singles—doubles—perturbativetriples (CCSD(T)) model.¹²⁴ Proceeding in this manner, we generate the coupled-cluster energy hierarchy consisting of the models MP2, CCSD, CCSD(T), CCSDT, with the same computational scaling as the coupled-cluster response hierarchy introduced above. Note that, in the energy hierarchy, the highest order contributions of the intermediate models are treated perturbatively rather than iteratively, thereby reducing the cost of the calculations relative to that of the response hierarchy.

In response theory, molecular properties are determined for ground and excited states and for transitions between these states. To apply response theory successfully, we therefore need to have a good zero-order description of these states. As most excited states are dominated by a single excitation from the ground-state configuration, response theory can only be applied confidently for electronic-structure models where single excitations are explicitly treated. In the MP2 model, single excitations do not contribute; consequently, this model does not constitute a useful starting point for determining molecular response functions. Likewise, in the CCSD(T) model, there is no direct interaction between the triple- and single-excitation spaces; the CCSD(T) model therefore also does not constitute a useful starting point for a response treatment. By contrast, in the intermediate models of the coupled-cluster response hierarchy the equation for the single excitations is not truncated, making these models suitable for response-function calculations.

Even though the intermediate models in the coupled-cluster energy hierarchy do not provide a sufficiently flexible description of single excitations for molecular properties, it is possible to improve the description by allowing the orbitals to relax, thereby effectively introducing single excitations. However, when response functions are determined with orbital relaxation included, the orbital-rotation parameters and amplitudes introduce singularities in the response function, rendering the pole and residue analysis problematic. Hence, only static molecular properties can be meaningfully obtained for models with orbital relaxation included. The static molecular properties obtained in this manner may be viewed as an analytic formulation of a finite-field energy determined in the presence of the perturbation, see section 4.4, where we describe how static molecular properties may be determined with orbital relaxation included. In the remainder of this section we consider time-dependent perturbations, thereby restricting ourselves to the coupled-cluster response hierarchy of wave functions.

4.3.2. Coupled-Cluster Quasi-Energy Lagrangian. In section 3 we set up a time-averaged quasi-energy Lagrangian for an intermediately normalized wave function, suitable for coupled-cluster theory. In the following, the optimized coupled-cluster wave function for the unperturbed system in eq 469 is written as

$$|\mathrm{CC}^{(0)}\rangle = \exp(T^{(0)})|\mathrm{HF}\rangle \tag{486}$$

By applying eq 193 with the identifications

$$R\rangle = |HF\rangle \tag{487}$$

$$|\hat{\mathbf{0}}^{(0)}\rangle = |\mathbf{C}\mathbf{C}^{(0)}\rangle \tag{488}$$

$$|\hat{0}\rangle = |CC(t)\rangle = \exp(T(t))|HF\rangle$$
 (489)

$$\langle \overline{0}| = \sum_{\mu} \overline{t}_{\mu}(t) \langle \overline{\mu}(t)| = \sum_{\mu} \overline{t}_{\mu}(t) \langle \mu| \exp(-T(t))$$
(490)

and noting that $\langle \overline{0} | \hat{0} \rangle = 0$ as assumed in the derivation of eq 193, we obtain the coupled-cluster quasi-energy Lagrangian⁶⁶

$$\mathscr{L} = \operatorname{Re}\left\{ \langle \mathrm{HF}|H|\mathrm{CC}(t) \rangle + \sum_{\mu} \overline{t}_{\mu}(t) \langle \overline{\mu}(t)|H - \mathrm{i}\partial_{t}|\mathrm{CC}(t) \rangle \right\}_{T}$$
(491)

with $H = H_0 + V(t)$, applicable to the standard hierarchy of coupled-cluster models CCS, CCSD, CCSDT, and so on. We note that the Lagrangian \mathscr{L} is the time-averaged sum of the coupled-cluster quasi-energy Q(t) and the (real part of the) time-dependent coupled-cluster amplitude equations with associated multipliers

$$Q(t) = \operatorname{Re}\langle \mathrm{HF}|H|\mathrm{CC}(t)\rangle \tag{492}$$

$$\overline{a}(t)|H - i\partial_t|CC(t)\rangle = 0$$
(493)

<

which are the coupled-cluster analogues of the expressions for exact theory in eqs 183 and 187. For later convenience, we also introduce the time-averaged complex quasi-energy Lagrangian

$$\mathscr{L}^{c} = \left\{ \langle \mathrm{HF}|H|\mathrm{CC}(t)\rangle + \sum_{\mu} \overline{t}_{\mu}(t)\langle \overline{\mu}(t)|H - \mathrm{i}\partial_{t}|\mathrm{CC}(t)\rangle \right\}_{T}$$
(494)

such that

$$\mathcal{L} = \operatorname{Re}\mathcal{L}^{c} \tag{495}$$

For the intermediate CC*n* models, the Lagrangian in eq 491 must be modified as discussed in section 4.3.6.

The cluster amplitudes and multipliers are now expanded in orders of the perturbation

$$T(t) = T^{(0)} + T^{(1)}(t) + T^{(2)}(t) + \dots$$
(496)

$$\overline{t}(t) = \overline{t}^{(0)} + \overline{t}^{(1)}(t) + \overline{t}^{(2)}(t) + \dots$$
(497)

where the zero-order cluster amplitudes are obtained by solving the amplitude equations in eq 478 and the zero-order multipliers are obtained as described in section 4.3.3. Likewise, the realvalued and complex quasi-energy Lagrangians are expanded in orders of the perturbation

$$\mathcal{L} = \mathcal{L}^{(0)} + \mathcal{L}^{(1)} + \mathcal{L}^{(2)} + \dots$$
 (498)

$$\mathcal{L}^{c} = \mathcal{L}^{c(0)} + \mathcal{L}^{c(1)} + \mathcal{L}^{c(2)} + \dots$$
 (499)

We here restrict ourselves to determining $\mathcal{L}^{(0)}$, $\mathcal{L}^{(1)}$, and $\mathcal{L}^{(2)}$; the latter will allow us to calculate the coupled-cluster linear response function. By the 2n + 1 and 2n + 2 rules we then need to determine the first-order amplitudes and zero-order multipliers. For future reference, we write the first-order amplitude operator in terms of its frequency components

$$T^{(1)}(t) = \sum_{B} \varepsilon_{B} T_{B}^{(1)} e^{-i\omega_{B}t}$$
(500)

where $T_B^{(1)}$ contains the first-order cluster amplitudes

$$T_B^{(1)} = \sum_B t_\mu^B \tau_\mu$$
 (501)

and where *B* is a combined operator-frequency index.

4.3.3. First-Order Molecular Properties. By analogy with eq 311, the zero-order coupled-cluster Lagrangian becomes

$$\mathscr{L}^{(0)} = \operatorname{Re}\langle \Lambda^{(0)} | H_0 | \operatorname{CC}^{(0)} \rangle = E_{\operatorname{CC}}$$
(502)

where we introduced the zero-order coupled-cluster wave function $|CC^{(0)}\rangle$ of eq 486 and, by analogy with eq 320, the zero-order coupled-cluster lambda state

$$\langle \Lambda^{(0)} | = \langle \mathrm{HF} | + \sum_{\mu} \overline{t}_{\mu}^{(0)} \langle \overline{\mu} |,$$

$$\langle \overline{\mu} | = \langle \mu | \exp(-T^{(0)})$$
(503)

The stationary conditions for the zero-order Lagrange multipliers give the coupled-cluster amplitude equations, while the stationary conditions for the amplitudes determine the zero-order multipliers

$$\frac{\mathrm{d}\mathscr{Z}^{c(0)}}{\mathrm{d}t_{\mu}^{(0)}} = \langle \bar{\mu} | H_0 | \mathrm{CC}^{(0)} \rangle = 0$$
(504)

$$\frac{\mathrm{d} \mathcal{Z}^{c(0)}}{\mathrm{d} t_{\mu}^{(0)}} = \langle \Lambda^{(0)} | [H_0, \tau_{\mu}] | \mathrm{CC}^{(0)} \rangle = 0$$
(505)

Whereas eq 504 is the cluster amplitude equation in eq 478, the linear equations in eq 505 determine the zero-order multipliers and may be written in the form

$$\mathbf{\bar{t}}^{(0)}\mathbf{A} = -\boldsymbol{\eta} \tag{506}$$

where A is the nonsymmetric Jacobian matrix and η a row vector

$$A_{\mu\nu} = \langle \overline{\mu} | [H_0, \tau_{\nu}] | CC^{(0)} \rangle$$
(507)

$$\eta_{\mu} = \langle \mathrm{HF} | [H_0, \tau_{\mu}] | \mathrm{CC}^{(0)} \rangle \tag{508}$$

According to the 2n + 1 and 2n + 2 rules for the wave function parameters and multipliers, respectively, the first-order molecular properties may be written as

$$\mathscr{L}_{B_0}^{(1)} = \frac{1}{2} C^{\pm} \langle \Lambda^{(0)} | V_{B_0} | CC^{(0)} \rangle, \quad \omega_{B_0} = 0$$
(509)

by analogy with eq 312.

4.3.4. Second-Order Molecular Properties. According to the 2n + 1 and 2n + 2 rules, the second-order coupled-cluster Lagrangian $\mathscr{L}^{(2)}$ may be constructed from the first-order amplitudes and the zero-order multipliers. Using the expansion in eq 302 we now consider the $\mathscr{L}^{(2)}_{B_0,B_1}$ component. To this end, we note that

$$|CC(t)\rangle = \exp(T(t))|HF\rangle$$
$$= \exp(T(t) - T^{(0)})|CC^{(0)}\rangle$$
(510)

$$\langle \mu | \exp(T(t)) = \langle \overline{\mu} | \exp(T^{(0)} - T(t))$$
(511)

and apply the BCH expansion in eq 438 to expand the quasienergy Lagrangian in eq 491, yielding

$$\mathcal{Z}_{B_{0}, B_{1}}^{(2)} = \frac{1}{2} C^{\pm} P_{[0, 1]} \Big(\langle \Lambda^{(0)} | \Big[V_{B_{0}}, T_{B_{1}}^{(1)} \Big] | CC^{(0)} \rangle \\ + \langle \Lambda^{(0)} | \Big[\Big[H_{0}, T_{B_{0}}^{(1)} \Big], T_{B_{1}}^{(1)} \Big] | CC^{(0)} \rangle \Big)$$
(512)

where we also used eq 471. Next, by inserting the expression for $T_{B_1}^{(1)}$ in eq 501 and introducing the notation

$$\eta_{\mu}^{B_0} = \langle \Lambda^{(0)} | [V_{B_0}, \tau_{\mu}] | CC^{(0)} \rangle$$
(513)

$$F_{\mu\nu} = \langle \Lambda^{(0)} | [[H_0, \tau_{\mu}], \tau_{\nu}] | CC^{(0)} \rangle$$
(514)

(where F should not be confused with the Fock matrix), we may express the second-order Lagrangian more compactly as

$$\mathscr{L}_{B_{0},B_{1}}^{(2)} = \frac{1}{2} C^{\pm} P_{[0,1]} \left(\sum_{\mu} \eta_{\mu}^{B_{0}} t_{\mu}^{B_{1}} + \sum_{\mu\nu} F_{\mu\nu} t_{\mu}^{B_{0}} t_{\nu}^{B_{1}} \right)$$
(515)

Finally, using eq 307 we arrive at the following expression for the coupled-cluster linear response function

$$\langle\langle A; B \rangle\rangle_{\omega_{B}} = C^{\pm} P_{[A, B]} \left(\sum_{\mu} \eta^{A}_{\mu} t^{B}_{\mu} + \sum_{\mu\nu} F_{\mu\nu} t^{A}_{\mu} t^{B}_{\nu} \right)$$
(516)

where we adopted the standard response notation, renaming V_{B_0} and V_{B_1} to A and B, respectively.

In the coupled-cluster response function of eq 516, the zeroorder multipliers are obtained from eq 506 while the first-order amplitudes are determined by solving eq 493 to first order. To obtain an explicit expression for the amplitude equations it is convenient to rewrite eq 493 in the form

$$\langle \overline{a} | \exp(T^{(0)} - T(t)) \Big(H_0 + V(t) - i\partial_t \exp(T(t) - T^{(0)}) | CC^{(0)} \rangle = 0$$

(517)

where we used eqs 510 and 511. By invoking the BCH expansion eq 438 in eq 517 and collecting first-order terms followed by differentiation with respect to a particular perturbation strength ε_{B_0} we obtain an equation for the first-order amplitudes

$$\sum_{\nu} \langle \overline{\mu} | [H_0, \tau_{\nu}] | CC^{(0)} \rangle t_{\nu}^{B_0} + \langle \overline{\mu} | V_{B_0} | CC^{(0)} \rangle$$
$$- \omega_{B_0} \sum_{\nu} \langle \overline{\mu} | \tau_{\nu} | CC^{(0)} \rangle t_{\nu}^{B_0} = 0$$
(518)

By introducing

$$\xi_{\mu}^{B_0} = \langle \overline{\mu} | V_{B_0} | CC^{(0)} \rangle \tag{519}$$

and noting that

$$\langle \overline{\mu} | \tau_{\nu} | \mathrm{CC}^{(0)} \rangle = \langle \mu | \nu \rangle = \delta_{\mu\nu}$$
(520)

we find that eq 518 may be written in the matrix form

$$\mathbf{A} - \omega_{B_0} \mathbf{I} \mathbf{t}^{B_0} = -\boldsymbol{\xi}^{B_0} \tag{521}$$

The response function and response equations developed above may be applied to the standard coupled-cluster models (CCS, CCSD, CCSDT, etc.) by introducing the appropriate truncations in the cluster operator and in the projection manifold.

4.3.5. Excitation Energies and Residues. Excitation energies correspond to the poles of the linear response function and may be determined by solving the Jacobian eigenvalue equation. Since the Jacobian is nonsymmetric, we have left and right eigenvalue equations

$$\mathbf{A}\mathbf{R}^{k} = \omega_{k}\mathbf{R}^{k}, \ \mathbf{L}^{k}\mathbf{A} = \mathbf{L}^{k}\omega_{k}, \ \mathbf{L}^{k}\mathbf{R}^{l} = \delta_{kl}$$
(522)

where \mathbf{L}^k is a left row eigenvector and \mathbf{R}^k is a right column eigenvector. In matrix form we may express the Jacobian eigenvalue problem as

$$\mathbf{LAR} = \mathbf{\Omega}, \ \mathbf{LR} = \mathbf{I}$$
(523)

where the diagonal matrix Ω contains the excitation energies. When all virtual excitations within a given orbital basis are included in *T* in eq 473, the Jacobian has the same real eigenvalues as the shifted Hamiltonian operator $H_0 - E_{\rm CC}$. Conversely, when *T* is truncated, the nonsymmetric Jacobian may have complex eigenvalues or even not be diagonalizable. Complex eigenvalues (and hence complex excitation energies) are unphysical and may occur because the coupled-cluster method is nonvariational. However, such eigenvalues are usually not encountered for electronic systems dominated by a single determinant and may always be removed by extending the excitation manifold.

Assuming that a diagonal representation exists, the linear response function in eq 516 may be expressed as

$$\langle\langle A;B\rangle\rangle_{\omega_{B}} = C^{\pm}P_{[A,B]}\left(\sum_{k}\frac{\mathrm{d}\eta_{k}^{Ad}\xi_{k}^{B}}{\omega_{B}-\omega_{k}} + \sum_{kl}\frac{\mathrm{d}\xi_{k}^{Ad}F_{kl}d\xi_{l}^{B}}{(\omega_{B}+\omega_{k})(\omega_{B}-\omega_{l})}\right)$$
(524)

where

$${}^{\mathbf{h}}\boldsymbol{\eta}_{k}^{A} = \boldsymbol{\eta}^{A}\mathbf{R}^{k}$$
(525)

$${}^{\mathrm{d}}\xi_k^A = \mathbf{L}^k \boldsymbol{\xi}^A \tag{526}$$

$${}^{\mathrm{d}}F_{kl} = (\mathbf{R}^k)^{\mathrm{T}} \mathbf{F} \mathbf{R}^l \tag{527}$$

In agreement with the response function for exact states in eq 278 the coupled-cluster linear response function has poles at $\omega_B = \pm \omega_k$. In addition, the coupled-cluster response function has an F-dependent term, not present in eq 278. This term vanishes in the limit of a full excitation manifold, as discussed in ref 125.

In coupled-cluster theory, transition strengths may be determined from the residue of the linear response function

$$\lim_{\omega_B \longrightarrow \omega_k} (\omega_B - \omega_k) \langle \langle A; B \rangle \rangle_{\omega_B} = \frac{1}{2} T^A_{0k} T^B_{k0} + \frac{1}{2} \left(T^B_{0k} T^A_{k0} \right)^*$$
(528)

where

$$T_{k0}^{A} = {}^{\mathrm{d}} \boldsymbol{\xi}_{k}^{A}, \quad T_{0k}^{A} = {}^{\mathrm{d}} \big(\boldsymbol{\eta}^{A} + \mathbf{F} \mathbf{t}^{A} (-\omega_{k}) \big)_{k}$$
(529)

Unlike for variational methods, $T_{0k}^A \neq (T_{k0}^A)^*$. However, as for exact states, the transition-strength matrix is Hermitian. The diagonal element of the transition-strength matrix $(1/2)T_{0k}^AT_{k0}^A + (1/2)(T_{0k}^AT_{k0}^A)^*$ is therefore real. However, since the transition strength matrix element is not a squared norm as in exact theory, it is not guaranteed to be positive. Again, this does not give any problems, in practice, when the coupled-cluster wave function is a good representation of the exact wave function.

4.3.6. Response Functions for the CCn Models. The response functions for the intermediate CCn models differ from those for the full coupled-cluster models discussed above. To illustrate this, we consider here the CC2 model, for which the time-dependent amplitude equations in eq 493 may be written as¹²¹

$$\langle \mu_1 | H^{T_1} + [H, T_2] | HF \rangle - i\dot{t}_{\mu_1} = 0$$
 (530)

$$\langle \mu_2 | H^{T_1} + [F + V^{T_1}(t), T_2] | HF \rangle - i\dot{t}_{\mu_2} = 0$$
 (531)

which are the time-dependent analogues of the CCSD singles equation in eq 481 and the CC2 doubles equation in eq 485,

respectively. The CC2 version of the quasi-energy Lagrangian in eq 491 may therefore be written in the form

$$\mathscr{S} = \operatorname{Re}\left\{ \langle \operatorname{HF} | H \exp(T_{1} + T_{2}) | \operatorname{HF} \rangle \right.$$
$$\left. + \sum_{\mu_{1}} \overline{t}_{\mu_{1}} \left(\langle \mu_{1} | H^{T_{1}} + [H, T_{2}] | \operatorname{HF} \rangle - i \dot{t}_{\mu_{1}} \right) \right.$$
$$\left. + \sum_{\mu_{2}} \overline{t}_{\mu_{2}} \left(\langle \mu_{2} | H^{T_{1}} + [F + V^{T_{1}}(t), T_{2}] | \operatorname{HF} \rangle - i \dot{t}_{\mu_{2}} \right) \right\}_{T}$$
(532)

The quasi-energy Lagrangian is stationary with respect to variations in the cluster amplitudes and multipliers. This condition may be used to determine an order expansion of the cluster amplitudes and multipliers, and the CC2 molecular response functions may subsequently be obtained by differentiating the quasi-energy Lagrangian as in eqs 307–309. The CC2 linear response function was first derived in ref 121, whereas the first derivation of the CC3 linear response function was given in ref 122.

4.3.7. Equation of Motion Coupled-Cluster Method. As an alternative to coupled-cluster response theory, molecular properties may be calculated using the equation of motion coupled-cluster (EOM-CC) method.^{126,127} In EOM-CC theory it is assumed that a coupled-cluster ground-state wave function has been determined that satisfies the cluster amplitude equations of eq 478 and has the energy $E_{\rm CC}$ of eq 479. However, unlike in response theory, no time-dependent wave function or equations are introduced. Instead, excitation energies and excited-state wave functions are determined by diagonalization of a shifted Hamiltonian matrix in a biorthonormal basis. Transition moments are subsequently determined directly as matrix elements between these states rather than as residues of response functions, and the response functions are obtained by inserting these matrix elements and excitation energies in the standard sum-over-state expressions for the response functions of an exact (variational) wave function as obtained in the quasi-energy formulation, see eqs 278-280. By comparing the EOM-CC and coupled-cluster response methods for truncated expansions we shall see that the two approaches give identical excitation energies for the standard coupled-cluster models, where the excitation operator is truncated at a given excitation level, whereas other molecular properties in the two approaches differ. For example, transition moments are only size intensive when calculated from the expressions obtained by response theory.

In Figure 2 the different strategies followed in EOM-CC and coupled-cluster response theories are summarized. In the EOM-CC method expressions for the response functions are first obtained using the time-dependent variation principle for the exact wave function. Subsequently, these expressions are used with truncated summations and approximate coupledcluster wave functions and excitation energies. By contrast, in response theory the wave function is approximated first and the time development of the approximate wave function is subsequently determined from the time-dependent variation principle.

To illustrate the differences between the two approaches, consider a CCSD calculation of a linear-response molecular property such as the frequency-dependent polarizability. In EOM-CC theory electronic states and excitation energies are obtained by solving the EOM-CCSD eigenvalue equation (with a



Figure 2. Comparison of EOM-CC and coupled-cluster response theory.

shifted Hamiltonian matrix). The resulting states and excitation energies are then inserted into the expression for the exact linear response function in eq 278. In CCSD response theory the linear response function is determined from eq 516. The main difference between eqs 278 and 516 is that the latter contains a term that is quadratic in the coupled-cluster amplitudes. This term is absent in eq 278, which was obtained using the fact that the exact wave function, unlike an approximate coupled-cluster wave function, is variational. The fact that EOM-CC theory combines equations for variational wave functions with nonvariational coupled-cluster theory thus leads to neglect of the quadratic term. The errors introduced by neglecting the quadratic term in eq 516 are most severe for the simplest coupledcluster methods, such as the CC2 model. In the limit of no truncation, EOM-CC theory and coupled-cluster response theory become identical.

In the coupled-cluster response formulation the response functions and their residues fulfill all relations that follow from the time-dependent variation principle, for example, the equivalence between various response functions and residues discussed for exact states in section 3.4.6 (see also ref 128). In particular, the response functions are size extensive and their residues are size intensive. In truncated EOM-CC theory such relations do not hold, in particular, the EOM-CC response functions are not size extensive¹²⁹ and their residues are not size intensive.¹²⁵ For example, EOM-CCSD theory gives the exact frequency-dependent polarizability for a two-electron system (the FCI result), while the polarizability for two noninteracting two-electron systems is not equal to the sum of the polarizabilities of the individual systems. Bartlett and co-workers^{130,131} proposed to obtain size-extensive EOM-CC models by complete elimination of unlinked terms in the EOM-CC equations. However, for the CCSD model the exact excitation energies (FCI result) are then not obtained for a two-electron system. These problems notwithstanding, calculations of different molecular properties for small molecules indicate that, in practice, the EOM-CC results are close to those obtained with response theory.

We are now ready to examine the EOM-CC model in more detail, limiting ourselves to excitation energies and transition moments. To obtain the EOM-CC excitation energies and states we write the eigenstate $|k\rangle$ of H_0 in the form

$$|k\rangle = s^{k}|CC^{(0)}\rangle + \sum_{\mu} R^{k}_{\mu}\tau_{\mu}|CC^{(0)}\rangle$$
(533)

The condition that $|k\rangle$ is an eigenstate of the shifted Hamiltonian $H_0 - E_{\rm CC}$ with excitation energy $\omega_k = E_k - E_{\rm CC}$ may then be expressed as

$$(H_0 - E_{\rm CC}) \left(s^k | {\rm CC}^{(0)} \rangle + \sum_{\mu} R^k_{\mu} \tau_{\mu} | {\rm CC}^{(0)} \rangle \right)$$
$$= \omega_k \left(s^k | {\rm CC}^{(0)} \rangle + \sum_{\mu} R^k_{\mu} \tau_{\mu} | {\rm CC}^{(0)} \rangle \right)$$
(534)

Projection of eq 534 from the left with the basis { $\langle HF|, \langle HF|$, $\langle HF|, \tau_{\mu}^{\dagger} \exp(-T^{(0)})$ }, which is biorthonormal to { $|CC^{(0)}\rangle, \tau_{\mu}|CC^{(0)}\rangle$ }, yields

$$\mathbf{H}^{\mathbf{b}}\begin{pmatrix}\mathbf{s}^{k}\\\mathbf{R}^{k}\end{pmatrix} = \omega_{k}\begin{pmatrix}\mathbf{s}^{k}\\\mathbf{R}^{k}\end{pmatrix}$$
(535)

where $\mathbf{H}^{\mathbf{b}}$ is the shifted Hamiltonian matrix

$$\mathbf{H}^{\mathrm{b}} = \begin{pmatrix} \mathbf{0} & -\boldsymbol{\eta} \\ \mathbf{0} & \mathbf{A} \end{pmatrix}$$
(536)

The vector $\boldsymbol{\eta}$ is defined in eq 508, **A** is the coupled-cluster Jacobian of eq 507, and the coupled-cluster energy and amplitude equations in eqs 478 and 479 have been used to identify the zero elements of the first column of the Hamiltonian. One solution to eq 535 is $(1,0)^{T}$ with zero eigenvalue, corresponding to the reference state $|CC^{(0)}\rangle$. The remaining solutions to eq 535 represent excited states, whose eigenvectors from eqs 535 and 536 satisfy the relations

$$s^k = \omega_k^{-1} \boldsymbol{\eta} \mathbf{R}^k \tag{537}$$

$$\mathbf{AR}^k = \omega_k \mathbf{R}^k \tag{538}$$

The vectors \mathbf{R}^k are thus the right eigenvectors of the Jacobian in eq 522, and the coefficient s^k may be obtained from this eigenvector using eq 537.

The left eigenvectors of \mathbf{H}^{b} , which are row vectors of the form $(L_{\mathrm{HF}}^{k}, L_{\mu}^{k})$, have the same eigenvalues ω_{k} as the right eigenvectors and are chosen to satisfy the biorthonormal normalization $\mathbf{L}^{k}\mathbf{R}^{k} = 1$. Specifically, the left eigenvector with zero eigenvalue fulfills the equation

$$\begin{pmatrix} \boldsymbol{L}_{\rm HF}^{0} & \boldsymbol{L}^{0} \end{pmatrix} \mathbf{H}^{\rm b} = \mathbf{0}$$
(539)

which, by comparison with eq 506, shows that $(L_{\text{HF}}^0, \mathbf{L}^0)$ corresponds to $(1, \mathbf{\bar{t}}^{(0)})$, which contains the zero-order multipliers. The remaining left eigenvectors have nonzero eigenvalues and are from eq 536 seen to have the form $(0, \mathbf{L}^k)$, where \mathbf{L}^k satisfies the condition

$$\mathbf{L}^{k}\mathbf{A} = \omega_{k}\mathbf{L}^{k} \tag{540}$$

and are therefore identical to the left eigenvectors of the Jacobian in eq 522. We conclude that the excitation energies and eigenvectors of EOM-CC and coupled-cluster response theories are identical.

The first column of the shifted Hamiltonian matrix in eq 536 vanishes since the coupled-cluster state satisfies the amplitude equations in eq 478. For the intermediate CCn models, the amplitudes satisfy modified equations such as the CC2 amplitude

equations in eqs 481 and 485; for these models the first column of the Hamiltonian therefore does not vanish. The first decision to be made in developing intermediate EOM-CC models is therefore whether to use a nonvanishing or vanishing first column. By contrast, in the response formulation the variation principle for the quasi-energy Lagrangian may be invoked to obtain the CCn excitation energies as described for the CC2 model in section 4.3.6. With this observation it is not difficult to construct various intermediate EOM-CC models, although we will not discuss this point further here.¹³²

We next consider EOM-CC response functions and transition moments, comparing these with their response-theory counterparts. In section 3.4 sum-over-states expressions for linear, quadratic, and cubic response functions of an exact state were given in eqs 278–280 with summations over the eigenstates of the zero-order Hamiltonian. In EOM-CC theory we obtain the expressions for the response functions by replacing these summations over exact states with summations over the biorthonormal eigenstates of the EOM-CC Hamiltonian in eqs 538 and 540 using $|CC^{(0)}\rangle$ and $\langle\Lambda^{(0)}|$ as the right and left reference states, respectively.

When molecular properties are evaluated in EOM-CC theory the exact response functions should be used where it is explicitly imposed that the response functions are real. Consider, for example, the transition-strength matrix elements, which in exact theory are given in eq 290 and in EOM-CC theory become

$$S_{AB}^{0k} = \langle \Lambda^{(0)} | A | k \rangle \langle k | B | CC^{(0)} \rangle$$
(541)

In EOM-CC theory the transition moments are given by

$$\langle k|B|CC^{(0)}\rangle = \sum_{\mu} L^{k}_{\mu} \langle \mu|\exp(-T^{(0)})B|CC^{(0)}\rangle$$
 (542)

$$\langle \Lambda^{(0)} | A | k \rangle = \sum_{\mu} \left(\langle \Lambda^{(0)} | A \tau_{\mu} | CC^{(0)} \rangle - \overline{t}_{\mu}^{(0)} \langle \Lambda^{(0)} | A | CC^{(0)} \rangle \right) R_{\mu}^{k}$$
(543)

A comparison of transition moments in EOM-CC and linearresponse theories shows that the right transition moments are identical in the two approaches, whereas the left transition moments differ unless the cluster operator contains the full excitation manifold.¹²⁵ The EOM-CC left transition moments may be written in the form

$$\langle \Lambda^{(0)} | A | k \rangle = \sum_{\mu} \left(\langle \Lambda^{(0)} | [A, \tau_{\mu}] | CC^{(0)} \rangle R^{k}_{\mu} + \langle \Lambda^{(0)} | \tau_{\mu} A | CC^{(0)} \rangle R^{k}_{\mu} - \overline{t}^{(0)}_{\mu} R^{k}_{\mu} \langle \Lambda^{(0)} | A | CC^{(0)} \rangle \right)$$
(544)

where the first term is identical to the first term in linear response theory. The last two terms become

$$\sum_{\mu} \left(\langle \Lambda^{(0)} | \tau_{\mu} A | CC^{(0)} \rangle R_{\mu}^{k} - \overline{t}_{\mu}^{(0)} R_{\mu}^{k} \langle \Lambda^{(0)} | A | CC^{(0)} \rangle \right)$$
$$= \sum_{\mu > \nu} \overline{t}_{\mu}^{(0)} R_{\nu}^{k} \langle \mu | \tau_{\nu} \exp(-T^{(0)}) A | CC^{(0)} \rangle$$
$$- \sum_{\mu} \overline{t}_{\mu}^{(0)} R_{\nu}^{k} \sum_{\nu} \overline{t}_{\nu}^{(0)} \langle \nu | \exp(-T^{(0)}) | CC^{(0)} \rangle$$
(545)

where only the first term is size intensive. Transition moments in EOM-CC theory are therefore not size intensive.

4.4. Orbital-Relaxed Response Functions

As discussed in section 4.3, the intermediate CCn models of the coupled-cluster response hierarchy are well suited to the study of frequency-dependent molecular properties; by contrast, the corresponding intermediate models of the coupled-cluster energy hierarchy, for example, the MP2 and CCSD(T) models, provide a poor description of such properties. However, with inclusion of orbital relaxation these models have become the method of choice for determining many static molecular properties at a lower computational cost than their counterparts in the coupled-cluster response hierarchy. In this section we describe the calculation of response properties with orbital relaxation for these perturbation-based intermediate models using the MP2 model as an example.

4.4.1. Orbital-Relaxed Response Functions. For a large class of wave function models the wave function is determined in two consecutive steps. First, a reference state is determined to obtain a qualitatively correct zero-order description of the electronic system; subsequently, dynamical correlation is recovered based on this zero-order description using a correlation method such as perturbation theory or coupled-cluster theory. For systems dominated by a single electronic configuration we may, for example, combine a Hartree—Fock zero-order description with MP2, CCSD, or CCSD(T) treatments of the correlation problem; for systems with several important configurations we may instead combine an MCSCF zero-order reference state with a CASPT2 treatment of the correlation. In both cases the MOs used for the correlation treatment are inherited from the zero-order wave function.

When molecular properties are calculated for such two-step models we must decide whether or not to incorporate the perturbation dependence of the MOs in the calculation of the response functions. In the discussion of the coupled-cluster response functions in section 4.3 the perturbation dependence of the MOs was not considered since orbital relaxation introduces singularities that destroy the pole structure of the response functions, making the calculated frequency-dependent properties unreliable. However, for static molecular properties relaxation of the MOs in the presence of the perturbation may be incorporated without introducing such artifacts; indeed, static molecular properties are typically (but not invariably) calculated from orbital-relaxed response functions. Determination of static molecular properties then becomes equivalent to performing finite-difference energy calculations in the presence of the perturbation.

For static molecular properties, where orbital relaxation is accounted for, the time-averaged quasi-energy Lagrangian in section 3.1.3 reduces to the (time-independent) energy Lagrangian for the two-step model. The orbital-relaxed energy Lagrangian may be expanded in the perturbation strengths as

$$L = E_{0} + \sum_{B_{0}} \langle (V_{B_{0}}) \rangle \varepsilon_{B_{0}} + \frac{1}{2} \sum_{B_{0}, B_{1}} \langle (V_{B_{0}}; V_{B_{1}}) \rangle \varepsilon_{B_{0}} \varepsilon_{B_{1}} + \frac{1}{6} \sum_{B_{0}, B_{1}, B_{2}} \langle (V_{B_{0}}; V_{B_{1}}, V_{B_{2}}) \rangle \varepsilon_{B_{0}} \varepsilon_{B_{1}} \varepsilon_{B_{2}} + \dots$$
(546)

where E_0 is the energy of the two-step model in the absence of the perturbation. No frequency subscript is attached to the response functions since these are always determined at zero frequency, and the notation $\langle (V_{B0}; V_{B1}) \rangle$ rather than $\langle \langle V_{B0}; V_{B1} \rangle \rangle$ indicates that orbital relaxation of the reference state is included in evaluation of the response function. The orbital-relaxed response functions are thus obtained by simple differentiation of the energy Lagrangian with respect to the relevant perturbation strengths. In section 4.4.2 we describe how orbital-relaxed response functions may be obtained for the MP2 model.

Orbital-relaxed response functions are used for many static molecular properties, for example, for molecular gradients and Hessians (which should be identical to the corresponding finitedifference quantities). A particular difficulty associated with these calculations is the need to take into account the effect of perturbation-dependent basis sets, as described in section 5.1 for geometrical derivatives. Also, when magnetic molecular properties associated with an external magnetic field are evaluated using London orbitals, for example, nuclear shielding constants, orbital-relaxed response functions have to be considered as described in section 5.2. It should be noted that many static molecular properties are best calculated without orbital relaxation. For example, in the calculation of indirect nuclear spin-spin coupling constants the inclusion of orbital relaxation introduces the effects of Hartree–Fock triplet instabilities into the CCSD(T) calculations and should therefore be avoided, ¹³³ as discussed in section 5.2.2.

4.4.2. Orbital-Relaxed MP2 Response Functions. In Møller–Plesset perturbation theory the Hamiltonian H_0 is partitioned as in eq 484. The Hartree–Fock wave function is the zero-order eigenfunction of the Fock operator (written here in the canonical representation)

$$F|\mathrm{HF}\rangle = \sum_{i} \varepsilon_{i}|\mathrm{HF}\rangle$$
 (547)

where ε_i are the energies of the occupied canonical orbitals. Applying standard perturbation theory we obtain, to first and second orders in the fluctuation potential, respectively, the Hartree–Fock energy

$$E_{\rm HF} = \langle \rm HF|F|\rm HF\rangle + \langle \rm HF|U|\rm HF\rangle + h_{\rm nuc} = \sum_{i} \varepsilon_{i} + \langle \rm HF|U|\rm HF\rangle + h_{\rm nuc}$$
(548)

and the second-order Møller–Plesset (MP2) correlation energy (given here for a closed-shell system)³⁰

$$E_{\rm MP2} = E_{\rm HF} - \sum_{aibj} \frac{(2g_{aibj} - g_{ajbi})g_{aibj}}{\varepsilon_a - \varepsilon_i + \varepsilon_b - \varepsilon_j}$$
(549)

where indices ij and a,b denote occupied and virtual orbitals, respectively. The MP2 model usually recovers a large part of the dynamical correlation energy, providing a useful correction to the Hartree–Fock energy at a cost significantly lower than that of CCSD theory (noniterative n^5 rather than iterative n^6). In this sense MP2 theory represents a highly successful approach to calculation of molecular correlation energies, providing, for many purposes, accurate, size-extensive energy corrections. Higher order energy corrections, such as those provided by third-order Møller–Plesset (MP3) theory and fourth-order Møller–Plesset (MP4) theory, may be derived but are considerably more expensive than the MP2 correction and do not always provide a smooth convergence toward the FCI energy, often oscillating or diverging.^{134,135}

For evaluation of the MP2 energy and its derivatives it is convenient to reformulate the theory somewhat. Thus, we write the second-order energy in the form

$$E_{\rm MP2} = \langle \rm HF | [U, T_2] | \rm HF \rangle \tag{550}$$

where T_2 is the double excitation operator in eq 475, whose amplitudes satisfy the equations

$$\sum_{\nu} \langle \mu | [F, \tau_{\nu}] | HF \rangle t_{\nu} = - \langle \mu | U | HF \rangle$$
(551)

where the Fock operator may be written in the form

$$F = \frac{1}{2} \sum_{pq, \sigma} \langle \mathrm{HF} | [a_{q\sigma}^{\dagger}, [a_{p\sigma}, H_0]]_+ | \mathrm{HF} \rangle E_{pq}$$
(552)

We do not insist on the canonical representation since degeneracies among the occupied or virtual orbitals may lead to instabilities in the derivatives.¹³⁶ We therefore only require that the Hartree–Fock state satisfies the Brillouin condition

$$\langle \mathrm{HF}|[E_{ai}, H_0]|\mathrm{HF}\rangle = 0 \tag{553}$$

which is a necessary and sufficient condition for stationarity of the Hartree–Fock energy.

We now introduce a static perturbation V such that the total Hamiltonian can be written as

$$H = H_0 + V = F + U + V$$
(554)

The perturbed Hartree–Fock wave function $|\text{HF}\rangle$, the perturbed excited state $|\tilde{\mu}\rangle$, and the perturbed creation operators $\tilde{a}_{p\sigma}^{\dagger}$ may then be expressed as¹³⁶

$$|\widetilde{HF}\rangle = \exp(-\kappa)|HF\rangle$$
 (555)

$$|\tilde{\mu}\rangle = \exp(-\kappa)|\mu\rangle$$
 (556)

$$\tilde{a}_{p\sigma}^{\dagger} = \exp(-\kappa) a_{p\sigma}^{\dagger} \exp(\kappa)$$
(557)

where κ is an anti-Hermitian orbital-rotation operator

$$\kappa = \sum_{ai} \kappa_{ai} (E_{ai} - E_{ia}) \tag{558}$$

where the excitation operators are given in eq 100, with summation over all pairs of occupied and virtual orbitals.

We may next construct the Møller–Plesset Lagrangian $L_{\rm MP2}$ by introducing the perturbed states and operators in eqs 555–557 into the energy expression in eq 550 and the constraints in eqs 551 and 553 with associated Lagrange multipliers \bar{t} and $\bar{\kappa}$, respectively, yielding

$$L_{MP2} = \langle HF | [U_{\kappa}, T_{2}] | HF \rangle + \langle \overline{t} | U_{\kappa} | HF \rangle + \langle \overline{t} | [F \{ H_{\kappa} \}, T_{2}] | HF \rangle + \sum_{ai} \overline{\kappa}_{ai} \langle HF | [E_{ai}, H_{\kappa}] | HF \rangle$$
(559)

where we introduced the notation (for a general operator *A*)

$$A_{\kappa} = \exp(\kappa)A \exp(-\kappa)$$
(560)

$$F\{A\} = \frac{1}{2} \sum_{pq, \sigma} \langle \mathrm{HF} | [a_{q\sigma}^{\dagger}, [a_{p\sigma}, A]]_{+} | \mathrm{HF} \rangle E_{pq}$$
(561)

$$\langle \overline{t} | = \sum_{\mu} \overline{t}_{\mu} \langle \mu | \tag{562}$$

We note that the orbitals are allowed to relax in the presence of the perturbation, orbital relaxation being described by the κ -transformed Hamiltonian H_{κ}^{136} From this variational Lagrangian the Møller–Plesset molecular properties are obtained in the usual manner, in accordance with the 2n + 1 and 2n + 2 rules

$$L_{MP2}^{(0)} = \langle HF | [U^{(0)}, T_2^{(0)}] | HF \rangle$$

$$L_{MP2}^{(1)} = \langle HF | \left[U_{\kappa}^{(1)}, T_2^{(0)} \right] | HF \rangle + \langle \overline{t}^{(0)} | U_{\kappa}^{(1)} | HF \rangle$$

$$+ \langle \overline{t}^{(0)} | \left[F \{ H^{(1)} \}, T_2^{(0)} \right] | HF \rangle$$

$$+ \langle HF | \left[\overline{\kappa}^{(0)}, H^{(1)} \right] | HF \rangle$$

$$L_{MP2}^{(2)} = \langle HF | \left[U_{\kappa}^{(2)}, T_2^{(0)} \right] | HF \rangle + \langle \overline{t}^{(0)} | U_{\kappa}^{(2)} | HF \rangle$$
(564)

$$+ \langle \overline{t}^{(0)} | \left[F\{H^{(2)}\}, T_{2}^{(0)} \right] | \text{HF} \rangle \\+ \langle \text{HF} | \left[\overline{\kappa}^{(0)}, H^{(2)} \right] | \text{HF} \rangle \\+ \langle \text{HF} | \left[U_{\kappa}^{(1)}, T_{2}^{(1)} \right] | \text{HF} \rangle \\+ \langle \overline{t}^{(0)} | \left[F\{H^{(1)}\}, T_{2}^{(1)} \right] | \text{HF} \rangle$$
(565)

The zero-order equations for the amplitudes and Lagrange multipliers are given by

$$\sum_{\nu} \langle \mu | [F, \tau_{\nu}] | \mathrm{HF} \rangle t_{\nu}^{(0)} = - \langle \mu | U^{(0)} | \mathrm{HF} \rangle$$
(566)

$$\sum_{\nu} \overline{t}_{\nu}^{(0)} \langle \nu | [f, \tau_{\mu}] | \mathrm{HF} \rangle = - \langle \mathrm{HF} | [U^{(0)}, \tau_{\mu}] | \mathrm{HF} \rangle$$
(567)

$$\sum_{j} \langle \mathrm{HF} | \left[\kappa_{i}, \left[\kappa_{j}, H^{(0)} \right] \right] | \mathrm{HF} \rangle \overline{\kappa}_{j}^{(0)}$$

$$= - \langle \mathrm{HF} | \left[\left[\kappa_{i}, U^{(0)} \right], T_{2}^{(0)} \right] | \mathrm{HF} \rangle - \langle \overline{t}^{(0)} | \left[\kappa_{i}, U^{(0)} \right] | \mathrm{HF} \rangle$$

$$- \langle \overline{t}^{(0)} | \left[F \left\{ \left[\kappa_{i}, H^{(0)} \right] \right\}, T_{2}^{(0)} \right] | \mathrm{HF} \rangle$$
(568)

whereas the first-order equations for the orbitals and amplitudes become

$$\sum_{j} \langle \mathrm{HF} | \left[\kappa_{i}, \left[\kappa_{j}, H^{(0)} \right] \right] | \mathrm{HF} \rangle \kappa_{j}^{(1)} = - \langle \mathrm{HF} | \left[\kappa_{i}, H^{(1)} \right] | \mathrm{HF} \rangle$$
(569)

$$\sum_{\nu} \langle \mu | [F, \tau_{\nu}] | \mathrm{HF} \rangle t_{\nu}^{(1)}$$
$$= - \langle \mu | U_{\kappa}^{(1)} | \mathrm{HF} \rangle - \langle \mu | \Big[F\{H^{(1)}\}, T_{2}^{(0)}] | \mathrm{HF} \rangle \qquad (570)$$

By substituting the solutions to the linear equations in eqs 566–570 into the energy expressions in eqs 563–565 we obtain MP2 static molecular properties up to second order.

4.5. Response Functions with Perturbation-Dependent Basis Sets

When molecular properties that depend on nuclear distortions or external magnetic fields are calculated, perturbationdependent basis sets are typically used, as discussed in sections 2.4 and 4.4.1. In the present section we discuss the evaluation of response functions for such properties, the most important of which are molecular gradients and Hessians (geometrical -

distortions) and nuclear shielding constants (magnetic field variations).

For a general perturbation parameter ε (which may represent a nuclear distortion or an applied magnetic field) we expand the energy in orders of the perturbation

$$E(\varepsilon) = E^{(0)} + E^{(1)}\varepsilon + \frac{1}{2}E^{(2)}\varepsilon^{2} + \dots$$
 (571)

and are particularly interested in the first- and second-order properties $E^{(1)}$ and $E^{(2)}$. The Hamiltonian is expanded in a similar manner

$$H(\varepsilon) = H_0 + H^{(1)}\varepsilon + \frac{1}{2}H^{(2)}\varepsilon^2 + \dots$$
 (572)

An important difference from the theory discussed before is the appearance of second-order and higher order perturbation operators, all of zero frequency. The appearance of nonlinear terms in the perturbation operator is a generalization of the operator considered in section 3.2, where $V(0) = 2V_0$ of eq 199 contains only a linear term. Expressing the energy in terms of a Lagrangian we then obtain the following expressions for the first-and second-order molecular properties

$$E^{(1)} = \frac{\mathrm{d}L}{\mathrm{d}\varepsilon} = \langle (H^{(1)}) \rangle \tag{573}$$

$$E^{(2)} = \frac{d^2 L}{d\epsilon^2} = \langle (H^{(2)}) \rangle + \langle (H^{(1)}; H^{(1)}) \rangle$$
(574)

where we used the notation for orbital-relaxed response functions introduced in eq 546, L is the variational Lagrangian, and the linear response function has been evaluated at zero frequency. These expressions are evaluated from the first- and second-order perturbed Hamiltonians

$$H^{(1)} = \sum_{pq} \tilde{h}_{pq}^{(1)} E_{pq} + \frac{1}{2} \sum_{pqrs} \tilde{g}_{pqrs}^{(1)} (E_{pq} E_{rs} - \delta_{qr} E_{ps})$$
(575)

$$H^{(2)} = \sum_{pq} \tilde{h}_{pq}^{(2)} E_{pq} + \frac{1}{2} \sum_{pqrs} \tilde{g}_{pqrs}^{(2)} (E_{pq} E_{rs} - \delta_{qr} E_{ps})$$
(576)

for which explicit expressions are given in section 2.4.4 for the first- and second-order OMO integrals expressed in terms of oneindex transformations. Here, we assumed singlet perturbations, which is sufficient for evaluation of molecular gradients and Hessians and also nuclear shielding constants of closed-shell systems. More generally, the first- and second-order operators may also include triplet excitation operators, see eqs 102–108.

Taking the expectation value of the operator in eq 575 we obtain for the first-order property

$$E^{(1)} = \sum_{pq} D_{pq} \tilde{h}_{pq}^{(1)} + \frac{1}{2} \sum_{pqrs} d_{pqrs} \tilde{g}_{pqrs}^{(1)} + h_{nuc}^{(1)}$$
(577)

where we introduced the one- and two-electron density matrices with elements

$$D_{pq} = \langle (E_{pq}) \rangle \tag{578}$$

$$\mathbf{d}_{pqrs} = \langle (E_{pq}E_{rs} - \delta_{qr}E_{ps}) \rangle \tag{579}$$

In the simplest cases the density matrices are expectation values of the wave functions, for example, in Hartree–Fock, FCI, and MCSCF theories. For nonvariational methods such as coupled-cluster theory, the density matrices are more complicated, involving contributions from the Lagrange multipliers of the Lagrangian. Inserting the expression for the first-derivative one-electron integrals in eq 134 and the corresponding expression for the two-electron integrals we obtain for all wave function models

$$E^{(1)} = \sum_{pq} D_{pq} h_{pq}^{(1)} + \frac{1}{2} \sum_{pqrs} d_{pqrs} g_{pqrs}^{(1)} - \sum_{pq} F_{pq} S_{pq}^{(1)} + h_{nuc}^{(1)}$$
(580)

where $S^{(1)}$ is the derivative of the overlap matrix, and we introduced the generalized Fock matrix

$$F_{pq} = \sum_{o} D_{po} h_{qo} + \sum_{ors} d_{pors} g_{qors}$$
(581)

For the important special case of molecular gradients the term involving the derivative overlap matrix and the generalized Fock matrix is known as the Pulay force. By re-expressing eq 580 in terms of AO integrals the gradient can be efficiently evaluated from derivative integrals in the AO basis, multiplying these integrals with the density matrices, avoiding storage of the large number of derivative integrals (for example, the Hartree—Fock molecular gradient in the AO basis is given by eq 386 in the time-independent limit).

The second-order property in eq 574 consists of two parts. The expectation value is calculated in the same manner as the molecular gradient in eq 577, replacing first derivatives by second derivatives

$$\langle (H^{(2)}) \rangle = \sum_{pq} D_{pq} \tilde{h}_{pq}^{(2)} + \frac{1}{2} \sum_{pqrs} d_{pqrs} \tilde{g}_{pqrs}^{(2)} + h_{nuc}^{(2)}$$
(582)

The resulting expression is more complicated than that of the firstorder property but can be straightforwardly and efficiently evaluated in the same manner. Evaluation of the linear-response part of the second-order property in eq 574 follows the general approach for a frequency-independent perturbation, keeping in mind that $H^{(1)}$ in eq 575 contains two-electron as well as one-electron contributions.

For a concrete example we consider the evaluation of the linear-response contribution in Hartree–Fock theory following the MCSCF formalism developed in section 4.2. Setting the frequency equal to zero in the response equations in eq 466 we obtain $\mathbf{E}^{[2]}\boldsymbol{\beta} = i\mathbf{V}^{[1]}$, where $\mathbf{E}^{[2]}$ is given in eq 453, $\boldsymbol{\beta}$ in eq 444, and $\mathbf{V}^{[1]}$ in eq 449, yielding

$$\begin{pmatrix} \langle \mathrm{HF}|[\mathbf{q}, [H_0, \mathbf{q}^{\dagger}]]|\mathrm{HF}\rangle & \langle \mathrm{HF}|[\mathbf{q}, [H_0, \mathbf{q}]]|\mathrm{HF}\rangle \\ \langle \mathrm{HF}|[\mathbf{q}^{\dagger}, [H_0, \mathbf{q}^{\dagger}]]|\mathrm{HF}\rangle & \langle \mathrm{HF}|[\mathbf{q}^{\dagger}, [H_0, \mathbf{q}]]|\mathrm{HF}\rangle \end{pmatrix} \begin{pmatrix} \kappa \\ \kappa^* \end{pmatrix} \\ = \mathrm{i} \begin{pmatrix} \langle \mathrm{HF}|[\mathbf{q}, H^{(1)}]|\mathrm{HF}\rangle \\ \langle \mathrm{HF}|[\mathbf{q}^{\dagger}, H^{(1)}]|\mathrm{HF}\rangle \end{pmatrix}$$
(583)

where we assumed real matrices A and B. Expanding the two components and assuming a real perturbation we obtain the following response equations for the real and imaginary components of κ

$$\langle \mathrm{HF} | [\mathbf{q}^{\dagger} - \mathbf{q}, [\mathbf{q}^{\dagger} - \mathbf{q}, H_0]] | \mathrm{HF} \rangle \mathrm{Im} \kappa$$
$$= - \langle \mathrm{HF} | [\mathbf{q}^{\dagger} - \mathbf{q}, \mathrm{Re} H^{(1)}] \mathrm{HF} \rangle$$
(584)

$$\langle \mathrm{HF} | [\mathbf{q}^{\dagger} + \mathbf{q}, [\mathbf{q}^{\dagger} + \mathbf{q}, H_0]] | \mathrm{HF} \rangle \mathrm{Re} \kappa$$

$$= - \langle \mathrm{HF} | [\mathbf{q}^{\dagger} + \mathbf{q}, \mathrm{Im} H^{(1)}] \mathrm{HF} \rangle$$
(585)

Taking into account spin symmetry, these equations may be reduced further, as discussed in ref 6.

4.6. Overview of Developments and Implementations

Having presented a theoretical framework for Hartree–Fock, MCSCF, coupled-cluster, and Møller–Plesset response theory we now review the extensive literature on this subject with emphasis on recent developments and implementations. First, we discuss in section 4.6.1 evaluation of expectation values and first-order properties. Section 4.6.2 is devoted to Hartree–Fock and MCSCF theories, while developments in coupled-cluster theory are described in section 4.6.3, and section 4.6.4 contains a brief discussion of explicitly correlated methods. In section 4.6.5 the second-order polarization-propagator approximation (SOPPA) is discussed, followed by the algebraic-diagrammatic-construction (ADC) method in section 4.6.6. Finally, section 4.6.7 contains a brief discussion of relativistic corrections.

4.6.1. Expectation Values and First-Order Properties. First-order molecular properties represent the first-order change in the electronic energy upon a perturbation of the system and therefore correspond to the first derivative of the energy with respect to the perturbation strength. For variational electronic-structure models the first derivative of the electronic energy is equal to the expectation value of the operator associated with the perturbation of the system, in accordance with the Hellmann–Feynman theorem, ^{68,69} see eq 177.

By contrast, nonvariational electronic-structure models do not satisfy the conditions of the Hellmann–Feynman theorem, as discussed in section 3.1.3. For such models, therefore, a molecular property calculated as an energy derivative and as a standard expectation value differ. The first-order property is then usually calculated as an energy derivative from a Lagrangian,⁷⁷ where it becomes a generalized expectation value (see eq 198), which, in coupled-cluster theory, for example, includes a contribution from the relaxation of the orbitals to the perturbation.

As an alternative to the Lagrangian method, Korona and Jeziorski presented in 2006 an approach for calculating oneelectron density matrices from the explicitly connected commutator expansion of the expectation value at the expectation-value CCSD (XCCSD) level of theory.¹³⁷ In their approach the density matrix is obtained at little additional cost beyond calculation of the CCSD energy, making it less demanding than Lagrangian-based CCSD theory by avoiding the multiplier equations, without compromising the quality of the calculated one-electron molecular properties, although we note that the resulting properties are not equivalent to finite-difference results.

Analytic first-order property schemes have a long history, which cannot be covered exhaustively here. Often they are obtained as a byproduct of analytic molecular-gradient implementations by replacing the differentiated Hamiltonian integrals with the integrals for the perturbation operator. Finite-field methods have also been extensively used to compute first-order properties, in particular, for molecular multipole moments, see, for example, ref 138. Thus far we restricted ourselves to considering first-order properties for the electronic ground state, but we note that these properties can also be computed for electronically excited states, as will be discussed in section 5.8. During the past few years attention has been directed toward reducing the computational scaling of ab initio methods, adapting them to larger molecular systems. As an example of such developments affecting the analytic calculation of first-order properties, Friedrich et al.¹³⁹ extended in 2009 their automated implementation of the incremental scheme for CCSD energies to the analytic computation of molecular (unrelaxed) first-order oneelectron properties, testing the convergence and accuracy of the incremental scheme for the dipole and quadrupole moments of a variety of chemically interesting systems. An analysis of the influence of local approximations in CCSD theory on electric dipole moments (and static dipole polarizabilities) was presented a few years earlier by Korona et al.¹⁴⁰ utilizing a finite-field approach.

4.6.2. Hartree-Fock and MCSCF Response Theory. Development of molecular response functions for a Hartree-Fock SCF state has a long history. Time-dependent Hartree-Fock equations were first derived by Dirac in 1930.¹⁴¹ Derivations and applications in the context of determining atomic and molecular properties were presented by McLachlan and Ball,¹⁴² Dalgarno and Victor,¹⁴³ and Dunning and McKoy¹⁴⁴ without explicitly setting up response functions. In this review we concentrate on modern developments, where the time evolution of the approximate state is used to set up response functions, from which molecular properties are determined. For Hartree-Fock theory this development was initiated by Dalgaard,¹⁴⁵ who determined the linear and quadratic response functions for Hartree-Fock states. The more general framework for carrying out response theory at the Hartree-Fock and MCSCF levels of theory was formulated by Olsen and Jørgensen in 1985,⁷⁰ who developed tractable expressions for Hartree-Fock and MCSCF linear and quadratic response functions, their poles and residues. Subsequently, efficient implementations of linear, 98 quadratic, 99 and cubic 100,101 response functions were presented.

In the formulation by Olsen and Jørgensen, response functions were obtained by applying the Ehrenfest theorem to determine the time evolution of an expectation value for the Hartree—Fock and MCSCF states. An alternative approach is the quasi-energy formalism, defining response functions as derivatives of the quasi-energy, as done in our discussion of response functions in Hartree—Fock (section 4.1) and MCSCF (section 4.2) theories. An advantage of this approach is that it allows computational expressions for frequency-dependent properties to be obtained by differentiation, thereby facilitating derivation of the working equations for computer implementations. This approach became popular with the work of Aiga, Sasagane, and Itoh^{64,65} and Christiansen, Hättig, and Jørgensen.⁶⁶ Frequency-dependent polarizabilities obtained as quasi-energy derivatives had been considered earlier by Rice and Handy.¹⁴⁶

The first open-ended approach for higher order molecular properties was that of Dykstra and Jasien¹⁴⁷ based on energy-derivative theory for static (time-independent) perturbations. Shortly thereafter Sekino and Bartlett presented an open-ended analytic approach for frequency-dependent Hartree–Fock (hyper)polarizabilities¹⁴⁸ based on Frenkel's variation principle for the Schrödinger equation.⁶⁷ A similar scheme for time-dependent Hartree–Fock theory was later presented by Karna and Dupuis.¹⁴⁹

Hartree—Fock response methods have continued to evolve in recent years, in particular, toward reducing the scaling of the computational cost, ideally to become of linear complexity. These developments have often been connected to those for Kohn—Sham theory, which is not discussed in this review. An essential step has been to abandon the MO basis, working instead

directly in the AO basis and in terms of the AO density matrix.^{13,14,102–106} The approach by Thorvaldsen et al.¹⁰⁶ is described in section 4.1 and is particularly convenient for calculations with perturbation-dependent basis sets, which are an integral part of this formulation. A historical account of response theory and quasi-energy approaches to the calculation of frequency-dependent magnetic properties is found in ref 150.

4.6.3. Coupled-Cluster Response Theory. Coupled-cluster response theory was first developed by Monkhorst,¹⁵¹ Dalgaard and Monkhorst,¹⁵² and Mukherjee and Mukherjee¹⁵³ in the late 1970s. However, development of coupled-cluster theory into the most accurate black-box approach for calculation of molecular properties took place in the 1990s, initiated by the derivation of the linear and quadratic response functions and their residues for the standard coupled-cluster models where the cluster operator is truncated at a given excitation level.¹⁵⁴ Response functions were obtained by examining the time evolution of a generalized coupled-cluster averaged value $\langle \Lambda | A | CC \rangle$, where the coupled-cluster state and the associated lambda state are required to satisfy the time-dependent Schrödinger equation. In ref 154 it was not imposed that $\langle \Lambda | A | CC \rangle$ is real. This requirement was later imposed by Pedersen and Koch,¹⁵⁵ thereby ensuring that the resulting response functions satisfy the symmetry relations of section 3.4.4 also for complex Hamiltonians. The quasi-energy Lagrangian approach was subsequently developed^{64,66,121} following the introduction a few years earlier of the Lagrangian technique.^{77,156} The quasi-energy Lagrangian approach simplified the derivation of the response functions, in particular, for intermediate coupled-cluster models such as the CC2 and CC3 models.

The first CCSD linear response implementation of excitation energies was reported in ref 157. Implementations of the CCSD linear,^{129,158,159} quadratic,¹⁶⁰ and cubic⁷⁵ response functions were subsequently presented. Introduction of the intermediate CCn hierarchy for frequency-dependent properties was an important development in the 1990s, including the $CC2^{121}$ and $CC3^{122,123}$ models as approximate CCSD and CCSDT models, respectively, see the discussion in section 4.3. Implementations of the CC3 model were first reported by Christiansen et al. for the linear response function,¹⁶¹ by Gauss et al. for the quadratic response function,¹⁶³ More recently, Kállay and co-workers derived and implemented linear¹⁶⁴ and quadratic¹⁶⁵ response functions for general coupled-cluster models with arbitrary excitation levels in the cluster operator.

Whereas the CC3 and higher order coupled-cluster models are still mostly used for small systems and benchmarking, the CC2 model, especially in combination with Cholesky decomposition^{166,167} and resolution of the identity^{168–170} techniques for the two-electron integrals, constitutes the only practical alternative to Kohn–Sham theory for computing response properties of larger systems (mainly excitation energies and other linear-response properties),^{171–177} although ongoing work on linear-scaling formulations of coupled-cluster theory appear promising with respect to future extensions to more accurate coupled-cluster models.

Very recently, Korona¹⁷⁸ proposed a noniterative correction to the (X)CCSD polarization propagator/linear response function, correct to third order in Møller–Plesset theory and of only n^6 complexity (compared with n^7 for the CC3 model). The same author also presented a variant of the CC2 linear response approach,¹⁷⁹ denoted XCC2, where the time-independent coupled-cluster polarization propagator of Moszynski et al.¹⁸⁰ is combined with CC2 excitation operators. For molecular properties the XCC2 model appears to be a practical alternative to the CC2 model.

Within the framework of coupled-cluster theory, molecular properties may also be calculated using the EOM-CC model discussed in section 4.3.7, developed primarily by Bartlett and co-workers, see refs 181 and 182, and the contribution from other authors to the present volume, for recent reviews. For the standard coupled-cluster hierarchy, EOM-CC theory yields the same excitation energies as those obtained with linear response theory, but there are differences for other properties such as the lack of size intensitivity of EOM-CC transition moments discussed in section 4.3.7.^{125,129}

Extensions of the EOM-CC model to higher order response properties have been proposed by Rozyczko and Bartlett.^{183,184} Importantly, the EOM-CC model provides an excellent framework for calculation of ionization potentials and electron-attachment energies, see, for example, refs 181, 185, and 186. Approximate triples models for excitation energies were introduced by Watts and Bartlett based on the CCSDT-1a model.^{187,188} The CCSDT-1a calculations showed that excitation energies dominated by a double excitation were significantly improved relative to the CCSD results, while excitation energies dominated by a single excitation did not show any improvement. Christiansen et al.¹⁸⁹ demonstrated that when the dominant triples contribution in the triples equation is considered, and importantly, no approximations are made in the singles and doubles amplitude equations; then excitation energies dominated by a single excitation are correct to third order in Møller-Plesset perturbation theory and are thus improved compared to the CCSD excitations (which are correct to second order). The CCSDT-1a model and most other proposed approximate triples models^{190,191} introduce approximations in the singles and doubles equations and therefore do not improve upon excitations dominated by a single excitation. By contrast, improvements are observed for the CC3^{192,193} and CCSDT-3¹⁹⁴ models, where no approximations are made in the singles and doubles equations.

Most of the development of coupled-cluster response theory has been concerned with the properties of singlet states, although extensions to properties of states of other spins have been made. The EOM-CC method in the spin-orbital basis as presented by Stanton and Bartlett allows evaluation of properties of states with general spin¹²⁷ within the limits of this approach. Hald et al.¹⁹⁵ presented an extension of the CC2 method to triplet excitation energies of closed-shell molecules using an integral-direct approach and explicitly spin-coupled triplet excitations in the orbital basis and with an implementation which has roughly the same operation count as for singlet excitations. The approach was subsequently extended to the CCSD¹⁹⁶ and CC3^{197,198} models. Several important implementations of the coupledcluster hierarchies to open-shell references have appeared over the years, and we refer to the review by Stanton and Gauss¹⁹⁹ for an account of earlier contributions on this subject up to 2003. In addition (limiting ourself to some of the most recent work) we mention here the spin-flip EOM-CC approach of Krylov and coworkers,^{181,200-203*} and the open-shell variant of the CC3 method by Crawford and co-workers.²⁰⁴

Concerning local coupled-cluster response methods we mention here the work of Russ and Crawford,²⁰⁵ which extends the local coupled-cluster approach of Pulay and Saebø^{206,207} to dipole polarizabilities²⁰⁸ and the optical rotation tensor,²⁰⁸ and the already mentioned analysis of Korona et al.¹⁴⁰ We also note the local multistate CC2 response method of Schütz and coworkers for calculating excitation energies and first-order properties of excited singlet^{209–212} and triplet²¹³ states in extended molecular systems.

4.6.4. Explicit Correlation. In recent years, explicitly correlated electronic-structure theory has established itself as an efficient and accurate alternative to the traditional treatment of correlation energies in quantum chemistry. In particular, explicitly correlated techniques are now being applied not just to calculation of electronic energies but also to calculation of molecular properties. The first analytic implementation of the calculation of first-order properties was presented in 2005 by Kordel et al. at the MP2-R12 level of theory²¹⁴ using an auxiliary basis for the resolution of the identity approximation with and without a frozen core. Two years later their approach was generalized to calculation of the full geometrical gradient.²¹⁵ Recently, Höfener et al. presented an analytical scheme for firstorder properties in MP2-F12 theory.²¹⁶ Smooth and rapid convergence toward the basis-set limit was observed for the dipole moments of small closed- and open-shell molecules in augmented correlation-consistent polarized-valence basis sets optimized for MP2-F12 theory. Finally, regarding response theory for explicitly correlated methods we refer to refs 217-219, noting here its recent extension to higher order response functions by Hanauer and Köhn.²²⁰

4.6.5. Second-Order Polarization-Propagator Approximation. A well-established framework for calculation of molecular response properties is the second-order polarization-propagator approximation (SOPPA).^{221–223} The SOPPA linear response function (the polarization propagator) was proposed as an extension to Hartree-Fock theory with the response function and its pole structure correct to second order in perturbation theory. It may be derived using a superoperator formalism;²²¹ alternatively, it may be obtained by an exponential parametrization of the time evolution, consisting of products of exponentials for orbital rotations and higher order excitations, truncated such that the response function and its poles are correct to second order in Møller-Plesset perturbation theory.²²³ The second strategy is particularly convenient for extension of SOPPA to quadratic and higher order response functions, making calculation of molecular properties such as hyperpolarizabilities, TPA cross sections, and excited-state properties possible within the SOPPA model, see ref 223 for details.

A few variants of the SOPPA approach have been proposed over the years, like the CCSDPPA (coupled-cluster polarizationpropagator approximation with single and double excitations) of Geertsen and Oddershede^{224,225} and the more recent SOPPA-(CCSD)^{226,227} and SOPPA(CC2)²²⁸ models, which have the same excitation orders as the SOPPA model but employ CCSD or CC2 amplitudes instead of the Møller-Plesset correlation coefficients. The idea is to retain as much as possible the n^{5} computational scaling of the SOPPA model in the propagator calculation (even though generation of the CCSD amplitudes in SOPPA(CCSD) theory still scales as n°). The performance of both the traditional SOPPA approach and of its coupled-clustermodified variants for calculation of various properties, including excitation energies, dipole oscillator strengths, shielding constants, indirect nuclear spin-spin coupling constants, C₆ dispersion coefficients, and rotational g tensors, has been the subject of several benchmark studies during the last 5 years.²²⁹⁻²³²

4.6.6. Algebraic-Diagrammatic-Construction Method. Another popular polarization-propagator method is the algebraic-diagrammatic-construction (ADC) approach of Schirmer.²³³ The second-order ADC(2) model^{233–237} allows for a theoretical description of single and double excitations consistently to second and first order, respectively, in perturbation theory. The computational scheme is essentially an eigenvalue problem of a Hermitian secular matrix defined with respect to the space of singly and doubly excited configurations. The configuration space is smaller (more compact) than that of comparable CI expansions, and the method leads to size-extensive results. As discussed by Hättig,²³⁸ the ADC(2) method is closely related to the $CIS(D_{\infty})^{239}$ and CC2 approximations. In the same paper the author reports an implementation of the analytic excitedstate gradients for the ADC(2) and $CIS(D_{\infty})$ models. A modified ADC(2) method, ^{234,237} with an implemented core-valence separation approximation, has been extensively applied to evaluation of excitation energies and transition moments in the X-ray region, see, for example, refs 237 and 240–242.

The third-order ADC(3) model, for direct computation of electronic excitation energies and transition moments, has been presented by Trofimov et al.²³⁶ based on a specific reformulation of the diagrammatic perturbation expansion for the polarization propagator. The computational scheme combines diagonalization of a Hermitian secular matrix and perturbation theory for the matrix elements. The relationship of the ADC(3) scheme to coupled-cluster theory, in particular, with respect to treatment of transition moments, was also discussed. Recently, calculation of linear and quadratic response functions using the ADC(2) and ADC(3) models has been presented, based on a Lanczos procedure.²⁴³

4.6.7. Relativistic Corrections. Over the years many approaches have been proposed and implemented to compute relativistic first-order and higher order properties. Referring to the excellent monograph of Dyall and Fægri Jr.²⁷ for details, we note that relativistic methods may be divided into perturbative and nonperturbative approaches. In the first group a nonrelativistic calculation of energies and properties is followed by a perturbation treatment with the squared fine-structure constant α^2 as perturbation strength. In the second group a relativistic Hamiltonian is used as the starting point. The distinction between the two groups is blurred by the fact that perturbation theory is used to determine the relativistic Hamiltonian and that, even in perturbative approaches, scalar relativistic terms, often using scalar-relativistic Hamiltonians, such as the Douglas-Kroll-Hess Hamiltonian²⁴⁴⁻²⁴⁷ or relativistic effective core potentials,^{248,249} may be included in a nonperturbative fashion, treating only the more complicated nonscalar relativistic terms such as spin-orbit terms perturbatively.

With respect to the perturbative approaches, direct perturbation theory (DPT) to second order (DPT2) has been applied by Stopkowicz et al. to compute corrections to electrical first-order properties for coupled-cluster methods.²⁵⁰ Very recently, also energy gradients were implemented using DPT to fourth order (DPT4), allowing for calculation of fourth-order relativistic corrections of the Hartree–Fock energy.²⁵¹ Combining the fourth-order gradients with numerical differentiation, these authors also determined relativistic corrections to various firstorder electrical properties at the Hartree–Fock level of theory for selected systems.²⁵² We also mention the general secondquantization formalism presented by Helgaker et al.²⁵³ for calculation of relativistic corrections to molecular electronic energies and properties; in their formalism a Hamiltonian valid for all values of the fine-structure constant α is used to set up a perturbation expansion in orders of α^2 using the general framework of time-independent response theory in the same manner as for geometrical and magnetic perturbations. Cheng and Gauss²⁵⁴ proposed an analytical scheme for calculation of first-order electrical properties using the spin-free Dirac–Coulomb Hamiltonian, which thus exploits density-matrix formulations in nonrelativistic coupled-cluster derivative theory, with benchmark calculations for first-order electrical properties of the hydrogen halides up to HAt and of a few selected iodo(fluoro)methanes.

For the nonperturbative approaches a method to calculate properties to arbitrary order within the framework of Douglas–Kroll–Hess theory has been suggested by Wolf and Reiher.²⁵⁵ By consistent use of unitary transformations of both wave functions and property operators the picture-change error is eliminated. This method has been applied to calculate expectation values of powers of 1/r.²⁵⁶ Regarding fully relativistic treatments we note that van Stralen et al. in 2005 presented the first implementation of analytical first-order one-electron molecular properties at the Dirac–Coulomb MP2 level of theory using a formalism that allows use of inactive spinors.²⁵⁷

Second-order and higher order electric and magnetic properties may also be calculated using the perturbative or nonperturbative relativistic approaches. Norman et al.²⁵⁸ compared Douglas-Kroll-Hess, effective core potentials, and Dirac-Coulomb Hartree-Fock calculations for the study of nonlinear optical processes. Klopper et al.²⁵⁹ presented an implementation of first-order relativistic corrections to electrical response properties (with applications to the static and frequency-dependent dipole polarizability and second dipole hyperpolarizability of Ne) at the level of closed-shell coupled-cluster theory within the DPT framework. Perturbative studies of spin-orbit effects have been presented in a number of works, in particular, in connection to nuclear magnetic shielding constants.^{43,260} Examples of nonperturbative approaches to calculation of molecular properties is the no-pair four-component linear response theory and implementation of Visscher et al.,²⁶¹ the four-component Hartree-Fock linear response theory and implementation by Saue and Jensen,²⁶² and the quadratic response theory and implementation of Norman and Jensen.²⁶³ More recently, the open-ended response theory of Thorvaldsen et al.¹⁰⁶ was extended to the twoand four-component levels of theory for calculation of arbitrary one-electron properties to any order,¹⁰⁹ including London atomic orbitals to first order.²⁶⁴

5. SURVEY OF MOLECULAR PROPERTIES

The present section contains a survey of molecular properties with emphasis on correlated electronic-structure methods and recent computational advances. The section is divided into eight subsections, covering molecular geometrical properties in section 5.1, NMR and EPR parameters in sections 5.2 and 5.3, respectively, electric multipole moments in section 5.4, linear and nonlinear response properties in sections 5.5 and 5.6, respectively, intermolecular interactions in section 5.7, and excitation energies and excited-state properties in section 5.8. Our survey is necessarily incomplete but should be sufficiently exhaustive and representative to illustrate the enormous impact that the ab initio evaluation of molecular properties now has in many areas of chemistry.

5.1. Geometrical Derivatives

Since the pioneering work of Bratož,²⁶⁵ Gerratt and Mills,²⁶⁶ Pulay,¹¹⁰ Moccia,²⁶⁷ and Thomsen and Swanstrøm²⁶⁸ the analytic calculation of geometrical derivatives such as molecular gradients and Hessians has revolutionized quantum chemistry by enabling efficient and automated calculation of molecular structure, spectroscopic constants, reaction paths, and classical trajectories. Indeed, there is still a vigorous activity in this field, new techniques and implementations being developed at all levels of ab initio theory, from linear-scaling techniques for large molecular systems to highly accurate force-field calculations on small systems with explicitly correlated wave functions. We do not discuss here the evaluation of specific spectroscopic constants related to geometrical derivatives and distortions but note that the combined development of advanced analytic derivative procedures and high-accuracy quantum-chemistry methods has had a great impact on, for example, rotational spectroscopy. A thorough account of the theory and application of state-ofthe-art quantum-chemical methods for accurate determination of the spectroscopic parameters relevant to rotational spectroscopy has very recently been reported by Puzzarini, Stanton, and Gauss;³ we refer the interested reader to this work for more information on the ab initio study of these molecular properties.

5.1.1. Molecular Gradients and Hessians. Geometrical derivatives may be calculated using the general methods of response theory, for example, denoting nuclear coordinates by $\mathbf{R}_{K'}$ we obtain for the molecular gradient and Hessian, respectively

$$\frac{\mathrm{d}E}{\mathrm{d}\mathbf{R}_{K}} = \left\langle \left(\frac{\mathrm{d}H}{\mathrm{d}\mathbf{R}_{K}}\right) \right\rangle$$

$$\frac{\mathrm{d}^{2}E}{\mathrm{d}\mathbf{R}_{K}\mathrm{d}\mathbf{R}_{I}} = \left\langle \left(\frac{\mathrm{d}^{2}H}{\mathrm{d}\mathbf{R}_{K}\mathrm{d}\mathbf{R}_{I}}\right) \right\rangle + \left\langle \left(\frac{\mathrm{d}H}{\mathrm{d}\mathbf{R}_{K}}; \left(\frac{\mathrm{d}H}{\mathrm{d}\mathbf{R}_{I}}\right)^{\mathrm{T}}\right) \right\rangle$$
(586)

$$\int_{K} d\mathbf{R}_{L} = \left\langle \left(d\mathbf{R}_{K} d\mathbf{R}_{L} \right) \right\rangle^{-1} \left\langle \left(d\mathbf{R}_{K}^{\prime} \left(d\mathbf{R}_{L} \right) \right) \right\rangle$$
(587)

Evaluation of these expressions was discussed in some detail in section 4.5 and is also discussed in many specialist reviews. In particular, we here refer to the review of Stanton and Gauss,²⁶⁹ who thoroughly reviewed the field up to the year 2000. In the following, we focus on developments during the past decade. Moreover, in the present section we restrict ourselves to calculation of forces, force constants, and property gradients, postponing discussion of excited-state gradients to section 5.8.5.

Coupled-cluster theory is today recognized as the method of choice for providing ab initio results of high accuracy, the CCSD(T) method being considered the gold standard of ab initio theory.^{30,270} In 2003, an integral-density direct implementation of the analytic CCSD(T) molecular gradient based on a Lagrangian formulation was presented. This implementation circumvented the bottleneck of storing either $O(N^4)$ two-electron integrals or $O(N^4)$ density matrix elements on disk and included a frozen-core variant.²⁷¹ However, for molecules with difficult electronic structures, more elaborate methods are needed for high accuracy, based on a multiconfigurational reference wave function or on inclusion of higher order excitations. Thus, Gauss and Stanton²⁷² implemented in 2000 analytic first and second derivatives for the CCSDT-n (n = 1, 2, 3) models, ^{273,274} CCSDT-1 first derivatives having been implemented already in 1998 by Scuseria and Schaefer.²⁷⁵ In ref 272 analytic derivatives were also presented for the CC3 model,¹²² later extended to include full triples corrections

CCSDT.²⁷⁶ Using the string-based many-body formalism of Kállay and co-workers,^{277–279} Kállay et al. implemented first and second analytic derivatives for these general coupled-cluster methods.^{280,281}

For molecules with a strong multireference character, correlated methods based on the dominance of a single reference determinant are insufficient for high-accuracy calculations. In recent years there has been increasing interest in the development of analytical derivatives for multireference wave functions. Lischka, Dallos, and Shepard developed analytic gradients for multistate MRCI wave functions for excited electronic states, 282,283 whereas Khait, Theis, and Hoffmann recently presented an implementation of MRCI single-doubles (MRCISD) gradients²⁸⁴ based on the Lagrangian formulation of Helgaker and Jørgensen,^{77,156} allowing them to calculate state-averaged MCSCF wave functions with varying weights on the electronic states, a restriction in the approach of Lischka et al.²⁸³ Celani and Werner, on the other hand, implemented analytic gradients for multireference perturbation theory,²⁸⁵ providing an efficient route to dynamic correlation for multireference wave functions.^{115,286,287} More recently, Gauss and co-workers^{288,289} implemented analytic gradients for two-determinant reference wave functions in Mukherjee's state-specific multireference coupled-cluster theory,²⁹⁰ later including orbital relaxation at the MCSCF rather than Hartree-Fock level of theory.²⁸⁹

A well-known difficulty with determinantal wave function expansions is their slow convergence with respect to the size of the orbital basis.^{30,291} This problem is solved by introducing an explicit dependence on interelectronic distances into the wave function in R12 theory,^{292,293} later generalized to F12 theory.^{294,295} For an overview of recent developments in this field, see also the review by Helgaker, Klopper, and Tew.²⁷⁰ The added complications of explicit correlation on top of the difficulties associated with nonvariational wave functions make it difficult to develop analytic techniques for geometrical derivatives and other properties. Still, an analytic gradient for the MP2-R12 method has been implemented by Kordel, Villani, and Klopper.²¹⁵

During the past decade, several approaches have been proposed to reduce the computational cost of correlated wave function methods, making them applicable to large molecular systems. One successful approach is the resolution of the identity approximation, $^{168-170}$ in which one of the overlap distributions of the twoelectron integrals is fitted in an auxiliary basis, reducing cost by avoiding calculation of four-center integrals. Distasio, Steele, and Head-Gordon implemented analytic gradients for this technique,²⁹⁶ in which the number of orbital responses required for gradients in their dual-basis resolution of the identity MP2 (DB-RI-MP2) method has been reduced to the product of the number of occupied and virtual orbitals determined by the rank of the small AO basis, leading to an efficient code for systems up to about 100 atoms. Kossmann and Neese recently presented an efficient implementation of gradients for the RI-J chain of spheres-exchange RI-MP2 (RIJCOSX-MP2) approach.^{297,298} We also note the RI implementation of gradients for the spin-component-scaled (SCS) methods by Hellweg, Grün, and Hättig.²⁹⁹

A related technique for reducing the cost of two-electron integral evaluation is Cholesky decomposition, recently applied to gradient integrals, but only in Kohn–Sham theory.³⁰⁰ Likewise, an efficient linear-scaling method for molecular forces has recently been presented in Kohn–Sham theory,³⁰¹ based on the combined use of density fitting, the continuous fast-multipole method,³⁰² and expansion of solid-harmonic Gaussians in Hermite

rather than Cartesian Gaussians;³⁰³ we also note the Kohn– Sham molecular gradient of Dominguez-Soria et al.³⁰⁴ for use in calculations on large molecules. For these molecular-gradient techniques to be useful in a wave function context more efficient methods must be developed for evaluation of exact-exchange integrals.

An important approach for reducing the computational scaling for large molecules is the use of local orbitals and local correlation methods.^{206,207} Schutz et al. presented a local-correlation implementation of analytic gradients utilizing the resolution of the identity approximation,³⁰⁵ demonstrating that the approach yields quadratic scaling with respect to molecule size and cubic scaling with respect to basis-set size. A local coupled-cluster gradient has been presented by Rauhut and Werner but appears not to be used in practical calculations.³⁰⁶ Regarding orbital localization, we also mention the work of Leininger et al.³⁰⁷ on preserving orbital locality during CASSCF geometry optimizations.

The frozen natural-orbital coupled-cluster (FNO-CC) approach^{308,309} allows for faster coupled-cluster calculations by reducing the size of the virtual space, see also related work by Klopper et al.³¹⁰ Taube and Bartlett presented an implementation of analytic gradients for the FNO-CC approach,³¹¹ including orbital relaxation for noncanonical and semicanonical perturbed orbitals. Although the method was successful in reproducing benchmark data for energies, geometries, and vibrational frequencies, it requires substantial disk storage, making it less attractive for larger systems. These authors also presented analytic derivatives for the variational $\Lambda CCSD(T)$ method^{312–316} (also known as the a-CCSD(T) method³¹³), demonstrating the improvements in the potential-energy surfaces with this method due to its better performance in bond-breaking situations.³¹⁶ Analytical $\Lambda CCSD(T)$ (T) derivatives were also discussed, but not implemented, by Crawford and Stanton in 1998.³¹³

In recent years, Gauss and co-workers developed highly correlated methods for relativistic corrections based on a perturbative treatment of the leading order relativistic corrections.^{317–320} Michauk and Gauss presented analytic gradients for such an approach, including relativistic corrections from the mass-velocity and (one- and two-electron) Darwin terms,³¹⁷ see eqs 39 and 40. More recently, Wang and Gauss presented analytic gradients³¹⁸ and Hessians³¹⁹ for two-component CCSD(T) wave functions including spin–orbit corrections,³²⁰ providing a cost-effective and highly accurate alternative to four-component coupled-cluster calculations.^{321,322} Also very recently, Zou et al.³²³ derived and implemented the analytical energy gradient of the normalized elimination of the small component (NESC) method,³²⁴ which allows calculation of NESC geometries and other first-order molecular properties in combination with Hartree-Fock theory, Kohn-Sham theory, coupled-cluster theory, or any electron correlation-corrected quantum-chemical method, provided the NESC Hamiltonian is determined in an efficient, yet accurate, way. Another interesting recent development in coupled-cluster theory has been the appearance of several parallel implementa-tions, also for molecular gradients and Hessians.^{325–327} Analytic derivatives are beginning to appear for nonstandard wave function models such as multiwavelets³²⁸ and the contracted anti-Hermitian Schrödinger equation.³²⁹ Finally, we note recent work on quantum algorithms for molecular properties and geometries. 330,331

5.1.2. Molecular Higher Order Derivatives. Analytic calculations of cubic and higher order contributions to molecular force fields are very limited. Handy and co-workers developed an

analytic approach for cubic³³² and quartic³³³ force fields as well as first and second geometric derivatives of the dipole moment and the electric polarizability.³³⁴ Despite the obvious advantages of such an analytic approach, the code has not been much used after the initial applications but we note a study of the anharmonic force field of benzene.³³⁵ To the best of our knowledge, higher order geometrical derivatives have not been explored by analytical techniques since these pioneering works. Instead, such derivatives have been obtained in a mixed numerical—analytical manner in which higher order derivatives are obtained by finite differences of analytically calculated gradients and Hessians, see, for example, refs 336–338.

5.1.3. Property Geometrical Derivatives. Besides energy derivatives, significant efforts have been directed toward development of analytical procedures for computing the geometric (and magnetic) derivatives of molecular response properties, which are important for (accurate) determination of various observables and spectroscopic constants. For instance, analytic computation of the gradient of the dipole moment, required for calculating infrared intensities in the double-harmonic approximation, was pioneered by Bratož in 1958 with efficient wave function implementations appearing from the mid-1980s.^{339,340} Likewise, the gradient of the electric dipole polarizability is needed for vibrational Raman intensities³³⁴ and for description of the related coherent anti-Stokes Raman (CARS)^{80,150,341} and vibrational Raman optical activity ROA^{2,342} techniques. Geometric derivatives of second-order and higher order response functions are needed to evaluate their vibrational corrections within the Born-Oppenheimer approximation, as discussed in more detail in section 6. The first geometric derivative of the transition dipole strength yields information on how the motion of the nuclei affects the UV spectrum (or the one-photon absorption) of a molecule through the Herzberg-Teller contribution-that is, the linear dependence of the transition dipole moment on the nuclear coordinates, which must be taken into account to describe how forbidden transitions become allowed by vibronic effects.³⁴³⁻³⁴⁵ Finally, the first geometric derivative of the excited-state energy can be used to determine and characterize the equilibrium geometry of a system in an excited electronic state, as discussed in section 5.8.5.

At present, most implementations of polarizabilities, hyperpolarizabilities, and transition-moment geometric gradients have been presented at the Hartree—Fock and Kohn—Sham levels of theory,^{79,80,334,345—347} with the notable exception of the coupledcluster implementation of the polarizability gradient presented by O'Neill et al. in 2007,³⁴⁸ which includes a general string-based program for calculation of Raman intensities for arbitrary coupled-cluster and CI methods.

5.2. Nuclear Magnetic Resonance

Evaluation of NMR parameters has become an important application of quantum chemistry over the last two decades. In the present subsection we consider the ab initio evaluation of the two basic parameters of high-resolution NMR: nuclear shielding constants and indirect nuclear spin—spin constants; in addition, we consider here the nuclear spin—rotation constants of microwave spectroscopy, which conceptually and computationally are closely related to the nuclear shielding constants.

Consider a closed-shell molecule in the presence of an external field *B* along the *z* axis, with nuclear spins I_K related to the nuclear magnetic moments M_K as in eq 55. Assuming free molecular rotation, the nuclear-magnetic energy levels can be reproduced

by the following high-resolution NMR spin Hamiltonian

$$H^{\text{NMR}} = -\sum_{K} \gamma_{K} \hbar (1 - \sigma_{K}) B I_{K,z} + \sum_{K > L} \gamma_{K} \gamma_{L} \hbar^{2} K_{KL} \mathbf{I}_{K} \cdot \mathbf{I}_{L}$$
(588)

where we introduced the nuclear shielding constants σ_K and the (reduced) indirect nuclear spin—spin coupling constants K_{KL} . Whereas the first term represents the Zeeman interaction of the nuclei with the external magnetic field, the second term represents the interactions between the nuclear moments. This is an effective nuclear spin Hamiltonian: it reproduces NMR spectra without considering the electrons explicitly. In experimental work the parameters σ_K and K_{KL} are adjusted to fit the observed spectra; here, we consider their evaluation using molecular electronic-structure theory. For specialized reviews, see refs 6 and 349.

5.2.1. Nuclear Shielding Constants. From a consideration of the NMR spin Hamiltonian in eq 588 we find that the shielding tensor of nucleus K is related to the second derivative of the molecular energy with respect to the applied field **B** and the nuclear magnetic moment M_K as

$$\frac{\mathrm{d}^2 E}{\mathrm{d}\mathbf{B}\mathrm{d}\mathbf{M}_K} = -\mathbf{I}_3 + \boldsymbol{\sigma}_K \tag{589}$$

The first term, which represents the coupling of the nucleus to the magnetic field in the absence of electrons, arises from the nuclear part of the Zeeman Hamiltonian H_z of eq 65. The second part of eq 589 is the nuclear shielding tensor σ_K and describes the modification to the Zeeman interaction introduced by the electrons. From an inspection of the molecular electronic Breit— Pauli Hamiltonian in eq 57 we note that there is a second-order diamagnetic coupling from $H_{\rm BM}$ in eq 72 and first-order couplings of the electrons with the field in the Zeeman operator H_z of eq 65 and with the nuclei in the paramagnetic spin—orbit operator H_{so} of eq 67, yielding (with the gauge origin at **O** and omitting summation over electrons for clarity)

$$\sigma_{K} = \left\langle \left(\frac{\mathrm{d}^{2} H_{\mathrm{BM}}}{\mathrm{d} \mathbf{B} \mathrm{d} \mathbf{M}_{K}} \right) \right\rangle + \left\langle \left(\frac{\mathrm{d} H_{z}}{\mathrm{d} \mathbf{B}}; \left(\frac{\mathrm{d} H_{so}}{\mathrm{d} \mathbf{M}_{K}} \right)^{\mathrm{T}} \right) \right\rangle$$
$$= \frac{\alpha^{2}}{2} \left\langle 0 \left| \frac{\mathbf{r}_{\mathbf{O}}^{\mathrm{T}} \mathbf{r}_{K} \mathbf{I}_{3} - \mathbf{r}_{\mathbf{O}} \mathbf{r}_{K}^{\mathrm{T}}}{r_{K}^{3}} \right| 0 \right\rangle - \alpha^{2} \sum_{n_{S}} \frac{\langle 0 | \mathbf{I}_{\mathbf{O}} | n_{S} \rangle \langle n_{S} | r_{K}^{-3} \mathbf{I}_{K}^{\mathrm{T}} | 0 \rangle}{E_{n_{S}} - E_{0}}$$
(590)

where the Rayleigh–Schrödinger expression, valid for closedshell states only, was derived by Ramsey in 1950.³⁵⁰ By symmetry there are no spin contributions from H_z and H_{so} for closed shells. The summations are therefore only over singlet excited states, as indicated by the summation indices n_s .

The first term in eq 590 is the diamagnetic contribution to the shielding. With the gauge origin at the nucleus in question its isotropic component is positive and represents the shielding arising from induced currents in the unperturbed electron density according to Lenz's law, opposing the applied field. The second, paramagnetic contribution in eq 590 arises from the induced electronic magnetic moment (wave function relaxation), which typically aligns with the field and hence opposes the diamagnetic term. Since the nuclear shielding constants arise from a hyperfine interaction between the electrons and the nuclei it is proportional to $\alpha^2 \approx 5 \times 10^{-5}$ and is measured in ppm. For

¹S systems (closed-shell atoms) the paramagnetic term vanishes completely and the shielding is given by the Lamb formula: $\sigma_K = \alpha^2 \langle 0 | r_K^{-1} | 0 \rangle / 3$.

In practice, the shielding constants are not evaluated from the sum-over-states expressions in eq 590 but from the responsetheory expression, taking the derivative of the energy with respect to the external magnetic field **B** and the nuclear magnetic moments \mathbf{M}_{K} . According to the 2n + 1 rule only first-order responses are needed; to reduce costs we evaluate these for the three magnetic field directions; for very large molecules other strategies may be important when only a few shielding constants are of interest.³⁵¹ To avoid gauge-origin problems, London orbitals^{44,61,352} (or some equivalent scheme^{353–359}) must be used for all but the smallest systems.

In the following we review recent computational advances in the study of shielding constants. Although this field is dominated by Kohn–Sham theory,^{360,361} MP2 theory with London orbitals³⁶² remains a standard method for shielding calculations, providing accurate results at a reasonable cost; moreover, as in many other areas of computational chemistry, the most accurate results are obtained using coupled-cluster theory. For developments up to 2002, see refs 6, 249, and 363; we also note the more recent perspective article by Vaara.³⁶⁴ In the following we restrict ourselves to the most recent developments in wave function theory for calculation of shielding constants.

During the past decade most developments in high-level calculations of NMR shielding constants have been carried out at the coupled-cluster level of theory. In particular, Kállay and Gauss implemented the calculation of NMR shieldings at arbitrary excitation levels for coupled-cluster and CI wave functions.²⁸¹ The quality of the results that can be obtained can be seen from Tables 1 and 2. Although higher order excitation levels give significant corrections to the shielding constants, these tables highlight the excellent performance of the CCSD(T) method for shielding constants.

For highly accurate calculations of nuclear shielding constants inclusion of zero-point vibrational corrections is important.³⁶⁶⁻³⁶⁸ These corrections can be on the order of 3-5% and larger than errors arising from approximations in the CCSD(T) model. Ruud et al.³⁶⁹ noted that the zero-point vibrational corrections for functional protons are transferable between molecules since the protons vibrate almost independently of the molecular skeleton. Both harmonic and anharmonic corrections to the zero-point vibrational corrections can be important. An approach such as the vibration mode following (VMF) method,³⁷⁰ which only includes harmonic contributions, often gives an incorrect description of the vibrational effects, both their sign and magnitude. An alternative to perturbation theory for zero-point vibrational corrections is the Feynman-path integral Monte Carlo method, but the large number of configurations needed to perform an accurate vibrational averaging have prevented the use of methods more accurate than Hartree–Fock theory for the vibrationally averaged shielding constants.^{371,372}

A number of CCSD(T) benchmark studies with perturbational zero-point vibrational corrections have been presented,^{373–379} yielding an agreement with experiment within 1–2 ppm. By parallelizing the CCSD(T) calculation of shielding constants,³²⁶ Harding et al. showed that only by combining highly accurate equilibrium geometries with CCSD(T) calculations in quadruple- ζ basis sets were the shielding constants of the 1-adamantyl cation obtained in good agreement with experimental observations.³²⁶ Table 1. Isotropic (σ) and Anisotropic ($\Delta \sigma$) NMR Shielding Constants (in ppm) of the BH Molecule Calculated at Different Levels of Electronic-Structure Theory at a Bond Distance of 123.24 pm in the TZP+ Basis with All Electrons Correlated^{281,365}

	$\sigma(^{11}B)$	$\Delta\sigma(^{11}\mathrm{B})$	$\sigma(^{1}H)$	$\Delta\sigma(^{1}\mathrm{H})$
HF	-261.3	690.1	24.21	14.15
MP2	-220.7	629.9	24.12	14.24
CCSD	-166.6	549.4	24.74	13.53
CCSD(T)	-171.5	555.2	24.62	13.69
CCSDT	-171.8	557.3	24.59	13.72
CCSDTQ	-170.1	554.7	24.60	13.70
CISD	-182.4	572.9	24.49	13.87
CISDT	-191.7	587.0	24.35	14.06
CISDTQ	-170.2	554.9	24.60	13.70
FCI	-170.1	554.7	24.60	13.70

Table 2. Coupled-Cluster Convergence of Isotropic (σ) and Anisotropic ($\Delta \sigma$) NMR Shielding Constants (in ppm) in CO Calculated in the cc-pVDZ Basis with a Frozen Core²⁸¹

	CCSD	CCSD(T)	CCSDT	CCSDTQ	CCSDTQ5	FCI
$\sigma(^{13}C)$	32.23	35.91	35.66	36.10	36.14	36.15
$\Delta \sigma$ (¹³ C)	361.30	356.10	356.47	355.85	355.80	355.79
$\Delta\sigma(^{17}\text{O})$	-13.93	-13.03	-13.16	-12.81	-12.91	-12.91
$\Delta\sigma(^{17}\text{O})$	636.01	634.55	634.75	634.22	634.52	634.35

Although London orbitals are today the most popular method for ensuring gauge-origin-independent calculations of magnetic properties, a number of different schemes have also been proposed.³⁵³⁻³⁵⁹ Among these the method of continuous transformation of the gauge origin of the current density by setting the diamagnetic contribution to zero (CTOCD-DZ)^{357,358} has gained popularity for correlated calculations since it ensures, for properties linear in the external magnetic field, gauge-origin independence without introducing field-dependent two-electron integrals. Sauer and co-workers implemented the CTOCD-DZ approach at various correlated levels, including the MCSCF and SOPPA³⁸¹ and CCSD³⁸² levels of theory. Garcia et al.³⁸³ compared the effect of coupled-cluster truncation on nuclear shieldings with an orbitally unrelaxed scheme using the CTOCD-DZ method, different from the energy-derivative approach used by Kállay and Gauss,²⁸¹ in which orbital relaxation is included in the calculation.

For high accuracy in the calculated shielding constants a carefully selected basis set that properly describes the outercore—inner-valence region is necessary; at the same time, it must be constructed in a manner that allows for a systematic improvement in the description of the correlation energy. Manninen and Vaara developed a technique for designing basis sets for molecular properties with special requirements in the core or valence regions³⁸⁴ based on the use of completeness profiles for measuring basis-set saturation in selected exponent ranges.³⁸⁵ The authors applied the developed basis sets to calculation of molecular magnetizabilities, nuclear magnetic shieldings, and indirect spin—spin coupling constants.³⁸⁴ We also note that several studies of basis-set extrapolation schemes for NMR shielding constants have been presented, ^{386–390} but these have not gained the same popularity as for energetics.^{270,391}

A novel application of the highly accurate methods that have been developed for nuclear magnetic shielding constants is the combination of measurement and theory to revisit and reassign values of nuclear magnetic moments. The tabulated magnetic moments have usually been derived from magnetic resonance measurements of atomic or molecular species with an assumed value for the diamagnetic shielding of the atomic species. As many of these values are from the 1950s, their accuracy is often low. Jackowski, Jaszuński, and co-workers performed important, pioneering work in revising and proposing new values for the nuclear magnetic moments of ¹³C, ¹⁴N, ¹⁵N, ¹⁷O, ¹⁹F, ³¹P, and ³³S, ³⁹² ²⁹Si and ⁷³Ge, ³⁹³ and ¹⁰B and ¹¹B.³⁹⁴

With the development of accurate methods for calculating NMR shielding constants there has been increased interest in smaller corrections to these constants. Whereas shielding polarizabilities (the derivatives of the shielding constants with respect to external electric fields) previously attracted interest as a means of rationalizing intermolecular and solvent effects on shielding constants, these effects are now calculated explicitly and we are aware of only one recent correlated study of shielding polarizabilities.³⁹⁵ To complement this study, Vaara, Manninen, and Lounila investigated the magnetic-field dependence of atomic systems to explore potential consequences for high-field NMR spectrometers.³⁹⁶ Recently, Kjær et al.³⁹⁷ explored the validity of the multipole shielding polarizabilities in combination with a reaction-field approach against the solvent effect induced by a polarizable molecular force field, showing that, by a fortuitous error cancellation, the best results are obtained by including only the linear electric-field effects.

Despite its smallness the effect of parity violation on nuclear magnetic shielding constants has been investigated to establish whether NMR spectroscopy is suitable for observing the effects of parity violation in chiral molecules.^{398,399} Both perturbation theory based on a nonrelativistic reference frame³⁹⁸ and four-component theory have been used to investigate these effects.³⁹⁹

Following the proposal of Buckingham and Parlett,⁴⁰⁰ Jaszuński and Rizzo, on the other hand, investigated the effect of circularly polarized light on nuclear magnetic moments (a magnetic moment-induced Faraday effect) using MCSCF quadratic response functions.⁴⁰¹

For heavier elements, relativistic effects are important, not only for the shielding of the heavy atoms but also for light elements in close proximity to a heavy element. This heavy-atom effect arises because spin—orbit coupling at the heavy atom induces a spin polarization of the electrons that couples (by the Fermi-contact and spin—dipole mechanisms) to the nuclear magnetic moment of the nearby light atom in the same way that this nuclear moment couples to the spin polarization induced by other nuclei; indeed, Kaupp et al.⁴³ demonstrated the close connection between the spin—spin coupling constants and the spin—orbit corrections to the shielding constant of light elements in the vicinity of heavy atoms.

Vaara and co-workers performed a complete perturbation analysis of the relativistic corrections to order a^4 to the shielding constants and presented calculations at the MCSCF level of theory.^{260,402–404} Nakatsuji and co-workers developed a generalized unrestricted quasirelativistic approach based on a mixed analytic and numerical scheme for calculation of relativistic nuclear magnetic shielding constants^{405,406} and extended it to the inclusion of electron correlation at the MP2 level of theory.⁴⁰⁷

The past decade has seen the emergence of two- and fourcomponent methods for calculation of nuclear magnetic shielding constants. Much of this development has taken place using Kohn-Sham theory, but we focus here on the wave function-based developments. In the relativistic domain the Dirac operator is linear in the momentum operator; consequently, the shielding constant is fully described by a linear response function, without an apparent diamagnetic contribution. However, as discussed by Aucar et al.,⁴⁰⁸ the diamagnetic term arises from a redressing of the electrons in the presence of the magnetic field, being described by the electron-positron rotations in the linear response equations. However, the basisset convergence of these rotations is very slow. It may therefore be advantageous to ignore these rotations and instead calculate the diamagnetic contribution through the Sternheim approximation.409

An important issue in four-component calculations is the balanced treatment of the large and small components. In the absence of magnetic fields it is useful to enforce that the small and large components are related through the restricted-kinetic balance condition,^{410,411} often enforced in the calculation and design of dual family basis sets⁴¹² or explicitly in the integral evaluation.413 Given that the small and large components are related by the momentum operator, application of an external magnetic field extends this condition to what is known as the restricted magnetic balance condition. Because it is difficult to design basis sets that fulfill this extended condition, unrestricted kinetic balance is instead often used to ensure accurate shielding constants,⁴⁰⁸ leading to the need for using much larger basis sets. An elegant solution to this problem was recently proposed by Komorovsky et al.,⁴¹³ who enforced relativistic kinetic balance in calculations at the integral level, extending the approach to include contributions from London orbitals.⁴¹⁴⁻⁴¹⁶ We here also note that Ilias et al.⁴¹⁷ presented gauge-origin-independent calculations of nuclear shielding constants at the four-component Dirac-Coulomb Hartree-Fock level of theory using London atomic orbitals.

Kudo et al.^{418,419} presented calculations of shielding constants using the method of NESC tensors by Filatov and Cremer,³²⁴ giving results in the zero-order regular approximation (ZORA)^{420,421} and in the second-order regular approximation (SORA) very similar to those obtained using the infinite-order two-component coupled Hartree—Fock method.⁴²² Kato et al.⁴²³ presented fourcomponent relativistic calculations of nuclear shielding constants at the CISD and CCSD levels of theory but without strictly enforcing gauge-origin invariance. We also note that Aucar and co-workers presented relativistic four-component calculations of nuclear shielding constants at the Dirac—Coulomb Hartree—Fock level of theory.^{424,425}

Calculation of current densities arising from application of an external magnetic field to a molecular electron density has been extended to the correlated levels of theory using London atomic orbitals, both for closed-⁴²⁶ and for open-shell species.⁴²⁷ It is noteworthy that London orbitals are not sufficient to ensure current conservation,⁶⁰ but they do lead to a significantly reduced current nonconservation.⁴²⁶ Bast et al.⁴²⁸ extended the scheme for calculating induced magnetic currents to the relativistic four-component Kohn—Sham level of theory but using a common gauge-origin approach. Jusélius and Sundholm advocated the use of calculated current densities to define the aromatic character of molecules⁴²⁹ as an alternative to the nucleus-independent chemical shift.⁴³⁰

We end the discussion of computational models for NMR shielding constants by considering developments aimed at large molecules based on Hartree-Fock and Kohn-Sham theories. Gauss and Werner presented in 2000 a local GIAO-MP2 approach for calculation of NMR shielding constants.⁴³¹ Although promising results were obtained, there does not appear to be many applications of the formalism. Ochsenfeld and coworkers developed linear-scaling methods for Hartree-Fock calculations of shielding constants using London atomic orbitals.^{12,432} Recently, Beer, Kussmann, and Ochsenfeld³⁵¹ extended this approach to achieve sublinear scaling by utilizing the locality of the perturbed density arising from the nuclear magnetic moment of interest. Loibl, Manby, and Schütz recently presented a density-fitting Hartree-Fock code for calculation of NMR shielding constants using London orbitals.⁴³³ We also note that Morokuma and co-workers extended their "our own *n*-layer integrated molecular orbital and molecular mechanics" (ONIOM) multilevel method to the two-level GIAO-MP2:GIAO-HF approach for shielding constants. 434,435

5.2.2. Nuclear Spin–Spin Couplings. To determine the coupling between the nuclear magnetic moments M_K and M_L we differentiate the molecular electronic energy with respect to these magnetic moments yielding

$$\frac{\mathrm{d}^2 E}{\mathrm{d}\mathbf{M}_K \mathrm{d}\mathbf{M}_L} = \mathbf{D}_{KL} + \mathbf{K}_{KL} \tag{591}$$

where D_{KL} and K_{KL} are the direct and indirect nuclear spin—spin coupling tensors, respectively. The direct coupling, which arises from the purely nuclear part of the spin—spin operator H_{ss} in eq 68, occurs by a classical dipole mechanism

$$\mathbf{D}_{KL} = \alpha^2 R_{KL}^{-5} (R_{KL}^2 \mathbf{I}_3 - 3 \mathbf{R}_{KL} \mathbf{R}_{KL}^{\mathrm{T}}) \approx 10^{-12} \text{ au}$$
(592)

However, since this coupling is anisotropic it vanishes in isotropic media such as gases and liquids and does not contribute to the high-resolution NMR spin Hamiltonian of eq 588. Instead, the spin—spin coupling in the NMR spin Hamiltonian is mediated by the electrons and is fully described by the indirect coupling tensor in eq 591.

From inspection of the Breit–Pauli Hamiltonian in eq 57 we find that the nuclear moments couple quadratically to the electrons in the diamagnetic operator $H_{\rm MM}$ of eq 73 and linearly to the electrons in the spin–orbit operator $H_{\rm so}$ of eq 67 and in the spin–spin operator $H_{\rm ss}$ of eq 68, yielding the following expression for the indirect nuclear spin–spin coupling constant⁴³⁶

$$\begin{split} \mathbf{K}_{KL} &= \left\langle \left(\frac{\mathrm{d}^{2} H_{\mathrm{MM}}}{\mathrm{d}\mathbf{M}_{K} \mathrm{d}\mathbf{M}_{L}} \right) \right\rangle + \left\langle \left(\frac{\mathrm{d} H_{so}}{\mathrm{d}\mathbf{M}_{K}}; \left(\frac{\mathrm{d} H_{so}}{\mathrm{d}\mathbf{M}_{L}} \right)^{\mathrm{T}} \right) \right\rangle \\ &+ \left\langle \left(\frac{\mathrm{d} H_{ss}}{\mathrm{d}\mathbf{M}_{K}}; \left(\frac{\mathrm{d} H_{ss}}{\mathrm{d}\mathbf{M}_{L}} \right)^{\mathrm{T}} \right) \right\rangle = \frac{\alpha^{4}}{2} \left\langle 0 | \frac{\mathbf{r}_{K}^{\mathrm{T}} \mathbf{r}_{L} \mathbf{I}_{3} - \mathbf{r}_{K} \mathbf{r}_{L}^{\mathrm{T}}}{\mathbf{r}_{N}^{3} \mathbf{r}_{L}^{3}} | \mathbf{0} \right\rangle \\ &- 2\alpha^{4} \sum_{n_{S}} \frac{\langle 0 | \mathbf{r}_{K}^{-3} \mathbf{I}_{K} | n_{S} \rangle \langle n_{S} | \mathbf{r}_{L}^{-3} \mathbf{I}_{L}^{\mathrm{T}} | \mathbf{0} \rangle}{E_{n_{S}} - E_{0}} \\ &- 2\alpha^{4} \sum_{n_{T}} \frac{\left\langle 0 | \frac{8\pi}{3} \delta(\mathbf{r}_{K}) \mathbf{s} + \frac{3\mathbf{r}_{K} \mathbf{r}_{K}^{\mathrm{T}} - \mathbf{r}_{K}^{2} \mathbf{I}_{3} \mathbf{s} | n_{T} \right\rangle \left\langle n_{T} | \frac{8\pi}{3} \delta(\mathbf{r}_{L}) \mathbf{s}^{\mathrm{T}} + \frac{3\mathbf{r}_{L} \mathbf{r}_{L}^{\mathrm{T}} - \mathbf{r}_{L}^{2} \mathbf{I}_{3} \mathbf{s}^{\mathrm{T}} | \mathbf{0} \right\rangle}{E_{n_{T}} - E_{0}} \\ \end{split}$$

where we omitted summations over electrons for clarity and where $n_{\rm S}$ and $n_{\rm T}$ denote singlet and triplet excited states, respectively. The first term is the diamagnetic spin—orbit contribution to the spin—spin coupling constant, which arises from the diamagnetic $H_{\rm MM}$ operator in eq 73. The second term in eq 593 is the paramagnetic spin-orbit contribution and arises from the part of the spin-orbit operator H_{so} in eq 67 that represents the coupling of the nuclear spins with the orbital motion of the electrons $\alpha^2 \mathbf{M}_K \cdot \mathbf{l}_{iK} / r_{iK}^3$. As this operator does not contain electron spin, the first summation in eq 593 is only over singlet states. The paramagnetic and especially diamagnetic orbital contributions are usually but not invariably small; for large internuclear separations they cancel.⁴³⁷ The second summation in eq 593 arises from the hyperfine part of the spin-spin operator H_{ss} in eq 68, with summation over triplet states only. With a large prefactor $(8\pi/3)^2 \approx 70.2$, the isotropic Fermicontact-Fermi-contact mechanism often dominates short-range spin-spin coupling constants, whereas the mixed Fermi-contact-spin-dipolar mechanism dominates the anisotropic part of the coupling tensor.³⁴⁹ However, there are many exceptions to these rules, and none of the contributions to the spin-spin coupling constants can be a priori neglected. Since the indirect nuclear spin-spin coupling constant arises from hyperfine interactions it is exceedingly small: $K_{KL} \approx 10^{-16}$ au. Experimentalists usually work in terms of the isotope-dependent indirect nuclear spin-spin coupling constants $J_{KL} = h\gamma_K \gamma_L K_{KL} / 4\pi^2 \approx 1$ Hz, see eq 588.

The indirect nuclear spin—spin coupling constants are evaluated using response theory, calculating the second derivative of the electronic energy with respect to the nuclear magnetic moments. According to the 2n + 1 and 2n + 2 rules, only first-order responses are needed. Thus, for a molecule containing $N_{\rm M}$ magnetic nuclei, $3N_{\rm M}$ response equations must be solved. However, to take advantage of spin symmetry, the Wigner—Eckart theorem is usually invoked, yielding three singlet and six triplet response equations for each nucleus, each of smaller dimension (the same as for a magnetic field perturbation, for example).⁶ Unlike for calculations of nuclear shielding constants, London orbitals are not needed for spin—spin calculations since no external magnetic field is involved.

Apart from requiring the solution of a large number of response equations, evaluation of indirect nuclear spin-spin coupling constants poses several challenges. First, nuclear spinspin coupling constants are extremely sensitive to an inadequate description of static electron correlation.⁶ As a result, the restricted Hartree-Fock model is unsuitable for spin-spin calculations, producing erratic results. Therefore, only correlated wave function models give reliable spin-spin constants, adding to the cost of such calculations. With the emergence of DFT as a computational model in quantum chemistry,⁴³⁸⁻⁴⁴¹ it became possible to perform spin-spin calculations reliably, at low cost, by providing a description less affected by triplet instabilities than Hartree-Fock theory; for a recent discussion of this point, see ref 442. In the present review we concentrate on high-accuracy calculations of nuclear spin-spin coupling constants by wave function theory. In Table 3 we listed calculated spin-spin coupling constants for a set of small molecules.

The first accurate calculations of indirect spin—spin coupling constants were the CI calculations on HD by Kowalewski et al.⁴⁴³ in 1974. More recently, the FCI model has been applied to evaluate the coupling constant of the helium dimer⁴⁴⁴ and the BH molecule.⁴⁴⁵ To study larger molecules, MCSCF theory has been widely used following the first implementation by Vahtras et al. in 1992.⁴⁴⁶ For small molecules such as C_2H_2 highly accurate results may be obtained with MCSCF theory;⁴⁴⁷ for larger systems it becomes difficult to choose an adequate active orbital space.⁴⁴⁸ Indeed, after introduction of Kohn—Sham

Table 3. Calculated Spin–Spin Coupling Constants (in Hertz) Compared with Experimental Values^{*a*}

		RHF	CAS	RAS	SOPPA	CCSD	CC3	exp*	vib
HF	$^{1}J_{\rm HF}$	59.2	48.0	48.1	46.8	46.1	46.1	47.6	-3.4
СО	¹ Jco	13.4	-28.1	-39.3	-45.4	-38.3	-37.3	-38.3	-1.7
N_2	${}^{1}J_{NN}$	175.0	-5.7	-9.1	-23.9	-20.4	-20.4	-19.3	-1.1
H_2O	$^{1}J_{OH}$	63.7	51.5	47.1	49.5	48.4	48.2	52.8	-3.3
	$^{2}J_{\rm HH}$	-1.9	-0.8	-0.6	-0.7	-0.6	-0.6	-0.7	0.1
$\rm NH_3$	${}^{1}J_{\rm NH}$	61.4	48.7	50.2	51.0	48.1		50.8	-0.3
	$^{2}J_{\rm HH}$	-1.9	-0.8	-0.9	-0.9	-1.0		-0.9	0.1
C_2H_4	$^{1}J_{CC}$	1672.0	99.6	90.5	92.5	92.3		87.8	1.2
	$^{1}J_{CH}$	249.7	51.5	50.2	52.0	50.7		50.0	1.7
	$^{2}J_{CH}$	-189.3	-1.9	-0.5	-1.0	-1.0		-0.4	-0.4
	$^{2}J_{\rm HH}$	-28.7	-0.2	0.1	0.1	0.0		0.2	0.0
	${}^{3}J_{cis}$	30.0	1.0	1.0	1.0	1.0		0.9	0.1
	${}^{3}J_{tn}$	33.3	1.5	1.5	1.5	1.5		1.4	0.2
$ \overline{\Delta} $	abs	180.3	3.3	1.6	1.8	1.2	1.6	*at l	Re
	%	5709	60	14	24	23	6		
^a For references to calculated and experimental values, see ref 349.									

theory for calculation of spin—spin coupling constants in the $1990s^{438-441}$ MCSCF theory has played a less prominent role in this area.

For high-accuracy calculations of spin-spin coupling constants, coupled-cluster theory is the preferred method. At the basic CCSD level of theory there are two implementations for spin-spin calculations: the EOM-CCSD implementation by Perera, Sekino, and Bartlett⁴⁴⁹ and the analytic second-derivative implementation by Stanton and Gauss.^{269,326} To avoid problems associated with triplet instabilities, derivative-based coupledcluster calculations of spin-spin coupling constants should be performed in an orbital-unrelaxed manner, without relaxing the orbitals in response to the triplet perturbations; the effects of orbital relaxation are instead described by means of the coupledcluster singles amplitudes.¹³³ Beyond the CCSD model the first implementations of coupled-cluster theory for calculation of spin-spin coupling constants are the CCSDT, CCSD(T), and CC3 implementations of Auer and Gauss.¹³³ Being suited to an unrelaxed orbital treatment, the CC3 model is more appropriate than the CCSD(T) model for spin-spin coupling constants.

The SOPPA model has played an important role in the study of indirect spin—spin coupling constants, being first applied by Geertsen and Oddershede already in 1984.⁴⁵⁰ In the related SOPPA(CCSD) model the MP2 amplitudes are replaced by CCSD amplitudes, yielding more accurate coupling constants.^{226,227} The MP2 model has also been used for spin—spin coupling constants but with much less success.^{451,452}

Indirect nuclear spin—spin coupling constants depend sensitively on the molecular geometry and often have large vibrational corrections. Indeed, with the quality of today's electronicstructure calculations the differences between the calculated and the experimental coupling constants are often smaller than the vibrational corrections. For example, in calculations of the C=C coupling constant in acetylene it was estimated that the residual error does not exceed 2–3 Hz,⁴⁴⁷ whereas the vibrational correction is greater than 10 Hz.⁴⁵³ We also note the work by Åstrand et al.⁴⁵⁴ who found a large zero-point vibrational correction of -25 Hz using MCSCF theory in the HF molecule. In similar studies, Wigglesworth et al.⁴⁵³ applied SOPPA(CCSD) theory to C₂H₂ while Jordan et al.⁴⁵⁵ used EOM-CCSD theory to study the NH···N coupling in CNH:NCH. Because of the high cost of such calculations, DFT provides an attractive alternative to wave function methods for vibrational corrections.^{456,457} Indeed, an attractive approach is to calculate the equilibrium spin—spin coupling constants using high-level wave function theory but to generate their rovibrational corrections using Kohn—Sham theory.

5.2.3. Nuclear Spin–Rotation Constants. The spin I_K of nucleus K can also interact with the magnetic moment generated by the rotation of the molecule. This rotationally induced magnetic moment is a non-Born–Oppenheimer effect, arising from a slight decoupling of the electronic and nuclear rotation moments.⁴⁵⁸ The resulting interaction is one of the mechanisms responsible (together with the nuclear quadrupole coupling) for the hyperfine structure (i.e., splitting or shift of the spectral lines) of molecular rotational spectra. To describe this splitting quantum mechanically the following effective Hamiltonian was introduced by Flygare⁴⁵⁸

$$H_{\rm NSR} = \sum_{K} \mathbf{I}_{K}^{\rm T} \mathbf{C}_{K} \mathbf{J}$$
(594)

where **J** is the total rotational angular momentum of the molecule and C_K is the nuclear spin—rotation (NSR) tensor of nucleus *K*, which contains both a nuclear and an electronic contribution

$$\mathbf{C}_K = \mathbf{C}_K^{\mathrm{nuc}} + \mathbf{C}_K^{\mathrm{el}} \tag{595}$$

Whereas the nuclear contribution depends only on the nuclear framework

$$\mathbf{C}_{K}^{\mathrm{nuc}} = \alpha \gamma_{K} \mathbf{I}^{-1} \sum_{L \neq K} \frac{R_{LK}^{2} \mathbf{I}_{3} - \mathbf{R}_{LK} \mathbf{R}_{LK}^{\mathrm{T}}}{R_{LK}^{3}}$$
(596)

where I is the inertia tensor, the electronic contribution can be expressed as a second derivative of the electronic energy

$$\mathbf{C}_{K}^{\text{el}} = \left(\frac{\mathrm{d}^{2}E}{\mathrm{d}\mathbf{I}_{K}\mathrm{d}\mathbf{J}}\right)_{\mathbf{I}_{K}, \ \mathbf{J}=\mathbf{0}}$$
(597)

It is most efficiently calculated using analytic derivative techniques, although, in their early work, Oddershede and co-workers calculated NSR constants using the polarization propagator approach.^{459–461} Use of perturbation-dependent basis functions, known as rotational London orbitals and defined as

$$\chi_{\mu}(\mathbf{r}, \mathbf{B}, \mathbf{J}) = \exp[-i(\mathbf{A}_{\mu}^{\mathbf{B}} + \mathbf{A}_{\mu}^{\mathbf{J}}) \cdot \mathbf{r}]\chi_{\mu}(\mathbf{r})$$
(598)

where $\chi_{\mu}(\mathbf{r})$ is a standard Gaussian and where

$$\mathbf{A}_{\mu}^{\mathbf{B}} = \frac{1}{2} \mathbf{B} \times (\mathbf{R}_{\mu} - \mathbf{O}), \quad \mathbf{A}_{\mu}^{\mathbf{J}} = -\mathbf{I}^{-1} \mathbf{J} \times \mathbf{R}_{\mu}$$
(599)

is recommended to improve basis-set convergence. Note, however, that there is no gauge-origin problem in calculations of NSR constants as the origin for the orbital magnetic dipole operator is dictated by the point about which the molecule rotates, that is, the center of mass of the molecule. Use of rotational London orbitals in calculations of NSR tensors and rotational g tensors (vide infra) was proposed in 1996 by Gauss et al.⁴⁶² at the Hartree–Fock level of theory, later extended to the coupledcluster⁴⁶³ and MCSCF^{464,465} methods.

The recent review by Puzzarini et al. contains a detailed discussion of the computational requirements for NSR tensors.³ In short, the CCSD(T) approach is the method of choice for

accurate NSR-tensor calculations; moreover, despite the use of rotational London orbitals, large basis sets may still be needed for high accuracy. Accurate calculations of NSR tensors have proven important in assisting in the assignment of experimental rotational lines in various systems; in ref 466, for instance, such calculations have guided the revision of the experimental result for CF₂.⁴⁶⁷ Other examples are reviewed in ref 3.

The NSR tensor is used to determine an absolute NMR scale. The need for an absolute NMR scale stems from the fact that, experimentally, only the chemical shift, that is, the difference in the magnetic shielding for a nucleus in two different chemical environments, is measured, preventing a direct comparison of experimental and computational results, see refs 468 and 469. An absolute shielding scale is established based on the following relation between the paramagnetic part of the shielding constant for a given nucleus and the electronic part of the NSR tensor

$$\boldsymbol{\sigma}_{\text{para}}^{K} = -\frac{m_{\text{p}}}{3m_{\text{e}}g_{K}}\mathbf{I}\mathbf{C}_{K}^{\text{el}}$$
(600)

Experimental determination of the absolute chemical-shielding constant thus consists of several steps, where the only purely experimental one is measurement of the molecular rotational spectrum and extraction of the NSR tensor, C_{K_i} for each nucleus K from its hyperfine structure.⁴⁶⁹ By subtracting rovibrational contributions from C_{K_i} the purely electronic part C_{K}^{el} is obtained, which in turn can be converted to the paramagnetic shielding σ_{para}^{K} . Adding the diamagnetic contribution σ_{dia}^{K} from accurate quantum-chemical calculations and correcting for rovibrational and temperature effects, the experimental absolute shielding constant for each nucleus can be determined and compared with the corresponding purely computational results, see, for instance, refs 366 and 470–473.

5.3. Electron Paramagnetic Resonance

The EPR effective spin Hamiltonian including contributions from the nuclear spins and ignoring contributions from the nuclear quadrupole moments may be written as⁴⁷⁴

$$H^{\text{EPR}} = H^{\text{ex}} + H^{\text{sZ}} + H^{\text{ZFS}} + H^{\text{hf}} + H^{\text{nZ}}$$
 (601)

Like the NMR effective Hamiltonian in eq 588 the EPR Hamiltonian contains no reference to the electronic structure of the molecule. The different contributions to the EPR Hamiltonian thus describe phenomenologically the interactions present in the molecule.

The first term H^{ex} in eq 601, the exchange contribution, accounts for the difference in energy between electronic states of different multiplicity, and the Hamiltonian

$$H^{\rm ex} = -2J\mathbf{s}_1 \cdot \mathbf{s}_2 \tag{602}$$

is often referred to as the Heisenberg Hamiltonian. The study of spin states in molecular complexes is an active research field in which correlated wave function methods play an important role because of the strong multireference character of intermediate-spin complexes. Although such spin states can be determined with the spin-flip coupled-cluster method of Krylov,²⁰⁰ most ab initio calculations are performed by calculating the energy differences between explicitly optimized states using multireference methods.^{475–478} As this review focuses on quasi-energy response methods we do not consider this contribution to the EPR effective spin Hamiltonian any further, referring instead to the review by Neese.⁷

The second contribution to the Hamiltonian in eq 601 is the electron spin-Zeeman interaction, describing the interaction

between the magnetic moment of the electron with the external magnetic induction

$$H^{\rm sZ} = \mu_{\rm B} \mathbf{S}^{\rm T} \mathbf{g} \mathbf{B} \tag{603}$$

where **g** is the electron *g* tensor and **S** the total effective spin of the system. For a free electron the strength of this interaction is determined by the free-electron *g* factor of eq 65, yielding $\mathbf{g} = g_e \mathbf{I}_3 = 2\mathbf{I}_3$, see eq 28. However, electrons in a molecule experience a local magnetic field, arising from a partial shielding (or deshielding) by the other electrons in the molecule, which leads to a shift relative to g_e

$$\mathbf{g} = g_{\mathbf{e}}\mathbf{I}_3 + \Delta \mathbf{g} \tag{604}$$

We note that **g** in general is a nonsymmetric matrix.

The third contribution to the EPR spin Hamiltonian in eq 601 is the spin-spin-interaction operator

$$H^{\rm ZFS} = S^{\rm T} DS \tag{605}$$

which gives rise to the zero-field splitting of the EPR spectrum, describing the dipole interaction between the spin magnetic moments of unpaired electrons. This contribution is thus only present in states with more than one unpaired electron, leading to a splitting of the spin sublevels of a given spin state.

The interactions of the spin magnetic moments of the electrons with the nuclear magnetic moments give rise to hyperfine structure in the EPR spectrum described by the operator

$$H^{\rm hf} = \sum_{K} \left(\mathbf{A}_{K}^{\rm fc} \mathbf{S}^{\rm T} \mathbf{I}_{K} + \mathbf{S}^{\rm T} \mathbf{A}_{K}^{\rm sd} \mathbf{I}_{K} \right)$$
(606)

The isotropic first part has the form of a contact interaction and measures the spin density at a given nucleus, whereas the anisotropic second part corresponds to the dipolar interaction of the electron spin magnetic moment and the nuclear magnetic moment. The last term in eq 601 is in general not observable in EPR spectroscopy because of the increased line widths due to the presence of the unpaired electrons. The nuclear Zeeman interaction H^{nZ} corresponds to the last term in eq 65. It is analogous to the interaction included in the NMR effective Hamiltonian eq 588, where instead the shift is thought of as arising with respect to the bare nuclear *g* factor as described by the shielding constant.

5.3.1. Electronic *g* **Tensors.** For molecules with a single open-shell electronic doublet state and high-spin radicals in the strong-field limit the energy difference between the eigenvalues of the effective EPR Hamiltonian can be written as⁴⁷⁴

$$\Delta E = \mu_{\rm B} \sqrt{\mathbf{B}^{\rm T} \mathbf{G} \mathbf{B}} \tag{607}$$

where we introduced the symmetric *g* tensor $\mathbf{G} = \mathbf{g}\mathbf{g}^{\mathrm{T}}$, which contains both the orbital and the electron spin contributions that contribute to the effective spin and thus correspond to the quantity measured in experiment. Because the unperturbed state is degenerate, degenerate perturbation theory needs to be employed for calculation of the electronic *g* tensor.^{479,480}

In a nonrelativistic treatment the electronic *g* tensor reduces to the free-electron *g* value. All corrections to the electronic *g* tensor are therefore relativistic in origin. Furthermore, in a relativistic two- or four-component framework the electronic *g* tensor can be calculated as an expectation value of the electronic spin density, making its calculation fairly straightforward.^{481–483} By contrast, for calculations that start from a nonrelativistic

reference wave function the leading order relativistic corrections are accounted for by perturbation theory. To order α^2 in the perturbation we obtain the following correction to the freeelectron *g* tensor

$$\Delta \mathbf{g} = \Delta \mathbf{g}^{\text{SO}} + \Delta \mathbf{g}^{\text{RMC}} + \Delta \mathbf{g}^{\text{GC}} + O(\alpha^4)$$
(608)

where the spin-orbit, relativistic mass-correction, and gaugecorrection contributions are given by

$$\Delta g_{\alpha\beta}^{\rm SO} = \frac{2}{\langle S_z(\text{max}) \rangle} \left\langle (\mathrm{d}H_z^{\rm orb}/\mathrm{d}\mathbf{B})_\alpha; H_{\rm so, \beta} \right\rangle \right\rangle \tag{609}$$

$$\Delta g_{\alpha\beta}^{\rm RMC} = \frac{2}{\langle S_z(\max) \rangle} \langle 0 | (dH_{\rm MVBs}/d\mathbf{B})_{\alpha\beta} | 0 \rangle \tag{610}$$

$$\Delta g_{\alpha\beta}^{\rm GC} = \frac{2}{\langle S_z(\max) \rangle} \langle 0 | (dH_{\rm Bs}/d\mathbf{B})_{\alpha\beta} | 0 \rangle \tag{611}$$

where H_z^{orb} is the orbital part of the Zeeman operator in eq 65, H_{MVBs} is the diamagnetic part of the mass—velocity operator in eq 74, and H_{Bs} is the diamagnetic part of the spin—orbit operator in eq 75. $\langle S_z(\text{max}) \rangle$ corresponds to the maximum spin projection on the *z* axis for the given spin state. The spin—orbit contribution $\Delta \mathbf{g}^{\text{SO}}$ usually dominates, representing the interaction of the spin—orbit operator H_{so} with the orbital part H^{B} of the Zeeman Hamiltonian H_z in eq 65. For more details, see refs 40 and 484–486. Higher order relativistic corrections have also been derived⁴⁸⁰ and shown to be important in oxo-molybdenium(V) and oxo-tungsten(V) complexes.⁴⁸²

Most calculations of electronic *g* tensors employ spin-unrestricted models for which conventional response theory can be used to calculate the relevant quantities from a nonrelativistic wave function, see refs 487 and 488. Spin-restricted models have also been applied, but it is then important to ensure that the triplet operators acting on the open-shell reference state properly take all possible excitations into account.⁴⁸⁰

Multireference CI and MCSCF wave functions have been used to calculate the g tensor of small molecules, see, for example, refs 489-495. In recent years there has been increasing interest in the calculation of electronic g tensors at the ab initio level of theory. Neese introduced spectroscopy-oriented multireference CI (SORCI) theory^{496,497} using a sum-over-states approach. A sum-over-states approach based on the spin-orbit-restricted active space-state interaction (SO-RASSI) model⁴⁹⁸ was introduced by Bolvin, 479,499,500 who showed that the invariance of the Zeeman Hamiltonian under symmetry transformations and its continuous change with the variations of the parameters of the system (geometry and crystal field) lead to a unique determination of the elements of the g tensor, in contrast to previous assumptions.⁵⁰¹ The approach of Bolvin, in which the spin-orbit interaction has been applied to the electronic states prior to evaluation of the electronic g tensor, was compared at the multistate CASPT2 level^{502,503} to the approach of Neese and Solomon, where an effective Hamiltonian approach is used to include spin-orbit effects to second order.⁵⁰

Very recently, Gauss, Kállay, and Neese presented a coupledcluster linear-response approach for calculating electronic *g* factors for arbitrary excitation levels, based on an unrestricted Hartree–Fock reference wave function.⁵⁰⁵ This development allows for a critical assessment of the ability of various Kohn– Sham exchange-correlation kernels to model the effect of correlation on electronic *g* tensors. In approximate theories the computed electronic g tensor will in general be origin dependent, since it depends on the external magnetic field. In exact theory the origin dependence of the orbital magnetic dipole operator is canceled by the origin dependence of the gauge-correction term. This means that in approximate calculations care should be exercised to ensure gauge-origin independence, for instance, using London orbitals.^{481,483,505,506} In general, however, electronic g tensors appear to be less sensitive than other properties involving magnetic-field perturbations to the choice of gauge origin (see, for instance, ref 507).

5.3.2. Zero-Field Splittings. In degenerate perturbation theory the zero-field splitting of a spin-degenerate energy level ${}^{2S+1}E_0$ is evaluated from the eigenvalues of the matrix

$$H_{ij}^{ZFS} = \langle^{2S+1} \Psi_{0}^{i} | H_{ss} |^{2S+1} \Psi_{0}^{j} \rangle - \sum_{n\lambda} \sum_{k} \frac{\langle^{2S+1} \Psi_{0}^{i} | H_{so} |^{\lambda} \Psi_{n}^{k} \rangle \langle^{\lambda} \Psi_{n}^{k} | H_{so} |^{2S+1} \Psi_{0}^{j} \rangle}{{}^{\lambda} E_{n} - {}^{2S+1} E_{0}}$$
(612)

where ${}^{\lambda}\Psi_{n}^{k}$ is a zero-order wave function of spin multiplicity λ and spin projection k. The zero-field splitting, which occurs only in molecules with two or more unpaired electrons, has two distinct contributions. The first contribution in eq 612 is an expectation value of the two-electron spin-spin dipole-dipole coupling operator of eq 68, whereas the other term is the indirect interaction between the spin-magnetic moments of two electrons, mediated by the one- and two-electron contributions of the spin-orbit operator in eq 67. In most cases the spin-orbit contribution dominates the zero-field splitting, in particular, if heavier elements are involved. For transition-metal complexes the spin-orbit effect, often dominated by a few close-lying electronic states, can be included in a limited sum-over-states expansion or using ligandfield theory.⁵⁰⁸ For molecules with weak spin-orbit interactions such as triplet-spin organic molecules, the electron spin-spin interaction may dominate the zero-field splitting.³⁸ However, in transition-metal complexes the indirect spin-spin coupling has also been shown to be important in some cases.⁵⁰⁹

The perturbation-theory approach of Neese and Solomon has also been applied to the calculation of zero-field splittings using a sum-over-states methodology to calculate the linear-response contribution to the zero-field splitting from CI-like wave functions.⁵⁰⁴ A linear-response approach for calculating zerofield splittings, also including the contributions from the direct spin—spin dipolar interaction was presented by Vahtras et al.,³⁸ later extended to include solvent effects.^{510,511} Ganyushin et al. demonstrated that the resolution of the identity approximation could be efficiently used to reduce the four-index spin—spin interaction integrals to three-index integrals, facilitating highly correlated calculations on large systems.⁵¹²

The group of Neese has been active both in developing efficient methodology for calculating zero-field splittings for multireference wave functions^{39,513} and in providing benchmarks for DFT approaches.^{509,514,515} Benchmark studies using CASPT2/NEVPT2 methods^{516,517} and multireference wave functions⁵¹⁸ have been presented in recent years. We note that the calculation of zero-field splittings in DFT has been controversial, with different expressions for the zero-field splitting constants having been proposed;^{509,513,519} this controversy was recently resolved by Schmitt et al.⁵²⁰

5.3.3. Hyperfine Coupling Tensors. The EPR hyperfine splitting constants

$$\mathbf{A}_{K} = A_{K}^{\mathrm{fc}} \mathbf{I}_{3} + \mathbf{A}_{K}^{\mathrm{sd}}$$
(613)

of $H^{\rm hf}$ in eq 606 couple the electron and nuclear spins, providing detailed information about the structure of radicals, in much the same manner as shielding constants do in NMR spectroscopy. The isotropic contribution to the hyperfine interaction $A_K^{\rm fc}$ and the anisotropic contribution $\mathbf{A}_K^{\rm sd40}$ in eq 613 arise from the electron–nuclear Fermi-contact and the spin–dipole operators in $H_{\rm ss}$ of eq 68, respectively. Both contributions are simple expectation values of the respective operators

$$A_{K}^{fc} = -\frac{4\pi}{3} g_{K} g_{e} \mu_{B} \mu_{N} \alpha^{2} \frac{1}{\langle S_{z}(\max) \rangle} \sum_{pq} \langle \phi_{p} | \delta(\mathbf{r}_{K}) | \phi_{q} \rangle \langle 0 | T_{pq}^{l, 0} | 0 \rangle$$

$$(614)$$

$$\mathbf{A}_{K}^{\mathrm{sd}} = -\frac{1}{2} g_{K} g_{e} \mu_{\mathrm{B}} \mu_{\mathrm{N}} \alpha^{2} \frac{1}{\langle S_{z}(\mathrm{max}) \rangle} \sum_{pq} \langle \phi_{p} | \frac{3 \mathbf{r}_{K} \mathbf{r}_{K}^{\mathrm{T}} - r_{K}^{2} \mathbf{I}_{3}}{r_{K}^{5}} | \phi_{q} \rangle \langle \mathbf{0} | \mathbf{T}_{pq} | \mathbf{0} \rangle$$

$$(615)$$

where the expectation values of the triplet excitation operators in eqs 98 and 101 are elements of the triplet density matrix. There are also relativistic corrections to the hyperfine coupling from the diamagnetic operator $H_{\rm Ms}$ in eq 76; we refer to refs 521 and 522 for more details.

The EPR hyperfine coupling is governed by triplet perturbations. Thus, as for the *g* tensor, it is common to use unrestricted wave functions for describing the reference state, in which case the hyperfine interactions can be calculated as regular expectation values, for example, from coupled-cluster wave functions in an orbital-relaxed formulation.^{523,524}

Correlated methods were presented early for the study of hyperfine interactions using, for instance, symmetry-adaptedcluster CI (SAC-CI) theory^{525,526} and MRCISD theory.^{527–529} Kossmann and Neese recently presented a correlated approach for calculating hyperfine coupling constants in larger molecules based on the orbital-optimized SCS-MP2 method.⁵³⁰ Orbital optimization significantly reduces spin contamination, whereas spin-component scaling⁵³¹ improves the dipolar hyperfine coupling constants relative to the orbital-optimized MP2 method.

Hyperfine coupling constants can also be calculated from spin-restricted wave functions, letting the system respond in an unrestricted manner to the triplet perturbation.⁵³² Within such a restricted—unrestricted approach first-order properties are calculated as a restricted expectation value but with an unrestricted correction for the response to the triplet perturbation. First implemented at the SCF and MCSCF levels of theory it has recently been extended to Kohn—Sham theory.⁵³³

In general, the most severe basis-set requirements in calculations of hyperfine coupling constants come from the isotropic hyperfine contribution, the anisotropic contribution being less demanding. We mention, in particular, that there are basis sets specifically designed for DFT calculations of hyperfine coupling constants (EPR-III).⁵³⁴ Alternatively, the Hiller–Sucher– Feinberg identity can be applied to the Fermi-contact operator to reduce basis-set requirements.⁵³⁵ This approach and its extensions have been explored by Rassolov and Chipman.^{536–538} and by Sundholm.⁵³⁹ There is a large computational activity on the study of hyperfine couplings, especially using DFT; see the review by Improta and Barone..⁵⁴⁰

5.4. Electric Multipole Moments

As discussed in section 2.2.5 the most important electric multipole moments are the electric dipole moment μ and the (traceless) quadrupole moment Θ . The molecular dipole moment is the first nonvanishing electric moment for neutral systems with an asymmetric charge distribution, whereas the quadrupole moment is the first nonvanishing moment of nonpolar systems. In the gas phase permanent dipole moments are usually determined from the shift in the rotational energy levels caused by an applied electric field in high-resolution microwave spectroscopy (the Stark effect^{541,542}) or from measurements of the dielectric constant.⁵⁴³ Electric quadrupole moments are typically obtained from measurements of an electric-field-gradient-induced-birefringence (EFGB) effect (anisotropy of the refractive index) known as the Buckingham effect.⁵⁴⁴ Alternatively, they can be deduced from the deflection of a molecular beam by an electric or inhomogeneous magnetic field or in an indirect manner from microwave Zeeman experiments.⁴⁵⁸

5.4.1. Electric Dipole Moments. Several benchmark studies of the dipole moment of small- and medium-sized molecules using high-accuracy methods, in particular, coupled-cluster theory, have appeared over the last 15 years, examining the basis-set and excitation-level requirements and the requirements for vibrational corrections.^{545–549} We also note a study of the divergence of Møller–Plesset theory of the dipole moments of BH and HF.^{550,551} The accuracy of the computed dipole moments with state-of-the-art coupled-cluster methods is comparable to the experimental accuracy, in a few instances, even superior to experiment, prompting a revision of experimental results.⁵⁴⁸

In a very recent benchmark study by Hellweg⁵⁵² the accuracy of dipole moments calculated from wave function methods based on second-order perturbation theory was investigated for ground and excited states, focusing on large systems, for which highly accurate methods were still out of reach.⁵⁵² The MP2, CC2, and ADC(2) methods were discussed, together with their SCS and scaled-opposite-spin (SOS) variants. It was concluded that an accuracy of about 0.2–0.1 D in the ground state (and about 0.3–0.2 D in the excited states) can be achieved with these approaches.

As reviewed in ref 3, dipole moments govern the intensity of rotational transitions in microwave spectroscopy. Theory can therefore help to predict rotational transitions and intensities. In general, reliable but not necessarily very accurate values are sufficient for such purposes. An interesting application of computed dipole moments is that presented in ref 553, where the theoretical values were used to determine the relative stability and abundance of the cis and trans forms of trisulfane and 1-oxatrisulfane, see also ref 3.

Accurate dipole moments are used in the investigation of interaction-induced properties, see section 5.7.2. For instance, the key role of the accuracy of the dipole-moment surface for prediction of the infrared transition intensities in the Ar–HF complex has been investigated by Jankowski and Ziolkowski.⁵⁵⁴

5.4.2. Electric Quadrupole Moments. Over the last two decades development of more accurate electronic-structure methods and advances in experimental techniques have re-kindled interest in the ab initio determination of molecular quadrupole moments.^{555–565} As an example, the quadrupole moment of N₂ determined by Graham and co-workers⁵⁶⁶ in 1998 was in noticeable disagreement with existing theoretical and experimental values. Taking advantage of newly developed high-accuracy methods for first-order molecular properties,

Halkier et al.⁵⁶⁷ performed a thorough investigation of the quadrupole moment of N₂, proposing a new reference value, $\Theta = (-4.93 \pm 0.03) \times 10^{-40}$ C m². Simultaneously, Coriani et al. identified the use of erroneous correction terms for the temperature-independent contribution in the experimental values as the main source of discrepancy between the theoretical and experimental values, and proposed a revised experimental dipole moment based on their CCSD value of the temperature-independent contribution to the EFGB.⁵⁵⁵ In 2003, a reinvestigation of the Buckingham effect in gaseous N₂ over a range of temperatures confirmed that the temperature-independent contribution to the effect is not negligible, yielding a new experimental value, $\Theta = -(4.97 \pm 0.16) \times 10^{-40}$ Cm²,⁵⁶² in excellent agreement with the ab initio value of Halkier et al. from 1998.⁵⁶⁷

5.5. Linear Response Properties

Many of the microscopic properties obtained from linear response theory are in the theory of electromagnetism defined by multipole expansions, that is, from the perturbative expansion of the real induced oscillating electric and magnetic multipole moments^{2,569,568} (in the Einstein summation convention)

$$\mu_{\alpha} = \alpha_{\alpha\beta}E_{\beta} + \frac{1}{\omega}\alpha'_{\alpha\beta}\dot{E}_{\beta} + \frac{1}{3}A_{\alpha,\beta\gamma}\nabla_{\beta}E_{\gamma} + \dots \quad (616)$$

$$\Theta_{\alpha\beta} = A_{\gamma,\ \alpha\beta} E_{\gamma} - \frac{1}{\omega} A'_{\gamma,\ \alpha\beta} \dot{E}_{\gamma} + C_{\alpha\beta,\ \gamma\delta} \nabla_{\gamma} E_{\delta} + \dots$$
(617)

$$m'_{\alpha} = \xi_{\alpha\beta}B_{\beta} + \frac{1}{\omega}\xi'_{\alpha\beta}\dot{B}_{\beta} + G_{\beta\alpha}E_{\beta} - \frac{1}{\omega}G'_{\beta\alpha}\dot{E}_{\beta} + \dots \quad (618)$$

In the following a few of the tensors entering the above expansions are discussed, due to the fundamental role they play in rationalizing a variety of physical and chemical phenomena. Because of its relation with the magnetizability $\xi_{\alpha\beta}$, the rotational g tensor will also be discussed below, even if it originates from a different mechanism, namely, the interaction of the rotationally induced magnetic moment with an external magnetic field.

5.5.1. Dipole Polarizabilities. Among the expansion coefficients introduced above, the electric dipole polarizability tensor

$$a_{\alpha\beta}(-\omega,\,\omega) = -\langle\langle\mu_{\alpha};\mu_{\beta}\rangle\rangle_{\omega} \tag{619}$$

is probably the best-known second-order property, being the leading term in the perturbative expansion of the (induced) dipole moment μ_{α} in the presence of a radiation field. The polarizability is used to rationalize, at the microscopic level, macroscopic observables such as the refractive index and the dielectric constant of a medium. It plays an important role in Raman spectroscopy and enters the temperature-dependent part of various birefringence effects (vide infra). The dipole polarizability gives information about molecular shape and charge complementary to that provided by the dipole moment and is important for characterization and elucidation of the molecular structure of various substances. Like the dipole moment, the dipole polarizability is a popular benchmark property for the many linear response methods that have been presented over the years. The numerous studies that have been carried out for the dipole polarizability have clarified the performance of the various computational approaches, see, for instance, refs 570 and 571. Moreover, high-accuracy ab initio dipole polarizabilities have been used to benchmark the performance of approximate Kohn-Sham exchange-correlation functionals.⁵⁷²

Owing to the importance of the polarizability itself through its connection to the refractive index and of the many observables derived from the polarizability, the literature on theoretical studies of polarizabilities is vast and would require a review of its own. For an up-to-date critical review of relevant literature of theoretical calculations of polarizabilities, we refer the reader to ref 573.

Apart from ground-state polarizabilities, excited-state polarizabilities can also be determined from response theory, for instance, as second residues of the cubic response function of the ground state, as done by Jonsson and co-workers at the Hartree—Fock^{574,575} and MCSCF¹⁰¹ levels of theory. For CCSD wave functions excited-state first-order properties were derived by Koch and Jørgensen¹⁵⁴ and by Stanton and Gauss,^{576,577} who also discussed the theory for static excited-state second-order properties. Alternatively, Hättig et al.⁵⁷⁸ presented an implementation based on an excited-state Lagrangian,⁵⁷⁹ which yields excited-state properties equivalent to those identified from ground-state double residues but with the artificial secular divergent terms removed, thereby remaining numerically stable in the limit of static external fields; the approach was applied to compute the dipole polarizabilities of the S₁ states of s-tetrazine and pyrimidine.

5.5.2. Magnetizabilities. The molecular magnetizability describes the magnetic moment induced in a molecule by an external magnetic field induction and how this induced moment may interact with the external field to yield an energy correction. It is computed as the second derivative of the electronic energy with respect to the components of an external magnetic field

$$\xi = \left. -\frac{\mathrm{d}^2 E}{\mathrm{d} \mathbf{B}^2} \right|_{\mathbf{B}=\mathbf{0}} \tag{620}$$

In the response-theory formalism this expression is equivalent to

$$\boldsymbol{\xi} = -\left\langle \left(\frac{\mathrm{d}^2 H_{\mathrm{BB}}}{\mathrm{d} \mathbf{B}^2} \right) \right\rangle - \left\langle (\mathbf{m}; \mathbf{m}^{\mathrm{T}}) \right\rangle_0 = \boldsymbol{\xi}^{\mathrm{dia}} + \boldsymbol{\xi}^{\mathrm{para}} \quad (621)$$

where the first contribution, known as the diamagnetic contribution, is the expectation value of the diamagnetic magnetizability operator in eq 71, differentiated twice with respect to the magnetic field. The second, paramagnetic contribution arises from the interaction of the magnetic dipole operator with itself. In closedshell systems, due to quenching of the magnetic moment, see eq 66, the magnetizability yields the first nonvanishing contribution to the energy in the presence of a static magnetic field

$$\Delta E(\mathbf{B}) = -\frac{1}{2} \mathbf{B}^{\mathrm{T}} \boldsymbol{\xi} \mathbf{B}$$
(622)

For closed-shell molecular systems the diamagnetic contribution usually dominates the magnetizability, making the molecule diamagnetic. For some closed-shell molecules with lowlying excited states it is the paramagnetic contribution that dominates, making the molecule paramagnetic. However, as discussed by Tellgren, Helgaker, and Soncini,⁵⁸⁰ all closed-shell molecules become diamagnetic in sufficiently strong magnetic fields.

As for other magnetic properties, calculation of the magnetizability is hampered by gauge-origin dependence. The most successful solution to this problem is the use of London orbitals.⁵⁸¹ When London orbitals are employed the paramagnetic contribution to the magnetizability is best defined as⁴⁶²

$$\boldsymbol{\xi}^{\text{para, LAO}}(\mathbf{O}) = \boldsymbol{\xi}^{\text{LAO}} - \boldsymbol{\xi}^{\text{dia}}(\mathbf{O})$$
(623)

which is equivalent to the definition in eq 621 in the limit of a complete basis set.

In addition to ensuring gauge-origin independence, use of London orbitals ensures fast basis-set convergence in calculations of magnetizabilities, as shown, for instance, in the study of the magnetizability anisotropy of PF_3 in ref 582. Even in the relatively large aug-cc-pVQZ basis the magnetizability anisotropy computed without London orbitals has the wrong sign.

Magnetizabilities have been extensively studied at the Hartree– Fock and MCSCF levels of theory.^{52,581,583–589}An implementation of the gauge-origin-independent calculation of magnetizabilities (and rotational *g* tensors) at the general coupled-cluster level with London orbitals has been presented by Gauss et al.⁵⁹⁰ An implementation of the polarizable-continuum model (PCM) in its integral-equation formulation for calculation of the magnetizabilities of solvated molecules at the Hartree–Fock, MCSCF, and (hybrid) Kohn–Sham levels of theory with London orbitals was reported in ref 591. The theory of a hybrid quantummechanics/molecular-mechanics (QM/MM) approach for gaugeorigin-independent calculations of the molecular magnetizability using Hartree–Fock and Kohn–Sham theories was presented in ref 592.

These studies have shown that the isotropic magnetizability is not particularly sensitive to electron correlation⁴⁶⁴ (which contributes less than 3% in most cases) nor to nonelectronic effects such as zero-point vibrational corrections (with aromatic and antiaromatic ring systems as exceptions)^{593,594} or medium effects. By contrast, for the anisotropic magnetizability these effects may be sizable.^{591,594} Despite the insensitivity to correlation and other effects for the isotropic magnetizability, the agreement between experimental and theoretical results is quite poor. Since most of the experimental results are relative to some reference compound it has been suggested that inaccurate reference magnetizabilities in the experimental determination is the reason for the disagreement and that, for small molecules at least, calibration of measurements should be based on accurate computed values.⁵⁹⁵

Recently, Lutnæs et al.⁵⁹⁶ established an accurate benchmark data set of magnetizabilities (and rotational g tensors) of 28 molecules using the CCSD and CCSD(T) models with extrapolation techniques to obtain estimates of the basis-set-limit quantities and used this set to examine the performance of Kohn–Sham theory for a wide variety of exchange-correlation functionals. None of the functionals examined proved competitive with the CCSD or CCSD(T) methods. The coupled-cluster results were also compared with the results of density-functional calculations constrained to give the same density, and the importance of current dependence in exchange-correlation functionals was discussed in light of this comparison.

The theory for the frequency-dependent magnetizability remains unresolved. A gauge-origin-independent theory for the frequency-dependent magnetizability for exact states has been proposed by Raab and de Lange.⁵⁹⁷ Although origin independent, the derivation of the frequency-dependent magnetizability is based on ad hoc (but physically reasonable) conditions to be fulfilled rather than on exact conditions. An apparently very similar expression can be obtained by considering the constitutive relations for the material constants,⁵⁹⁸ although we note that the definition of the inverse permeability in this case is not unique.⁵⁹⁹ Independent of the formalism, for approximate calculations also the frequency-dependent magnetizabilities require the use of London atomic orbitals. Krykunov et al. started from the expressions of Raab and de Lange⁵⁹⁷ and calculated static and dynamic linear magnetic responses in approximate time-dependent DFT (TDDFT).⁶⁰⁰

5.5.3. Rotational *g* **Tensors.** We already noted that molecular rotation leads to an induced magnetic moment, as described in section 5.2.3. The rotational *g* tensor arises from the interaction of this rotationally induced magnetic moment with an external magnetic field and is often referred to as the rotational Zeeman effect.⁴⁵⁸ The rotational *g* tensor is closely related to the molecular magnetizability in the nonrelativistic description of magnetic interactions,⁴⁵⁸ being governed by the paramagnetic part of the magnetizability

$$\mathbf{g} = -4m_{\mathrm{p}}\boldsymbol{\xi}^{\mathrm{para}}(\mathbf{R}_{\mathrm{CM}})\mathbf{I}^{-1} + \mathbf{g}^{\mathrm{nuc}}$$
(624)

where we used the definition of the paramagnetic magnetizability in eq 623 with the center of mass as gauge origin. The nuclear contribution to the rotational *g* tensor is given by

$$\mathbf{g}^{\text{nuc}} = \frac{1}{2\mu_{\text{N}}} \sum_{K} Z_{K} [R_{K}^{2} \mathbf{I}_{3} - \mathbf{R}_{K} \mathbf{R}_{K}^{\text{T}}] \mathbf{I}^{-1}$$
(625)

where the nuclear positions \mathbf{R}_{K} are given relative to the center of mass of the molecule.

Because of its close connection to the magnetizability the rotational *g* tensor inherits many of the computational characteristics of the magnetizability, including its gauge-origin dependence and sensitivity to the choice of basis set. Gauss et al.⁴⁶² demonstrated that, by introducing rotational London orbitals, a computationally efficient scheme for calculation of rotational *g* tensors is obtained, where basis-set and gauge-origin problems are largely removed. An extreme case of the strong basis-set dependence observed for the rotational *g* tensor was provided by Ruud and Helgaker⁵⁸² for PF₃, for which the g_{\perp} values obtained with and without rotational London AOs in the aug-cc-pVDZ basis are of the same magnitude but of opposite sign. The results obtained in this basis with London AOs are within 4% of the basis-set limit for this molecule, in marked contrast to the non-London results.

Unlike the magnetizability, rotational *g* tensors can be determined with high accuracy in molecular-beam⁶⁰¹ and microwave Zeeman experiments.⁴⁵⁸ These experiments thus serve as an important source of high-quality data against which accurate ab initio methods can be benchmarked. Cybulski and Bishop presented two early studies of correlated calculations of rotational *g* tensors of diatomic molecules at the MP2 and the linearized-CCD levels of theory.^{471,602} Large (but non-London) basis sets were used to ensure convergence and vibrational corrections were applied to get results within the narrow experimental error bars.

Sauer and co-workers presented a number of studies of the rotational *g* tensor of small molecules using the SOPPA method^{603–605} and more recently the SOPPA(CCSD) method.^{606–609} In many of these studies special attention has been given to the vibrational and temperature dependence of the rotational *g* tensors.⁴⁶¹ Enevoldsen et al.⁶¹⁰ presented calculations of the rotational *g* tensor at the relativistic four-component Dirac–Coulomb Hartree–Fock level of theory using the same definition of the rotational *g* tensor at the relativistic and nonrelativistic levels of theory. We note, however, that a relativistic treatment of the induced rotational magnetic moments has not yet been presented, and it is unclear whether the direct connection between the magnetizability and the rotational *g* tensor also holds in a relativistic framework.

With introduction of rotational London AOs⁴⁶² larger molecules came within the reach of wave function-based calculations, and highly accurate MCSCF studies of rotational *g* tensors have been presented by Ruud and co-workers, ^{454,464,587,593,611,612} including studies of the rotational *g* tensor of electronically excited states. ^{613,614} Gauss et al. ⁵⁹⁰ implemented the calculation of rotational *g* tensors at the coupled-cluster level of theory for arbitrary excitation levels using the string-based approach of Kállay and Surjan.²⁷⁸ This implementation has been used to provide high-level theoretical data for rotational *g* tensors (and magnetizabilities) that have served as a benchmark set for exploring the suitability of different exchange-correlation functionals in calculations of magnetic properties.^{596,615}

An important application of rotational g tensors is for determination of molecular structures. It can be shown^{3,458,616} that the leading order electronic correction ΔB_{el} to the rotational tensor B_e of a molecule is proportional to the rotational g tensor

$$\Delta \mathbf{B}_{\rm el} = \frac{m_{\rm e}}{m_{\rm p}} \mathbf{g} \mathbf{B}_{\rm e} \tag{626}$$

The importance of this correction varies from molecule to molecule. For HF, N₂, and HOF it is 1-2 orders of magnitude smaller than the vibrational correction, whereas it is sizable and nonnegligible for BH, CH⁺, and SiC₃.^{3,617} We refer to the recent review by Puzzarini et al.³ for a detailed discussion of this correction to rotational constants.

5.5.4. Optical Rotation Tensors. The trace of the mixed magnetic–electric G' tensor is proportional to the magnitude of the specific optical rotation $[\alpha]_{\omega}$ in (isotropic) chiral systems^{618–620}

$$G_{\alpha\beta}^{'} = i\langle\langle m_{\alpha}, \mu_{\beta} \rangle\rangle_{\omega}, \ [\alpha]_{\omega} \propto \sum_{\alpha} G_{\alpha\alpha}^{'}$$
 (627)

Calculations of the G' tensor are routinely used to help in the assignment of the absolute configuration of chiral species, see, for instance, refs 619 and 621–623. Owing to the presence of the magnetic dipole operator, determination of optical rotation is hampered by the problem of gauge-origin dependence. Thus, even if this tensor is easily accessible from the linear response function by replacing one electric dipole operator in the electric dipole polarizability with the magnetic dipole operator special care must be exercised to remove the unphysical gauge-origin dependence. For the variational Hartree-Fock, Kohn-Sham, and MCSCF models, translational invariance of the optical rotation (i.e., the trace of G' but not of the individual tensor components) can be achieved using London orbitals.^{54,107,624,625} London orbitals are therefore routinely employed for optical-rotation calculations at the Hartree-Fock and Kohn-Sham levels of theory. Alternatively, the velocity gauge can be used in such calculations.626

The first CCSD calculations of optical rotation were performed by Ruud and Helgaker⁶²⁷ as part of a benchmark study of Kohn–Sham exchange-correlation functionals. However, a fundamental problem remains concerning gauge-origin independence of optical rotation for the truncated coupled-cluster model and other models that do not satisfy the hypervirial relation; for such models, use of London orbitals does not ensure invariance of the optical rotation.^{155,619,628} As a solution, Pedersen et al. proposed to compute the optical rotation instead from the velocity-gauge expression, subtracting its static limit from the G' tensor.⁶²⁹ Their approach avoids use of London orbitals and allows use of the same velocity-gauge formula for variational and nonvariational electronic-structure models. Moreover, the basis-set convergence in their formulation is similar to that of the conventional length-gauge method with London orbitals.

Crawford and co-workers have been particularly active in the area of determination of chiroptical properties using coupledcluster (and Kohn–Sham) linear response methods.^{208,619,630–636} In a study similar to that of Ruud et al.,⁶³⁷ Crawford and Stephens compared the performance of the CCSD and Becke-3parameter–Lee–Yang–Parr^{638,639} (B3LYP) models (at the sodium D line) for 13 molecules, observing a good agreement (in sign) except for norbornanone.⁶³⁵ At lower frequencies the B3LYP and CCSD models often differ substantially owing to the more accurate prediction of excitation energies and rotatory strengths by the latter.

Russ and Crawford²⁰⁸ presented a new scheme for constructing localized correlation domains for reducing the scaling of coupled-cluster response calculations for optical rotation. The approach exploits an atom-based decomposition of the coupledperturbed Hartree-Fock response to both external electric and magnetic fields to select the correlation domain of each occupied orbital, as an extension of their earlier work on calculations of dipole polarizabilities using local coupled-cluster approaches.²⁰⁵ Applying their domain-selection scheme to a series of chiral molecules (including pseudolinear structures, cage-like structures, and aromatic rings) the authors found that the crossover points between the canonical- and the local-correlation approaches are larger than for the conventional scheme of Boughton–Pulay domains,⁶⁴⁰ in agreement with their earlier analysis for dipole polarizabilities. Localization errors are reasonably small (a few percent) for pseudolinear structures with domain sizes of 6-8 atoms, whereas cage-like molecules are significantly more problematic, requiring natural domain sizes of 10 or more atoms to obtain the most reliable localization errors.

For the reliable prediction of optical rotation, inclusion of vibrational, conformational, and solvent corrections is often crucial.^{620,641–648} An extreme case is methyloxirane,^{633,649–651} where vibrational effects must be taken into account to reproduce the experimental gas-phase optical rotation to within a few degrees and for which the vibrational corrections may even lead to a change in the sign of the optical rotation. In ref 633 Pedersen et al. presented gas-phase optical rotations calculated from coupled-cluster theory with zero-point vibrational corrections calculated using Kohn–Sham theory, concluding that it is more important to use high-level electronic-structure methods than to include zero-point vibrational corrections. For the importance of conformational flexibility, see, for example, refs 645, 652, and 653.

For optical rotation of nonisotropic (oriented) systems an additional tensor quantity must be accounted for in calculations of the optical rotation, namely, the dipole, quadrupole polarizability tensor²

$$A_{\alpha, \beta\gamma} = -\langle \langle \mu_{\alpha}, \Theta_{\beta\gamma} \rangle \rangle_{\omega}$$
(628)

obtained by replacing one dipole operator with the quadrupole operator in the linear response function for the dipole polarizability. The $A_{\alpha,\beta\gamma}$ and $G'_{\alpha\beta}$ tensors are also important for other

tensor	frequency matching	NLO process	acronym
$\beta_{lphaeta\gamma}(\omega_{lpha};\omega_{eta},\omega_{\gamma})$	$\omega_{\beta} = \omega_{\gamma} = \omega$	second-harmonic generation	SHG
	$\omega_{\beta} = \omega, \omega_{\gamma} = 0$	electro-optic Pockels effect	EOPE
	$\omega_{\beta} = -\omega_{\gamma} = \omega$	optical rectification	OR
$\gamma_{\alpha\beta\gamma\delta}(\omega_{\alpha};\omega_{\beta},\omega_{\gamma},\omega_{\delta})$	$\omega_{\beta} = \omega_{\gamma} = \omega_{\delta}$	third-harmonic generation	THG
	$\omega_\beta=\omega,\omega_\gamma=\omega_\delta=0$	electro-optical Kerr effect	dc-Kerr
	$\omega_{\beta} = \omega_{\gamma} = \omega, \omega_{\delta} = -\omega$	intensity-dependent refractive index	IDRI
		degenerate four-wave mixing	DFWM
	$\omega_{\beta} = \omega_{\gamma} = \omega, \omega_{\delta} = 0$	electric-field induced SHG	ESHG
	$\omega_{\beta} = 0, \omega_{\gamma} = \omega, \omega_{\delta} = -\omega$	dc-optical rectification	dc-OR

optical effects, for instance, the Buckingham birefringence^{560,654} and for ROA.^{2,342}

5.6. Nonlinear Response Properties

Interest in molecular nonlinear properties has increased enormously in the last few decades, reflecting the growing importance of nonlinear optics in several fields of advanced technology.^{655–657} One example is the design of new optical materials and optical devices for data storage, since many nonlinear optical processes are mediated by the third-order nonlinear susceptibility, which corresponds to the second electric dipole hyperpolarizability. These properties have therefore always attracted much attention from theoreticians as well as experimentalists. On the one hand, when (accurate) experimental results are available they are often used as a test bed for benchmarking new computational approaches and approximations. On the other hand, given that their experimental determination is often difficult, theoretical predictions, in particular, when made with methods whose accuracy can be assessed with confidence, often help in the calibration of experimental measurements, sometimes even prompting the design of novel experimental set-ups.

5.6.1. Hyperpolarizabilities and Nonlinear Optics. The first and second dipole hyperpolarizabilities

$$\beta_{\alpha\beta\gamma}(\omega_{\alpha};\omega_{\beta},\omega_{\gamma}) = \langle \langle \mu_{\alpha};\mu_{\beta},\mu_{\gamma} \rangle \rangle_{\omega_{\beta},\omega_{\gamma}},$$

$$\omega_{\alpha} = -\omega_{\beta} - \omega_{\gamma}$$
(629)

$$\begin{split} \gamma_{\alpha\beta\gamma\delta}(\omega_{\alpha};\omega_{\beta},\,\omega_{\gamma},\,\omega_{\delta}) &= -\langle\langle\mu_{\alpha};\mu_{\beta},\,\mu_{\gamma},\,\mu_{\delta}\rangle\rangle_{\omega_{\beta},\,\omega_{\gamma},\,\omega_{\delta}},\\ \omega_{\alpha} &= -\omega_{\beta} - \omega_{\gamma} - \omega_{\delta} \end{split} \tag{630}$$

are the best known examples of molecular properties obtained from the quadratic and cubic response functions, respectively. Depending on the choice of frequencies ω_i associated with the perturbations, a wealth of nonlinear optical (NLO) phenomena⁶⁵⁵⁻⁶⁵⁷ may be addressed, see Table 4.

For NLO properties, highly accurate ab initio methods such as coupled-cluster theory have been applied primarily to small molecules, where there has been a significant interplay between theory and experiment, see ref 74 for a detailed review. This interplay has been essential for illuminating the role of various mechanisms in NLO processes and for identification and design of molecules, chromophores, and functional groups with specific properties. Indeed, as for the polarizability, the literature on theoretical (and experimental) studies of hyperpolarizabilities is vast, with several dedicated collective contributions. We refer the interested reader to the book edited by Papadopoulos, Leszczynski, and Sadlej¹⁰ for contributions covering various aspects related to NLO properties: current advances in the computation of the NLO properties of molecules, crystalline solids, and nanoparticles, methods employed to compute the properties of both microscopic and macroscopic forms of matter, studies of NLO properties of organometallic compounds, rotaxanes, glasses, Langmuir—Blodget films, gold, and silver nanoparticles, and more. Strategies for developing new NLO materials are discussed in connection with the hyper-Rayleigh scattering technique.

We here note that the four-component Hartree–Fock quadratic-response code of Norman and Jensen²⁶³ has been applied to compute the second-harmonic generation (SHG) hyperpolarizabilities of CsAg and CsAu. Kussman and Ochsenfeld extended their linear-scaling Hartree–Fock and Kohn–Sham code to calculation of first hyperpolarizabilities for molecules with a nonvanishing band gap, presenting calculations of the first hyperpolarizability of α -D-glucose polymers with up to 48 monomer units.¹³

5.6.2. Mixed Hyperpolarizabilities and Birefringences. Response theory has played an important role in the study of birefringences, that is, the optical anisotropy of the refractive index Δn for two different directions, often induced when the probing light (either polarized or unpolarized) passes through the sample in the presence of additional fields in specific geometrical setups with respect to the direction of propagation of the probing beam. Interest in these birefringences (and their absorptive counterparts, the dichroisms) has steadily increased during the last two decades, reflecting the steady progress in optics and detection techniques from the experimental side and rapid advances of methods and computational power from the theoretical side.⁶⁵⁸ Among the birefringences studied computationally in recent years are the linear birefringences induced by an externally applied magnetic induction field (Cotton–Mouton effect),^{658–662} by an electric field gradient (the Buckingham effect or EFGB),⁵⁴⁴ and by mixed electric and magnetic induction fields (Jones and magneto-electric birefringences),⁶⁶³ axial magnetochiral birefringence,^{664,665} and circular birefringence (Faraday rotation).^{666–669}

At a fixed pressure the general form of an optical anisotropy is⁶⁵⁸

$$\Delta n \propto A_0 + \frac{1}{T}A_1 + \dots \tag{631}$$

The temperature-dependent contribution A_1 arises from different mechanisms of molecular reorientation and involves the interaction of the fields with permanent electric or magnetic moments. The temperature-independent contribution A_0 arises from electronic

reorganization and involves different higher order tensors (hyperpolarizabilities and hypermagnetizabilities). Terms with more than a linear inverse temperature dependence are possible and usually connected to the presence of permanent magnetic dipole moments or higher order processes involving more complicated interactions between fields and multipoles. The temperature-independent terms are the only nonzero terms for systems of spherical symmetry.

At the molecular level the terms A_0 , A_1 , ... contain isotropic averages of molecular, both static and frequency-dependent, tensor properties such as permanent multipole moments, polarizabilities, and hyperpolarizabilities. In particular, we find for the Faraday rotation ϕ , usually discussed in terms of the Verdet constant $V(\omega)^{669-672}$

$$\phi = V(\omega)B_z \propto B_z \varepsilon_{\alpha\beta\gamma} \left(\alpha_{\alpha\beta,\gamma}^{'(m)} + \frac{1}{kT} m_{\gamma}^n \alpha_{\alpha\beta}^{'} \right)$$
(632)

$$\alpha_{\alpha\beta,\gamma}^{(m)} = i\langle\langle \mu_{\alpha}; \mu_{\beta}, m_{\gamma} \rangle\rangle_{\omega,0}$$
(633)

$$\alpha'_{\alpha\beta} = -i\langle\langle \mu_{\alpha}; \mu_{\beta}\rangle\rangle_{\omega} \tag{634}$$

for the Buckingham effect^{544,555,557,673}

$${}_{m}Q \propto b(\omega) + \frac{1}{kT} \left[\Theta_{\alpha\beta} \alpha_{\alpha\beta} - \mu_{\alpha} \left(A_{\beta, \ \alpha\beta} + \frac{5}{\omega} \varepsilon_{\alpha\beta\gamma} G'_{\beta\gamma} \right) \right]$$
(635)

$$b(\omega) = \frac{1}{15} \left(B_{\alpha\beta,\ \alpha\beta} - \mathcal{B}_{\alpha,\ \alpha\beta,\ \beta} - \frac{5}{\omega} \varepsilon_{\alpha\beta\gamma} J'_{\alpha,\ \beta,\gamma} \right) \quad (636)$$

$$B_{\alpha\beta,\ \gamma\delta} = \langle \langle \mu_{\alpha}; \mu_{\beta}, \Theta_{\gamma\delta} \rangle \rangle_{\omega,\ 0}$$
(637)

$$\begin{aligned}
\mathscr{B}_{a, \beta\gamma, \delta} &= \langle \langle \mu_{a}; \Theta_{\beta\gamma}, \mu_{\delta} \rangle \rangle_{\omega, 0} \\
&= \langle \langle \mu_{a}; \mu_{\delta}, \Theta_{\beta\gamma} \rangle \rangle_{0, \omega}
\end{aligned} (638)$$

$$J'_{\alpha, \beta, \gamma} = i\langle\langle \mu_{\alpha}; m_{\beta}, \mu_{\gamma} \rangle\rangle_{\omega, 0}$$
(639)

for the Cotton–Mouton effect^{659–662}

$${}_{m}C \propto \Delta \eta + \frac{1}{15kT} (3\alpha_{\alpha\beta}\xi_{\alpha\beta} - \alpha_{\alpha\beta}\xi_{\alpha\beta})$$
(640)

$$\eta_{\alpha\beta\gamma\delta} = -\langle\langle\mu_{\alpha};\mu_{\beta}, m_{\gamma}, m_{\delta}\rangle\rangle_{\omega, 0, 0} + \langle\langle\mu_{\alpha};\mu_{\beta}, \xi_{\gamma\delta}\rangle\rangle_{\omega, 0}$$
(641)

and for the magnetoelectric Jones effect⁶⁶³

$$n_{-45^{\circ}} - n_{+45^{\circ}} \propto 3G_{\alpha\beta\alpha\beta} + 3G_{\alpha\beta\beta\alpha} - 2G_{\alpha\alpha\beta\beta} - \frac{\omega}{2} \varepsilon_{\alpha\beta\gamma} \left(a'_{\alpha\beta\delta\delta\gamma} + a'_{\alpha\beta\delta\gamma\delta}\right) + \frac{1}{kT} \left[\mu_{\alpha} (3G_{\alpha\beta\beta} + 3G_{\beta\alpha\beta} - 2G_{\beta\beta\alpha}) - \frac{\omega}{2} \varepsilon_{\alpha\beta\gamma} \left(\mu_{\gamma} a'_{\alpha\beta\delta\delta} + \mu_{\delta} a'_{\alpha\beta\delta\gamma}\right)\right]$$
(642)

$$G_{\alpha\beta\gamma} = \langle \langle \mu_{\alpha}; m_{\beta}, m_{\gamma} \rangle \rangle_{\omega, 0} - \langle \langle \mu_{\alpha}; \xi_{\beta\gamma}^{\text{dia}} \rangle \rangle_{\omega}$$
(643)

$$G_{\alpha\beta\gamma\delta} = -\langle\langle\mu_{\alpha}; m_{\beta}, m_{\gamma}, \mu_{\delta}\rangle\rangle_{\omega, 0, 0} + \langle\langle\mu_{\alpha}; \xi_{\beta\gamma}^{dia}, \mu_{\delta}\rangle\rangle_{\omega, 0}$$
(644)

$$a'_{\alpha\beta\gamma\delta} = i\langle\langle \mu_{\alpha}; Q_{\beta\gamma}, m_{\delta}\rangle\rangle_{\omega, 0}$$
(645)

$$a'_{\alpha\beta\gamma\delta\varepsilon} = -i\langle\langle\mu_{\alpha}; Q_{\beta\gamma}, m_{\delta}, \mu_{\varepsilon}\rangle\rangle_{\omega, 0, 0}$$
(646)

Accurate theoretical investigations of birefringences have proven important in several instances, both in relation to the interplay between experiment and theory and in clarification of the fundamental theories of the effects. For instance, calculations of the Cotton–Mouton effect may be used for calibration of the experiment, 659,674 whose accuracy needs to be assessed with great precision if one hopes to be able to detect the magnetic birefringence of vacuum.^{674–678} State-of-the-art computations of the Buckingham birefringence have been used to revise the experimentally derived quadrupole moments of various molecules from single-temperature measurements and, in particular, identify inconsistencies in the experimentally derived quadrupole moment of N2.555,567 Calculations on CO557,560 and other species helped settle a discrepancy between the Buckingham-Longuet-Higgins⁶⁵⁴ and the Imrie-Raab⁶⁷⁹ theories of the linear birefringence induced in a gas of dipolar molecules by an electric-field gradient in favor of the former theory. A later revision brought the theories into agreement, confirming the validity of the original result obtained by Buckingham and Longuet-Higgins.680,681

Among the recent methodological advances that affect the ability to compute birefringences is the gauge-origin-independent formulation and implementation of the Verdet constant at the CCSD level of theory using London orbitals,⁶⁶⁸ where the Verdet constant was reformulated as a magnetic-field derivative of the dipole polarizability. A similar formulation was used in the AO-based Hartree—Fock (and Kohn—Sham) implementation by Kjærgaard et al.⁶⁶⁹ A quasi-energy formulation of Hartree—Fock and Kohn—Sham response theories enabled the temperature-independent term of the Buckingham effect.⁶⁸² and the Cotton—Mouton effect at nonzero frequencies¹⁰⁶ to be determined in a gauge-origin-independent manner. A relativistic extension of the latter formulation has recently been used to investigate the importance of relativity for the Buckingham effect of carbon dichalcogenides.²⁶⁴

5.6.3. Dispersion Coefficients. The dispersion (i.e., frequency dependence) of response functions in the nonabsorptive region is often accounted for by an expansion in powers of the frequency. For instance, the electric dipole polarizability is expanded according to the Cauchy series

$$\alpha(\omega) = \sum_{k=0}^{\infty} \omega^2 S(-2k-2)$$
(647)

where the dispersion coefficients S(-2k - 2) are known as Cauchy moments

$$S(-k) = \sum_{n \neq 0} \frac{f_{0n}^{\alpha\beta}}{\omega_{0n}^{k}}, \quad f_{0n}^{\alpha\beta} = 2\omega_{0n} \langle 0|\mu_{\alpha}|n\rangle \langle n|\mu_{\beta}|0\rangle$$
(648)

with $S(-2) = \alpha(0)$. Similar expansions can be introduced for the hyperpolarizabilities.

Hättig and co-workers^{683–685} derived analytic expressions for the dispersion coefficients of linear, quadratic, and cubic response functions using the time-averaged quasi-energy Lagrangian technique within coupled-cluster response theory for the CCS, CC2, and CCSD models, demonstrating that the dispersion coefficients are sensitive to basis-set effects and correlation treatment in a manner similar to that of static (hyper)polarizabilities. The analytic dispersion coefficients for frequencydependent properties are calculated from the derivatives of the properties with respect to their frequency arguments using Padé approximants to improve the results. A derivation and implementation of the analytic expressions for the second-order dispersion coefficients of second-, third-, and fourth-order properties in FCI linear, quadratic, and cubic response theory have been presented by Larsen et al.⁶⁸⁶

5.7. Interaction Properties

Many molecular properties are related to intermolecular interactions and can be evaluated using the general techniques of response theory. In the following we consider first evaluation of dispersion-interaction coefficients and next evaluation of interaction-induced properties of van der Waals complexes.

5.7.1. Weak Interactions and Long-Range Dispersion-Interaction Coefficients. At large separations the forces between two neutral systems are dispersive, arising from a weak correlation of the motion of the electrons in the two systems. The dispersion energy, the dominant contribution to the interaction energy at large intermolecular distances, can be computed from response functions. Denoting two closed-shell atoms by A and B the dispersion energy has the form

$$E^{AB} = \sum_{n} \sum_{m} \frac{|\langle \mathbf{0}_{A}\mathbf{0}_{B} | V^{AB} | n_{A}m_{B} \rangle|^{2}}{\omega_{n0}^{A} + \omega_{m0}^{B}}$$
(649)

which may be expanded in terms of dispersion coefficients as

$$E^{AB} = -\frac{C_6^{AB}}{R^6} - \frac{C_8^{AB}}{R^8} - \frac{C_{10}^{AB}}{R^{10}} + \dots$$
(650)

According to the Casimir–Polder formula, 687 the C_6 coefficient, for which accurate experimental results are available, can be computed from the integral

$$C_6^{AB} = \frac{3}{\pi} \int_0^\infty \alpha^A (-i\omega; i\omega) \alpha^B (-i\omega; i\omega) d\omega$$
(651)

which involves the dipole polarizability (linear response function) at imaginary frequencies (i ω). Higher order coefficients are similarly obtained from expressions involving higher order multipole polarizabilities.

Direct evaluation of the polarizability at imaginary frequencies in eq 651 by regular response theory may be cumbersome since it involves complex arithmetics, standard nonrelativistic quantumchemical codes being based on real arithmetics. One typical solution is to exploit the expansion formula for the polarizability in terms of the real Cauchy moments in eq 647, which has a radius of convergence in the complex plane equal to the lowest dipole excitation frequency. Evaluation of the C_6 dispersion coefficients in MCSCF linear response theory was presented by Fowler, Jørgensen, and Olsen in 1990.⁸⁷ An alternative strategy, proposed by Norman and co-workers,⁸⁸ is based on the complex polarization propagator, where the dynamical polarizabilities at imaginary frequencies are obtained directly from the complex polarization propagator and the C_6 coefficients are determined from the Casimir–Polder relation, see also refs 688 and 689 for Hartree–Fock and Kohn–Sham applications to C_{60} , sodium clusters, and *n*-alkanes.

In ref 231 Paidarova and Sauer used the SOPPA, SOPPA. (CCSD), CC2, and CCSD models to determine the dipole oscillator strength sum rules of the hydrogen halides HX (with X = F, Cl, Br, and I) and the C_6 dispersion coefficient for all pairs of interacting HX molecules by numerical integration of the Casimir–Polder formula. The dependence of the polarizabilities, their dispersion, and C_6 coefficients on the level of correlation and the dependence of the C_6 coefficients on the two intramolecular bond lengths were studied.

5.7.2. Interaction-Induced Properties. Coupled-cluster (response) methods have been extensively employed to determine intermolecular potential energies, interaction-induced first-order properties, and polarizability and hyperpolarizability surfaces of van der Waals complexes (dimers).^{690–694} Indeed, van der Waals dimers have received much attention during the last two decades, both experimentally and theoretically. One of the most important research objectives of the many studies that have appeared is to obtain a better understanding of the nature of pair interactions. Interaction-induced polarizabilities are an important source of information on intermolecular forces; therefore, much work has been carried out over recent years aimed at their accurate experimental determination.^{695–700}

From a theoretical point of view, evaluation of interactioninduced properties in van der Waals complexes is difficult, given that dispersion places substantial demands on the level of electron-correlation treatment and on the choice of basis set.^{661,690,692,701-703} Interaction-induced properties are usually computed according to

$$\Delta P(\mathbf{R}) = P_{AB}(S_{AB}|\mathbf{R}) - P_A(S_{AB}|\mathbf{R}) - P_B(S_{AB}|\mathbf{R})$$
(652)

where $P_{AB}(S_{AB}|\mathbf{R})$ is the property of the A–B dimer (at geometry **R**) and $P_A(S_{AB}|\mathbf{R})$ and $P_B(S_{AB}|\mathbf{R})$ are the properties of monomers A and B at the same geometry. All quantities are calculated in the dimer basis S_{AB} to reduce basis-set superposition errors.

A typical study of interaction-induced properties is determination of the effect of many-body collisions or, in macroscopic terms, of the density on some property P such as the refractive index or the EFGB of a real gas. To this end, the property is expressed in terms of virial expansions^{704,705}

$$P = A_P + B_P \rho + C_P \rho^2 + \dots$$
(653)

where ρ is the density and A_P , B_P , C_P , ..., are the first, second, third, ... virial coefficients. Usually only the first and second virial coefficients are considered; for the latter, a semiclassical expression of the form^{706,707}

$$B_P \propto 4\pi N^2 \int \Delta P(R_{AB}) R_{AB}^2 e^{-V(R_{AB})/kT} dR_{AB}$$
(654)

is often adopted,^{706,707} where R_{AB} is the distance from A to B, $V(R_{AB})$ is the interaction potential, and $\Delta P(R_{AB})$ the relevant interaction-induced property.

During the past decade, several coupled-cluster studies have been performed of the second dielectric and second refractivity virial coefficients, the Kerr and hyperpolarizability second virial coefficients, as well as the second virial coefficients for the Cotton—Mouton and Buckingham birefringences in various systems, see, for example, refs 661, 690, 691, 694, and 708.

5.8. Excited-State Energies, Properties, and Transition Properties

An attractive feature of response theory is the fact that excitedstate molecular properties, for example, excitation energies from the ground state and between excited states, (multiphoton) transition strengths, excited-state first-order properties, and excited-state structure parameters, may be calculated without the explicit optimization of the excited-state wave function. Vertical excitation energies are obtained as poles of the linear response function by solving a generalized eigenvalue problem, one-photon transition strengths such as UV oscillator strengths and ECD rotational strengths are obtained from the residues of the linear response function, while multiphoton strengths and excited-state (hyper)polarizabilities are accessible from residues of higher order response functions. Excited-state polarizabilities were briefly discussed in section 5.5.1; in the following, some more excited-state properties (including transition properties) are discussed.

5.8.1. Excitation Energies and One-Photon Transition Strengths. Implementations of response theory for (vertical) excitation energies have been reported at many levels of wave function theory: Hartree-Fock theory or the random-phase approximation (RPA), CI-singles (CIS) theory or the Tamm-Dancoff approximation (TDA), MCSCF theory, coupled-cluster theory (CC2,¹²¹ CCSD,¹⁵⁷ CC3,^{122,123} and CCSDR(3)¹⁸⁹), EOM-CC theory,^{126,181,182,203} propagator theories such as ADC-(2), ADC(3),^{233,236,709} SOPPA and SOPPA(CCSD),²²⁶ secondorder RPA (RPA(D)) theory,⁷¹⁰ and CIS(D) theory. The corresponding dipole transition strengths can be obtained from the residues of the linear response function (the dipole polarizability), available for most of these methods. The choice of the electronicstructure method for excitation energies and transition strengths depends on the problem at hand and the preferences of the user; we do not discuss these aspects here, referring instead to recent benchmark studies.^{230,711-713} We also refer to refs 714 and 715 for a discussion of the relation between the ADC approach, coupled-cluster linear response theory, EOM-CC theory, multireference coupled-cluster MRCC schemes, and the SAC-CI approach.

As an illustration we mention here a recent coupled-cluster study of the electronic gas-phase spectrum of glycine, alanine, related amines, and carboxylic acids by Osted et al.⁷¹⁶ The systems were chosen in order to investigate the origin of the common electronic excitations in amino acids, paying special attention to the valence excitation from the nonbonding lone pair on the CO oxygen atom to the antibonding π orbital ($n(O) \rightarrow$ $\pi^*(CO)$) and to the first Rydberg excitation from the nonbonding lone pair on the nitrogen atom $(n(N) \rightarrow 3s)$. Excellent agreement with all reliable experimental values was achieved. Predictions for vertical excitation energies were given for all molecules, including glycine and alanine, for which no gas-phase experimental results are available. Finally, calculations on protonated amino acids were presented, showing an isolation of the $n(O) \rightarrow \pi^*(CO)$ transition from higher lying states by as much as 1.9 eV for alanine. As another example of excitation energy studies, vertical excitation energies and transition dipole moments between excited electronic states were in ref 717 calculated for the *trans*-polyene series C_4H_6 to $C_{12}H_{14}$ to study formation of excited-state absorption spectra of these species, applying quadratic response theory in conjunction with Hartree-Fock and coupled-cluster theories. As a final example, Åstrand et al. performed an ab initio SOPPA investigation of the electronic spectra of azobenzene dyes, suggesting that the information obtained about spectra in this nonempirical manner may be useful for identifying promising diazo components for development of data-storage devices.⁷¹⁸

5.8.2. Electronic Circular Dichroism. Residues of the optical rotation tensor G' give access to electronic circular dichroism (ECD). The key molecular quantity is in this case the ECD rotational strength, which in the length gauge is given by

$$R_{n0} = \operatorname{Im}\langle 0|\boldsymbol{\mu}|n\rangle \cdot \langle n|\mathbf{m}|0\rangle \tag{655}$$

ECD studies based on wave function methods are still somewhat limited, the field being dominated by TDDFT. The status of the ab initio determination of ECD (and optical rotation) was reviewed a few years ago by Pecul and Ruud, 620 and by Crawford and coworkers. 619,636

The use of London orbitals for gauge-origin-independent ECD calculations was proposed by Bak et al. in 1995 for the variational Hartree—Fock and MCSCF methods, later extended to TDDFT by Pecul et al.⁷¹⁹ Alternatively, as for optical rotation, origin invariance of the ECD rotational strengths may be achieved using a velocity-gauge representation for the electric-dipole operator

$$R_{n0} = \frac{1}{\omega_{n0}} \operatorname{Re}\langle 0|\mathbf{p}|n\rangle \cdot \langle n|\mathbf{m}|0\rangle$$
(656)

Several modern ECD implementations utilize the velocity-gauge formulation, which, for variational models, becomes equivalent to its length-gauge counterpart in the complete-basis-set limit.

The lack of gauge invariance of the truncated coupled-cluster model affects calculation of ECD in the same manner as that of optical rotation discussed earlier.⁷²⁰ In particular, London orbitals do not suffice to ensure invariance. Pedersen and Koch proposed a reformulation of coupled-cluster theory that would allow gauge-invariant results to be obtained in coupled-cluster theory,^{721,722} although this proposal has not yet been implemented. The work of Pedersen et al. in ref 720 is the first ECD implementation and calculation at the CCSD level of theory. The authors considered gauge and origin invariance in the scalar rotational strength and rotational-strength tensor, the latter which provides the ECD intensity of oriented samples. Subsequently, EOM-CC (and DFT) studies of ECD have been carried out by Crawford and co-workers.^{636,723,724} Moreover, Diedrich and Grimme⁷²⁵ systematically investigated the ability of different quantum-chemical methods, TDDFT, CC2, MRMP2, and DFT/MRCI, to predict experimental ECD on a test suite of seven molecules containing a range of difficult chromophores and to three model systems for which accurate ab initio MRCI reference data were used for comparison.

5.8.3. Multiphoton Absorption and Dichroism. Twophoton absorption (TPA) and two-photon circular dichroism (TPCD)⁷²⁶ are further examples of optical phenomena accessible by response theory. We mention here a coupled-cluster study of formaldehyde, diacetylene, and water by Paterson et al.,⁷²⁷ in which the effect of triple excitations on TPA cross sections was determined for the first time. This study presented a detailed comparison of the coupled-cluster results with those obtained using Kohn–Sham theory with a variety of exchangecorrelation functionals. Best results were obtained with the Coulomb attenuating method B3LYP (CAM-B3LYP) functional,⁷²⁸ although it was noted that care must be exercised with diffuse Rydberg states. Nielsen et al.⁷²⁹ presented a detailed investigation of TPA cross sections of water, employing different series of correlation-consistent basis sets in combination with several coupled-cluster, CI, and Kohn–Sham models, finding it difficult to reach convergence. TPCD has so far only been investigated at the Hartree–Fock and TDDFT levels of theory, see, for example, refs 730–733. Calculation of three-photon absorption cross-sessions using response theory have been reported at the Hartree–Fock level in refs 734–736.

On the basis of an analysis of the first residues of coupledcluster response functions, Hättig et al.⁷³⁷ devised variational functionals from which the transition moments for *n*-photon excitations can be calculated as *n*th-order derivatives. Combining these functionals with variational perturbation theory, these authors developed a new approach for derivation of multiphoton transition moments, allowing them to utilize the full strength of variational perturbation theory directly rather than through residues. Coupled-cluster multiphoton transition moments derived by this approach were shown to be formally equivalent to those identified from the first residues of the ground-state response functions. The strength of the new approach was demonstrated by derivation of three- and four-photon transition moments.⁷³⁷

A similar approach, based on an analysis of the second residues of the coupled-cluster ground-state response functions, yielded general expressions for multiphoton transition moments between two excited states and for excited-state response functions in ref 579. The second residues were rewritten as derivatives of variational functionals, and computational efficient expressions were obtained in accordance with the 2n + 1 and 2n + 2 rules. For the excited-state response functions, secular divergencies were removed by reformulating the expressions identified from the second residues of the ground-state response functions as derivatives of an excited-state quasi-energy Lagrangian. Explicit expressions were given for coupled-cluster one-, two-, and threephoton transition moments between two excited states and for excited-state first-order properties and linear and quadratic response functions.

5.8.4. Magnetic Circular Dichroism. The single and double residues of the mixed dipole–dipole–magnetic-dipole quadratic response function have been shown to correspond to, respectively, the \mathcal{B} and \mathcal{A} terms of magnetic circular dichorism (MCD), the absorption counterpart of Faraday rotation discussed in section 5.6.2.^{96,669,738} The \mathcal{A} and \mathcal{B} terms can also be formulated as derivatives with respect to the strength of an external magnetic field of, respectively, the excitation energy^{96,739} and the dipole oscillator strength, ^{668,669} that is, as derivatives of the pole and residues of the magnetic-field-dependent dipole polarizability. In this way, the field-dependent contributions from London orbitals are more easily incorporated and gauge-origin independence ensured, even for the nonvariational coupled-cluster method.^{668,669} Most MCD methodological developments (in a response-function theory context) have taken place within TDDFT, except for the original MCSCF implementation, ^{668,740} both by Coriani and co-workers.

5.8.5. Excited-State Molecular Gradients and First-Order Properties. By combining analytic gradient techniques with response theory, excited-state molecular gradients can be calculated and hence excited-state potential-energy surfaces efficiently explored without explicitly calculating the excited-state wave function.

Analytic derivative techniques for the gradient and other properties of excited-state energy surfaces began to appear in the early 1990s, with the CIS implementation of Foresman et al. in 1992.⁷⁴¹ Formulas for the EOM-CCSD excited-state gradient were presented by Stanton in 1993⁷⁴² and subsequently implemented by Stanton and Gauss,^{576,577} whereas an alternative derivation and implementation were presented by Szalay.⁷⁴³ An extension to general coupled-cluster and CI models using string-based methods was given by Kállay and Gauss in 2004.⁷⁴⁴ Spin-flip EOM-CCSD gradients were presented by Levchenko et al.⁷⁴⁵ and molecular gradients for the similarity-transformed EOM-CC method by Gwaltney et al.⁷⁴⁶ The automated algebraic derivation of excited-state gradients for EOM-CC and similarity-transformed EOM-CC techniques was also discussed by Wladyslawski and Noojien in 2005.⁷⁴⁷

The CIS model has for a long time been considered the standard approach for excited-state optimizations of large systems.^{748,749} However, Köhn and Hättig¹⁷⁴ presented in 2003 a derivation and implementation of excited-state molecular gradients at the CC2 level of theory, employing the resolution of the identity approximation for the electron repulsion integrals. Adiabatic excitation energies, excited-state-structure constants, and vibrational frequencies could be calculated. Generalizations of the SCS and SOS modifications of MP2 perturbation theory to the CC2 model (termed SCS-CC2 and SOS-CC2) were discussed, and a preliminary implementation was presented in ref 299. In ref 750 a quartic-scaling algorithm for evaluating the analytical gradient of quasidegenerate SOS second-order perturbation corrections to the CIS energy method $(SOS-CIS(D_0))$ was presented, where the low-order scaling was attained using the resolution-of-the-identity approximation and the Laplace transform. The efficiency of the method was demonstrated by calculating the excited-state gradients of molecules of varying sizes. An implementation of the analytic excited-state gradients for the ADC(2) and CIS(D $_{\infty}$) models was presented in 2005 by Hättig.²³⁸ Owing to the non-Hermiticity of its Jacobian, the CC2 method appears to give a physically incorrect description of conical intersections between states of the same symmetry. This problem does not arise in ADC(2) theory, where a Hermitian secular matrix is used. An AO-based Lagrangian Hartree–Fock (and Kohn–Sham) implementation was recently presented by Coriani et al.³⁴⁵

Regarding excited-state first-order properties other than molecular gradients, Devarajan et al.⁷⁵¹ reported an investigation of the dipole moments of low-lying singlet and triplet excited states of ozone and the ozone cation radical, obtained using the Fockspace multireference-coupled-cluster (FS-MRCC) analytical linear response approach. A benchmark study of the accuracy of excited-state dipole moments of furan and pyrrole calculated in coupled-cluster theory has been presented by King⁷⁵² and compared with TDDFT results by Burcl et al.⁷⁵³

Kats et al.²¹⁰ presented a method for calculating transition strengths and first-order properties of singlet ground and excited states of extended molecular systems based on the CC2 model, with local approximations in the doubles part of the wave function and density fitting for the electron repulsion integrals. The Boughton–Pulay domains⁶⁴⁰ for local correlation were found to be inadequate for excited-state properties and a new scheme proposed.

Tellgren et al.⁷⁵⁴ reported an implementation of the secondorder residue of the quadratic response function in the fourcomponent Hartree–Fock approximation, from which firstorder properties of electronically excited states can be obtained. The scheme was used to compute the excited-state electric dipole moments of the valence excited states in CsAg and CsAu, for which nonscalar relativistic effects were found to be substantial. **5.8.6.** Phosphorescence. Whereas transition moments between singlet and triplet states are obtained directly from the residue of the linear response function when relativistic two- or four-component wave functions are used,^{262,755} we must in the nonrelativistic case consider the spin—orbit-perturbed groundand excited-state wave functions to make these dipole transitions allowed

$$|{}^{3}0^{(1)}\rangle = -\sum_{3_{n}} \omega_{n}^{-1} |{}^{3}n^{(0)}\rangle \langle {}^{3}n^{(0)} | H_{\rm so} |{}^{1}0^{(0)}\rangle \tag{657}$$

$$|{}^{1}f^{(1)}\rangle = \sum_{n} (\omega_{f} - \omega_{n})^{-1} |{}^{1}n^{(0)}\rangle \langle {}^{1}n^{(0)} | H_{\rm so} |{}^{3}f^{(0)}\rangle$$
(658)

Including these first-order corrections in the expression for the transition moments we find that the first-order contribution to the dipole transition moment between singlet and triplet states may be written as a residue of a quadratic response function^{70,756}

$$\langle {}^{1}0|\mu|^{3}f\rangle^{(1)} = \lim_{\omega \to \omega_{f}} (\omega - \omega_{f})\langle\langle\mu; H_{\rm so}, V^{\omega}\rangle\rangle_{0, \omega}$$
(659)

where V^{ω} is an arbitrary triplet operator (that determines the excitation vector) and ω matches the singlet-triplet excitation energy. The phosphorescent radiative lifetime τ_k of the *k*th component of $|^{3}f\rangle$ can be obtained from the relation

$$\frac{1}{\tau_k} = \frac{4\omega_f^3 \alpha^3}{3} \sum_{\nu} |\langle^1 0 | r_{\nu} |^3 f_k \rangle|^2$$
(660)

and depends not only on the transition moment but also on the excitation energy. The transition moment may vary for different polarizations of the light. The average phosphorescent lifetime is dominated by the shortest lifetime, corresponding to the polarization with the largest partial transition rate. Implementations of single residues of the triplet quadratic response function have been presented by Vahtras et al.⁷⁵⁶ for MCSCF theory and Christiansen and Gauss for coupled-cluster theory.⁷⁵⁷ A number of studies of phosphorescent lifetimes, including heavy-atom effects on these lifetimes, have been presented by Minaev and coworkers using MCSCF wave functions.^{510,511,755,758}

5.8.7. Spin—**Orbit Coupling Constants.** Spin—orbit coupling constants describe the coupling between singlet and triplet states and correspond to the residue of the triplet linear response function⁷⁵⁹

$$\lim_{\omega \longrightarrow \omega_f} (\omega - \omega_f) \langle \langle H_{\rm so}; V^{\omega_f} \rangle \rangle_{\omega} \tag{661}$$

Although the spin—orbit operator in eq 67 has both one- and two-electron parts, the one-electron mean-field⁷⁶⁰ or scaled⁷⁶¹ spin—orbit operators are commonly used; the differences in these approximate spin—orbit operators have been analyzed by Neese.⁷⁶² Most calculations of spin—orbit coupling constants involve the variational CI optimization of the relevant states using multireference theory.^{763—767} However, spin—orbit matrix elements have also been calculated using MCSCF and coupled-cluster linear response theory.^{759,768} Within a relativistic framework, the spin—orbit interaction arises naturally and is included in the optimization of the wave function. Fedorov and Gordon have given an extensive overview of the different methods used for calculating spin—orbit coupling constants up to 2003.⁷⁶⁹ An interesting recent development is the extension of the Columbus spin—orbit graphical unitary group approach (GUGA) for multireference wave functions^{282,770} to a hybrid MRCI/DFT level⁷⁷¹

in the spirit of the DFT/MRCI approach of Grimme and Waletzke. $^{772}\,$

The spin-orbit coupling is the main intramolecular interaction mechanism responsible for intersystem crossings. Moreover, spin-orbit effects often differ substantially in their behavior from the scalar relativistic effects, leading, for instance, to an interesting correlation between spin-orbit effects on nuclear shielding constants and indirect spin-spin coupling constants for molecules with heavy nuclei.43 By enabling singlet-triplet transitions it provides a very different pole structure for relativistic linear response functions than for nonrelativistic ones.²⁶² It is therefore important to consider spin-orbit effects separately from other relativistic effects, even in four-component calculations. A scheme for removing spin-orbit effects in relativistic calculations has been proposed by Dyall⁷⁷³ and applied at the four-component level of theory by Visscher and Saue.⁷⁷⁴ More recently, Cheng and Gauss applied a spin-free Dirac-Coulomb Hamiltonian to calculate the electric properties at the coupledcluster level of theory.²⁵⁴

5.8.8. Finite Lifetimes. As discussed in section 3.7, an important methodological development within response theory in the past decade is the implementation of damped response approaches. Even though the idea of introducing a phenomenological damping factor in the perturbative expressions for the (hyper)polarizability to account for the finite lifetime of the excited states is not new,^{71,775} its most successful formulation and implementation in the context of response theory is probably the complex polarization propagator method presented about 10 years ago by Norman et al. for the linear response function at the Hartree-Fock and MCSCF levels of theory.⁹³ Extension of the method to the quadratic response function along with a more rigorous justification of the methodology appeared a few years later.⁸⁹ In the same year a damped response approach was presented by Jensen et al.⁹¹ within DFT. A different but equivalent formulation of the complex polarization propagator based on the quasi-energy formulation, called damped response theory, was recently proposed by Kristensen et al.⁹⁰ Damped response formulations are now available at the Hartree-Fock, MCSCF, and Kohn-Sham levels of theory. Similar approaches have also been proposed in the context of vibrational CI,^{776,777} based on a Lanczos method, to evaluate the pure vibrational contributions to the polarizabilities and first hyperpolarizabilities of molecules and to compute infrared spectra from the imaginary part of the damped vibrational linear response function.

The phenomenological introduction of empirical lifetimes for the excited states in the response functions is useful for several reasons. It allows us, for instance, to study properties and spectra in resonant regions without the need to compute individual excitation energies and transition strengths and the subsequent convolution with line-shape functions. Thus, absorption, ECD, and MCD are computed directly from the imaginary or real part of the relevant response functions over the entire frequency range, including the highly energetic region of X-ray absorption.^{83,84,91,95,778–780} Applications, mainly at the TDDFT level, include, for instance, the first calculations of the complete optical-rotation dispersion curves using origin-independent DFT by Norman et al.⁸⁵ Another application, already mentioned in section 5.7.1, is the direct evaluation of the polarization propagator (linear response function) on the whole imaginary frequency axis, avoiding the expansion of the polarizability in a series of the Cauchy moments. In this manner, Casimir-Polder interaction potentials (C_6 coefficients) can be determined.^{88,688,781}

6. PURE VIBRATIONAL CORRECTIONS

In this section we consider an additional, nonelectronic contribution to molecular properties that, in principle, appears for any response function, namely, pure vibrational contributions, arising from excitations within the vibrational domain of the electronic ground state. The corresponding pure rotational contributions are usually assumed to be small and neglected.⁷⁸² In the sum-over-states expressions for the response functions in section 3.4 the summations are, in principle, over the vibronic rather than electronic states of the molecule. For the polarizability, for example, these summations may be written in the form (following the notation in ref 783)

$$\alpha_{\alpha\beta}(-\omega_{\sigma};\omega_{1}) = \sum P_{-\sigma, 1} \sum_{k, K \neq 0, 0} \frac{\langle 0, 0 | \mu_{\alpha} | k, K \rangle \langle k, K | \mu_{\beta} | 0, 0 \rangle}{\omega_{kK, 00} - \omega_{\sigma}}$$
(662)

where *k* and *K* denote electronic and vibrational states, respectively. In the Born–Oppenheimer approximation^{784,785} the vibronic wave function is written as a product of a nuclear wave function C_K^N and an electronic wave function $\psi_k^{\rm el}$ that depends parametrically on the nuclear positions

$$\Psi_{k, K}(\mathbf{R}, \mathbf{r}) = C_{K}^{N}(\mathbf{R})\psi_{k}^{\text{el}}(\mathbf{r}, \mathbf{R})$$
(663)

Separating the sum in eq 662 into a sum over the vibrational manifold of the electronic ground state and a sum over all the electronic excited states we obtain the expression

$$\begin{aligned} \alpha_{\alpha\beta}(-\omega_{\sigma};\omega_{1}) \\ &= \sum P_{-\sigma, 1} \left(\sum_{K \neq 0} \frac{\langle 0, 0 | \mu_{\alpha} | 0, K \rangle \langle 0, K | \mu_{\beta} | 0, 0 \rangle}{\omega_{0K, 00} - \omega_{\sigma}} \right. \\ &+ \sum_{k \neq 0} \frac{\langle 0, 0 | \mu_{\alpha} | k, 0 \rangle \langle k, 0 | \mu_{\beta} | 0, 0 \rangle}{\omega_{k, 0} - \omega_{\sigma}} \right) \end{aligned}$$
(664)

where the first term is the pure vibrational contribution to the polarizability whereas the second term corresponds to the zero-point vibrationally averaged electronic polarizability, simplified by application of the closure approximation for the vibrations. In Table 5 we collected the pure vibrational contributions to polarizabilities and hyperpolarizabilities for exact nuclear—electronic wave functions in the Born—Oppenheimer approximation.^{783,786}

For diatomic molecules we may evaluate the pure vibrational corrections in Table 5 using the vibrational sum-over-states expressions.^{570,787–789} The literature on pure vibrational contributions to diatomic molecules has been surveyed by Bishop and Norman.⁷⁸³ For polyatomic molecules a number of approximate schemes have been introduced, the most popular being the perturbation-theory approach of Bishop and Kirtman,^{783,790} whose accuracy was investigated for diatomic molecules by Bishop and Norman⁷⁸⁹ and for polyatomic molecules by Torrent-Sucarrat et al.⁷⁹¹ Bishop et al. introduced an alternative scheme, where the molecular geometry relaxes in the presence of applied static electric fields;⁷⁹² subsequently, Luis et al. proposed a scheme involving electric-field-relaxed coordinates, significantly reducing the computational cost.⁷⁹³ An advantage of the latter two approaches is the implicit inclusion of the leading order anharmonic corrections, whose inclusion otherwise requires expensive calculation of cubic force constants (for mechanical anharmonicity) or geometric second derivatives of the (hyper)polarizabilities.

A very different approach has been taken by Christiansen et al.,^{776,794} who calculated the pure vibrational contributions

from vibrational linear response theory using the vibrational CI method.^{795,796} Unlike for the perturbation-based methods, the potential-energy and property surfaces are calculated on a grid. Their approach has been extended to the quadratic response theory level to recover anharmonic effects.⁷⁹⁷ An interesting analysis of the vibrational pole structure has been published.⁷⁷⁶

The theory and calculation of pure vibrational corrections have been reviewed several times.^{782,798,799} In this review, with its focus on molecular electronic response theory, we restrict ourselves to the perturbative treatment of pure vibrational contributions. Representing the zero-order vibrational wave function as a product of harmonic-oscillator wave functions, one for each vibrational normal mode, the molecular properties are expanded around the equilibrium geometry in the nuclear displacements. Evaluation of the pure vibrational corrections thus requires calculation of geometric derivatives of the (hyper)polarizabilities. In Table 6, we list the expressions for the contributions to the pure vibrational (hyper)polarizabilities.⁷⁸³

Although a Hartree-Fock implementation of analytic derivatives for cubic and quartic force fields^{332,333} and dipole moment and polarizability derivatives³³⁴ was presented by Handy and coworkers in the early 1990s, it is only with the recent Hartree-Fock implementation of Champagne and co-workers^{346,800-804} that analytic calculations of pure vibrational contributions have appeared; recently, an AO-based Kohn-Sham implementation was presented by Thorvaldsen et al.⁷⁹ and later extended to two- and four-component relativistic theory.¹⁰⁹ In post-Hartree–Fock theory, calculations of pure vibrational contributions involves taking finite differences along normal coordinates of analytically calculated hyperpolarizabilities. Following an early correlated study by Hättig and Jørgensen,⁸⁰⁵ Norman et al. considered the combined effect of electron correlation and solvation on the electronic and vibrational hyperpolarizabilities in methanol.⁸⁰⁶ Using CCSD(T) theory, Sadlej and Papadopoulos examined the effects of electron correlation and relativity on electronic and vibrational hyperpolarizabilities, demonstrating that these are not always additive. $^{807-809}$

Pure vibrational contributions to nonlinear properties involving magnetic fields are typically less important than their electric analogues, often vanishing by symmetry as illustrated in eq 66. Indeed, only properties with a quadratic dependence on the magnetic field exhibit pure vibrational corrections, such as the hypermagnetizability that determines the Cotton—Mouton effect.^{810,811} Correlated calculations of pure vibrational contributions to hypermagnetizabilities have been presented.⁸¹² Rizzo and Cappelli also recently presented CCSD investigations of the pure vibrational contributions to EFGB and Jones birefringence.⁸¹³

Whereas pure vibrational contributions are negligible for processes at optical frequencies, they may be significant and even dominate for static hyperpolarizabilities, see, for example, ref 814. Considering the known difficulties of Kohn–Sham theory in treating electric properties of extended systems⁸¹⁵ and the importance of electron correlation for electronic and vibrational hyperpolarizabilities,^{816,817} the lack of analytic correlated wave function-based methods for calculating geometrical derivatives of electric properties beyond the dipole gradient is unfortunate; with the recent implementation of CCSD(T) polarizability gradients,³⁴⁸ this situation is beginning to change.

7. MOLECULAR PROPERTIES IN THE LIQUID AND SOLID PHASES

The changes observed in a molecule when going from the gas phase to the liquid or solid phase are conveniently divided into Table 5. Pure Vibrational Contributions to the Polarizability $\alpha^{pv} = [\mu^2]$, First Hyperpolarizability $\beta^{pv} = [\mu\alpha] + [\mu^3]$, and Second Hyperpolarizability $\gamma^{pv} = [\mu\beta] + [\alpha^2] + [\mu^2\alpha] + [\mu^4]$ for Exact Nuclear–Electronic Wave Functions in the Born–Oppenheimer Approximation in the Notation $(P)_{MN} = \langle M|P|N \rangle^a$

type	vibrational contribution
$[\mu^2]$	$\sum P_{-\sigma,\sigma} \sum_{K}' (\mu_{\alpha})_{0K} (\mu_{\beta})_{K0} (\omega_{K0} - \omega_{\sigma})^{-1}$
$[\mu\alpha]$	$\frac{1}{2}\sum P_{-\sigma;1,-2}\sum_{K}'(\mu_{\alpha})_{0K}(\alpha_{\beta\gamma})_{K0}(\omega_{K0}\pm\omega_{\sigma})^{-1}$
$[\mu^3]$	$\sum P_{-\omega;1,2} \sum_{KL}' (\mu_{\alpha})_{0K} (\bar{\mu}_{\beta})_{KL} (\mu_{\gamma})_{L0} (\omega_{K0} - \omega_{\sigma})^{-1} (\omega_{1} - \omega_{2})^{-1}$
$[\mu\beta]$	$\frac{1}{6}\sum P_{-\sigma, -1, -2, -3} \sum_{K} '(\mu_{\alpha})_{0K} (\beta_{\beta\gamma\delta})_{K0} (\omega_{K0} \pm \omega_{\sigma})^{-1}$
$[\alpha^2]$	$\frac{1}{4}\sum P_{-\sigma, 1, 2, 3} \sum_{K}' (\alpha_{\alpha\beta})_{0K} (\alpha_{\gamma\delta})_{K0} (\omega_{K0} - \omega_2 - \omega_3)^{-1}$
$[\mu^2 \alpha]$	$\frac{1}{2}\sum P_{-\sigma, 1, 2, 3} \sum_{KL}' [(\mu_{\alpha})_{0K}(\overline{\mu}_{\beta})_{KL}(\alpha_{\gamma\delta})_{L0}(\omega_{K0} \pm \omega_{\sigma})^{-1} \{\omega_{L0} \pm (\omega_{2} + \omega_{3})\}^{-1} + (\mu_{\alpha})_{0K}(\overline{\alpha}_{\beta\gamma})_{KL}(\mu_{\delta})_{L0}(\omega_{K0} - \omega_{\sigma})^{-1}(\omega_{L0} - \omega_{3})^{-1}]$
$[\mu^4]$	$\sum P_{-\sigma, -1, -2, -3} [\sum_{KLM} '(\mu_{\alpha})_{0K} (\bar{\mu}_{\beta})_{KL} (\bar{\mu})_{LM} (\mu_{\delta})_{M0} (\omega_{K0} - \omega_{\sigma})^{-1} (\omega_{L0} - \omega_{2} - \omega_{3})^{-1} (\omega_{M} - \omega_{3})^{-1}$
	$-\sum_{\nu\mu}'(\mu_{\alpha})_{0K}(\mu_{\beta})_{K0}(\mu_{\gamma})_{0L}(\mu_{\delta})_{L0}(\omega_{K0}-\omega_{\sigma})^{-1}(\omega_{L0}-\omega_{3})^{-1}(\omega_{L0}+\omega_{2})^{-1}]$

^{*a*} The notation Σ_{K} ' indicates that the summations are over excited states only.

direct and indirect effects.⁸¹⁸ The direct effects arise from the polarization of the solute's electron density by the surrounding solvent molecules or by the surrounding crystal molecules; the indirect effects arise from relaxation of the solute's geometry induced by this polarization. We here focus on the direct solvent effects, noting that the indirect effects are straightforwardly accounted for by allowing the molecular structure to relax in the presence of the solvent.

Solvent effects are usually treated by one of the following solvation models: the supermolecule model, the multilevel model, or the implicit model. In the following these models are discussed briefly; for more detailed discussions we refer to the review by Tomasi, Mennucci, and Cammi,⁸¹⁸ where calculation of molecular properties of solvated molecules is also treated. As an illustration of the accuracy of explicit, discrete, and continuum approaches for calculation of the excitation energies, we refer to the recent study on camphor by Kongsted et al.⁸¹⁹

7.1. Supermolecular Models

The supermolecular model is conceptually the simplest solvation model. The properties of the solute are calculated in the presence of the neighboring solvent molecules, which are treated at the same theoretical level as the solute. For correlated wave function methods this approach has severe limitations with respect to the size of the systems that can be treated, although recent advances in linear-scaling techniques, local correlation methods, and incremental schemes have made this approach more practical.^{139,208,209,211,820–822} A contributing factor to the high cost of such studies is the need to perform a statistical averaging over solute—solvent configurations, a procedure that may require sampling over 100–500 configurations depending on the property of interest.⁸²³

In discussing supermolecular property calculations it is important to distinguish between intensive and extensive properties. Asymptotically, intensive molecular properties are independent of the size of the system, whereas extensive properties are proportional to the system size. Intensive properties are easily extracted from supermolecular calculations, as the solvent molecules only act as perturbers on the property of interest. Extensive properties, by contrast, depend on the response of all molecules present in the system; a polarizability calculation, for example, provides the polarizability of all molecules in the system rather than that of the solute alone. Indeed, the response may actually be dominated by the solvent molecules and by surface effects.

For extensive properties a differential-shell approach has been proposed,⁸²⁴ where the molecular property of interest is calculated as the difference between the property of the full system (solute and solvent) and the property of the system with the solute removed. The approach has been used with some success, reproducing experimental solvent shifts⁸²⁴ for the first hyperpolarizability, for which it reproduces the experimentally observed sign change for water going from the gas to the liquid phase; it has also been applied to the magnetizability⁵⁸⁸ and the Cotton-Mouton effect.⁸²⁵ In some cases the solute itself may produce an imprint on the solvent configuration, for instance, for a chiral molecule in an achiral solvent. In these cases a supermolecular calculation is necessary to recover these effects. Perhaps the most striking example of such an effect was provided by Zuber, Beratan, and co-workers, who showed that the largest effect on the optical rotation of methyloxirane solvated in water or benzene arises from the response of the solute's imprint on the solvent structure.^{826,827} This observation also explained the strong solvent dependence observed for methyloxirane and, in particular, the change of sign of its optical rotation when changing the solvent from benzene to water.⁸²⁷

From a methodological point of view, supermolecular models do not require special computational technology beyond those methods developed for studying isolated molecules in the gas phase. We therefore do not discuss these models further here but note that, because of the large systems considered, DFT is the method of choice for such calculations. Furthermore, given the

Table 6. Contributions to Dynamic Vibrational Polarizabilities and Hyperpolarizabilities^a

$$\left|\mu^{2}\right|^{0, 0} = \frac{1}{2} \sum P_{\alpha, \beta} \sum_{a} \frac{\partial \mu_{\alpha}}{\partial Q_{a}} \frac{\partial \mu_{\beta}}{\partial Q_{a}} \frac{1}{a^{\pm} \sigma} \\ \left|\mu^{2}\right|^{2, 0} = \frac{1}{8} \sum P_{\alpha, \beta} \sum_{a, b} \left[\frac{\partial^{2} \mu_{\alpha}}{\partial Q_{a}} \frac{\partial^{2} \mu_{\beta}}{\partial Q_{a} \partial Q_{b}} \left(\frac{1}{\omega_{a}} + \frac{1}{\omega_{b}} \right) \lambda_{ab}^{\pm \sigma} + \frac{\partial^{2} \mu_{\alpha}}{\partial Q_{a} \partial Q_{a} \partial Q_{b}} \frac{\partial \mu_{\beta}}{\omega_{a}} \frac{\lambda_{a}^{\pm \sigma}}{\partial Q_{a} \partial Q_{a} \partial Q_{b}} \frac{\lambda_{a}^{\pm \sigma}}{\omega_{a}} \right] \\ \left|\mu^{2}\right|^{1, 1} = -\frac{1}{4} \sum P_{\alpha, \beta} \sum_{a, b, c} \left[F_{abc} \frac{\partial^{2} \mu_{\alpha}}{\partial Q_{a} \partial Q_{b}} \frac{\partial \mu_{\beta}}{\partial Q_{c}} \left(\frac{1}{\omega_{a}} + \frac{1}{\omega_{b}} \right) \lambda_{ab}^{\pm \sigma} \lambda_{c}^{\pm \sigma} + F_{bc} \frac{\partial^{2} \mu_{\alpha}}{\partial Q_{a} \partial Q_{b}} \frac{\partial \mu_{\beta}}{\partial Q_{a}} \frac{\lambda_{a}^{\pm \sigma}}{\partial Q_{a}} \frac{\partial \mu_{\beta}}{\partial Q_{a}} \frac{\lambda_{a}^{\pm \sigma}}{\partial Q_{a} \partial Q_{b}} \frac{\lambda_{a}^{\pm \sigma}}{\partial Q_{a}} \frac{\partial \mu_{\beta}}{\partial Q_{a}} \frac{\lambda_{a}^{\pm \sigma}}{\partial Q_{a} \partial Q_{b}} \frac{\partial \mu_{\beta}}{\partial Q_{a}} \frac{\lambda_{a}^{\pm \sigma}}{\partial Q_{a} \partial Q_{b}} \frac{\lambda_{a}^{\pm \sigma}}{\partial Q_{a}} \frac{\lambda_{a}^{\pm \sigma}}{\partial Q_{a} \partial Q_{b}} \frac{\partial \mu_{\beta}}{\partial Q_{a}} \frac{\lambda_{a}^{\pm \sigma}}{\partial Q_{a} \partial Q_{b}} \frac{\lambda_{a}^{\pm \sigma}}{\partial Q_{a}} \frac{\lambda_{a}^{\pm \sigma}}}{\lambda_{a}^{\pm \sigma$$

. .

^{*a*} (i) For the $[\mu^2]$ terms, $\omega_{\sigma} = \omega_1$; for the $[\mu^3]$ and $[\mu^2 \alpha]$ terms, $\omega_{\sigma} = \omega_1 + \omega_2$; for the $[\mu^4]$ terms, $\omega_{\sigma} = \omega_1 + \omega_2 + \omega_3$. (ii) $\Sigma P_{\alpha,\beta}$ indicates the sum over terms generated by the permutation of the pairs of optical frequencies and indices: $(-\omega_{\alpha}\alpha)$ and $(-\omega_{1\beta}\beta)$; $\Sigma P_{\alpha,\beta,\gamma}$ indicates the permutations of $(-\omega_{\alpha}\alpha)$ and $(\omega_{1\beta}\beta)$; and $(\omega_{2\beta}\gamma)$; $\Sigma P_{\alpha\beta,\gamma,\delta}$ indicates the permutations of $(-\omega_{\alpha}\alpha)$ and $(\omega_{1\beta}\beta)$, and $(\omega_{2\beta}\gamma)$, and $(\omega_{3\beta}\delta)$. (iii) $\lambda_x^{\pm i} = (\omega_x^2 - \omega_i^2)^{-1}$; $\lambda_{xy}^{\pm i} = [(\omega_x + \omega_y)^2 - \omega_i^2]^{-1}$; $\lambda_x^{\pm ij} = [\omega_x^2 - (\omega_i + \omega_j)^2]^{-1}$.

dynamic nature of the liquid system, a conformational averaging is in general needed. This can be achieved either by a Boltzmann averaging over energetically accessible equilibrium structures or by a sampling of instantaneous solute—solvent configurations generated by Car—Parrinello⁸²⁸ or molecular-dynamics⁸²⁹ simulations.

7.2. Multilevel Models

In the quantum-mechanics/molecular-mechanics (QM/MM) approach $^{830-832}$ the solute is treated quantum mechanically by a suitable electronic-structure method whereas the solvent is treated by molecular mechanics. Phenomenologically, we may write the energy of the QM/MM system as

$$E^{\rm QM/MM} = E^{\rm QM} + E^{\rm MM} + E^{\rm QM/MM} \tag{665}$$

where $E^{\rm QM}$ is the energy of the quantum-mechanical part of the system, $E^{\rm MM}$ the classical energy of the solvent described by molecular mechanics, and $E^{\rm QM/MM}$ the energy arising from the interaction between the solute and the solvent. The molecular mechanics energy $E^{\rm MM}$ contains contributions from parametrized force fields involving bond stretching and bending

motions, and van der Waals interactions are typically modeled using a Lennard–Jones potential.⁸³³

The coupling term $E^{QM/MM}$ requires special attention. In the simplest case the MM surroundings may be represented by (distributed) multipoles (charges, dipoles, quadrupoles, etc.)^{834,835}

$$E^{\rm QM/MM(unpol)} = E^{\rm nuc/mul} + E^{\rm el/mul}$$
(666)

$$= E^{\mathrm{nuc/mul}} + \langle 0 | G^{\mathrm{es}} | 0 \rangle$$

where $E^{\text{nuc/mul}}$ is the interaction energy between the nuclear charges of the solute and the distributed multipoles used to describe the surrounding nuclei while G^{es} describes the electrostatic (Coulomb) interaction between the electrons of the solute with the MM multipoles. In practice, the QM/MM calculations are performed in the same manner as conventional quantum-chemical calculations but with a modified one-electron part due to the additional Coulomb interactions, see, for example refs 592, 835, and 836.

Although the multipolar description of the solvent often works well for energies, it is in most cases inadequate for molecular properties, where the perturbation associated with the property of interest polarizes both solvent and solute, thereby affecting the

$$\boldsymbol{\mu}^{\text{ind}} = \boldsymbol{\alpha} \mathbf{F}^{\text{tot}} \tag{667}$$

where the induced dipole moment μ^{ind} and the polarizability α collect all dipole vectors and polarizability tensors at the MM sites and where $F^{\text{tot}} = F^{\text{sta}} + F^{\text{ind}}$ represents the total electric field (at the MM sites) as composed of a static component (due to the QM nuclei, QM electrons, and the MM multipoles) and an induced component (due to the induced dipoles themselves).

At the correlated level of electronic-structure theory, Kongsted and co-workers developed a coupled-cluster/molecularmechanics (CC/MM) method for accurate studies of molecular properties in solution, within the framework of a polarizable molecular-mechanics force field. Expectation values and dipole moments were first studied,⁸³⁷ followed shortly by implementation of the linear response function at the CC/MM level of theory.^{838–840} Although polarizabilities have been studied, the main focus in these studies is the excitation energies and transition moments, for which a detailed comparison with experiment is possible. We note, however, that the CC/MM methodology was able to reproduce the sign change of the optical rotation of (R)-methyloxirane when going from the gas phase to solution.⁶⁴⁷

Kongsted and co-workers later extended their CC/MM approach to quadratic response theory, presenting calculations of first hyperpolarizabilities⁸⁴¹ and TPA cross sections⁸⁴² of solvated molecules. Given the high cost of coupled-cluster theory, in particular, with triples included, it becomes important to reduce the number of solute – solvent configurations that need to be averaged. Osted et al.⁸⁴³ performed an investigation on liquid water, paying particular attention to how the computational cost can be reduced while maintaining accuracy.

Recently, the polarizable QM/MM method was significantly extended by incorporating self-consistent polarization effects in a more efficient fashion, appropriate for systems with thousands of MM sites. The new strategy is referred to as the polarizable embedding (PE) model.⁸³⁴ Sneskov et al.⁸³⁵ extended the PE method to linear and quadratic response functions at the CC2 and CCSD levels of theory, including also an approximate treatment of triple excitations through the CCSDR(3)/MM model.⁸⁴⁴ In the PE approach the environmental effects are conveniently incorporated into a coupled-cluster QM/MM quasi-energy Lagrangian

$$\mathscr{L}_{\text{QM/MM}} = \mathscr{L} + \langle \Lambda^{(0)} | G^{\text{es}} | \text{CC}^{(0)} \rangle - \frac{1}{2} \left(\boldsymbol{\mu}^{\text{ind}} \right)^{\text{T}} \mathbf{F}^{\text{sta}} + U$$
(668)

where \mathscr{X} is the standard QM part defined in eq 491 whereas U represents the remaining contributions to the total energy, which do not have an explicit dependence on the electronic parameters and thus are of minor importance from a coupled-cluster response point of view. Employing the same strategy as in the vacuum case (see section 4.3), we obtain modified amplitude and multiplier equations

$$0 = \langle \mu | \exp(-T)(H + G) \exp(T) | \text{HF} \rangle$$
(669)

$$0 = \langle \Lambda^{(0)} | [H + G, \tau_{\nu_j}] | CC^{(0)} \rangle$$
(670)

The only difference compared with the vacuum equations is the presence of an effective solvent operator

$$G = G^{\rm es} - \sum_{a, \nu} \mu_{\nu}^{a, \, \rm ind} \varepsilon_{\nu}^{a} \tag{671}$$

which couples the amplitude and multiplier equations. Here, \mathcal{E}_{ν}^{a} is a second-quantized operator representing the field at an MM site due to the QM electrons, the summation running over all Cartesian components (index ν) of all MM sites (index *a*). It is important to note the prominent role of the Lagrangian in this solvent model. In conventional vacuum coupled-cluster theory the Lagrangian is not needed for calculating energies but convenient for calculating other properties. In coupled-cluster QM/ MM theory it is the cornerstone of the model: it is only the evaluation of the energy Lagrangian itself that guarantees convergence to the FCI limit. A related difference to vacuum theory is the coupled nature of eqs 669 and 670, arising from the selfconsistent inclusion of polarization. For details, see ref 835.

Recently, Steindal et al.⁸⁴⁵ presented a fully polarizable threelevel model in which the polarizable QM/MM approach was combined with the PCM of Tomasi and Mennucci.^{846–848} The method was used to study excitation energies in solution, exploring the convergence of the excitation energies with the radius of the cutoff sphere used to define the solute– solvent clusters. An alternative three-level model is based on the use of the effective fragment potential (EFP) model in combination with the PCM.^{849,850} Both these methodologies thus allow for fully polarizable multiscale models. However, only gradients have so far been implemented for the EFP/ PCM model.

In the reference-interaction-site model (RISM) the liquid is represented by a set of atoms where chemical bonds are described by strong intramolecular correlations. The approach was first proposed by Chandler and Anderson⁸⁵¹ and applied to molecules by inclusion of charge distributions and molecular structure by Hirata and Rossky.^{852,853} The methodology was put in a quantum-mechanical framework by Hirata, Kato, and Ten-No,^{854,855} referred to as the RISM-SCF method. The statistical nature of the solute—solvent interactions are included through correlation functions, and the interactions between the solute and the solvent are expressed in the form of an integral equation. A comparison of the performance of the RISM-SCF and PCM approaches has been presented by Sato and Sakaki.⁸⁵⁶ The RISM-SCF method has been extended to treatment of nondynamical solvation of excited states (vide infra) by Ishida et al.⁸⁵⁷

An early application of the RISM-SCF approach was calculation of NMR chemical shieldings and the gas-to-liquid solvation shift of water⁸⁵⁸ using London orbitals for gauge-origin independence. An interesting property of the statistical nature of the RISM approach is the possibility to study temperature and density effects; however, the results reported in ref 858 were in poor agreement with experiment, and the method has since not been used for such purposes. This is unfortunate as the poor agreement may have arisen from a need to include the nearestneighbor solvent molecules in the quantum-mechanical description of the shielding constants.^{588,859,860} The RISM-SCF method has been extended to calculation of molecular gradients, including CCSD(T) gradients, allowing for an automated optimization of molecular structures in solution.^{861,862}

In the EFP method of Gordon and Jensen⁸⁶³ exchangerepulsion effects are included in the ground state in addition to
	transition energies		excitation-energy shift relative to cyclohexane		
	equilibrium	nonequilibrium	equilibrium	nonequilibrium	exp.
cyclohexane	37 386	37 386	0	0	0
CH_2Cl_2	37 529	37 611	143	225	570
ethanol	37 919	38 256	524	870	1360
water	37 960	38 342	574	956	2000
^{<i>a</i>} Results taken from ref 883 and obtained at the CAS(4,3)/6-31G(d) level of theory.					

Table 7. Electrostatic Contributions to the $n \rightarrow \pi^*$ Transition Energies (cm⁻¹) in Acetone in Various Solvents, Comparing Equilibrium and Nonequilibrium Solvation^{*a*}

the electrostatic and polarization contributions included in the QM/MM methods. As such, for the ground-state energy the EFP approach is somewhat more complicated to implement than the polarized QM/MM approach but includes effects that are important, in particular, for description of surfaces and interfaces and for the solvation process itself. As for the RISM-SCF method, one of the first applications of the EFP method was to model NMR shielding constants.⁸⁶⁴ In the most recent EFP formulations molecular gradients have been implemented at the Hartree–Fock and Kohn–Sham levels of theory;^{850,865,866} Kohn–Sham linear response functions⁸⁶⁷ and excited-state gradients have also been implemented.⁸⁶² In the Kohn–Sham implementation, exchange repulsion is treated by a set of fitting parameters, in contrast to the Hartree–Fock implementations where the exchange contribution is treated more rigorously, allowing for a simpler theoretical framework and application to response properties.

An emerging technique that may prove important for property calculations is the density embedding of wave function models of Carter et al., ^{868,869} recently extended by Gomes et al. to calculation of solvatochromic shifts.⁸⁷⁰ The method is based on the embedding scheme of Wesolowski and Warshel⁸⁷¹ for weakly interacting systems, where the solvent is described by DFT. This technique is useful in that it enables us to study the solute with high-level quantum-chemistry methods while retaining a quantum-mechanical description of the solvent. The approach has been generalized to coupled chromophores within a fully DFT-in-DFT model;⁸⁷² it would be interesting to see an extension of this method to a wave function treatment of the solute.

We finally note that the three-level multiscale ONIOM approach^{873,874} may also be used for solvation studies. As for all methods involving specific solvent molecules, dynamical averaging must be taken into account. The ONIOM approach is a difference approach, where the result for the solute or molecule of interest is obtained by a set of energy differences between different model calculations. It is therefore straightforward to calculate molecular properties for any computational model that allows for calculation of molecular properties. As such, the developments discussed in earlier sections for calculation of gas-phase molecular properties can be straightforwardly applied in the ONIOM approach.

7.3. Continuum Models

A different approach to solvation is to assume that because of dynamical averaging the solvent can be represented as a structureless, homogeneous, polarizable dielectric medium, with the solute contained inside a cavity in the medium. This assumption forms the basis for the continuum models.⁸¹⁸ In this way we do not have to consider the internal degrees of freedom of the

solvent and the Hamiltonian becomes

$$H^{\rm eff} = H + H^{\rm int} \tag{672}$$

where H is the isolated solute Hamiltonian and H^{int} describes the interaction between the solute and the solvent. The solvent is represented by a dielectric constant, which may be tensorial for nonisotropic media such as liquid crystals⁸⁷⁵ or position dependent for inhomogeneous systems such as surfaces or interfaces. $^{876-879}$ Given that the Hamiltonian $H^{\rm eff}$ describes the interaction between the solute and a polarizable dielectric medium it depends on the solute density. Therefore, as for the polarizable QM/MM models, the Schrödinger equation must be solved in a self-consistent manner. Special care needs to be given to methods that are nonvariational.^{880,881} However, the framework presented in section 3 can straightforwardly be extended to include the effects of a dielectric continuum by addition of a contribution from the interaction Hamiltonian, whose form depends on the description of the solute-solvent interactions (in terms of, for instance, boundary-element methods or multipole expansions).⁸¹⁸

In continuum calculations the properties of the solvent are described by a dielectric constant that has both a static and an optical component. Whereas the static dielectric constant describes both nuclear and electronic polarization of the dielectric medium, the optical dielectric constant describes only the electronic polarization. The optical dielectric constant is appropriate for processes involving dynamic electromagnetic fields, where only the electronic degrees of freedom are able to respond to the applied perturbation (with the nuclear degrees of freedom remaining in their unperturbed state), resulting in nonequilibrium solvation.⁸⁸² While there is a large variation in the static dielectric constant for different compounds (from about 2 for nonpolar molecules to about 80 for highly polar solvents such as water), the variation in the optical dielectric constant is small (between 1.5 and 2.5 depending on the electronic polarizability of the solvent). Nonequilibrium solvation is of little concern for nonpolar solvents (because of the small differences between the static and optical dielectric constants) but can be significant for polar solvents. This is illustrated for the excitation energies of acetone in Table 7, where large shifts in the excitation energies also arise from nonelectrostatic interactions, which can be accounted for by including the neighboring solvent molecules in the quantummechanical system.^{883,884}

The origin of nonequilibrium solvation is illustrated in Figure 3. From an equilibrated ground-state electronic structure the molecule is excited into a solvated structure where only the electronic degrees of freedom of the solvent are able to adjust



Figure 3. Schematic representation of nondynamical solvation effects in absorption and emission processes. Reprinted with permission of Luca Frediani (University of Tromsø), 2011.

instantaneously to the solute excitation. Given sufficient time the solvent molecules readjust to the excited-state electron density of the solute, leading to a new excited-state equilibrated structure, from which the molecule may be de-excited into a nonequilibrated ground-state electronic structure. Because of the induced restructuring of the solute in the excited state, absorption and emission energies are no longer identical. Note that during the equilibration time for the excited state, the solute geometry may relax, giving rise to an additional indirect solvation effect on the difference between the absorption and the emission spectra of the solute. Even without considering nuclear relaxation the separation of the solvent response into static and dynamic parts means that there is no longer a one-to-one correspondence between excitation energies obtained by an explicit optimization of the excited state and by linear response theory, as demonstrated by Cammi.885

Over the years a variety of continuum models, differing in the choice of H^{int} , have been proposed; for a recent review, see ref 818. We consider here continuum models that have been extended to calculation of molecular properties and comment on their suitability for studies of solvation effects, the main difference between these being the shape of the cavity in which the solute is placed.

Mikkelsen and co-workers developed a continuum model with a spherical cavity.⁸⁸⁶ Because of the simple cavity structure the solute—solvent interactions can be represented by a multipole expansion of the solute density. This model has been adapted to MCSCF cubic response theory with nonequilibrium solvation,^{887–893} to gaugeorigin-independent calculations of NMR shielding constants and magnetizabilities with London orbitals,^{588,894} to mixed electric—magnetic properties,^{825,895} to nonequilibrium solvation with triplet perturbations,⁸⁹¹ and to inhomogeneous solvation (e.g., from a metal surface) up to cubic response functions.^{876,877} The spherical-continuum model has recently been extended to linear and quadratic coupled-cluster response theory^{841,896,897} with applications to optical rotation.⁶⁴⁷ A review of the field has been provided by Mikkelsen.⁸⁹⁸

The cavity shape and distance between the nuclear positions and the cavity boundary are nearly always empirical factors in continuum calculations. An interesting exception to this rule was presented by Luo et al.⁸⁹⁹ By combining the classical Maxwell field theory with quantum-chemical reaction-field theory using spherical cavities these authors showed that the cavity radius is uniquely determined by the calculated polarizability and dielectric constant of the neat liquid. The radius for a spherical cavity of a neat liquid is therefore uniquely defined and no longer a parameter in the calculation.

Use of a spherical cavity leads to a simple interaction Hamiltonian, requiring only calculation of multipole integrals (and their derivatives for perturbation-dependent basis sets). However, accurate results can only be expected for molecules of a nearly spherical shape or for properties that depend on the response along the longest molecular axis, for example, nonlinear optical properties of conjugated push-pull molecules.900 In general, a more realistic, molecule-shaped cavity must be used, as provided by the polarizable-continuum model (PCM) in its various flavors ^{846,848,901–905} and by the conductor-screening model (COSMO).^{906–910} In these models the cavity surface is partitioned into small surface elements, onto which charges are attached to describe the induced polarization. For calculations of properties with wave function methods, the PCM is used more often than the COSMO model; given the focus on wave function methods in this review our discussion of solvent calculations with molecule-shaped cavities is therefore biased toward PCM calculations.

The PCM has been extended to the calculation of linear, quadratic, and cubic response functions using MCSCF wave functions,^{911–914} including nonequilibrium solvation and local-field effects.⁹¹⁵ Ferrighi et al. implemented the calculation of magnetizabilities using London orbitals.⁵⁹¹ Also, zero-field splittings have been studied using continuum models within the framework of a spherical-cavity approach and using MCSCF wave functions.^{510,511} However, solvent effects on magnetic properties are in general poorly reproduced both by continuum^{588,860,916} and by QM/MM^{917–921} models. It appears mandatory to include the nearest solvent molecules in the quantum system, indicating the presence of quantum effects that cannot be recovered by the simple electrostatic PCM and QM/MM models. It would be of interest to investigate whether visualization of ring currents⁴²⁶ in weakly interacting systems would reveal effects of ring-current interactions in solvated complexes.

Finally, we note that Caricato et al. combined the EOM-CC method with PCM⁹²² as an alternative to the coupled-cluster linear-response approach of Christiansen and Mikkelsen.^{881,896} Cammi also described evaluation of molecular gradients for excited states using the EOM-CC method.⁹²³

7.4. Molecular Properties with Periodic Boundary Conditions

Restricting ourselves to wave function methods, the number of studies of molecular properties in the solid state are limited, mainly because few codes have been developed for calculations subject to periodic boundary conditions, although we note the recently presented projected-augmented wave code by Marsman et al.⁹²⁴ More fundamentally, the representation of the electric dipole operator is nontrivial for periodic systems, noting that the form given in eq 62 breaks the periodicity of the sample.^{925,926}

Hartree—Fock codes for treating one-dimensional systems were presented independently by Champagne and Andre⁹²⁷ and Otto⁹²⁸ in 1992. Otto applied the code to the calculation of polarizabilities and hyperpolarizabilities of linear chains of hydrogen, water, and lithium hydride molecules, whereas Champagne and Andre focused on the polarizabilities of polyethene and polysilane.⁹²⁷ An alternative formulation was later presented by Kirtman et al.⁹²⁹ On the basis of the approaches for one- and twodimensional systems by Del Re et al.⁹³⁰ and by André et al.,^{931,932} Hirata and Iwata implemented analytic molecular gradients at the MP2 level of theory⁹³³ and force fields at the Hartree—Fock level of theory.⁹³⁴ An implementation of time-dependent Hartree—Fock theory for one-dimensional periodic systems has also been developed.⁹³⁵

For two- and three-dimensional systems Kudin and Scuseria first presented a finite-field approach to the calculation of polarizabilities of periodic systems at the Hartree—Fock level of theory,⁹³⁶ later extended to the analytic calculation of Hartree—Fock and Kohn—Sham frequency-dependent polarizabilities by Izmaylov et al.⁹³⁷ The group of Scuseria also developed analytic approaches for calculation of forces,⁹³⁸ force fields,⁹³⁹ and dipole gradients.⁹⁴⁰ A geometry optimization method based on internal coordinates has also been presented by the same group.⁹⁴¹ In a parallel development, Dovesi and Orlando reported similar extensions in the CRYSTAL code for molecular gradients,⁹⁴² static polarizabilities,⁹⁴³ and first hyperpolarizabilities^{944,945} at the Hartree—Fock and Kohn—Sham levels of theory. Maschio et al. furthermore presented the first optimized structures at the local-MP2 level of theory for periodic systems using a mixed analytic and numerical scheme.⁹⁴⁶

8. CONCLUDING REMARKS

Molecular properties can nowadays be calculated for all important wave function models using the sophisticated methods developed in quantum chemistry over the last decades. The underlying framework for such calculations is molecular response theory. In the present review we have given a uniform treatment of this theory, applicable to time-dependent and time-independent perturbations, to variational and nonvariational wave function models, for perturbation-dependent and perturbationindependent basis sets. This general framework will undergo further developments and adaptations in the future, for example, to the linear-scaling calculation of molecular properties of large molecular systems at the correlated and uncorrelated levels of theory, to calculation of molecular properties using explicitly correlated methods, and to calculation of properties using multiconfigurational coupled-cluster techniques. The future will also see a stronger merging of wave function and density-functional methods, for example, through development of range-separated and orbital-dependent functionals for exchange and correlation; such developments will necessitate a merging of property techniques for wave function and density-functional methods. We believe the property techniques presented and reviewed here are sufficiently flexible to be adapted to all these developments of quantum chemistry, although the increasing accuracy of quantumchemical calculations may lead to development of techniques that go beyond the regular perturbation-theory approach within the minimal-coupling scheme discussed by us.947

As demonstrated in the present review, molecular response theory has been applied to a vast range of systems and perturbations. Apart from calculation of molecular forces and force constants, essential to explore chemistry, molecular response theory is today applied to all forms of molecular spectroscopy for interpretation and prediction of molecular spectra and spectroscopic constants. Important theoretical developments over the last two decades have been application of quantum chemistry to magnetic resonance spectroscopies, optical activity and birefringence, nonlinear optics, and excited-state properties. In the future, the range of molecular properties routinely studied theoretically will broaden further to reflect new experimental developments in nonlinear and multidimensional electronic and vibrational spectroscopies. Although these and related applications of quantum chemistry have not all been fully reviewed here, we believe that our review is sufficiently exhaustive to direct the reader to the most important developments in wave function-based molecular response theory over the last two decades.

AUTHOR INFORMATION

Corresponding Author

*E-mail: trygve.helgaker@kjemi.uio.no.

BIOGRAPHIES



Trygve Helgaker is a professor of theoretical chemistry at the University of Oslo (Norway). He received his Ph.D. degree from the University of Oslo in 1986 (supervisor Jan Almlöf) and spent 1 year as a postdoctoral fellow at the University of Minneapolis and 2 years at the University of Aarhus (Denmark), before taking a position as Lecturer at the University of Oslo in 1989, where he became Full Professor in 1993. His research interests are in the area of highaccuracy quantum chemistry, density-functional theory, molecular properties, and linear-scaling techniques. He has spent sabbatical years at the University of Cambridge and University of Durham, United Kingdom. He is an author of the Dalton and Dirac programs.



Sonia Coriani is a researcher in physical chemistry at the University of Trieste (Italy). She graduated from the University of Modena (Italy) in 1993 and received her Ph.D. degree from the University of Aarhus (Denmark) in 2000. She was a research fellow at the ICQEM in Pisa in 1995, associate professor II at the Centre for Theoretical and Computation Chemistry (Norway) in 2007–2011, and Marie-Curie IEF fellow at the University of Aarhus in 2010–2011. Her research focuses on the development and application of wave-function- and density-functional-based methods for accurate determination of molecular properties, with emphasis on structural properties, nonlinear optics,

birefringence, and dichroism effects, and, more recently, X-ray related phenomena. She is an author of the Dalton program.



Poul Jørgensen is Professor at Aarhus University (Denmark), where he received his doctoral degree in 1976. He is coauthor of more than 300 papers and 3 monographs. His major interest is electronic structure theory, in particular, determination of molecular properties using response function theory. Recently, his interest has turned toward a description of large molecular systems using linear-scaling algorithms.



Kasper Kristensen is a postdoctoral fellow at the Lundbeck Foundation Center for Theoretical Chemistry, Aarhus University (Denmark), where he also received his Ph.D. degree (2011). His research topics are response theory for self-consistent field models and local coupled-cluster theory.



Jeppe Olsen is a professor of chemistry at Aarhus University (Denmark). He received his Ph.D degree in Chemistry from University of Aarhus and was a postdoctoral fellow at Texas A&M University with Danny L. Yeager and the University of Utah with Jack Simons. From 1986 to 1998 he was employed at Lund University and returned to Aarhus University in 1998. His main research interests are development of ab initio methods for the study of molecular structure and properties, including high-accuracy methods for small molecules, methods for large molecules, and relativistic contributions. He is an author of the Dalton, Dirac, MOLCAS, and LUCIA programs.



Kenneth Ruud is a professor of theoretical chemistry at the University of Tromsø (Norway). He received his Ph.D. degree from the University of Oslo in 1998 (supervisor Trygve Helgaker) and spent 2 years as a postdoctoral fellow in San Diego with Peter Taylor before moving to Tromsø in 2001. His main research interests are development of new ab initio methods for the study of molecular properties. Additional interests include solvent effects (continuum and QM/MM methods) and vibronic and relativistic effects. He is an author of the Dalton and Dirac programs.

ACKNOWLEDGMENT

This work received support from the Lundbeck Foundation and the Norwegian Research Council. S.C. acknowledges financial support within the Marie-Curie Actions Intra-European Fellowship (IEF) funding scheme (FP7-PEOPLE-2009-IEF), grant agreement no. 254326. K.R. and T.H. have been supported by the Research Council of Norway through a Centre of Excellence Grant (grant no. 179568/V30). The authors thank Trond Saue, Michał Jaszuński, and Kristian Sneskov for discussions and comments and Luca Frediani for providing them with Figure 3.

ACRONYMS

algebraic diagrammatic construction [model]
<i>n</i> th-order ADC [model]
atomic orbital
Baker–Campbell–Hausdorff [expansion]
Becke-3-parameter-Lee-Yang-Parr
[functional]
Coulomb-attenuating-method B3LYP
[functional]
coherent anti-Stokes Raman scattering
complete active space
second-order CAS perturbation theory
coupled cluster [model]

CCn	CC [rank] <i>n</i> [hybrid model]
CCS	CC singles [model]
CCSD	CC singles doubles [model]
CCSDPPA	CCSD polarization-propagator approximation
CCSDR(3)	CC singles_doubles_response-triples [model]
CCSDT	CC singles_doubles_triples [model]
CCSD(T)	CC singles_doubles_perturbative_triples
CC3D(1)	[model]
CCSDTQ	CC singles-doubles-triples-quadruples
	[model]
CCSDTQ5	CC singles-doubles-triples-quadruples-
	quintuples [model]
CD	circular dichroism
CI	configuration interaction [model]
CIS	CI singles [model]
CISD	CI singles-doubles [model]
CIS(D)	CI singles-perturbative-doubles [model]
CISDT	CI singles-doubles-triples [model]
CISDTQ	CI singles-doubles-triples-quadruples [model]
$CIS(D_{m})$	iterative CIS(D) [model]
COSMO	conductor-screening model
CTOCD-DZ	continuous transformation of the origin of the
01000002	current density by setting diamagnetic contribution
	to zero [model]
DB_RI_MP2	dual-basis_BLMP2 [model]
DET	density functional theory
DEMM	degenerate four wave mining
	direct porturbation theory
DPT.	urect perturbation theory
DPIn	<i>n</i> -order DP1
ECD	electronic CD [spectroscopy]
EFGB	electric-field-gradient-induced birefringence
EFP	effective fragment potential [model]
EOM-CC	equation of motion CC [model]
EOPE	electro-optic Pockels effect
EPR	electron paramagnetic resonance
ESHG	electric-field-induced SHG
FCI	full-CI [model]
FNO-CC	frozen-natural-orbital-CC
GIAO	gauge-origin-including AO
GUGA	graphical unitary group approach
HF	Hartree—Fock [model]
IDRI	intensity-dependent refractive index
$\Lambda CCSD(T)$	λ -based CCSD(T) [model]
MCD	magnetic CD [spectroscopy]
MCSCF	multiconfigurational SCF [model]
MM	molecular mechanics
МО	molecular orbital
MPn	<i>n</i> -order Møller—Plesset [model]
MRCI	multireference CI [model]
MRCISD	multireference CISD [model]
MRMP2	multireference MP2 [model]
NESC	normalized elimination of the small component
NLOC	[model]
NEVDT2	second order <i>u</i> electron valence state
INLVI 12	porturbation theory
NIO	perturbation theory
NLO ND (D	nonlinear optics
NOD	nuclear magnetic resonance
NQK	nuclear quadrupole resonance
INSK	nuclear spin rotation
OMO	ortnonormalized MO
ONIOM	our own <i>n</i> -layer integrated molecular orbital and
	molecular mechanics [model]

OR	optical rectification			
PCM	polarizable-continuum model			
RISM	reference-interaction-site model			
QM	quantum mechanics			
RAS	restricted active space			
RHF	restricted HF [model]			
RI	resolution of the identity			
RIJCOSX-MP2	RI-J chain of spheres exchange MP2 [model]			
ROA	Raman optical activity			
SAC-CI	symmetry-adapted-cluster CI [model]			
SCF	self-consistent field			
SCS	spin-component-scaled [model]			
SHG	second-harmonic generation			
SOPPA	second-order polarization-propagator approxima-			
	tion			
SOPPA(CCSD)SOPPA with CCSD amplitudes				
SOPPA(CC2)	SOPPA with CC2 amplitudes			
SORA	second-order regular approximation			
SO-RASSI	spin—orbit RAS state interaction [model]			
SORCI	spectroscopy-oriented MRCI [model]			
SOS	scaled opposite spin [model]; sum-over-states			
	[expression]			
TDDFT	time-dependent DFT			
TDSCF	time-dependent self-consistent field [method]			
THG	third-harmonic generation			
TPA	two-photon absorption			
TPCD	two-photon CD			
UMO	unmodified MO			
UV	ultraviolet			
VCD	vibrational CD			
VMF	vibration-mode-following [model]			
XCCn	expectation-value CCn [model]			
XCCSD	expectation-value CCSD [model]			
(X)CCSD	noniterative XCCSD [model]			
ZORA	zero-order regular approximation			

REFERENCES

(1) Mukamel, S. Principles of Nonlinear Optics and Spectroscopy; Oxford University Press: New York, 1995.

(2) Barron, L. D. *Molecular light scattering and optical activity*, 2nd ed.; Cambridge University Press: Cambridge, UK, 2004.

(3) Puzzarini, C.; Stanton, J. F.; Gauss, J. Int. Rev. Phys. Chem. 2010, 29, 273.

(4) Fleming, G. R. Chemical Applications of Ultrafast Spectroscopy; Oxford University Press: New York, 1986.

(5) Laane, J.; Takahashi, H.; Bandrauk, A. D. *Structure and Dynamics of Electronic Excited States*; Springer: Heidelberg, 1998.

(6) Helgaker, T.; Jaszuński, M.; Ruud, K. Chem. Rev. 1999, 99, 293.
(7) Neese, F. Coord. Chem. Rev. 2009, 253, 526.

(8) Pedone, A.; Biczysko, M.; Barone, V. ChemPhysChem 2010, 11, 1812.

(9) Boyd, R. W. Nonlinear Optics; Academic Press: San Diego, 1992. (10) Challenges and Advances in Computational Chemistry and Physics; In Non-linear Optical Properties of Matter: From Molecules to Condensed Phases; Papadopoulos, M. G., Leszczynski, J., Sadlej, A., Eds.; Springer: New York, 2006; Vol. I

(11) Niklasson, A. M. N.; Weber, V. J. Chem. Phys. 2007, 127, 064105.
(12) Kussmann, J.; Ochsenfeld, C. J. Chem. Phys. 2007, 127, 054103.

(13) Kussmann, J.; Ochsenfeld, C. J. Chem. Phys. 2007, 127, 204103.

(14) Coriani, S.; Høst, S.; Jansik, B.; Thøgersen, L.; Olsen, J.; Jørgensen, P.; Reine, S.; Pawlowski, F.; Helgaker, T.; Sałek, P. J. Chem. Phys. 2007, 126, 154108. (15) Moss, R. E. Advanced Molecular Quantum Mechanics; Chapman and Hall: London, 1973.

- (16) Whiffen, D. H. Pure Appl. Chem. 1978, 50, 75.
- (17) Saue, T. Adv. Quantum Chem. 2005, 48, 383.
- (18) Ferraro, M. B.; Herr, T. E.; Lazzeretti, P.; Malagoli, M.; Zanasi, R. *Phys. Rev. A* **1992**, *45*, 6272.
 - (19) Lazzeretti, P. Theor. Chim. Acta 1993, 87, 59.

(20) Wolinski, K.; Hinton, J. F.; Pulay, P. In Nuclear Magnetic Shieldings and Molecular Structure; Tossell, J. A., Ed.; NATO ASI Series

- C; Kluwer: Dordrecht, 1993; p 576. (21) Ferraro, M. B.; Caputo, M. C.; Zitto, M. E. *Chem. Phys.* 1999,
- 241, 1.
 - (22) Schwarzschild, K. Gött. Nach. Math.-Phys. Kl 1903, 126.

(23) Gell-Mann, M. Nuovo Cimento Suppl. 1956, 4, 848.

(24) Reiher, M.; Wolf, A. Relativistic Quantum Chemistry: The Fundamental Theory of Molecular Science; Wiley-VCH: Weinheim, 2009.

(25) Pauli, W. Z. Phys. 1927, 43, 601.

(26) Mohr, P. J.; Taylor, B. N.; Newell, D. B. *Rev. Mod. Phys.* 2008, 80, 633.

(27) Dyall, K. G.; Fægri, K., Jr. Introduction to relativistic quantum chemistry; Oxford University Press: New York, 2007.

- (28) Berger, R.; Quack, M. J. Chem. Phys. 2000, 112, 3148.
- (29) Visscher, L.; Dyall, K. G. Atom. Data Nucl. Data 1997, 67, 207.
- (30) Helgaker, T.; Jørgensen, P.; Olsen, J. Molecular Electronic-Structure Theory; John Wiley & Sons: Chichester, 2000.
 - (31) Pyykkö, P. Mol. Phys. 2008, 106, 1965.
 - (32) Kutzelnigg, W. Z. Phys. D 1990, 15, 27.

(33) van Lenthe, E.; Baerends, E. J.; Snijders, J. G. J. Chem. Phys. 1993, 99, 4597.

- (34) Coriani, S.; Helgaker, T.; Jørgensen, P.; Klopper, W. J. Chem. Phys. 2004, 121, 6591.
- (35) Buslaev, Y. A.; Kravcenko, E. A.; Kolditz, L. *Coord. Chem. Rev.* **1987**, 82, 7.
- (36) Abragam, A. The Principles of Nuclear Magnetic Resonance; Oxford University Press: Oxford, 1961.
- (37) Stone, A. *The Theory of Intermolecular Forces;* Clarendon Press: Oxford, 2000.
- (38) Vahtras, O.; Loboda, O.; Minaev, B. F.; Ågren, H.; Ruud, K. Chem. Phys. 2002, 279, 133.
- (39) Ganyushin, D.; Neese, F. J. Chem. Phys. 2006, 125, 024103.
- (40) Harriman, J. E. *Theoretical foundations of electron spin resonance;* Academic Press: New York, 1978.
- (41) Nakatsuji, H.; Takashima, H.; Hada, M. *Chem. Phys. Lett.* **1995**, 233, 95.

(42) Malkin, V. G.; Malkina, O. L.; Salahub, D. R. Chem. Phys. Lett. 1996, 261, 335.

- (43) Kaupp, M.; Malkina, O. L.; Malkin, V. G.; Pyykkö, P. *Chem. Eur. J.* **1998**, *4*, 118.
 - (44) London, F. J. Phys. Radium 1937, 8, 397.
 - (45) Sadlej, A. J. Chem. Phys. Lett. 1977, 47, 50.
 - (46) Darling, C. L.; Schlegel, H. B. J. Phys. Chem. 1994, 98, 5855.
- (47) Olsen, J.; Bak, K. L.; Ruud, K.; Helgaker, T.; Jørgensen, P. Theor. Chim. Acta 1995, 90, 421.

(48) Helgaker, T. U.; Almlöf, J. Int. J. Quantum Chem. 1984, 26, 275.

(49) Helgaker, T. U. In *Geometrical Derivatives of Energy Surfaces and Molecular Properties*; Jørgensen, P., Simons, J., Eds.; NATO ASI Series; Clarendon Press: Dordrecht, 1986; Vol. 166

(50) Lengsfield, B. H., III In *Geometrical Derivatives of Energy Surfaces* and Molecular Properties; Jørgensen, P., Simons, J., Eds.; NATO ASI Series; Clarendon Press: Dordrecht, 1986; Vol. 166

(51) Löwdin, P.-O. J. Chem. Phys. 1950, 18, 365.

(52) Ruud, K.; Helgaker, T.; Olsen, J.; Jørgensen, P.; Bak, K. L. Chem. Phys. Lett. 1995, 235, 47.

- (53) Bak, K. L.; Jørgensen, P.; Jensen, H. J. A.; Olsen, J.; Helgaker, T. J. Chem. Phys. **1992**, 97, 7573.
- (54) Bak, K. L.; Hansen, A. E.; Ruud, K.; Helgaker, T.; Olsen, J.; Jørgensen, P. *Theor. Chim. Acta* **1995**, *90*, 441.

- (55) Bak, K. L.; Jørgensen, P.; Helgaker, T.; Ruud, K.; Jensen, H. J. A. J. Chem. Phys. **1993**, *98*, 8873.
- (56) Bak, K. L.; Jørgensen, P.; Helgaker, T.; Ruud, K. Faraday Discuss. 1994, 99, 121.
 - (57) Simons, J.; Jørgensen, P.; Helgaker, T. Chem. Phys. 1984, 86, 413.
- (58) Helgaker, T. U.; Almlöf, J.; Jensen, H. J. A.; Jørgensen, P. J. Chem. Phys. **1986**, 84, 6266.
 - (59) Helgaker, T.; Jørgensen, P. J. Chem. Phys. 1991, 95, 2595.
 - (60) Epstein, S. T. J. Chem. Phys. 1973, 58, 1592.
 - (61) Ditchfield, R. J. Chem. Phys. 1972, 56, 5688.
 - (62) Dalgaard, E. Chem. Phys. Lett. 1977, 47, 279.
- (63) Langhoff, P. W.; Epstein, S. T.; Karplus, M. Rev. Mod. Phys. 1972, 44, 602.
 - (64) Aiga, F.; Sasagane, K.; Itoh, R. J. Chem. Phys. 1993, 99, 3779.
 - (65) Sasagane, K.; Aiga, F.; Itoh, R. J. Chem. Phys. 1993, 99, 3738.
- (66) Christiansen, O.; Jørgensen, P.; Hättig, C. Int. J. Quantum Chem. 1998, 68, 1.
- (67) Frenkel, J. Wave Mechanics, Advanced General Theory; Clarendon: Oxford, 1934.
- (68) Hellmann, H. Einführung in die Quantenchemie; Deuticke: Leipzig, 1937.
 - (69) Feynman, R. P. Phys. Rev. 1939, 56, 340.
 - (70) Olsen, J.; Jørgensen, P. J. Chem. Phys. 1985, 82, 3235.
 - (71) Orr, B. J.; Ward, J. F. Mol. Phys. 1971, 20, 513.
- (72) Linderberg, J.; Ohrn, Y. Propagators in quantum chemistry; Wiley: Hoboken, NJ, 2004.
 - (73) Ehrenfest, P. Z. Phys. 1927, 45, 455.

(74) Christiansen, O.; Coriani, S.; Gauss, J.; Hättig, C.; Jørgensen, P.; Pawlowski, F.; Rizzo, A. Non-linear Optical Properties of Matter: From Molecules to Condensed Phases; In *Challenges and Advances in Computational Chemistry and Physics*; Papadopoulos, M. G., Leszczynski, J., Sadlej, A., Eds.; Springer: New York, 2006; Vol. I, p 51.

(75) Hättig, C.; Christiansen, O.; Jørgensen, P. Chem. Phys. Lett. 1998, 282, 139.

(76) Helgaker, T.; Jørgensen, P. In *Methods in Computational Molecular Physics*; Wilson, S., Diercksen, G. H. F., Eds.; Plenum Press: New York, 1992.

(77) Helgaker, T.; Jørgensen, P. Theor. Chim. Acta 1989, 75, 111.

(78) Kristensen, K.; Jørgensen, P.; Thorvaldsen, A. J.; Helgaker, T.

J. Chem. Phys. 2008, 129, 214103.

(79) Thorvaldsen, A. J.; Ruud, K.; Jaszuński, M. J. Phys. Chem. A 2008, 112, 11942.

(80) Thorvaldsen, A. J.; Ferrighi, L.; Ruud, K.; Ågren, H.; Coriani, S.; Jørgensen, P. *Phys. Chem. Chem. Phys.* **2009**, *11*, 2293.

(81) Sellers, H. Int. J. Quantum Chem. 1986, 30, 433. Erratum: ibid. 1986, 30, 713

- (82) Koch, H.; Harrison, R. J. J. Chem. Phys. 1991, 95, 7479.
- (83) Ekström, U.; Norman, P. Phys. Rev. A 2006, 74, 042722.
- (84) Ekström, U.; Norman, P.; Carravetta, V.; Ågren, H. Phys. Rev. Lett. 2006, 97, 143001.
- (85) Norman, P.; Ruud, K.; Helgaker, T. J. Chem. Phys. 2004, 120, 5027.
- (86) Visser, F.; Wormer, P. E. S.; Stam, P. J. Chem. Phys. 1983, 79, 4973.
- (87) Fowler, P. W.; Jørgensen, P.; Olsen, J. J. Chem. Phys. 1990, 93, 7256.
- (88) Norman, P.; Jiemchooroj, A.; Sernelius, B. E. J. Chem. Phys. 2003, 118, 9167.
- (89) Norman, P.; Bishop, D. M.; Jensen, H. J. A.; Oddershede, J. J. Chem. Phys. 2005, 123, 194103.
- (90) Kristensen, K.; Kauczor, J.; Kjærgaard, T.; Jørgensen, P. J. Chem. Phys. **2009**, 131, 044112.
- (91) Jensen, L.; Autschbach, J.; Schatz, G. C. J. Chem. Phys. 2005, 122, 224115.
- (92) Jensen, L.; Zhao, L. L.; Autschbach, J.; Schatz, G. C. J. Chem. Phys. 2005, 123, 174110.

(93) Norman, P.; Bishop, D. M.; Jensen, H. J. A.; Oddershede, J. J. Chem. Phys. 2001, 115, 10323.

(94) Solheim, H.; Ruud, K.; Coriani, S.; Norman, P. J. Phys. Chem. A 2008, 112, 9615.

- (95) Solheim, H.; Ruud, K.; Coriani, S.; Norman, P. J. Chem. Phys. 2008, 128, 094103.
- (96) Kjærgaard, T.; Kristensen, K.; Kauczor, J.; Jørgensen, P.; Coriani, S.; Thorvaldsen, A. J. *J. Chem. Phys.* **2011**, *135*, 024112.
- (97) Kristensen, K.; Kauczor, J.; Thorvaldsen, A. J.; Jørgensen, P.; Kjærgaard, T.; Rizzo, A. J. Chem. Phys. 2011, 134, 214104.
- (98) Jørgensen, P.; Jensen, H. J. A.; Olsen, J. J. Chem. Phys. 1988, 89, 3654.
- (99) Hettema, H.; Jensen, H. J. A.; Jørgensen, P.; Olsen, J. J. Chem. Phys. **1992**, 97, 1174.
- (100) Norman, P.; Jonsson, D.; Vahtras, O.; Ågren, H. Chem. Phys. Lett. 1995, 242, 7.
- (101) Jonsson, D.; Norman, P.; Ågren, H. J. Chem. Phys. 1996, 105, 6401.
- (102) Larsen, H.; Jørgensen, P.; Olsen, J.; Helgaker, T. J. Chem. Phys. **2000**, *113*, 8908.
- (103) Niklasson, A. M. N.; Challacombe, M. Phys. Rev. Lett. 2004, 92, 193001.
- (104) Weber, V.; Niklasson, A. M. N.; Challacombe, M. Phys. Rev. Lett. 2004, 92, 193002.
- (105) Kjærgaard, T.; Jørgensen, P.; Olsen, J.; Coriani, S.; Helgaker, T. J. Chem. Phys. **2008**, *129*, 054106.
- (106) Thorvaldsen, A. J.; Ruud, K.; Kristensen, K.; Jørgensen, P.; Coriani, S. J. Chem. Phys. **2008**, 129, 214108.
- (107) Krykunov, M.; Autschbach, J. J. Chem. Phys. 2005, 123, 114103.
- (108) Ekström, U.; Visscher, L.; Bast, R.; Thorvaldsen, A. J.; Ruud, K. J. Chem. Theory Comput. **2010**, *6*, 1971.
- (109) Bast, R.; Thorvaldsen, A. J.; Ringholm, M.; Ruud, K. Chem. Phys. 2009, 356, 177.
 - (110) Pulay, P. Mol. Phys. 1969, 17, 197.
- (111) Pople, J. A.; Krishnan, R.; Schlegel, H. B.; Binkley, J. S. Int. J. Quantum Chem. 1978, 14, 545.
- (112) Larsen, H.; Helgaker, T.; Olsen, J.; Jørgensen, P. J. Chem. Phys. 2001, 115, 10344.
 - (113) Roos, B. O. Adv. Chem. Phys. 1987, 69, 399.
 - (114) Werner, H.-J. Adv. Chem. Phys. 1987, 69, 1.
- (115) Andersson, K.; Malmqvist, P.-A.; Roos, B. O.; Sadlej, A. J.; Wolinski, K. J. Phys. Chem. **1990**, *94*, 5483.
- (116) Angeli, C.; Cimiraglia, R.; Evangelisti, S.; Leininger, T.; Malrieu, J.-P. J. Chem. Phys. 2001, 114, 10252.
- (117) Shavitt, I.; Bartlett, R. J. *Many-Body Methods in Chemistry and Physics. MBPT and Coupled-Cluster Theory;* Cambridge University Press: Cambridge, 2009.
 - (118) Purvis, G.; Bartlett, R. J. J. Chem. Phys. 1982, 76, 1910.
- (119) Noga, J.; Bartlett, R. J. J. Chem. Phys. **1987**, 86, 7041. Erratum: ibid. **1988**, 89, 3401.
- (120) Scuseria, G. E.; Schaefer, H. F. Chem. Phys. Lett. 1988, 152, 382.
- (121) Christiansen, O.; Koch, H.; Jørgensen, P. Chem. Phys. Lett. 1995, 243, 409.
- (122) Christiansen, O.; Koch, H.; Jørgensen, P. J. Chem. Phys. 1995, 103, 7429.
- (123) Koch, H.; Christiansen, O.; Jørgensen, P.; de Meras, A. M. J. S.; Helgaker, T. J. Chem. Phys. **1997**, *106*, 1808.
- (124) Raghavachari, K.; Trucks, G. W.; Pople, J. A.; Head-Gordon, M. Chem. Phys. Lett. **1989**, 157, 479.
- (125) Koch, H.; Kobayashi, R.; de Meras, A. M. J. S.; Jørgensen, P. J. Chem. Phys. **1994**, 100, 4393.
- (126) Geertsen, J.; Rittby, M.; Bartlett, R. J. Chem. Phys. Lett. 1989, 164, 57.
 - (127) Stanton, J. F.; Bartlett, R. J. J. Chem. Phys. 1993, 98, 7029.
 - (128) Olsen, J.; Jørgensen, P. In Modern Electronic Structure Theory
- II; Yarkoni, D., Ed.; Wiley-VCH: New York, 1995.
 (129) Kobayashi, R.; Koch, H.; Jørgensen, P. Chem. Phys. Lett. 1994, 219, 30.

- (130) Sekino, H.; Bartlett, R. J. Adv. Quantum Chem. 1999, 35, 149.
- (131) Musial, M.; Bartlett, R. J. J. Chem. Phys. 2011, 134, 034106.
- (132) Hättig, C.; Koch, H.; Jørgensen, P. J. Chem. Phys. 1998, 109, 3293.
 - (133) Auer, A. A.; Gauss, J. J. Chem. Phys. 2001, 115, 1619.
- (134) Olsen, J.; Christiansen, O.; Koch, H.; Jørgensen, P. J. Chem. Phys. **1996**, 105, 5082.
- (135) Christiansen, O.; Olsen, J.; Jørgensen, P.; Koch, H.; Malmqvist, P.-A. Chem. Phys. Lett. 1996, 261, 369.
- (136) Helgaker, T.; Jørgensen, P.; Handy, N. C. Theor. Chim. Acta 1989, 76, 227.
 - (137) Korona, T.; Jeziorski, B. J. Chem. Phys. 2006, 125, 184109.
 - (138) Maroulis, G. Chem. Phys. Lett. 2001, 334, 214.
- (139) Friedrich, J.; Coriani, S.; Helgaker, T.; Dolg, M. J. Chem. Phys. 2009, 131, 154102.
- (140) Korona, T.; Pfluger, K.; Werner, H.-J. Phys. Chem. Chem. Phys. 2004, 6, 2059.
- (141) Dirac, P. Proc. Camb. Philos. Soc. 1930, 26, 376.
- (142) McLachlan, A. D.; Ball, M. A. Rev. Mod. Phys. 1964, 36, 844.
- (143) Dalgarno, A.; Victor, G. A. Proc. R. Soc. London A 1966,
- 291, 291.
 - (144) Dunning, T. H.; McKoy, V. J. Chem. Phys. 1967, 47, 1735.
 - (145) Dalgaard, E. Phys. Rev. A 1982, 26, 42.
 - (146) Rice, J. E.; Handy, N. C. J. Chem. Phys. 1991, 94, 4959.
 - (147) Dykstra, C. E.; Jasien, P. G. Chem. Phys. Lett. 1984, 109, 388.
 - (148) Sekino, H.; Bartlett, R. J. J. Chem. Phys. 1986, 85, 976.
 - (149) Karna, S. P.; Dupuis, M. J. Comput. Chem. 1991, 12, 487.
 - (150) Bast, R.; Ekström, U.; Gao, B.; Helgaker, T.; Ruud, K.;
- Thorvaldsen, A. J. Phys. Chem. Chem. Phys. 2011, 13, 2627.
 - (151) Monkhorst, H. J. Int. J. Quantum Chem. 1977, 12, 421.
 - (152) Dalgaard, E.; Monkhorst, H. J. Phys. Rev. A 1983, 28, 1217.
 - (153) Mukherjee, D.; Mukherjee, P. K. Chem. Phys. **1979**, 39, 325.
 - (154) Koch, H.; Jørgensen, P. J. Chem. Phys. **1990**, 93, 3333.
 - (155) Pedersen, T. B.; Koch, H. J. Chem. Phys. 1997, 106, 8059.
 (156) Jørgensen, P.; Helgaker, T. J. Chem. Phys. 1988, 89, 1560.
 - (150) Jørgensen, r.; Heigaker, 1. J. Chem. Phys. **1966**, 89, 1500.
- (157) Koch, H.; Jensen, H. J. A.; Jørgensen, P.; Helgaker, T. J. Chem. Phys. **1990**, *93*, 3345.
- (158) Christiansen, O.; Koch, H.; Halkier, A.; Jørgensen, P.; Helgaker, T.; de Meras, A. M. J. S. *J. Chem. Phys.* **1996**, *105*, 6921.
- (159) Christiansen, O.; Halkier, A.; Koch, H.; Jørgensen, P.; Helgaker, T. J. Chem. Phys. **1998**, 108, 2801.
- (160) Hättig, C.; Christiansen, O.; Koch, H.; Jørgensen, P. Chem. Phys. Lett. **1997**, 269, 428.
- (161) Christiansen, O.; Gauss, J.; Stanton, J. F. Chem. Phys. Lett. 1998, 292, 437.
- (162) Gauss, J.; Christiansen, O.; Stanton, J. F. Chem. Phys. Lett. 1998, 296, 117.
- (163) Pawlowski, F.; Jørgensen, P.; Hättig, C. Chem. Phys. Lett. 2004, 391, 27.
- (164) Kállay, M.; Gauss, J. J. Mol. Struct. (THEOCHEM) 2006, 768, 71.
- (165) O'Neill, D. P.; Kállay, M.; Gauss, J. J. Chem. Phys. 2007, 127, 134109.
- (166) Beebe, N. H. F.; Linderberg, J. Int. J. Quantum Chem. 1977, 12, 683.
- (167) Koch, H.; de Meras, A. M. J. S.; Pedersen, T. B. J. Chem. Phys. 2003, 118, 9481.
 - (168) Whitten, J. L. J. Chem. Phys. 1973, 58, 4496.
- (169) Dunlap, B. I.; Connolly, J. W. D.; Sabin, J. R. J. Chem. Phys. 1979, 71, 3396.
- (170) Vahtras, O.; Almlöf, J.; Feyereisen, M. W. Chem. Phys. Lett. 1993, 213, 514.
- (171) Hättig, C.; Weigend, F. J. Chem. Phys. 2000, 113, 5154.
- (172) Hättig, C.; Köhn, A. J. Chem. Phys. 2002, 117, 6939.
- (173) Hättig, C.; Hald, K. Phys. Chem. Chem. Phys. 2002, 4, 2111.
- (174) Köhn, A.; Hättig, C. J. Chem. Phys. 2003, 119, 5021.
- (175) Pedersen, T. B.; de Meras, A. M. J. S.; Koch, H. J. Chem. Phys. 2004, 120, 8887.

- (176) Fliegl, H.; Hättig, C.; Klopper, W. J. Chem. Phys. 2006, 124, 044112.
- (177) Pabst, M.; Köhn, A. J. Chem. Phys. 2008, 129, 214101.
- (178) Korona, T. Mol. Phys. 2010, 108, 343.
- (179) Korona, T. Phys. Chem. Chem. Phys. 2010, 12, 14977.
- (180) Moszynski, R.; Zuchowski, P. S.; Jeziorski, B. Collect. Czech. Chem. Commun. 2005, 70, 1109.
 - (181) Krylov, A. I. Annu. Rev. Phys. Chem. 2008, 59, 433.
 - (182) Bartlett, R. J. Mol. Phys. 2010, 108, 2905.
- (183) Rozyczko, P. B.; Perera, S. A.; Nooijen, M.; Bartlett, R. J. J. Chem. Phys. **1997**, 107, 6736.
 - (184) Rozyczko, P. B.; Bartlett, R. J. J. Chem. Phys. 1998, 109, 9201.
 - (185) Nooijen, M.; Bartlett, R. J. J. Chem. Phys. 1995, 102, 3629.
- (186) Musial, M.; Kucharski, S. A.; Bartlett, R. J. J. Chem. Phys. 2003, 118, 1128.
 - (187) Lee, Y. S.; Bartlett, R. J. J. Chem. Phys. 1984, 80, 4371.
- (188) Lee, Y. S.; Kucharski, S. A.; Bartlett, R. J. J. Chem. Phys. **1984**, *81*, 5906.
- (189) Christiansen, O.; Koch, H.; Jørgensen, P. J. Chem. Phys. 1996, 105, 1451.
 - (190) Watts, J. D.; Bartlett, R. J. J. Chem. Phys. 1994, 101, 3073.
 - (191) Watts, J. D.; Bartlett, R. J. Chem. Phys. Lett. 1995, 233, 81.
- (192) Koch, H.; Christiansen, O.; Jørgensen, P.; Olsen, J. Chem. Phys. Lett. 1995, 244, 75.
- (193) Christiansen, O.; Koch, H.; Jørgensen, P.; Olsen, J. Chem. Phys. Lett. 1996, 256, 185.
- (194) Watts, J. D.; Bartlett, R. J. Chem. Phys. Lett. 1996, 258, 581.
- (195) Hald, K.; Hättig, C.; Yeager, D. L.; Jørgensen, P. Chem. Phys. Lett. 2000, 328, 291.
- (196) Hald, K.; Hättig, C.; Jørgensen, P. J. Chem. Phys. 2000, 113, 7765.
 (197) Hald, K.; Hättig, C.; Olsen, J.; Jørgensen, P. J. Chem. Phys.
- 2001, 115, 3545.
 (198) Hald, K.; Jørgensen, P.; Christiansen, O.; Koch, H. J. Chem.
- *Phys.* **2002**, *116*, 5963.
 - (199) Stanton, J. F.; Gauss, J. Adv. Chem. Phys. 2003, 125, 101.
 - (200) Krylov, A. I. Chem. Phys. Lett. 2001, 338, 375.
 - (201) Levchenko, S. V.; Krylov, A. I. J. Chem. Phys. 2004, 120, 175.
- (202) Slipchenko, L. V.; Krylov, A. I. J. Chem. Phys. 2005, 123, 084107.
- (203) Krylov, A. I. Acc. Chem. Res. 2006, 39, 83.
- (204) Smith, C. E.; King, R. A.; Crawford, T. D. J. Chem. Phys. 2005, 122, 054110.
- (205) Russ, N. J.; Crawford, T. D. Chem. Phys. Lett. 2004, 400, 104.
- (206) Saebø, S.; Pulay, P. Annu. Rev. Phys. Chem. 1993, 44, 213.
- (207) Saebø, S.; Pulay, P. J. Chem. Phys. 1988, 88, 1884.
- (208) Russ, N. J.; Crawford, T. D. Phys. Chem. Chem. Phys. 2008, 10, 3345.
- (209) Kats, D.; Korona, T.; Schütz, M. J. Chem. Phys. 2006, 125, 104106.
- (210) Kats, D.; Korona, T.; Schütz, M. J. Chem. Phys. 2007, 127, 064107.
 - (211) Kats, D.; Schütz, M. J. Chem. Phys. 2009, 131, 124117.
 - (212) Kats, D.; Schütz, M. Z. Phys. Chem. 2010, 224, 601.
- (213) Freundorfer, K.; Kats, D.; Korona, T.; Schütz, M. J. Chem. Phys. 2010, 133, 244110.
- (214) Kordel, E.; Villani, C.; Klopper, W. J. Chem. Phys. 2005, 122, 214306.
- (215) Kordel, E.; Villani, C.; Klopper, W. Mol. Phys. 2007, 105, 2565.
- (216) Höfener, S.; Hättig, C.; Klopper, W. Z. Phys. Chem. 2010, 224, 695.
 - (217) Neiss, C.; Hättig, C. J. Chem. Phys. 2007, 126, 154101.
 - (218) Yang, J.; Hättig, C. J. Chem. Phys. 2009, 131, 074102.
 - (219) Yang, J.; Hättig, C. Z. Phys. Chem. 2010, 224, 383.
 - (220) Hanauer, M.; Köhn, A. J. Chem. Phys. 2009, 131, 124118.
- (221) Nielsen, E. S.; Jørgensen, P.; Oddershede, J. J. Chem. Phys. 1980, 73, 6238.
- (222) Bak, K. L.; Koch, H.; Oddershede, J.; Christiansen, O.; Sauer, S. P. A. J. Chem. Phys. **2000**, 112, 4173.

- (223) Olsen, J.; Jørgensen, P.; Helgaker, T.; Oddershede, J. J. Phys. Chem. A 2005, 109, 11618.
 - (224) Geertsen, J.; Oddershede, J. J. Chem. Phys. 1986, 85, 2112.
- (225) Geertsen, J.; Eriksen, S.; Oddershede, J. Adv. Quantum Chem. 1991, 22, 167.
- (226) Sauer, S. P. A. J. Phys. B 1997, 30, 3773.
- (227) Enevoldsen, T.; Oddershede, J.; Sauer, S. P. A. Theor. Chem. Acc. 1998, 100, 275.
- (228) Kjær, H.; Sauer, S. P. A.; Kongsted, J. J. Chem. Phys. 2010, 133, 144106.
- (229) Kjær, H.; Sauer, S. P. A.; Kongsted, J.; Rusakov, Y. Y.; Krivdin, L. B. *Chem. Phys.* **2011**, *381*, 35.
- (230) Falden, H. H.; Falster-Hansen, K. R.; Bak, K. L.; Rettrup, S.; Sauer, S. P. A. J. Phys. Chem. A 2009, 113, 11995.
- (231) Paidarova, I.; Sauer, S. P. A. Collect. Czech. Chem. Commun. 2008, 73, 1415.
- (232) Sauer, S. P. A. Adv. Quantum Chem. 2005, 48, 185.
- (233) Schirmer, J. Phys. Rev. A 1982, 26, 2395.
- (234) Barth, A.; Schirmer, J. J. Phys. B 1985, 18, 867.
- (235) Trofimov, A. B.; Schirmer, J. J. Phys. B 1995, 28, 2299.
- (236) Trofimov, A. B.; Stelter, G.; Schirmer, J. J. Chem. Phys. 1999, 111, 9982.
- (237) Trofimov, A. B.; Moskovskaya, T. E.; Gromov, E. V.; Vitkovskaya, N. M.; Schirmer, J. J. Struct. Chem. 2000, 41, 483.
- (238) Hättig, C. Adv. Quantum Chem. 2005, 50, 37.
- (239) Head-Gordon, M.; Oumi, M.; Maurice, D. R. *Mol. Phys.* 1999, 96, 593.
- (240) Schirmer, J.; Barth, A.; Tarantelli, F. Chem. Phys. 1988, 122, 9.
- (241) Schmitt, A.; Schirmer, J. Chem. Phys. 1992, 164, 1.
- (242) Schirmer, J.; Trofimov, A. B.; Randall, K. J.; Feldhaus, J.;
- Bradshaw, A. M.; Lemaire, J.; Chen, C. T.; Sette, F. Phys. Rev. A 1993, 47, 1136.
- (243) Trofimov, A. B.; Krivdina, I. L.; Weller, J.; Schirmer, J. Chem. Phys. 2006, 329, 1.
 - (244) Douglas, M.; Kroll, N. Ann. Phys.-New York 1974, 82, 89.
 - (245) Hess, B. A. Phys. Rev. A 1985, 32, 756.
 - (246) Hess, B. A. Phys. Rev. A 1986, 33, 3742.
- (247) Wolf, A.; Reiher, M.; Hess, B. A. J. Chem. Phys. 2002, 117, 9215.
- (248) McMurchie, L. E.; Davidson, E. R. J. Comput. Phys. 1981, 44, 289.
- (249) Pitzer, R. M.; Winter, N. W. Int. J. Quantum Chem. 1991, 40, 773.
 - (250) Stopkowicz, S.; Gauss, J. J. Chem. Phys. 2008, 129, 164119.
 - (251) Stopkowicz, S.; Gauss, J. J. Chem. Phys. 2011, 134, 064114.
 - (252) Stopkowicz, S.; Gauss, J. J. Chem. Phys. 2011, 134, 204106.
 - (253) Helgaker, T.; Hennum, A. C.; Klopper, W. J. Chem. Phys.

2006, 125, 024102.

- (254) Cheng, L.; Gauss, J. J. Chem. Phys. 2011, 134, 244112.
- (255) Wolf, A.; Reiher, M. J. Chem. Phys. 2006, 124, 064102.
- (256) Wolf, A.; Reiher, M. J. Chem. Phys. 2006, 124, 064103.
- (257) van Stralen, J. N. P.; Visscher, L.; Larsen, C. V.; Jensen, H. J. A. *Chem. Phys.* **2005**, *311*, 81.
- (258) Norman, P.; Schimmelpfennig, B.; Ruud, K.; Jensen, H. J. A.; Ågren, H. J. Chem. Phys. **2002**, *116*, 6914.
- (259) Klopper, W.; Coriani, S.; Helgaker, T.; Jørgensen, P. J. Phys. B 2004, 37, 3753.
- (260) Vaara, J.; Ruud, K.; Vahtras, O. J. Chem. Phys. 1999, 111, 2900.
- (261) Visscher, L.; Saue, T.; Oddershede, J. Chem. Phys. Lett. 1997, 274, 181.
 - (262) Saue, T.; Jensen, H. J. A. J. Chem. Phys. 2003, 118, 522.
 - (263) Norman, P.; Jensen, H. J. A. J. Chem. Phys. 2004, 121, 6145.
 - (264) Bast, R.; Ruud, K.; Rizzo, A.; Helgaker, T. Theor. Chem. Acc.
- **2011**, *129*, 685.
 - (265) Bratož, S. Colloques Int. CNRS 1958, 82, 287.
 - (266) Gerratt, J.; Mills, I. M. J. Chem. Phys. 1968, 49, 1719.
 - (267) Moccia, R. Chem. Phys. Lett. 1970, 5, 260.

- (268) Thomsen, K.; Swanstrøm, P. Mol. Phys. 1973, 26, 735.
- (269) Stanton, J. F.; Gauss, J. Int. Rev. Phys. Chem. 2000, 19, 61.
 (270) Helgaker, T.; Klopper, W.; Tew, D. P. Mol. Phys. 2008,
- (270) Heighter, H., Heipper, H., Fein, D. F. Han, H., 2000, 106, 2107.
- (271) Hald, K.; Halkier, A.; Jørgensen, P.; Coriani, S.; Hättig, C.; Helgaker, T. J. Chem. Phys. 2003, 118, 2985.
- (272) Gauss, J.; Stanton, J. F. Phys. Chem. Chem. Phys. 2000, 2, 2047.
- (273) Urban, M.; Noga, J.; Cole, S. J.; Bartlett, R. J. J. Chem. Phys. 1985, 83, 4041.
- (274) Noga, J.; Bartlett, R. J.; Urban, M. Chem. Phys. Lett. 1987, 134, 126.
- (275) Scuseria, G. E.; Schaefer, H. F. Chem. Phys. Lett. 1988, 146, 23.
- (276) Gauss, J.; Stanton, J. F. J. Chem. Phys. 2002, 116, 1773.
- (277) Kállay, M.; Surjan, P. R. J. Chem. Phys. 2000, 113, 1359.
- (278) Kállay, M.; Surjan, P. R. J. Chem. Phys. 2001, 115, 2945.
- (279) Kállay, M.; Szalay, P. G.; Surjan, P. R. J. Chem. Phys. 2002, 117, 980.
- (280) Kállay, M.; Gauss, J.; Szalay, P. G. J. Chem. Phys. 2003, 119, 2991.
 - (281) Kállay, M.; Gauss, J. J. Chem. Phys. 2004, 120, 6841.
- (282) Lischka, H.; Shepard, R.; Pitzer, R. M.; Shavitt, I.; Dallos, M.;
- Müller, T.; Szalay, P. G.; Seth, M.; Kedziora, G. S.; Yabushita, S.; Zhang, Z. Y. Phys. Chem. Chem. Phys. 2001, 3, 664.
- (283) Lischka, H.; Dallos, M.; Shepard, R. Mol. Phys. 2002, 100, 1647.
- (284) Khait, Y. G.; Theis, D.; Hoffmann, M. R. Mol. Phys. 2010, 108, 2703.
 - (285) Celani, P.; Werner, H.-J. J. Chem. Phys. 2003, 119, 5044.
- (286) Roos, B. O.; Andersson, K.; Fulscher, M. P.; Malmqvist, P.-A.; Serrano-Andres, L.; Pierloot, K.; Merchan, M. Adv. Chem. Phys. **1996**,
- 93, 219.
 - (287) Celani, P.; Werner, H.-J. J. Chem. Phys. 2000, 112, 5546.
- (288) Prochnow, E.; Evangelista, F. A.; Schaefer, H. F.; Allegrini, W. D.; Gauss, J. J. Chem. Phys. **2009**, 131, 064109.
- (289) Jagau, T.-C.; Prochnow, E.; Evangelista, F. A.; Gauss, J. J. Chem. Phys. 2010, 132, 144110.
- (290) Mahapatra, U. S.; Datta, B.; Mukherjee, D. Mol. Phys. 1998, 94, 157.
- (291) Bak, K. L.; Halkier, A.; Jørgensen, P.; Olsen, J.; Helgaker, T.; Klopper, W. J. Mol. Struct. 2001, 567, 375.
- (292) Kutzelnigg, W. Theor. Chim. Acta 1985, 68, 445.
- (293) Klopper, W.; Kutzelnigg, W. Chem. Phys. Lett. 1987, 134, 17.
- (294) Ten-no, S. J. Chem. Phys. 2004, 121, 117.
- (295) May, A. J.; Valeev, E. F.; Polly, R.; Manby, F. R. Phys. Chem. Chem. Phys. 2005, 7, 2710.
- (296) Distasio, R. A.; Steele, R. P.; Head-Gordon, M. Mol. Phys. 2007, 105, 2731.
- (297) Neese, F.; Wennmohs, F.; Hansen, A.; Becker, U. *Chem. Phys.* 2009, 356, 98.
- (298) Kossmann, S.; Neese, F. J. Chem. Theory Comput. 2010, 6, 2325.
- (299) Hellweg, A.; Grün, S. A.; Hättig, C. Phys. Chem. Chem. Phys. 2008, 10, 4119.
- (300) Aquilante, F.; Lindh, R.; Pedersen, T. B. J. Chem. Phys. 2008, 129, 034106.
- (301) Reine, S.; Krapp, A.; Iozzi, M. F.; Bakken, V.; Helgaker, T.; Pawlowski, F.; Sałek, P. J. Chem. Phys. **2010**, *133*, 044102.
- (302) White, C.; Johnson, B. G.; Gill, P. M. W.; Head-Gordon, M. Chem. Phys. Lett. **1994**, 230, 8.
- (303) Reine, S.; Tellgren, E. I.; Helgaker, T. Phys. Chem. Chem. Phys. 2007, 9, 4771.
- (304) Dominguez-Soria, V. D.; Geudtner, G.; Morales, J. L.; Calaminici, P.; Köster, A. M. J. Chem. Phys. 2009, 131, 124102.
- (305) Schütz, M.; Werner, H.-J.; Lindh, R.; Manby, F. R. J. Chem. Phys. 2004, 121, 737.
- (306) Rauhut, G.; Werner, H.-J. Phys. Chem. Chem. Phys. 2001, 3, 4853.

- (307) Leininger, T.; Angeli, C.; Evangelisti, S.; Cimiraglia, R.; Maynau, D. Chem. Phys. Lett. 2003, 371, 49.
- (308) Sosa, C.; Geertsen, J.; Trucks, G. W.; Bartlett, R. J.; Franz, J. A. Chem. Phys. Lett. **1989**, 159, 148.
- (309) Taube, A. G.; Bartlett, R. J. Collect. Czech. Chem. Commun. 2005, 70, 837.
- (310) Klopper, W.; Noga, J.; Koch, H.; Helgaker, T. Theor. Chem. Acc. 1997, 97, 164.
 - (311) Taube, A. G.; Bartlett, R. J. J. Chem. Phys. 2008, 128, 164101.
 - (312) Kucharski, S. A.; Bartlett, R. J. J. Chem. Phys. 1998, 108, 5243.
- (313) Crawford, T. D.; Stanton, J. F. Int. J. Quantum Chem. 1998, 70, 601.
- (314) Perera, S. A.; Rozyczko, P. B.; Bartlett, R. J.; Hirata, S. Mol. Phys. 2005, 103, 2081.
 - (315) Taube, A. G.; Bartlett, R. J. J. Chem. Phys. 2008, 128, 044110.
 - (316) Taube, A. G.; Bartlett, R. J. J. Chem. Phys. 2008, 128, 044111.
 - (317) Michauk, C.; Gauss, J. J. Chem. Phys. 2007, 127, 044106.
 - (318) Wang, F.; Gauss, J. J. Chem. Phys. 2008, 129, 174110.
 - (319) Wang, F.; Gauss, J. J. Chem. Phys. 2009, 131, 164113.
- (320) Wang, F.; Gauss, J.; van Wüllen, C. J. Chem. Phys. 2008,
- 129, 064113. (321) Visscher, L.; Dyall, K. G.; Lee, T. J. Int. J. Quantum Chem.
- (321) Visscher, L.; Dyan, K. G.; Lee, T. J. Int. J. Quantum Chem. 1995, 411.
- (322) Visscher, L.; Lee, T. J.; Dyall, K. G. J. Chem. Phys. 1996, 105, 8769.
- (323) Zou, W.; Filatov, M.; Cremer, D. J. Chem. Phys. 2011, 134, 244117.
 - (324) Filatov, M.; Cremer, D. J. Chem. Phys. 2005, 122, 064104.
- (325) Ishimura, K.; Pulay, P.; Nagase, S. J. Comput. Chem. 2007, 28, 2034.
- (326) Harding, M. E.; Metzroth, T.; Gauss, J.; Auer, A. A. J. Chem. Theory Comput. 2008, 4, 64.
- (327) Lotrich, V. F.; Flocke, N.; Ponton, M.; Yau, A. D.; Perera, S. A.; Deumens, E.; Bartlett, R. J. *J. Chem. Phys.* **2008**, *128*, 194104.
- (328) Yanai, T.; Fann, G. I.; Gan, Z. T.; Harrison, R. J.; Beylkin, G. J. Chem. Phys. 2004, 121, 2866.
 - (329) Foley, J. J.; Mazziotti, D. A. Mol. Phys. 2010, 108, 2543.
 - (330) Kassal, I.; Aspuru-Guzik, A. J. Chem. Phys. 2009, 131, 224102.
 - (331) Lanyon, B. P.; Whitfield, J. D.; Gillett, G. G.; Goggin, M. E.;
- Almeida, M. P.; Kassal, I.; Biamonte, J. D.; Mohseni, M.; Powell, B. J.;
- Barbieri, M.; Aspuru-Guzik, A.; White, A. G. Nat. Chem. 2010, 2, 106.
- (332) Colwell, S. M.; Jayatilaka, D.; Maslen, P. E.; Amos, R. D.; Handy, N. C. Int. J. Quantum Chem. 1991, 40, 179.
- (333) Maslen, P. E.; Jayatilaka, D.; Colwell, S. M.; Amos, R. D.; Handy, N. C. J. Chem. Phys. **1991**, 95, 7409.
- (334) Jayatilaka, D.; Maslen, P. E.; Amos, R. D.; Handy, N. C. Mol. Phys. **1992**, 75, 271.
- (335) Maslen, P. E.; Handy, N. C.; Amos, R. D.; Jayatilaka, D. J. Chem. Phys. **1992**, 97, 4233.
- (336) Breidung, J.; Thiel, W.; Gauss, J.; Stanton, J. F. J. Chem. Phys. 1999, 110, 3687.
 - (337) Barone, V. J. Chem. Phys. 2004, 120, 3059.
 - (338) Barone, V. J. Chem. Phys. 2005, 122, 014108.
 - (339) Amos, R. D. Chem. Phys. Lett. 1984, 108, 185.
- (340) Helgaker, T.; Jensen, H. J. A.; Jørgensen, P. J. Chem. Phys. 1986, 84, 6280.
- (341) Mohammed, A.; Ågren, H.; Thorvaldsen, A. J.; Ruud, K. *Chem. Phys. Lett.* **2010**, *485*, 320.
 - (342) Ruud, K.; Thorvaldsen, A. J. *Chirality* **2010**, *21*, E54.
 - (343) Herzberg, G.; Teller, E. Z. Phys. Chem. B **1933**, 21, 410.
 - (344) Santoro, F.; Lami, A.; Improta, R.; Bloino, J.; Barone, V.
- J. Chem. Phys. 2008, 128, 224311.
- (345) Coriani, S.; Kjærgaard, T.; Jørgensen, P.; Ruud, K.; Huh, J.; Berger, R. J. Chem. Theory Comput. **2010**, *6*, 1028.
 - (346) Quinet, O.; Champagne, B. J. Chem. Phys. 2001, 115, 6293.
- (347) Rappoport, D.; Furche, F. J. Chem. Phys. 2007, 126, 201104.
- (348) O'Neill, D. P.; Kállay, M.; Gauss, J. Mol. Phys. 2007, 105,
- 2447.

(349) Helgaker, T.; Jaszuński, M.; Pecul, M. Prog. Nucl. Mag. Resonan. Spectrosc. 2008, 53, 249.

- (350) Ramsey, N. F. Phys. Rev. 1950, 78, 699.
- (351) Beer, M.; Kussmann, J.; Ochsenfeld, C. J. Chem. Phys. 2011, 134, 074102.
- (352) Wolinski, K.; Hinton, J. F.; Pulay, P. J. Am. Chem. Soc. 1990, 112.8251.
 - (353) Kutzelnigg, W. Isr. J. Chem. 1980, 19, 193.
 - (354) Schindler, M.; Kutzelnigg, W. J. Chem. Phys. 1982, 76, 1919.
- (355) Hansen, A. E.; Bouman, T. D. J. Chem. Phys. 1985, 82, 5035.
- (356) Keith, T. A.; Bader, R. F. W. Chem. Phys. Lett. 1992, 194, 1.
- (357) Lazzeretti, P.; Malagoli, M.; Zanasi, R. Chem. Phys. Lett. 1994, 220, 299.
- (358) Coriani, S.; Lazzeretti, P.; Malagoli, M.; Zanasi, R. Theor. Chim. Acta 1994, 89, 181.
- (359) Zanasi, R.; Lazzeretti, P.; Malagoli, M.; Piccinini, F. J. Chem. Phys. 1995, 102, 7150.
- (360) Bühl, M.; Kaupp, M.; Malkina, O. L.; Malkin, V. G. J. Comput. Chem. 1999, 20, 91.
 - (361) Autschbach, J. Coord. Chem. Rev. 2007, 251, 1796.
 - (362) Gauss, J. Chem. Phys. Lett. 1992, 191, 614.
 - (363) Gauss, J.; Stanton, J. F. Adv. Chem. Phys. 2002, 123, 355.
 - (364) Vaara, J. Phys. Chem. Chem. Phys. 2007, 9, 5399.
- (365) Gauss, J.; Ruud, K. Int. J. Quantum Chem. 1995, 55, 437.
- (366) Vaara, J.; Lounila, J.; Ruud, K.; Helgaker, T. J. Chem. Phys. 1998, 109, 8388.
- (367) Sundholm, D.; Gauss, J.; Schäfer, A. J. Chem. Phys. 1996, 105, 11051.
- (368) Ruud, K.; Åstrand, P.-O.; Taylor, P. R. J. Chem. Phys. 2000, 112, 2668.
- (369) Ruud, K.; Åstrand, P.-O.; Taylor, P. R. J. Am. Chem. Soc. 2001, 123, 4826.
- (370) Dransfeld, A. Chem. Phys. 2004, 298, 47.
- (371) Böhm, M. C.; Schulte, J.; Ramirez, R. Int. J. Quantum Chem. 2002, 86, 280.
 - (372) Böhm, M. C.; Ramirez, R.; Schulte, J. Chem. Phys. 2007, 342, 1.
- (373) Auer, A. A.; Gauss, J.; Stanton, J. F. J. Chem. Phys. 2003, 118.10407.
- (374) Prochnow, E.; Auer, A. A.; Banert, K. J. Phys. Chem. A 2007, 111, 9945.
- (375) Harding, M. E.; Lenhart, M.; Auer, A. A.; Gauss, J. J. Chem. Phys. 2008, 128, 244111.
 - (376) Auer, A. A. J. Chem. Phys. 2009, 131, 024116.
- (377) Prochnow, E.; Auer, A. A. J. Chem. Phys. 2010, 132, 064109.
- (378) Perera, S. A.; Gregusova, A.; Bartlett, R. J. J. Phys. Chem. A 2009, 113, 3197.
- (379) Gregusova, A.; Perera, S. A.; Bartlett, R. J. J. Chem. Theory Comput. 2010, 6, 1228.
- (380) Harding, M. E.; Gauss, J.; von Rague Schleyer, P. J. Phys. Chem. A 2011, 115, 2340.
- (381) Ligabue, A.; Sauer, S. P. A.; Lazzeretti, P. J. Chem. Phys. 2003, 118, 6830.
- (382) Ligabue, A.; Sauer, S. P. A.; Lazzeretti, P. J. Chem. Phys. 2007, 126, 154111.
- (383) Cuesta, I. G.; Marin, J. S.; de Meras, A. M. J. S.; Pawlowski, F.; Lazzeretti, P. Phys. Chem. Chem. Phys. 2010, 12, 6163.
 - (384) Manninen, P.; Vaara, J. J. Comput. Chem. 2006, 27, 434.
 - (385) Chong, D. P. Can. J. Chem. 1995, 73, 79.
- (386) Kupka, T.; Ruscic, B.; Botto, R. E. J. Phys. Chem. A 2002, 106, 10396.
- (387) Kupka, T.; Ruscic, B.; Botto, R. E. Solid State Nucl. Magn. Reson. 2003, 23, 145.
- (388) Kupka, T.; Lim, C. J. Phys. Chem. A 2007, 111, 1927.
- (389) Kupka, T.; Stachow, M.; Nieradka, M.; Kaminsky, J.; Pluta, T.; Sauer, S. P. A. Magn. Reson. Chem. 2011, 49, 231.
- (390) de Oliveira, P. J. P.; Jorge, F. E. J. Phys. B 2008, 41, 145101.

- (391) Helgaker, T.; Klopper, W.; Koch, H.; Noga, J. J. Chem. Phys. 1997, 106, 9639.
- (392) Antušek, A.; Jackowski, K.; Jaszuński, M.; Makulski, W.; Wilczek, M. Chem. Phys. Lett. 2005, 411, 111.
- (393) Makulski, W.; Jackowski, K.; Antušek, A.; Jaszuński, M. J. Phys. Chem. A 2006, 110, 11462.
- (394) Jackowski, K.; Makulski, W.; Szyprowska, A.; Antušek, A.;
- Jaszuński, M.; Juselius, J. J. Chem. Phys. 2009, 130, 044309.
 - (395) Rizzo, A.; Gauss, J. J. Chem. Phys. 2002, 116, 869.
- (396) Vaara, J.; Manninen, P.; Lounila, J. Chem. Phys. Lett. 2003, 372, 750.
- (397) Kjær, H.; Sauer, S. P. A.; Kongsted, J. J. Chem. Phys. 2011, 134, 044514.
 - (398) Laubender, G.; Berger, R. Phys. Rev. A 2006, 74, 032105.
- (399) Weijo, V.; Bast, R.; Manninen, P.; Saue, T.; Vaara, J. J. Chem. Phys. 2007, 126, 074107.
- (400) Buckingham, A. D.; Parlett, L. C. Mol. Phys. 1997, 91, 805. (401) Jaszuński, M.; Rizzo, A. Mol. Phys. 1999, 96, 855.
- (402) Vaara, J.; Ruud, K.; Vahtras, O.; Ågren, H.; Jokisaari, J. J. Chem.
- Phys. 1998, 109, 1212.
- (403) Manninen, P.; Lantto, P.; Vaara, J.; Ruud, K. J. Chem. Phys. 2003, 119, 2623.
- (404) Manninen, P.; Ruud, K.; Lantto, P.; Vaara, J. J. Chem. Phys. 2005, 122, 114107.
- (405) Fukuda, R.; Hada, M.; Nakatsuji, H. J. Chem. Phys. 2003, 118, 1015.
- (406) Fukuda, R.; Hada, M.; Nakatsuji, H. J. Chem. Phys. 2003, 118, 1027.
 - (407) Fukuda, R.; Nakatsuji, H. J. Chem. Phys. 2005, 123, 044101.
- (408) Aucar, G. A.; Saue, T.; Visscher, L.; Jensen, H. J. A. J. Chem. Phys. 1999, 110, 6208.
- (409) Sternheim, M. M. Phys. Rev. 1962, 128, 676.
- (410) Stanton, R. E.; Havriliak, S. J. Chem. Phys. 1984, 81, 1910.
- (411) Dyall, K. G.; Fægri, K., Jr. Chem. Phys. Lett. 1990, 174, 25.
- (412) Fægri, K., Jr. Theor. Chem. Acc. 2001, 105, 252.
- (413) Komorovsky, S.; Repisky, M.; Malkina, O. L.; Malkin, V. G.; Ondik, I. M.; Kaupp, M. J. Chem. Phys. 2008, 128, 104101.
- (414) Cheng, L.; Xiao, Y.; Liu, W. J. Chem. Phys. 2009, 131, 244113.
- (415) Komorovsky, S.; Repisky, M.; Malkina, O. L.; Malkin, V. G. I. Chem. Phys. 2010, 132, 154101.
 - (416) Hamaya, S.; Fukui, H. Bull. Chem. Soc. Jpn. 2010, 83, 635.
- (417) Ilias, M.; Saue, T.; Enevoldsen, T.; Jensen, H. J. A. J. Chem. Phys. 2009, 131, 124119.
- (418) Kudo, K.; Maeda, H.; Kawakubo, T.; Ootani, Y.; Funaki, M.; Fukui, H. J. Chem. Phys. 2006, 124, 224106.
- (419) Ootani, Y.; Yamaguti, H.; Maeda, H.; Fukui, H. J. Chem. Phys. 2006, 125, 164106.
 - (420) Chang, C.; Pelissier, M.; Durand, P. Phys. Scr. 1986, 34, 394. (421) Heully, J. L.; Lindgren, I.; Lindroth, E.; Lundqvist, S.;
- Mårtensson-Pendrill, A. M. J. Phys. B 1986, 19, 2799.
- (422) Kudo, K.; Fukui, H. J. Chem. Phys. 2005, 123, 114102. Erratum: ibid. 2006, 124, 209901.
- (423) Kato, M.; Hada, M.; Fukuda, R.; Nakatsuji, H. Chem. Phys. Lett. 2005, 408, 150.
- (424) Gomez, S. S.; Romero, R. H.; Aucar, G. A. J. Chem. Phys. 2002, 117, 7942.
- (425) Gomez, S. S.; Maldonado, A.; Aucar, G. A. J. Chem. Phys. 2005, 123, 214108.
- (426) Juselius, J.; Sundholm, D.; Gauss, J. J. Chem. Phys. 2004, 121, 3952.
- (427) Taubert, S.; Sundholm, D.; Juselius, J. J. Chem. Phys. 2011, 134.054123.
 - (428) Bast, R.; Juselius, J.; Saue, T. Chem. Phys. 2009, 356, 187.
- (429) Juselius, J.; Sundholm, D. Phys. Chem. Chem. Phys. 1999, 1.3429.
- (430) von Rague Schleyer, P.; Maerker, C.; Dransfeld, A.; Jiao, H. J.; van Eikema Hommes, N. J. R. J. Am. Chem. Soc. 1996, 118, 6317.

(431) Gauss, J.; Werner, H.-J. Phys. Chem. Chem. Phys. 2000, 2, 2083.

(432) Ochsenfeld, C.; Kussmann, J.; Koziol, F. Angew. Chem., Int. Ed. 2004, 43, 4485.

- (433) Loibl, S.; Manby, F. R.; Schütz, M. Mol. Phys. 2010, 108, 477.
- (434) Karadakov, P. B.; Morokuma, K. Chem. Phys. Lett. 2000, 317, 589.

(435) Rickard, G. A.; Karadakov, P. B.; Webb, G. A.; Morokuma, K. J. Phys. Chem. A **2003**, 107, 292.

- (436) Ramsey, N. F. Phys. Rev. 1953, 91, 303.
- (437) Watson, M. A.; Sałek, P.; Lemaire, J.; Jaszuński, M.; Helgaker, T. *Chem.—Eur. J.* **2004**, *10*, 4627.
- (438) Malkin, V. G.; Malkina, O. L.; Salahub, D. R. Chem. Phys. Lett. 1994, 221, 91.
- (439) Malkina, O. L.; Salahub, D. R.; Malkin, V. G. J. Chem. Phys. 1996, 105, 8793.
- (440) Sychrovsky, V.; Gräfenstein, J.; Cremer, D. J. Chem. Phys. 2000, 113, 3530.
- (441) Helgaker, T.; Watson, M. A.; Handy, N. C. J. Chem. Phys. 2000, 113, 9402.
- (442) Lutnæs, O. B.; Helgaker, T.; Jaszuński, M. Mol. Phys. 2010, 108, 2579.
- (443) Kowalewski, J.; Roos, B. O.; Siegbahn, P.; Vestin, R. Chem. Phys. **1974**, 3, 70.
 - (444) Pecul, M. J. Chem. Phys. 2000, 113, 10835.
- (445) Auer, A. A.; Gauss, J.; Pecul, M. Chem. Phys. Lett. 2003, 368, 172.
- (446) Vahtras, O.; Ågren, H.; Jørgensen, P.; Jensen, H. J. A.; Padkjær, S. B.; Helgaker, T. *J. Chem. Phys.* **1992**, *96*, 6120.
- (447) Jaszuński, M.; Ruud, K. Chem. Phys. Lett. **2001**, 336, 473.
- (448) Pecul, M.; Dodziuk, H.; Jaszuński, M.; Lukin, O.; Leszczynski,
- J. Phys. Chem. Chem. Phys. 2001, 3, 1986.
- (449) Perera, S. A.; Sekino, H.; Bartlett, R. J. J. Chem. Phys. 1994, 101, 2186.
 - (450) Geertsen, J.; Oddershede, J. Chem. Phys. 1984, 90, 301.
- (451) Fukui, H.; Miura, K.; Matsuda, H.; Baba, T. J. Chem. Phys. 1992, 97, 2299.
 - (452) Fukui, H.; Baba, T.; Kurogi, Y. J. Chem. Phys. 2000, 112, 3532.
 - (453) Wigglesworth, R. D.; Raynes, W. T.; Kirpekar, S.; Sauer, S.P.
- A; Oddershede, J. J. Chem. Phys. 2000, 112, 3735. Erratum ibid. 2001, 114, 9192.
- (454) Åstrand, P.-O.; Ruud, K.; Mikkelsen, K. V.; Helgaker, T. J. Chem. Phys. **1999**, 110, 9463.
- (455) Jordan, M. J. T.; Toh, J. S. S.; Bene, J. E. D. Chem. Phys. Lett. 2001, 346, 288.
- (456) Ruden, T. A.; Lutnæs, O. B.; Helgaker, T.; Ruud, K. J. Chem. Phys. 2003, 118, 9572.
- (457) Ruden, T. A.; Helgaker, T.; Jaszuński, M. Chem. Phys. 2004, 296, 53.
 - (458) Flygare, W. H. Chem. Rev. 1974, 74, 653.
 - (459) Oddershede, J.; Geertsen, J. J. Chem. Phys. 1990, 92, 6036.
- (460) Paidarova, I.; Komasa, J.; Oddershede, J. Mol. Phys. **1991**, 72, 559.
- (461) Sauer, S. P. A; Spirko, V.; Oddershede, J. Chem. Phys. 1994, 184, 1.
- (462) Gauss, J.; Ruud, K.; Helgaker, T. J. Chem. Phys. 1996, 105, 2804.
 - (463) Gauss, J.; Sundholm, D. Mol. Phys. 1997, 91, 449.
- (464) Ruud, K.; Helgaker, T.; Jørgensen, P. J. Chem. Phys. 1997, 107, 10599.
- (465) Coriani, S.; Jaszuński, M.; Rizzo, A.; Ruud, K. Chem. Phys. Lett. 1998, 287, 677.
- (466) Puzzarini, C.; Coriani, S.; Rizzo, A.; Gauss, J. Chem. Phys. Lett. 2005, 409, 118.
- (467) Hansen, N.; Mäder, H.; Temps, F. Chem. Phys. Lett. 2005, 409, 149.
 - (468) Jameson, C. J. Chem. Rev. 1991, 91, 1375.
 - (469) Bryce, D. L.; Wasylishen, R. E. Acc. Chem. Res. 2003, 36, 327.
- (470) Wasylishen, R. E.; Bryce, D. L. J. Chem. Phys. 2002, 117, 10061.

- (471) Cybulski, S. M.; Bishop, D. M. J. Chem. Phys. 1997, 106, 4082.
- (472) Puzzarini, C. Theor. Chem. Acc. 2008, 121, 1.
- (473) Puzzarini, C.; Cazzoli, G.; Harding, M. E.; Vazquez, J.; Gauss, J. J. Chem. Phys. **2009**, 131, 234304.
- (474) Harriman, J. E. Theoretical Foundations of Electron Spin Resonance; Academic Press: New York, 1978.
- (475) Moreira, I. D. P. R; Suaud, N.; Guihery, N.; Malrieu, J.-P.; Caballol, R.; Bofill, J. M.; Illas, F. *Phys. Rev. B* **2002**, *66*, 134430.
- (476) Ghosh, A.; Taylor, P. R. Curr. Opin. Chem. Biol. 2003, 7, 113.
- (477) Munoz, D.; Graaf, C. D.; Illas, F. J. Comput. Chem. 2004, 25, 1234.
 - (478) Gutlich, P.; Goodwin, H. A. Top. Curr. Chem. 2004, 233, 1.
 - (479) Bolvin, H. ChemPhysChem 2006, 7, 1575.
- (480) Rinkevicius, Z.; de Almeida, K. J.; Oprea, C. I.; Vahtras, O.; Ågren, H.; Ruud, K. J. Chem. Theory Comput. **2008**, *4*, 1810.
- (481) van Lenthe, E.; Wormer, P. E. S.; van der Avoird, A. J. Chem. Phys. **1997**, 107, 2488.
- (482) Hrobarik, P.; Malkina, O. L.; Malkin, V. G.; Kaupp, M. Chem. Phys. 2009, 356, 229.
- (483) Repisky, M.; Komorovsky, S.; Malkin, E.; Malkina, O. L.; Malkin, V. G. Chem. Phys. Lett. 2010, 488, 94.
 - (484) Lushington, G. H.; Grein, F. Theor. Chim. Acta 1996, 93, 259.
 - (485) Lushington, G. H.; Grein, F. J. Chem. Phys. 1997, 106, 3292.
- (486) Manninen, P.; Vaara, J.; Ruud, K. J. Chem. Phys. 2004, 121, 1258.
- (487) Geurts, P. J. M.; Bouten, P. C. P.; van der Avoird, A. J. Chem. Phys. **1980**, 73, 1306.
- (488) Ding, Z.; Gulla, A. F.; Budil, D. J. Chem. Phys. 2001, 115, 10685.
- (489) Tachikawa, H. Chem. Phys. Lett. 1996, 260, 582.
- (490) Lushington, G. H.; Grein, F. Int. J. Quantum Chem. 1996, 60, 1679.
- (491) Engström, M.; Minaev, B. F.; Vahtras, O.; Ågren, H. Chem. Phys. **1998**, 237, 149.
- (492) Engström, M.; Himo, F.; Gräslund, A.; Minaev, B. F.; Vahtras, O.; Ågren, H. J. Phys. Chem. A **2000**, 104, 5149.
 - (493) Lushington, G. H. J. Phys. Chem. A 2000, 104, 2969.
 - (494) Bruna, P. J.; Grein, F. Chem. Phys. Lett. 2000, 318, 263.
- (495) Vahtras, O.; Engström, M.; Schimmelpfennig, B. Chem. Phys. Lett. 2002, 351, 424.
 - (496) Neese, F. Magn. Reson. Chem. 2004, 42, S187.
- (497) van Gastel, M.; Lubitz, W.; Lassmann, G.; Neese, F. J. Am. Chem. Soc. 2004, 126, 2237.
- (498) Malmqvist, P.-A.; Roos, B. O.; Schimmelpfennig, B. Chem. Phys. Lett. 2002, 357, 230.
- (499) Gerloch, M.; McMeeking, R. F. J. Chem. Soc., Dalton Trans. 1975, 2443.
- (500) Bolvin, H. Inorg. Chem. 2007, 46, 417.
- (501) Chibotaru, L. F.; Ceulemans, A.; Bolvin, H. Phys. Rev. Lett. 2008, 101, 033003.
- (502) Vancoillie, S.; Malmqvist, P.-A.; Pierloot, K. ChemPhysChem 2007, 8, 1803.
- (503) Vancoillie, S.; Pierloot, K. J. Phys. Chem. A 2008, 112, 4011.
- (504) Neese, F.; Solomon, E. I. Inorg. Chem. 1998, 37, 6568.
- (505) Gauss, J.; Kállay, M.; Neese, F. J. Phys. Chem. A 2009, 113, 11541.
 - (506) Quiney, H. M.; Belanzoni, P. Chem. Phys. Lett. 2002, 353, 253.
 - (507) Kacprzak, S.; Kaupp, M. J. Phys. Chem. B 2004, 108, 2464.
 - (508) Neese, F.; Solomon, E. I. In Magnetism: Molecules to Materials
- IV; Miller, J. S., Drillon, M., Eds.; Wiley-VCH: Weinheim, 2003; p 345.
 (509) Neese, F. J. Am. Chem. Soc. 2006, 128, 10213.
- (510) Loboda, O.; Minaev, B. F.; Vahtras, O.; Ruud, K.; Ågren, H.
- J. Chem. Phys. 2003, 119, 3120.
- (511) Minaev, B. F.; Loboda, O.; Vahtras, O.; Ruud, K.; Ågren, H. *Theor. Chem. Acc.* **2004**, *111*, 168.
- (512) Ganyushin, D.; Gilka, N.; Taylor, P. R.; Marian, C. M.; Neese, F. J. Chem. Phys. **2010**, *132*, 144111.
 - (513) Neese, F. J. Chem. Phys. 2007, 127, 164112.

625

- (514) Sinnecker, S.; Neese, F. J. Phys. Chem. A 2006, 110, 12267.
- (515) Zein, S.; Neese, F. J. Phys. Chem. A 2008, 112, 7976.
- (516) Duboc, C.; Ganyushin, D.; Sivalingam, K.; Collomb, M.-N.; Neese, F. J. Phys. Chem. A **2010**, *114*, 10750.
- (517) Vancoillie, S.; Chalupsky, J.; Ryde, U.; Solomon, E. I.; Pierloot, K.; Neese, F.; Rulisek, L. J. Phys. Chem. B **2010**, *114*, 7692.
- (518) Gilka, N.; Taylor, P. R.; Marian, C. M. J. Chem. Phys. 2008, 129, 044102.
 - (519) Pederson, M.; Khanna, S. Phys. Rev. B 1999, 60, 9566.
- (520) Schmitt, S.; Jost, P.; van Wüllen, C. J. Chem. Phys. 2011, 134, 194113.
- (521) Neese, F. J. Chem. Phys. 2003, 118, 3939.
- (522) Arbuznikov, A. V.; Vaara, J.; Kaupp, M. J. Chem. Phys. 2004, 120, 2127.
- (523) Sekino, H.; Bartlett, R. J. J. Chem. Phys. 1985, 82, 4225.
- (524) Perera, S. A.; Watts, J. D.; Bartlett, R. J. J. Chem. Phys. 1994, 100, 1425.
- (525) Momose, T.; Nakatsuji, H.; Shida, T. J. Chem. Phys. 1988, 89, 4185.
- (526) Momose, T.; Yamaguchi, M.; Shida, T. J. Chem. Phys. 1990, 93, 7284.
 - (527) Feller, D.; Davidson, E. R. J. Chem. Phys. 1984, 80, 1006.
- (528) Engels, B.; Peyerimhoff, S. D.; Davidson, E. R. *Mol. Phys.* **1987**, 62, 109.
 - (529) Engels, B.; Peyerimhoff, S. D. Mol. Phys. 1989, 67, 583.
 - (530) Kossmann, S.; Neese, F. J. Phys. Chem. A **2010**, 114, 11768.
 - (531) Grimme, S. J. Chem. Phys. 2003, 118, 9095.
- (532) Fernandez, B.; Jørgensen, P.; Byberg, J. R.; Olsen, J.; Helgaker, T.; Jensen, H. J. A. *J. Chem. Phys.* **1992**, *97*, 3412.
- (533) Rinkevicius, Z.; Telyatnyk, L.; Vahtras, O.; Ågren, H. J. Chem. Phys. 2004, 121, 7614.
- (534) Barone, V. In Recent Advances in Density Functional Methods, Part 1; D. P. Chong,, Ed.; World Scientific: Singapore, 1995; p 287.
- (535) Hiller, J.; Sucher, J.; Feinberg, G. Phys. Rev. A 1978, 18, 2399.
- (536) Rassolov, V. A.; Chipman, D. M. Theor. Chim. Acta 1994, 88, 339.
- (537) Rassolov, V. A.; Chipman, D. M. J. Chem. Phys. **1996**, 105, 1470. (538) Rassolov, V. A.; Chipman, D. M. J. Chem. Phys. **1996**,
- 105, 1479.
 - (539) Sundholm, D. J. Chem. Phys. 1995, 102, 4895.
 - (540) Improta, R.; Barone, V. Chem. Rev. 2004, 104, 1231.
- (541) Condon, E. U.; Shortley, G. H. *The Theory of Atomic Spectra*; Cambridge University Press: Cambridge, 1935.
- (542) Kroto, H. W. Molecular Rotation Spectra; Dover: New York, 1992.
- (543) Kremer, F.; Schönhals, A. Broadband Dielectric Spectroscopy; Springer: Berlin, 2002.
 - (544) Buckingham, A. D. J. Chem. Phys. 1959, 30, 1580.
- (545) Halkier, A.; Larsen, H.; Olsen, J.; Jørgensen, P.; Gauss, J. J. Chem. Phys. **1999**, 110, 734.
- (546) Halkier, A.; Klopper, W.; Helgaker, T.; Jørgensen, P. J. Chem. Phys. **1999**, 111, 4424.
- (547) Bak, K. L.; Gauss, J.; Helgaker, T.; Jørgensen, P.; Olsen, J. Chem. Phys. Lett. 2000, 319, 563.
 - (548) Halkier, A.; Taylor, P. R. Chem. Phys. Lett. 1998, 285, 133.
 - (549) Puzzarini, C. Phys. Chem. Chem. Phys. 2004, 6, 344.
- (550) Halkier, A.; Larsen, H.; Olsen, J.; Jørgensen, P. J. Chem. Phys. 1999, 110, 7127.
- (551) Larsen, H.; Halkier, A.; Olsen, J.; Jørgensen, P. J. Chem. Phys. 2000, 112, 1107.
 - (552) Hellweg, A. J. Chem. Phys. 2011, 134, 064103.
- (553) Koerber, M.; Baum, O.; Hahn, U.; Gauss, J.; Giesen, T. F.; Schlemmer, S. J. Mol. Spectrosc. **2009**, 257, 34.
- (554) Jankowski, P.; Ziolkowski, M. J. Chem. Phys. 2008, 128, 034308.
- (555) Coriani, S.; Hättig, C.; Jørgensen, P.; Rizzo, A.; Ruud, K. J. Chem. Phys. **1998**, 109, 7176.
 - (556) Halkier, A.; Coriani, S. Chem. Phys. Lett. 1999, 303, 408.

- (557) Rizzo, A.; Coriani, S.; Halkier, A.; Hättig, C. J. Chem. Phys. 2000, 113, 3077.
- (558) Coriani, S.; Halkier, A.; Rizzo, A.; Ruud, K. Chem. Phys. Lett. 2000, 326, 269.
 - (559) Halkier, A.; Coriani, S. Chem. Phys. Lett. 2001, 346, 329.
- (560) Coriani, S.; Halkier, A.; Jonsson, D.; Gauss, J.; Rizzo, A.; Christiansen, O. J. Chem. Phys. 2003, 118, 7329.

(561) Buckingham, A. D.; Coriani, S.; Rizzo, A. Theor. Chem. Acc. 2007, 117, 969. Erratum ibid. 2007, 117, 979.

- (562) Ritchie, G. L. D.; Watson, J. N.; Keir, R. I. Chem. Phys. Lett. 2003, 370, 376.
 - (563) Chetty, N.; Couling, V. W. Mol. Phys. 2011, 109, 655.
 - (564) Chetty, N.; Couling, V. W. J. Chem. Phys. 2011, 134, 144307.
 - (565) Chetty, N.; Couling, V. W. J. Chem. Phys. 2011, 134, 164307.
 - (566) Graham, C.; Imrie, D.; Raab, R. E. Mol. Phys. 1998, 93, 49.
- (567) Halkier, A.; Coriani, S.; Jørgensen, P. Chem. Phys. Lett. 1998, 294, 292.
- (568) Buckingham, A. D. Intermolecular Forces. In *Advances in Chemical Physics*; Hirschfelder, J. O., Ed.; Wiley: New York, 1967; Vol. 12.
- (569) Raab, R. E.; de Lange, O. L. In *Multipole Theory in Electromagnetism. Classical, Quantum and Symmetry Aspects, with Applications;* Birman, J., Edwards, S. F., Friend, R., Rees, M., Sherrington, D., Veneziano, G., Eds.; International Series of Monographs in Physics; Clarendon Press: Oxford, 2005; Vol. 128.
- (570) Christiansen, O.; Hättig, C.; Gauss, J. J. Chem. Phys. 1998, 109, 4745.
- (571) Larsen, H.; Olsen, J.; Hättig, C.; Jørgensen, P.; Christiansen, O.; Gauss, J. J. Chem. Phys. **1999**, 111, 1917.
- (572) Sałek, P.; Helgaker, T.; Vahtras, O.; Ågren, H.; Jonsson, D.; Gauss, J. *Mol. Phys.* **2005**, *103*, 439.
- (573) Maroulis, G. Computational Aspects Of Electric Polarizability Calculations: Atoms, Molecules and Clusters; IOS Press: Amsterdam, 2006.
- (574) Jonsson, D.; Norman, P.; Luo, Y.; Ågren, H. J. Chem. Phys. 1996, 105, 581.
- (575) Norman, P.; Jonsson, D.; Ågren, H. Chem. Phys. Lett. 1997, 268, 337.
 - (576) Stanton, J. F.; Gauss, J. J. Chem. Phys. 1994, 100, 4695.
 - (577) Stanton, J. F.; Gauss, J. Theor. Chim. Acta 1995, 91, 267.
- (578) Hättig, C.; Christiansen, O.; Coriani, S.; Jørgensen, P. J. Chem. Phys. **1998**, 109, 9237.
 - (579) Hättig, C.; Jørgensen, P. J. Chem. Phys. 1998, 109, 9219.
- (580) Tellgren, E. I.; Helgaker, T.; Soncini, A. Phys. Chem. Chem. Phys. 2009, 11, 5489.
- (581) Ruud, K.; Helgaker, T.; Bak, K. L.; Jørgensen, P.; Jensen, H. J. A. J. Chem. Phys. **1993**, 99, 3847.
 - (582) Ruud, K.; Helgaker, T. Chem. Phys. Lett. 1997, 264, 17.
- (583) Ruud, K.; Helgaker, T.; Jørgensen, P.; Bak, K. L. Chem. Phys. Lett. 1994, 223, 12.
- (584) Ruud, K.; Skaane, H.; Helgaker, T.; Bak, K. L.; Jørgensen, P. J. Am. Chem. Soc. **1994**, 116, 10135.
- (585) Ruud, K.; Helgaker, T.; Bak, K. L.; Jørgensen, P.; Olsen, J. Chem. Phys. 1995, 195, 157.
- (586) Åstrand, P.-O.; Mikkelsen, K. V.; Ruud, K.; Helgaker, T. J. Phys. Chem. **1996**, 100, 19771.
- (587) Ruud, K.; Åstrand, P.-O.; Helgaker, T.; Mikkelsen, K. V. J. Mol. Struct. (THEOCHEM) 1996, 388, 231.
- (588) Mikkelsen, K. V.; Ruud, K.; Helgaker, T. Chem. Phys. Lett. 1996, 253, 443.
- (589) Ruud, K.; Ågren, H.; Helgaker, T.; Dahle, P.; Koch, H.; Taylor, P. R. *Chem. Phys. Lett.* **1998**, 285, 205.
- (590) Gauss, J.; Ruud, K.; Kállay, M. J. Chem. Phys. 2007, 127, 074101.
- (591) Ferrighi, L.; Marchesan, D.; Ruud, K.; Frediani, L.; Coriani, S. J. Chem. Phys. **2005**, 123, 204104.
- (592) Aidas, K.; Kongsted, J.; Nielsen, C. B.; Mikkelsen, K. V.; Christiansen, O.; Ruud, K. *Chem. Phys. Lett.* **200**7, 442, 322.

(593) Ruud, K.; Vaara, J.; Lounila, J.; Helgaker, T. Chem. Phys. Lett. 1998, 297, 467.

- (594) Ruud, K.; Åstrand, P.-O.; Taylor, P. R. J. Phys. Chem. A 2001, 105, 9926.
- (595) Ruud, K.; Taylor, P. R.; Jaszuński, M. J. Phys. Chem. A 2000, 104, 168.

(596) Lutnæs, O. B.; Teale, A. M.; Helgaker, T.; Tozer, D. J.; Ruud, K.; Gauss, J. J. Chem. Phys. **2009**, 131, 144104.

- (597) Raab, R. E.; de Lange, O. L. Mol. Phys. 2006, 104, 1925.
- (598) Graham, E. B.; Raab, R. E. Proc. R. Soc. London A 2000, 456, 1193.
- (599) Post, E. J. Formal structure of electromagnetism; North-Holland: Amsterdam, 1962.
 - (600) Krykunov, M.; Autschbach, J. J. Chem. Phys. 2007, 126, 024101.
- (601) Ramsey, N. F. Molecular Beams; Oxford University Press: New York, 1956.
- (602) Cybulski, S. M.; Bishop, D. M. J. Chem. Phys. 1994, 100, 2019.
- (603) Sauer, S. P. A; Spirko, V.; Oddershede, J. Chem. Phys. 1991, 153, 189.
- (604) Sauer, S. P. A; Oddershede, J.; Geertsen, J. Mol. Phys. 1992, 76, 445.
 - (605) Sauer, S. P. A; Ogilvie, J. F. J. Phys. Chem. 1994, 98, 8617.
 - (606) Sauer, S. P. A Chem. Phys. Lett. 1996, 260, 271.
- (607) Ogilvie, J. F.; Oddershede, J.; Sauer, S. P. A. Adv. Chem. Phys. 2000, 111, 475.
 - (608) Sauer, S. P. A. Adv. Quantum Chem. 2005, 48, 469.
 - (609) Sauer, S. P. A. J. Chem. Phys. 2010, 133, 171101.
- (610) Enevoldsen, T.; Rasmussen, T.; Sauer, S. P. A. J. Chem. Phys. 2001, 114, 84.
- (611) Åstrand, P.-O.; Ruud, K.; Mikkelsen, K. V.; Helgaker, T. *Mol. Phys.* **1997**, *92*, 89.
- (612) Åstrand, P.-O.; Ruud, K.; Mikkelsen, K. V.; Helgaker, T. *Chem. Phys. Lett.* **1997**, 271, 163.
- (613) Helgaker, T.; Jaszuński, M.; Ruud, K. Pol. J. Chem. 1998, 72, 1405.
 - (614) Jaszuński, M.; Ruud, K. Chem. Phys. Lett. 1999, 306, 64.
- (615) Mohn, C. E.; Wilson, D. J. D.; Lutnæs, O. B.; Helgaker, T.; Ruud, K. Adv. Quantum Chem. **2005**, *50*, *77*.
- (616) Puzzarini, C.; Heckert, M.; Gauss, J. J. Chem. Phys. 2008, 128, 194108.
- (617) Stanton, J. F.; Gauss, J.; Christiansen, O. J. Chem. Phys. 2001, 114, 2993.
 - (618) Rosenfeld, L. Z. Phys. 1929, 52, 161.
 - (619) Crawford, T. D. Theor. Chem. Acc. 2006, 115, 227.
 - (620) Pecul, M.; Ruud, K. Adv. Quantum Chem. 2005, 50, 185.
 - (621) Polavarapu, P. L. Chirality 2002, 14, 768.
 - (622) Polavarapu, P. L. Chirality 2003, 15, 284.
 - (623) Polavarapu, P. L. Chirality 2006, 18, 348.
- (624) Helgaker, T.; Ruud, K.; Bak, K. L.; Jørgensen, P.; Olsen, J. *Faraday Discuss.* **1994**, *99*, 165.
- (625) Cheeseman, J. R.; Frisch, M. J.; Devlin, F. J.; Stephens, P. J. J. Phys. Chem. A **2000**, 104, 1039.
- (626) Grimme, S.; Furche, F.; Ahlrichs, R. Chem. Phys. Lett. 2002, 361, 321.
 - (627) Ruud, K.; Helgaker, T. Chem. Phys. Lett. 2002, 352, 533.
 - (628) Pedersen, T. B.; Koch, H. Chem. Phys. Lett. 1998, 293, 251.
- (629) Pedersen, T. B.; Koch, H.; Boman, L.; de Meras, A. M. J. S. Chem. Phys. Lett. **2004**, 393, 319.
- (630) Tam, M. C.; Russ, N. J.; Crawford, T. D. J. Chem. Phys. 2004, 121, 3550.
 - (631) Tam, M. C.; Crawford, T. D. J. Phys. Chem. A 2006, 110, 2290.
- (632) Pedersen, T. B.; Kongsted, J.; Crawford, T. D.; Ruud, K. J. Chem. Phys. **2009**, 130, 034310.
- (633) Pedersen, T. B.; Kongsted, J.; Crawford, T. D. *Chirality* **2009**, *21*, E68.
- (634) Crawford, T. D.; Allegrini, W. D. Mol. Phys. 2009, 107, 1041.
- (635) Crawford, T. D.; Stephens, P. J. J. Phys. Chem. A 2008, 112, 1339.

- (636) Crawford, T. D.; Tam, M. C.; Abrams, M. L. J. Phys. Chem. A 2007, 111, 12057.
- (637) Ruud, K.; Stephens, P. J.; Devlin, F. J.; Taylor, P. R.; Cheeseman, J. R.; Frisch, M. J. *Chem. Phys. Lett.* **2003**, 373, 606.
- (638) Becke, A. D. J. Chem. Phys. 1993, 98, 5648.
- (639) Stephens, P. J.; Devlin, F. J.; Chabalowski, C. F.; Frisch, M. J. J. Phys. Chem. **1994**, 98, 11623.
 - (640) Boughton, J. W.; Pulay, P. J. Comput. Chem. 1993, 14, 736.
 - (641) Mort, B. C.; Autschbach, J. J. Phys. Chem. A 2005, 109, 8617.
 - (642) Wiberg, K. B.; Vaccaro, P.; Cheeseman, J. R. J. Am. Chem. Soc.
- 2003, 125, 1888.
 (643) Wiberg, K. B.; Wang, Y.-G.; Vaccaro, P.; Cheeseman, J. R.;
 Trucks, G.; Frisch, M. J. J. Phys. Chem. A 2004, 108, 32.
- (644) Wiberg, K. B.; Wang, Y.-G.; Vaccaro, P.; Cheeseman, J. R.; Luderer, M. J. Phys. Chem. A **2005**, 109, 3405.
- (645) Marchesan, D.; Coriani, S.; Forzato, C.; Nitti, P.; Pitacco, G.; Ruud, K. J. Phys. Chem. A **2005**, *109*, 1449.
 - (646) Kongsted, J.; Ruud, K. Chem. Phys. Lett. 2008, 451, 226.
- (647) Kongsted, J.; Pedersen, T. B.; Strange, M.; Osted, A.; Hansen,
- A. E.; Mikkelsen, K. V.; Pawlowski, F.; Jørgensen, P.; Hättig, C. Chem. Phys. Lett. 2005, 401, 385.
- (648) Wiberg, K. B.; Wilson, S. M.; Wang, Y.-G.; Vaccaro, P. H.; Cheeseman, J. R.; Luderer, M. R. J. Org. Chem. **2007**, 72, 6206.
- (649) Ruud, K.; Taylor, P. R.; Åstrand, P.-O. Chem. Phys. Lett. 2001, 337, 217.
 - (650) Ruud, K.; Zanasi, R. Angew. Chem., Int. Ed. 2005, 44, 3594.
- (651) Kongsted, J.; Pedersen, T. B.; Jensen, L.; Hansen, A. E.; Mikkelsen, K. V. J. Am. Chem. Soc. 2006, 128, 976.
- (652) Pecul, M.; Ruud, K.; Rizzo, A.; Helgaker, T. J. Phys. Chem. A 2004, 108, 4269.
- (653) Coriani, S.; Baranowska, A.; Ferrighi, L.; Forzato, C.; Marchesan,
- D.; Nitti, P.; Pitacco, G.; Rizzo, A.; Ruud, K. Chirality 2006, 18, 357.
- (654) Buckingham, A. D.; Longuet-Higgins, H. C. Mol. Phys. 1968, 14, 63.
- (655) Shen, Y. R. Principles of Nonlinear Optics; Wiley: New York, 1984.
- (656) Bloembergen, N. Nonlinear Optics, 4th ed.; World Scientific: Singapore, 1996.
- (657) Butcher, P. N.; Cotter, D. The Elements of Nonlinear Optics, Cambridge Studies in Modern Optics, reprint ed.; Cambridge University Press: Cambridge, 1993.
- (658) Rizzo, A.; Coriani, S. Adv. Quantum Chem. 2005, 50, 143.
- (659) Rizzo, A.; Helgaker, T.; Ruud, K.; Barszczewicz, A.; Jaszuński, M.; Jørgensen, P. J. Chem. Phys. **1995**, 102, 8953.
- (660) Rizzo, C.; Rizzo, A.; Bishop, D. M. Int. Rev. Phys. Chem. 1997, 16, 81.
- (661) Rizzo, A.; Ruud, K.; Bishop, D. M. Mol. Phys. 2002, 100, 799.
- (662) Rizzo, A.; Kállay, M.; Gauss, J.; Pawlowski, F.; Jørgensen, P.; Hättig, C. J. Chem. Phys. **2004**, 121, 9461.
 - (663) Rizzo, A.; Coriani, S. J. Chem. Phys. 2003, 119, 11064.
- (664) Coriani, S.; Pecul, M.; Rizzo, A.; Jørgensen, P.; Jaszuński, M. J. Chem. Phys. **2002**, 117, 6417.
- (665) Jansik, B.; Rizzo, A.; Frediani, L.; Ruud, K.; Coriani, S. J. Chem. Phys. **2006**, 125, 234105.
- (666) Jaszuński, M.; Jørgensen, P.; Rizzo, A.; Ruud, K.; Helgaker, T. Chem. Phys. Lett. **1994**, 222, 263.
- (667) Coriani, S.; Hättig, C.; Jørgensen, P.; Halkier, A.; Rizzo, A. Chem. Phys. Lett. **1998**, 293, 324.
- (668) Coriani, S.; Hättig, C.; Jørgensen, P.; Helgaker, T. J. Chem. Phys. 2000, 113, 3561.
- (669) Kjærgaard, T.; Jørgensen, P.; Thorvaldsen, A. J.; Sałek, P.; Coriani, S. J. Chem. Theory Comput. **2009**, *5*, 1997.
 - (670) Verdet, E. M. Annal. Chim. (3rd Ser.) 1854, 41, 370.
- (671) Caldwell, D. J.; Eyring, H. *The Theory of Optical Activity*; Wiley-Interscience: New York, 1971; Chapter: The Faraday Effect, p 167.

(672) Buckingham, A. D.; Stephens, P. J. Annu. Rev. Phys. Chem. 1966, 17, 399. (674) Bregant, M.; Cantatore, G.; Carusotto, S.; Cimino, R.; Valle, F. D.; Domenico, G. D.; Gastaldi, U.; Karuza, M.; Lozza, V.; Milotti, E.; Polacco, E.; Raiteri, G.; Ruoso, G.; Zavattini, E.; Zavattini, G. *Chem. Phys. Lett.* **2009**, 471, 322.

(675) Berceau, P.; Battesti, R.; Fouche, M.; Rizzo, C. Can. J. Phys. 2011, 89, 153.

(676) Zavattini, E.; Zavattini, G.; Ruoso, G.; Raiteri, G.; Polacco, E.; Milotti, E.; Lozza, V.; Karuza, M.; Gastaldi, U.; Domenico, G. D.; Valle, F. D.; Cimino, R.; Carusotto, S.; Cantatore, G.; Bregant, M. *Phys. Rev. D* **2008**, 77, 032006.

(677) Zavattini, E.; Zavattini, G.; Ruoso, G.; Polacco, E.; Milotti, E.; Karuza, M.; Gastaldi, U.; Domenico, G. D.; Valle, F. D.; Cimino, R.; Carusotto, S.; Cantatore, G.; Bregant, M. *Phys. Rev. Lett.* **200**7, *99*, 129901.

(678) Zavattini, E.; Zavattini, G.; Ruoso, G.; Polacco, E.; Milotti, E.; Karuza, M.; Gastaldi, U.; Domenico, G. D.; Valle, F. D.; Cimino, R.; Carusotto, S.; Cantatore, G.; Bregant, M. *Phys. Rev. Lett.* **2006**, *96*, 110406.

(679) Imrie, D.; Raab, R. E. Mol. Phys. 1991, 74, 833.

- (680) Raab, R. E.; de Lange, O. L. Mol. Phys. 2003, 101, 3467.
- (681) de Lange, O. L.; Raab, R. E. Mol. Phys. 2004, 102, 125.
- (682) Shcherbin, D.; Thorvaldsen, A. J.; Ruud, K.; Coriani, S.; Rizzo, A. Phys. Chem. Chem. Phys. 2009, 11, 816.
- (683) Hättig, C.; Christiansen, O.; Jørgensen, P. J. Chem. Phys. 1997, 107, 10592.

(684) Hättig, C.; Jørgensen, P. Theor. Chem. Acc. 1998, 100, 230.

- (685) Hättig, C.; Jørgensen, P. Adv. Quantum Chem. 1999, 35, 111.
- (686) Larsen, H.; Olsen, J.; Hättig, C. Chem. Phys. Lett. 1999, 307, 235.
 - (687) Casimir, H. B. G.; Polder, D. Phys. Rev. 1948, 73, 360.
- (688) Jiemchooroj, A.; Sernelius, B. E.; Norman, P. Phys. Rev. A 2004, 69, 044701.
- (689) Jiemchooroj, A.; Norman, P.; Sernelius, B. E. J. Chem. Phys. 2006, 125, 124306.
- (690) Rizzo, A.; Hättig, C.; Fernandez, B.; Koch, H. J. Chem. Phys. 2002, 117, 2609.
- (691) Koch, H.; Hättig, C.; Larsen, H.; Olsen, J.; Jørgensen, P.; Fernandez, B.; Rizzo, A. J. Chem. Phys. **1999**, *111*, 10108.
- (692) Hättig, C.; Cacheiro, J. L.; Fernandez, B.; Rizzo, A. *Mol. Phys.* **2003**, *101*, 1983.
- (693) Cacheiro, J. L.; Fernandez, B.; Rizzo, A.; Jansik, B.; Pedersen, T. B. *Mol. Phys.* **2008**, *106*, 881.
- (694) Baranowska, A.; Fernandez, B.; Rizzo, A.; Jansik, B. Phys. Chem. Chem. Phys. 2009, 11, 9871.
- (695) Stone, A. J. The Theory of Intermolecular Forces; Clarendon Press: Oxford, 1996.
 - (696) Moldover, M. R. J. Res. Natl. Inst. Stand. 1998, 103, 167.
- (697) Hurly, J. J.; Moldover, M. R. J. Res. Natl. Inst. Stand. 2000, 105, 667.
- (698) Moldover, M. R.; Buckley, T. J. Int. J. Thermophys. 2001, 22, 859.
- (699) Schmidt, J. W.; Gavioso, R. M.; May, E. F.; Moldover, M. R. *Phys. Rev. Lett.* **2007**, *98*, 254504.
- (700) Moszynski, R. Challenges and Advances in Computational Chemistry and Physics. In *Molecular Materials with Specific Interactions—Modeling and Design*; Papadopoulos, M., Sadlej, A., Leszczynski, J., Eds.; Springer: Berlin, 2007.
- (701) Rizzo, A.; Coriani, S.; Marchesan, D.; Cacheiro, J. L.; Fernandez, B.; Hättig, C. *Mol. Phys.* **2006**, *104*, 305.
- (702) Antušek, A.; Jaszuński, M.; Rizzo, A. J. Chem. Phys. 2007, 126, 074303.
- (703) Bancewicz, T.; Maroulis, G. Chem. Phys. Lett. 2010, 498, 349.
- (704) Buckingham, A. D.; Pople, J. A. Trans. Faraday Soc. 1955, 51, 1029.

- (706) Buckingham, A. D.; Pople, J. A. Faraday Discuss. 1956, 17.
- (707) Buckingham, A. D.; Raab, R. E. Trans. Faraday Soc. 1958, 54, 623.
- (708) Marchesan, D.; Coriani, S.; Rizzo, A. Mol. Phys. 2003, 101, 1851.
- (709) Trofimov, A. B.; Stelter, G.; Schirmer, J. J. Chem. Phys. 2002, 117, 6402.
- (710) Christiansen, O.; Bak, K. L.; Koch, H.; Sauer, S. P. A. Chem. Phys. Lett. **1998**, 284, 47.
- (711) Schreiber, M.; Sauer, S. P. A.; Thiel, W. J. Chem. Phys. 2008, 128, 134110.
- (712) Silva-Junior, M. R.; Schreiber, M.; Sauer, S. P. A.; Thiel, W. J. Chem. Phys. **2008**, 129, 104103.
- (713) Silva-Junior, M. R.; Sauer, S. P. A.; Schreiber, M.; Thiel, W. *Mol. Phys.* **2010**, *108*, 453.
 - (714) Schirmer, J.; Mertins, F. Theor. Chem. Acc. 2010, 125, 145.
 - (715) Mertins, F.; Schirmer, J. Phys. Rev. A 1996, 53, 2140.
- (716) Osted, A.; Kongsted, J.; Christiansen, O. J. Phys. Chem. A 2005, 109, 1430.
- (717) Cronstrand, P.; Christiansen, O.; Norman, P.; Ågren, H. Phys. Chem. Chem. Phys. 2001, 3, 2567.
- (718) Åstrand, P.-O.; Ramanujam, P. S.; Hvilsted, S.; Bak, K. L.; Sauer, S. P. A. J. Am. Chem. Soc. **2000**, 122, 3482.
- (719) Pecul, M.; Ruud, K.; Helgaker, T. Chem. Phys. Lett. 2004, 388, 110.
- (720) Pedersen, T. B.; Koch, H.; Ruud, K. J. Chem. Phys. 1999, 110, 2883.
- (721) Pedersen, T. B.; Koch, H.; Hättig, C. J. Chem. Phys. 1999, 110, 8318.
- (722) Pedersen, T. B.; Fernandez, B.; Koch, H. J. Chem. Phys. 2001, 114, 6983.
- (723) Kowalczyk, T. D.; Abrams, M. L.; Crawford, T. D. J. Phys. Chem. A 2006, 110, 7649.
- (724) Crawford, T. D.; Tam, M. C.; Abrams, M. L. Mol. Phys. 2007, 105, 2607.
 - (725) Diedrich, C.; Grimme, S. J. Phys. Chem. A 2003, 107, 2524.

(726) Toro, C.; Boni, L. D.; Lin, N.; Santoro, F.; Rizzo, A.; Hernandez, F. E. *Chem.*—*Eur. J.* **2010**, *16*, 3504.

- (727) Paterson, M. J.; Christiansen, O.; Pawlowski, F.; Jørgensen, P.; Hättig, C.; Helgaker, T.; Sałek, P. J. Chem. Phys. **2006**, 124, 054322.
- (728) Yanai, T.; Tew, D. P.; Handy, N. C. Chem. Phys. Lett. 2004, 393, 51.
- (729) Nielsen, C. B.; Rettrup, S.; Sauer, S. P. A. J. Chem. Phys. 2006, 124, 114108.
- (730) Jansik, B.; Rizzo, A.; Ågren, H. Chem. Phys. Lett. 2005, 414, 461.
- (731) Rizzo, A.; Jansik, B.; Pedersen, T. B.; Ågren, H. J. Chem. Phys. 2006, 125, 064113.
 - (732) Jansik, B.; Rizzo, A.; Ågren, H. J. Phys. Chem. B 2007, 111, 446.
 - (733) Rizzo, A.; Lin, N.; Ruud, K. J. Chem. Phys. 2008, 128, 164312.
- (734) Cronstrand, P.; Luo, Y.; Norman, P.; Ågren, H. Chem. Phys. Lett. 2003, 375, 233.
- (735) Cronstrand, P.; Norman, P.; Luo, Y.; Ågren, H. J. Chem. Phys. **2004**, *121*, 2020.
- (736) Cronstrand, P.; Luo, Y.; Ågren, H. Adv. Quantum Chem. 2005, 50, 1.
- (737) Hättig, C.; Christiansen, O.; Jørgensen, P. J. Chem. Phys. 1998, 108, 8331.
- (738) Coriani, S.; Jørgensen, P.; Rizzo, A.; Ruud, K.; Olsen, J. Chem. Phys. Lett. **1999**, 300, 61.
- (739) Seth, M.; Ziegler, T.; Banerjee, A.; Autschbach, J.; van Gisbergen, S. J. A.; Baerends, E. J. *Chem. Phys.* **2004**, *120*, 10942.
- (740) Kjærgaard, T.; Jansik, B.; Jørgensen, P.; Coriani, S.; Michl, J. J. Phys. Chem. A 2007, 111, 11278.
- (741) Foresman, J. B.; Head-Gordon, M.; Pople, J. A.; Frisch, M. J. J. Phys. Chem. **1992**, 96, 135.

- (742) Stanton, J. F. J. Chem. Phys. 1993, 99, 8840.
- (743) Szalay, P. G. Int. J. Quantum Chem. 1995, 55, 151.
- (744) Kállay, M.; Gauss, J. J. Chem. Phys. 2004, 121, 9257.
- (745) Levchenko, S. V.; Wang, T.; Krylov, A. I. J. Chem. Phys. 2005, 122, 224106.
- (746) Gwaltney, S. R.; Bartlett, R. J.; Nooijen, M. J. Chem. Phys. 1999, 111, 58.
- (747) Wladyslawski, M.; Nooijen, M. Adv. Quantum Chem. 2005, 49, 1.
 - (748) Maurice, D. R.; Head-Gordon, M. Mol. Phys. 1999, 96, 1533.
 - (749) Dreuw, A.; Head-Gordon, M. Chem. Rev. 2005, 105, 4009.
- (750) Rhee, Y. M.; Casanova, D.; Head-Gordon, M. J. Chem. Theory Comput. 2009, 5, 1224.
- (751) Devarajan, A.; Hirao, K.; Pal, S. Collect. Czech. Chem. Commun. 2003, 68, 47.
 - (752) King, R. A. J. Phys. Chem. A 2008, 112, 5727.
- (753) Burcl, R.; Amos, R. D.; Handy, N. C. *Chem. Phys. Lett.* **2002**, 355, 8.
- (754) Tellgren, E. I.; Henriksson, J.; Norman, P. J. Chem. Phys. 2007, 126, 064313.
- (755) Jansson, E.; Norman, P.; Minaev, B. F.; Ågren, H. J. Chem. Phys. 2006, 124, 114106.
- (756) Vahtras, O.; Ågren, H.; Jørgensen, P.; Jensen, H. J. A.; Helgaker, T.; Olsen, J. J. Chem. Phys. **1992**, *97*, 9178.
- (757) Christiansen, O.; Gauss, J. J. Chem. Phys. 2002, 116, 6674.
- (758) Minaev, B. F.; Vahtras, O.; Ågren, H. Chem. Phys. 1996, 208, 299.
- (759) Vahtras, O.; Ågren, H.; Jørgensen, P.; Jensen, H. J. A.; Helgaker,T.; Olsen, J. J. Chem. Phys. 1992, 96, 2118.
- (760) Hess, B. A.; Marian, C. M.; Wahlgren, U.; Gropen, O. Chem. Phys. Lett. **1996**, 251, 365.
- (761) Koseki, S.; Gordon, M. S.; Schmidt, M. W.; Matsunaga, N. J. Phys. Chem. **1995**, *99*, 12764.
 - (762) Neese, F. J. Chem. Phys. 2005, 122, 034107.
- (763) Buenker, R. J.; Alekseyev, A. B.; Liebermann, H. P.; Lingott, R.; Hirsch, G. J. Chem. Phys. **1998**, 108, 3400.
- (764) Vallet, V.; Maron, L.; Teichteil, C.; Flament, J. P. J. Chem. Phys. 2000, 113, 1391.
- (765) Yabushita, S.; Zhang, Z. Y.; Pitzer, R. M. J. Phys. Chem. A **1999**, 103, 5791.
 - (766) Sjøvoll, M.; Gropen, O.; Olsen, J. Theor. Chem. Acc. 1997, 97, 301.
- (767) Kleinschmidt, M.; Tatchen, J.; Marian, C. M. J. Chem. Phys. 2006, 124, 124101.
- (768) Christiansen, O.; Gauss, J.; Schimmelpfennig, B. Phys. Chem. Chem. Phys. 2000, 2, 965.
- (769) Fedorov, D. G.; Koseki, S.; Schmidt, M. W.; Gordon, M. S. Int. Rev. Phys. Chem. 2003, 22, 551.
- (770) Tilson, J. L.; Shepard, R.; Naleway, C.; Wagner, A. F.; Ermler,
 W. C. J. Chem. Phys. 2000, 112, 2292.
- (771) Beck, E. V.; Stahlberg, E. A.; Burggraf, L. W.; Blaudeau, J.-P. Chem. Phys. 2008, 349, 158.
 - (772) Grimme, S.; Waletzke, M. J. Chem. Phys. 1999, 111, 5645.
 - (773) Dyall, K. G. J. Chem. Phys. 1994, 100, 2118.
 - (774) Visscher, L.; Saue, T. J. Chem. Phys. 2000, 113, 3996.
- (775) Bloembergen, N.; Lotem, H.; Lynch, R. T. Indian J. Pure Appl. Phys. **1978**, *16*, 151.
- (776) Hansen, M. B.; Seidler, P.; Györffy, W.; Christiansen, O. J. Chem. Phys. **2010**, 133, 114102.
- (777) Seidler, P.; Hansen, M. B.; Györffy, W.; Toffoli, D.; Christiansen, O. J. Chem. Phys. 2010, 132, 164105.
- (778) Jiemchooroj, A.; Norman, P. J. Chem. Phys. 2007, 126, 134102.
- (779) Jiemchooroj, A.; Ekström, U.; Norman, P. J. Chem. Phys. 2007, 127, 165104.
 - (780) Villaume, S.; Norman, P. Chirality 2010, 21, E13.
- (781) Jiemchooroj, A.; Norman, P.; Sernelius, B. E. J. Chem. Phys. 2005, 123, 124312.
 - (782) Bishop, D. M. Rev. Mod. Phys. 1990, 62, 343.

- (783) Bishop, D. M.; Norman, P. In *Handbook of Advanced Electronic and Photonic Materials*; Nalwa, H. S., Ed.; Academic Press: San Diego, 2000.
- (784) Born, M.; Oppenheimer, R. Ann. Phys.-Berlin 1927, 84, 457.
- (785) Born, M.; Huang, K. Dynamical Theory of Crystal Lattices; Clarendon Press: Oxford, 1954; Appendix 8.
 - (786) Bishop, D. M.; Kirtman, B. J. Chem. Phys. 1991, 95, 2646.
- (787) Archibong, E. F.; Thakkar, A. J. Chem. Phys. Lett. 1993, 201, 485.
 - (788) Bishop, D. M.; Cybulski, S. M. J. Chem. Phys. 1994, 101, 2180.
 - (789) Bishop, D. M.; Norman, P. J. Chem. Phys. 1999, 111, 3042.
 - (790) Bishop, D. M.; Kirtman, B. J. Chem. Phys. 1992, 97, 5255.
 - (791) Torrent-Sucarrat, M.; Sola, M.; Duran, M.; Luis, J. M.; Kirt-
- man, B. J. Chem. Phys. 2002, 116, 5363.
- (792) Bishop, D. M.; Hasan, M.; Kirtman, B. J. Chem. Phys. 1995, 103, 4157.
- (793) Luis, J. M.; Duran, M.; Champagne, B.; Kirtman, B. J. Chem. Phys. 2000, 113, 5203.
- (794) Christiansen, O.; Kongsted, J.; Paterson, M. J.; Luis, J. M. J. Chem. Phys. **2006**, 125, 214309.
- (795) Bowman, J. M.; Christoffel, K.; Tobin, F. J. Phys. Chem. 1979, 83, 905.
- (796) Gerber, R. B.; Ratner, M. A. Adv. Chem. Phys. 1988, 70, 97.
- (797) Hansen, M. B.; Christiansen, O.; Hättig, C. J. Chem. Phys. 2009, 131, 154101.
 - (798) Bishop, D. M. Adv. Quantum Chem. 1994, 25, 1.
- (799) Bishop, D. M. Adv. Chem. Phys. 1998, 104, 1.

(800) Quinet, O.; Champagne, B.; Kirtman, B. J. Comput. Chem. 2001, 22, 1920.

- (801) Quinet, O.; Champagne, B. J. Chem. Phys. 2002, 117, 2481.
- (802) Quinet, O.; Kirtman, B.; Champagne, B. J. Chem. Phys. 2003, 118, 505.
- (803) Quinet, O.; Liegeois, V.; Champagne, B. J. Chem. Theory Comput. 2005, 1, 444.
- (804) Liegeois, V.; Ruud, K.; Champagne, B. J. Chem. Phys. 2007, 127, 204105.
 - (805) Hättig, C.; Jørgensen, P. J. Chem. Phys. 1998, 109, 2762.
 - (806) Norman, P.; Luo, Y.; Ågren, H. J. Chem. Phys. 1998, 109, 3580.
- (807) Raptis, S. G.; Papadopoulos, M. G.; Sadlej, A. J. J. Chem. Phys. 1999, 111, 7904.
- (808) Avramopoulos, A.; Ingamells, V.; Papadopoulos, M. G.; Sadlej, A. J. J. Chem. Phys. 2001, 114, 198.
- (809) Avramopoulos, A.; Papadopoulos, M. G.; Sadlej, A. J. J. Chem. Phys. 2002, 117, 10026.
- (810) Ruud, K.; Jonsson, D.; Taylor, P. R. Phys. Chem. Chem. Phys. 2000, 2, 2161.
- (811) Jonsson, D.; Norman, P.; Ågren, H.; Rizzo, A.; Coriani, S.; Ruud, K. J. Chem. Phys. **2001**, 114, 8372.

(812) Jonsson, D.; Norman, P.; Vahtras, O.; Ågren, H.; Rizzo, A. J. Chem. Phys. **1997**, 106, 8552.

- (813) Rizzo, A.; Cappelli, C. Int. J. Quantum Chem. 2011, 111, 760.
- (814) Bishop, D. M.; Champagne, B.; Kirtman, B. J. Chem. Phys. 1998, 109, 9987.
- (815) Champagne, B.; Perpete, E. A.; van Gisbergen, S. J. A.; Baerends, E. J.; Snijders, J. G.; Soubra-Ghaoui, C.; Robins, K. A.;
- Kirtman, B. J. Chem. Phys. **1998**, 109, 10489.
- (816) Champagne, B.; Perpete, E. A. Int. J. Quantum Chem. 1999, 75, 441.
- (817) Torrent-Sucarrat, M.; Sola, M.; Duran, M.; Luis, J. M.; Kirtman, B. J. Chem. Phys. **2003**, 118, 711.
- (818) Tomasi, J.; Mennucci, B.; Cammi, R. Chem. Rev. 2005, 105, 2999.
- (819) Kongsted, J.; Mennucci, B.; Coutinho, K.; Canuto, S. Chem. Phys. Lett. 2010, 484, 185.
- (820) Sumowski, C. V.; Ochsenfeld, C. J. Phys. Chem. A 2009, 113, 11734.
- (821) Zienau, J.; Kussmann, J.; Ochsenfeld, C. Mol. Phys. 2010, 108, 333.

(822) Ziolkowski, M.; Jansik, B.; Kjærgaard, T.; Jørgensen, P. J. Chem. Phys. 2010, 133, 014107.

- (823) Hopmann, K. H.; Ruud, K.; Pecul, M.; Kudelski, A.; Dracinsky, M.; Bouř, P. J. Phys. Chem. B **2011**, *115*, 4128.
- (824) Mikkelsen, K. V.; Luo, Y.; Ågren, H.; Jørgensen, P. J. Chem. Phys. **1995**, 102, 9362.
- (825) Ruud, K.; Ågren, H.; Dahle, P.; Helgaker, T.; Rizzo, A.; Coriani, S.; Koch, H.; Sylvester-Hvid, K. O.; Mikkelsen, K. V. J. Chem. Phys. **1998**, 108, 599.
- (826) Mukhopadhyay, P.; Zuber, G.; Goldsmith, M.-R.; Wipf, P.; Beratan, D. N. *ChemPhysChem* **2006**, *7*, 2483.
- (827) Mukhopadhyay, P.; Zuber, G.; Wipf, P.; Beratan, D. N. Angew. Chem., Int. Ed. 2007, 46, 6450.
 - (828) Car, R.; Parrinello, M. Phys. Rev. Lett. 1985, 55, 2471.
- (829) Leach, A. Molecular Modelling: Principles and Applications; Prentice Hall: Harlow, 2001.
- (830) Warshel, A.; Levitt, M. J. Mol. Biol. 1976, 103, 227.
- (831) Weiner, S. J.; Kollman, P. A.; Nguyen, D. T.; Case, D. A.
- J. Comput. Chem. 1986, 7, 230. (832) Field, M. J.; Bash, P. A.; Karplus, M. J. Comput. Chem. 1990,
- 700.
 (833) Lennard-Jones, J. E. Proc. R. Soc. London A 1924, 738, 463.
- (834) Olsen, J. M.; Aidas, K.; Kongsted, J. J. Chem. Theory Comput.
 2010, 6, 3721.
- (835) Sneskov, K.; Schwabe, T.; Kongsted, J.; Christiansen, O. J. Chem. Phys. 2011, 134, 104108.
- (836) Kongsted, J.; Nielsen, C. B.; Mikkelsen, K. V.; Christiansen, O.; Ruud, K. J. Chem. Phys. **200**7, 126, 034510.
- (837) Kongsted, J.; Osted, A.; Mikkelsen, K. V.; Christiansen, O. Chem. Phys. Lett. 2002, 364, 379.
- (838) Kongsted, J.; Osted, A.; Mikkelsen, K. V.; Christiansen, O. Mol. Phys. 2002, 100, 1813.
- (839) Kongsted, J.; Osted, A.; Mikkelsen, K. V.; Christiansen, O. J. Chem. Phys. 2003, 118, 1620.
- (840) Osted, A.; Kongsted, J.; Mikkelsen, K. V.; Christiansen, O. *Mol. Phys.* **2003**, *101*, 2055.
- (841) Kongsted, J.; Osted, A.; Mikkelsen, K. V.; Christiansen, O. J. Chem. Phys. 2003, 119, 10519.
- (842) Paterson, M. J.; Kongsted, J.; Christiansen, O.; Mikkelsen, K. V.; Nielsen, C. B. J. Chem. Phys. **2006**, 125, 184501.
- (843) Osted, A.; Kongsted, J.; Mikkelsen, K. V.; Åstrand, P.-O.; Christiansen, O. J. Chem. Phys. **2006**, *124*, 124503.
- (844) Sneskov, K.; Matito, E.; Kongsted, J.; Christiansen, O. J. Chem. Theory Comput. 2010, 6, 839.
- (845) Steindal, A. H.; Ruud, K.; Frediani, L.; Aidas, K.; Kongsted, J. J. Phys. Chem. B 2011, 115, 3027.
- (846) Cances, E.; Mennucci, B.; Tomasi, J. J. Chem. Phys. 1997, 107, 3032.
 - (847) Cances, E.; Mennucci, B. J. Math. Chem. 1998, 23, 309.
- (848) Mennucci, B.; Cances, E.; Tomasi, J. J. Phys. Chem. B 1997, 101, 10506.
 - (849) Li, H.; Gordon, M. S. J. Chem. Phys. 2007, 126, 124112.
- (850) Li, H.; Fedorov, D. G.; Nagata, T.; Kitaura, K.; Jensen, J. H.; Gordon, M. S. *J. Comput. Chem.* **2010**, *31*, 778.
 - (851) Chandler, D.; Andersen, H. C. J. Chem. Phys. 1972, 57, 1930.
 - (852) Hirata, S.; Rossky, P. J. Chem. Phys. Lett. 1981, 83, 329.
- (853) Hirata, S.; Pettitt, B. M.; Rossky, P. J. J. Chem. Phys. 1982, 77, 509.
- (854) Ten-no, S.; Hirata, S.; Kato, S. Chem. Phys. Lett. 1993, 214, 391.
 - (855) Ten-no, S.; Hirata, S.; Kato, S. J. Chem. Phys. 1994, 100, 7443.
 - (856) Sato, H.; Sakaki, S. J. Phys. Chem. A 2004, 108, 1629.
 - (857) Ishida, T.; Hirata, S.; Kato, S. J. Chem. Phys. 1999, 110, 11423.
- (858) Yamazaki, T.; Sato, H.; Hirata, S. Chem. Phys. Lett. 2000, 325, 668.
- (859) Mennucci, B.; Martinez, J. M.; Tomasi, J. J. Phys. Chem. A 2001, 105, 7287.
 - (860) Mennucci, B.; Martinez, J. M. J. Phys. Chem. B 2005, 109, 9830.

- (861) Minezawa, N.; Kato, S. J. Chem. Phys. 2007, 126, 054511.
- (862) Yokogawa, D.; Sato, H.; Sakaki, S. J. Chem. Phys. 2009, 131, 214504.
- (863) Day, P. N.; Jensen, J. H.; Gordon, M. S.; Webb, S. P.; Stevens, W. J.; Krauss, M.; Garmer, D.; Basch, H.; Cohen, D. *J. Chem. Phys.* **1996**, *105*, 1968.
- (864) Freitag, M. A.; Hillman, B.; Agrawal, A.; Gordon, M. S. J. Chem. Phys. 2004, 120, 1197.

(865) Nagata, T.; Brorsen, K.; Fedorov, D. G.; Kitaura, K.; Gordon, M. S. J. Chem. Phys. **2011**, 134, 124115.

- (866) Nagata, T.; Fedorov, D. G.; Sawada, T.; Kitaura, K.; Gordon, M. S. J. Chem. Phys. **2011**, *134*, 034110.
- (867) Yoo, S.; Zahariev, F.; Sok, S.; Gordon, M. S. J. Chem. Phys. 2008, 129, 144112.
- (868) Govind, N.; Wang, Y. A.; da Silva, A. J. R.; Carter, E. A. Chem. Phys. Lett. **1998**, 295, 129.
- (869) Govind, N.; Wang, Y. A.; Carter, E. A. J. Chem. Phys. 1999, 110, 7677.
- (870) Gomes, A. S. P.; Jacob, C. R.; Visscher, L. Phys. Chem. Chem. Phys. 2008, 10, 5353.
- (871) Wesolowski, T. A.; Warshel, A. J. Phys. Chem. 1993, 97, 8050.
 (872) Neugebauer, J. J. Chem. Phys. 2007, 126, 134116.
- (873) Svensson, M.; Humbel, S.; Froese, R. D. J.; Matsubara, T.; Sieber, S.; Morokuma, K. J. Phys. Chem. **1996**, 100, 19357.
- (874) Dapprich, S.; Komaromi, I.; Byun, K. S.; Morokuma, K.; Frisch, M. J. J. Mol. Struct. (THEOCHEM) **1999**, 461, 1.
- (875) Mennucci, B.; Cammi, R. Int. J. Quantum Chem. 2003, 93, 121.
- (876) Jørgensen, S.; Ratner, M. A.; Mikkelsen, K. V. J. Chem. Phys. 2001, 115, 8185.
- (877) Jørgensen, S.; Ratner, M. A.; Mikkelsen, K. V. J. Chem. Phys. 2002, 116, 10902.
- (878) Frediani, L.; Cammi, R.; Corni, S.; Tomasi, J. J. Chem. Phys. 2004, 120, 3893.
- (879) Frediani, L.; Mennucci, B.; Cammi, R. J. Phys. Chem. B 2004, 108, 13796.
- (880) Cammi, R.; Mennucci, B.; Tomasi, J. J. Phys. Chem. A 1999, 103, 9100.
- (881) Christiansen, O.; Mikkelsen, K. V. J. Chem. Phys. 1999, 110, 1365.
 - (882) van der Zwan, G.; Hynes, J. T. J. Phys. Chem. 1985, 89, 4181.
 - (883) Cossi, M.; Barone, V. J. Chem. Phys. 2000, 112, 2427.
- (884) Aidas, K.; Kongsted, J.; Osted, A.; Mikkelsen, K. V.; Christiansen,
 O. J. Phys. Chem. A 2005, 109, 8001.
- (885) Cammi, R.; Corni, S.; Mennucci, B.; Tomasi, J. J. Chem. Phys. 2005, 122, 104513.
- (886) Mikkelsen, K. V.; Ågren, H.; Jensen, H. J. A.; Helgaker, T. J. Chem. Phys. **1988**, 89, 3086.
- (887) Mikkelsen, K. V.; Jørgensen, P.; Jensen, H. J. A. J. Chem. Phys. 1994, 100, 6597.
- (888) Mikkelsen, K. V.; Sylvester-Hvid, K. O. J. Phys. Chem. 1996, 100, 9116.
- (889) Mikkelsen, K. V.; Luo, Y.; Ågren, H.; Jørgensen, P. J. Chem. Phys. **1994**, 100, 8240.
- (890) Jonsson, D.; Norman, P.; Ågren, H.; Luo, Y.; Sylvester-Hvid, K. O.; Mikkelsen, K. V. J. Chem. Phys. **1998**, 109, 6351.
- (891) Poulsen, T. D.; Ogilby, P. R.; Mikkelsen, K. V. J. Chem. Phys. 1999, 111, 2678.
- (892) Sylvester-Hvid, K. O.; Mikkelsen, K. V.; Jonsson, D.; Norman, P.; Ågren, H. J. Chem. Phys. **1998**, 109, 5576.
- (893) Sylvester-Hvid, K. O.; Mikkelsen, K. V.; Jonsson, D.; Norman, P.; Ågren, H. J. Phys. Chem. A **1999**, 103, 8375.
- (894) Mikkelsen, K. V.; Jørgensen, P.; Ruud, K.; Helgaker, T. J. Chem. Phys. **1997**, 106, 1170.
- (895) Ruud, K.; Helgaker, T.; Rizzo, A.; Coriani, S.; Mikkelsen, K. V. J. Chem. Phys. **1997**, 107, 894.
- (896) Christiansen, O.; Mikkelsen, K. V. J. Chem. Phys. 1999, 110, 8348.

- (897) Kongsted, J.; Osted, A.; Mikkelsen, K. V.; Christiansen, O. J. Chem. Phys. 2004, 120, 3787.
- (898) Mikkelsen, K. V. Annu. Rev. Phys. Chem. 2006, 57, 365.
- (899) Luo, Y.; Ågren, H.; Mikkelsen, K. V. Chem. Phys. Lett. 1997, 275, 145.
- (900) Bartkowiak, W.; Misiaszek, T. Chem. Phys. 2000, 261, 353.
- (901) Miertus, S.; Scrocco, E.; Tomasi, J. Chem. Phys. 1981, 55, 117.
- (902) Amovilli, C.; Barone, V.; Cammi, R.; Cances, E.; Cossi, M.;
- Mennucci, B.; Pomelli, C. S.; Tomasi, J. Adv. Quantum Chem. 1999, 32, 227.
 (903) Chipman, D. M. J. Chem. Phys. 1997, 106, 10194.
 - (904) Chipman, D. M. J. Chem. Phys. **1999**, 110, 8012.
- (905) Zhan, C. G.; Bentley, J.; Chipman, D. M. J. Chem. Phys. 1998, 108, 177.
- (906) Klamt, A.; Schuurmann, G. J. Chem. Soc., Perkin Trans. 2 1993, 799.
- (907) Klamt, A. J. Phys. Chem. 1995, 99, 2224.
- (908) Barone, V.; Cossi, M. J. Phys. Chem. A 1998, 102, 1995.
- (909) Pye, C. C.; Ziegler, T. Theor. Chem. Acc. 1999, 101, 396.
- (910) Chipman, D. M.; Dupuis, M. Theor. Chem. Acc. 2002, 107, 90.
- (911) Cammi, R.; Frediani, L.; Mennucci, B.; Ruud, K. J. Chem. Phys. 2003, 119, 5818.
- (912) Ruud, K.; Frediani, L.; Cammi, R.; Mennucci, B. Int. J. Mol. Sci. 2003, 4, 119.
- (913) Frediani, L.; Ågren, H.; Ferrighi, L.; Ruud, K. J. Chem. Phys. 2005, 123, 144117.
- (914) Ferrighi, L.; Frediani, L.; Ruud, K. J. Chem. Phys. 2010, 132, 024107.
- (915) Ferrighi, L.; Frediani, L.; Cappelli, C.; Sałek, P.; Ågren, H.; Helgaker, T.; Ruud, K. *Chem. Phys. Lett.* **2006**, 425, 267.
- (916) Mennucci, B. J. Am. Chem. Soc. 2002, 124, 1506.
- (917) Cui, Q.; Karplus, M. J. Phys. Chem. B 2000, 104, 3721.
- (918) Benzi, C.; Crescenzi, O.; Pavone, M.; Barone, V. Magn. Reson. Chem. **2004**, 42, S57.
- (919) Sebastiani, D.; Röthlisberger, U. J. Phys. Chem. B 2004, 108, 2807.
- (920) Aidas, K.; Møgelhøj, A.; Kjær, H.; Nielsen, C. B.; Mikkelsen, K. V.; Ruud, K.; Christiansen, O.; Kongsted, J. *J. Phys. Chem. A* **2007**, *111*, 4199.
 - (921) Kongsted, J.; Mennucci, B. J. Phys. Chem. A 2007, 111, 9890.
- (922) Caricato, M.; Mennucci, B.; Scalmani, G.; Trucks, G. W.; Frisch, M. J. J. Chem. Phys. **2010**, 132, 084102.
- (923) Cammi, R. Int. J. Quantum Chem. 2010, 110, 3040.
- (924) Marsman, M.; Grüneis, A.; Paier, J.; Kresse, G. J. Chem. Phys. 2009, 130, 184103.
 - (925) Blount, E. I. Solid State Phys. 1962, 13, 305.
 - (926) Resta, R. Rev. Mod. Phys. 1994, 66, 899.
- (927) Champagne, B.; André, J. M. Int. J. Quantum Chem. 1992, 42, 1009.
 - (928) Otto, P. Phys. Rev. B 1992, 45, 10876.
- (929) Kirtman, B.; Gu, F. L.; Bishop, D. M. J. Chem. Phys. 2000, 113, 1294.
- (930) Re, G. D.; Ladik, J.; Biczo, G. Phys. Rev. 1967, 155, 997.
- (931) André, J. M.; Gouverne, L.; Leroy, G. Bull. Soc. Chim. Belg. 1967, 76, 661.
 - (932) André, J. M.; Leroy, G. Theor. Chim. Acta 1967, 9, 123.
- (933) Hirata, S.; Iwata, S. J. Mol. Struct. (THEOCHEM) 1998, 451, 121.
 - (934) Hirata, S.; Iwata, S. J. Chem. Phys. 1998, 109, 4147.
- (935) Hirata, S.; Head-Gordon, M.; Bartlett, R. J. J. Chem. Phys. 1999, 111, 10774.
 - (936) Kudin, K. N.; Scuseria, G. E. J. Chem. Phys. 2000, 113, 7779.
- (937) Izmaylov, A. F.; Brothers, E. N.; Scuseria, G. E. J. Chem. Phys. 2006, 125, 224105.
 - (938) Kudin, K. N.; Scuseria, G. E. Phys. Rev. B 2000, 61, 16440.
- (939) Izmaylov, A. F.; Scuseria, G. E. J. Chem. Phys. 2007, 127, 144106.
- (940) Izmaylov, A. F.; Scuseria, G. E. Phys. Rev. B 2008, 77, 165131.

- (941) Kudin, K. N.; Scuseria, G. E.; Schlegel, H. B. J. Chem. Phys. 2001, 114, 2919.
- (942) Civalleri, B.; D'Arco, P.; Orlando, R.; Saunders, V. R.; Dovesi, R. Chem. Phys. Lett. **2001**, 348, 131.
- (943) Ferrero, M.; Rerat, M.; Orlando, R.; Dovesi, R. J. Comput. Chem. 2008, 29, 1450.
- (944) Lacivita, V.; Rerat, M.; Kirtman, B.; Ferrero, M.; Orlando, R.; Dovesi, R. J. Chem. Phys. **2009**, 131, 204509.
- (945) Orlando, R.; Lacivita, V.; Bast, R.; Ruud, K. J. Chem. Phys. 2010, 132, 244106.
- (946) Maschio, L.; Usvyat, D.; Schütz, M.; Civalleri, B. J. Chem. Phys. 2010, 132, 134706.
- (947) Masiello, D.; Deumens, E.; Öhrn, Y. Phys. Rev. A 2005, 71, 032108.
- (948) Hammond, J. R.; Kowalski, K.; deJong, W. A. J. Chem. Phys. 2007, 127, 144105.
- (949) Hammond, J. R.; deJong, W. A.; Kowalski, K. J. Chem. Phys. 2008, 128, 224102.
- (950) Hammond, J. R.; Kowalski, K. J. Chem. Phys. 2009, 130, 194108.
- (951) Kowalski, K.; Krishnamoorthy, S.; Villa, O.; Hammond, J. R.; Govind, N. J. Chem. Phys. **2010**, *132*, 154103.

NOTE ADDED IN PROOF

Reference to relatively recent work of Kowalski and coworkers on coupled-cluster response properties (see, e.g., refs 948–950 and references therein) and on the completely-renormalized EOM-CC formalism for excited-state energies (see, e.g., ref 951 and references therein) was inadvertently omitted in sections 4.6.3, 5.5, and 5.8.